United States Air Force

 $\frac{1}{2}$ 

Environmental Restoration Program

Eielson Air Force Base, Alaska



Operable Units 3, 4, and 5 Record of Decision

# FINAL

September 1995

1

÷.,

## Eielson Air Force Base Operable Units 3, 4, and 5 Record of Decision

## **Declaration of the Record of Decision**

## Site Name and Location

Operable Units 3, 4, and 5 Eielson Air Force Base, Alaska

## Statement of Basis and Purpose

This decision document presents the selected remedial actions and the no action decisions for Operable Units 3, 4, and 5 at Eielson Air Force Base, Alaska, chosen in accordance with the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), as amended by the Superfund Amendments and Reauthorization Act (SARA), the May 1991 Federal Facility Agreement Under CERCLA Section 120 entered into by the U.S. Air Force, the U.S. Environmental Protection Agency, and the State of Alaska, and to the extent practicable, the National Contingency Plan. This document also presents the decision that no further action is required for three other source areas at Eielson AFB. This decision is based on the administrative record for this site.

The State of Alaska concurs with the selected remedies and the no action decisions.

## Assessment of the Source Areas

OUs 3, 4, and 5 consist of 20 source areas that have been combined because of commonalities in contamination. These source areas include solvent- and fuel-spill sites, fuel storage tanks, drum storage areas, asphalt-cement mixing areas, landfills, and a fire-training area.

The OU 3 source areas are

- DP44 Battery Shop Leach Field
- WP45/SS57 Photo Lab/Fire Station Parking Lot
- ST56 Engineer Hill Spill Site
- SS61 Vehicle Maintenance Building 3213

#### The OU 4 source areas are

- DP25 E-6 Fuel Storage Tank Area
- ST27 E-11 Fuel Storage Tank Area
- WP33 Wastewater Plant Effluent Infiltration Pond
- SS35 Asphalt Mixing and Drum Burial Area
- SS36 Drum Storage Area
- SS37 Drum Storage Area
- SS39/SS63 Asphalt Lake/Asphalt Lake Spill Site

- ST58 Old Quartermaster Service Station Site
- SS64 Transportation Maintenance Drum Storage Site

The OU 5 source areas are

<ul> <li>LF02</li> </ul>	Old Base Landfill
--------------------------	-------------------

- LF03/FT09 Inactive Base Landfill/Fire Training Area
- LF04 Old Army Landfill and Explosive Ordnance Disposal Area
- LF06 Old Landfill

Three additional source areas were assessed, using a screening process, and are included in this report.

#### No Further Action Under CERCLA

Ten source areas (ST27, WP33, SS36, SS37, SS39/SS63, SS64, LF02, LF04, and LF06) will receive no further remedial action under CERCLA because they present little risk to human health and the environment. The no further action determination was based on a remedial investigation/baseline risk assessment and a sitewide draft ecological risk assessment. Although no further action is required, the groundwater at these source areas will continue to be monitored as part of the Sitewide Program to confirm the results of the remedial investigation. Under a separate federal program, the Air Force has submitted a closure plan for the ordnance area at LF04 under the Resource Conservation and Recovery Act (RCRA), Section 3008(a).

In addition, source areas LF01, WP32, and DP55 will receive no further action because, based on existing information, they do not present an unacceptable risk to human health and the environment. Data from these sites were compared to risk-based screening criteria (e.g., maximum contaminant levels [MCLs], EPA guidance documents) to evaluate the hazards. If contamination at a site was below the screening level or the affected pathway was incomplete, no further action was required. These three source areas meet these requirements and, therefore, require no further action under CERCLA.

#### Limited Action

Groundwater constituents in five of the source areas (WP45/SS57, ST56, SS61, and DP25) exceed maximum contaminant levels. These areas are isolated, have no significant contamination or inaccessible residual contamination in the vadose zone, and are characterized by a stable plume configuration. In the case of DP25, the plume is limited to an active tank farm. Action for these five source areas is limited to continued groundwater monitoring and restrictions on the use of the groundwater.

#### Active Remediation

The five source areas, DP44, SS35, ST58, and LF03/FT09, will be actively remediated as described in the following section.

In summary, actual or threatened releases and exposure of people to hazardous substances from DP44, WP45/SS57, ST56, SS61, DP25, SS35, ST58, and LF03/FT09 within OUs 3, 4, and 5, if not addressed by implementing the response action selected in this record of decision, may present a substantial endangerment to public health, welfare, or the environment.

## **Descriptions of Selected Remedies**

Selected remedies for the 23 source areas fall into 3 categories: 1) no further action under CERCLA, 2) limited action, and 3) active remediation.

Cleanup alternatives will be implemented using a phased approach, where design data gathering and ongoing monitoring will continue to be evaluated to confirm the appropriateness of the selected remedy or, once a remedy is implemented, to determine the effectiveness of the technology. This phased approach will accommodate needed selected remedy or system modifications.

For source areas where the selected remedy is limited action or active remediation, Air Force directives will restrict the use of groundwater until it meets federal and state standards. The Air Force will monitor the groundwater to evaluate contaminant movement and concentrations until compliance with state and federal regulations is attained.

#### Source Areas Requiring No Further Action

These 13 source areas (ST27, WP33, SS36, SS37, SS39/SS63, SS64, LF02, LF04, LF06, LF01, WP32, and DP55) will receive no further remedial action under CERCLA because they do not pose an unacceptable risk to human health or the environment. The groundwater at or near these sites will continue to be monitored as part of the Sitewide monitoring program, as appropriate, to verify that contaminant concentrations, if any, remain within acceptable screening levels.

#### Sources Requiring Limited Action

Five of the source areas (WP45/SS57, ST56, SS61, and DP25) will receive limited action including:

- Institutional controls to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.
- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.
- For groundwater at ST56, wellhead treatment using carbon adsorption or air stripping will be applied, as appropriate, to prevent human exposure to contaminants above regulatory levels.
- If future developments in bioventing technology make implementation practical at DP25, installation of a bioventing system will be re-evaluated at that time.

#### **Active Remediation**

Five source areas, DP44, SS35, ST58, and LF03/FT09 will be actively remediated. The major components of the selected remedies for each area are described in the following subsections. The selected remedies for DP44 and ST58 use treatment to address the principal threat of soil contamination that is posing a threat to groundwater through leaching.

## DP44 - Battery Shop Leach Field

- Installation of a soil vapor extraction to remove solvent contamination in soil that is posing a threat to groundwater through leaching.
- Institutional controls to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.
- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.

#### SS35 - Asphalt Mixing and Drum Burial Site

- Installation of a soil cover over the surface soil contamination to prevent direct contact by humans and animals and surface water runoff into Garrison Slough.
- Removal of drums in the future, if it is determined that they are a continuing source of contamination.
- Monitoring of surface water, sediments, and aquatic organisms in this area, as required to verify
  effectiveness of the cover, and monitoring of the groundwater to verify that levels remain below
  acceptable screening levels.

#### ST58 - Old Quartermaster Service Station Site

- Installation of a bioventing system to remove fuels contamination in the soil that poses a threat to groundwater through leaching. This system may include air injection within the upper part of the groundwater table and smear zone to volatilize and promote bioremediation of the contaminants. The system may also include air extraction, if deemed appropriate.
- Institutional controls to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.
- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.

#### LF03/FT09 - Inactive Base Landfill/Fire Training Area

- For the portion of the landfill where disposal occurred before 1980, RCRA Part 264 is relevant and appropriate. Currently, no groundwater at the edge of the waste management area exceeds regulatory levels; the residual contamination poses a direct contact threat. A cover to address the direct contact threat will be installed and maintained in accordance with relevant and appropriate requirements of Part 264. Groundwater at the landfill will continue to be monitored, as appropriate, to verify that the contaminant concentrations, if any, remain within acceptable screening levels.
- For the portion of the landfill where disposal occurred after 1980, RCRA Part 264 is applicable. The final cover will be constructed to: (1) provide long-term minimization of migration of liquids,

(2) function with minimum maintenance. (3) promote drainage and minimize erosion, (4) accommodate settling and subsidence, and (5) have a permeability less than or equal to the natural subsoils present. Post-closure care, including maintenance and monitoring, will be conducted in accordance with 40 CFR 264.117 and 264.228(b).

• Institutional controls will be implemented to restrict land use. In the event of base closure, any remaining contamination will be addressed in accordance with CERCLA Section 120.

## **Statutory Determination**

The selected remedies protect human health and the environment, comply with federal and state requirements that are legally applicable or relevant and appropriate to the remedial actions, and are cost effective. The remedies use permanent solutions and alternative treatment (or resource recovery) technologies to the maximum extent practicable and satisfy the statutory preference for remedies that employ treatment that reduces toxicity, mobility, or volume as a principal element.

Because these remedies will result in hazardous substances remaining onsite above health-based levels, reviews will be conducted at DP44, WP45/SS57, ST56, SS61, DP25, SS35, ST58, and LF03/FT09 no less often than every 5 years after the initiation of the remedial action, in accordance with Section 121(c) of CERCLA to ensure the remedies continue to provide adequate protection of human health and the environment.

Signature sheet for the foregoing Record of Decision for the final remedial action for Operable Units 3, 4 and 5 at Eielson Air Force Base, Alaska, between the United States Department of the Air Force and the United States Environmental Protection Agency, with concurrence by the State of Alaska.

JOHN G. JORBER, General, USAF Commander, Pacific Air Forces

30 40 95

# Signature and Support Agency Acceptance of the Remedies for OU 3, 4, and 5

<u>9-19-95</u> Date

KURT FREDRIKSSON Director of the Division of Spill Prevention and Response Alaska Department of Environmental Conservation

## Signature and Support Agency Acceptance of the Remedies for OU 3, 4, and 5

lach

9/22/98 Date

CHUCK CLARKE Regional Administrator Region 10 U.S. Environmental Protection Agency

## Contents

1.0	Site !	ame, Location, and Description	1
2.0	Site History and Enforcement Activities		
3.0	Highlights of Community Participation		
4.0	Scope	and Role of Operable Units	1
	4.1	Source Evaluation Areas	2
5.0	Sum	ary of OU 3, 4, and 5 Site Characteristics	1
	5.1	Contaminants of Concern Identified	1
		5.1.1 Meteorology	15
		5.1.2 Soils 5.1	15
		5.1.3 Background Soils	16
		5.1.4 Permafrost 5.1	16
		5.1.5 Groundwater 5.1	16
		5.1.6 Surface Water 5	17
		J. D. Surface Water Control Co	.,
6.0	Natut	and Extent of Contamination at Operable Unit 3	1
	6.1	Source Area DP44	1
		6.1.1 Soil Contamination at DP44	1
		6.1.2 Groundwater Contamination	2
	6.2	Source Area WP45/SS57	3
		6.2.1 Soil Contamination 64	4
		6.2.2 Groundwater Contamination 6.4	4
	63	Source Area ST56	ς.
	0.5	6.3.1 Soil Contamination 64	, ,
		6.3.2 Groundwater Contamination	5
	6 4		э Т
	0.4	$6.4 \pm \text{Soll Contomication} $	7
			/
		6.4.2 Groundwater Contamination 6.8	5
7.0	Natu	and Extent of Contamination at Operable Unit 4	1
	7.1	Contaminants of Concern	1
	7.2	Source Area DP25	1
		7.2.1 Soil Contamination 7.2	2
		7.2.2 Groundwater Contamination 7.3	ĩ
	73	Source Area ST27	đ
		7 3 1 Soil Contamination 74	ă
		7.3.2 Groundwater Contamination and Floating Fuel	т И
	7 4	Source Area WD33 74	+ <
	7.4		נ כ
	1.3	300000 Alice 3333         1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	3 7
			/ 0
		7.5.2 Groundwater Contamination	5
	/.6	Source Area SS30	5
		7.6.1 Soil Contamination	9

	7.7	<ul> <li>7.6.2 Groundwater Contamination</li> <li>Source Area SS37</li> <li>7.7.1 Soil Contamination</li> <li>7.7.2 Groundwater Contamination</li> </ul>	7.9 7.9 7.9 7.10
	7.8	Source Areas SS39 and SS63 7.8.1 Soil Contamination 7.8.2 Groundwater Contamination	7.10 7.10 7.11
	7.9	Source Area ST58         7.9.1 Soil Contamination         7.9.2 Groundwater Contamination	7.11 7.12 7.12
	7.10	Source Area SS64	7.13 7.13 7.13
8.0	Natur	re and Extent of Contamination at Operable Unit 5	8.1
	8.I	Contaminants of Concern	8.1
	8.2	Source Area LF02	8.1
		8.2.1 Soil Contamination	8.1
		8.2.2 Groundwater Contamination	8.2
-	8.3	Source Area LF03/FT09	8.2
		8.3.1 Soil Contamination	8.2
		8.3.2 Groundwater Contamination	8.3
	8.4	Source Area LF04	8.4
	8.5	Source Area LF06	8.4
		8.5.1 Soil Contamination	8.5
		8.5.2 Groundwater Contamination	8.5
9.0	Sumn	nary of Operable Unit Risks	9.1
	9.1	Human Health Risks	9.1
		9.1.1 Source Evaluation Report Areas	9.1
		9.1.2 Operable Unit 3, 4, and 5 Source Areas	9.2
	9.2	Site Data, Screening, and Identification of Contaminants of Potential Concern	9.2
	9.3	Exposure Assessment	9.3
		9.3.1 Conceptual Site Model	9.3
		9.3.2 Exposure Factors	9.4
	9.4	Evaluation of Lead Contamination	9.5
	9.5	Toxicity Assessment	9.5
	9.6	Risk Characterization	9.6
	9.7	Uncertainty Evaluation for the Human Health Risk Assessment	9.7
	9.8	Environmental Risks	9.9
	9.9	Summary of Source Area Disposition	9.9
10.0	Desci	ription of Alternatives – OU 3	10.1
	10.1	Remedial Action Objectives	10.1
	10.2	Remedial Action Components	10.2
		10.2.1 No Action	10.2
		10.2.2 Institutional Controls and Groundwater Monitoring	10.3
		10.2.3 Groundwater Extraction and Treatment	10.4
		10.2.4 Soil Vapor Extraction/Bioventing	10.4
		10.2.5 Soil Excavation	10.4

]	10.3 Source Area DP44	. 10.5
	10.3.1 Site Conditions	. 10.5
	10.3.2 Remedial Alternatives	. 10.5
]	10.4 Source Area WP45/SS57	. 10.7
	10.4.1 Site Conditions	. 10.7
	10.4.2 Remedial Alternatives	10.7
1	10.5 Source Area SS61	10.11
	10.5.1 Site Conditions	10.11
	10.5.2 Remedial Alternatives	. 10.12
11.0 I	Description of Alternatives – OU 4	11.1
1	11.1 Remedial Action Objectives	. 11.1
3	11.2 Remedial Action Components	11.2
	11.2.1 No Action	11.2
	11.2.2 Institutional Controls and Groundwater Monitoring	11.3
	11.2.2 Groundwater Extraction and Treatment	11.0
	11.2.9 Ground mater Excluderion and Treatments 17.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	11 4
	11.2.4 Excavation/Removal	11.4
	11.2.5 Cover	11.4
•	11.2.0 Blovening	11.4
1	11.2 L Six Conditions	11.5
		11.5
	11.3.2 Remedial Alternatives	11.5
1	11.4 Source Area SS35	11.8
	11.4.1 Site Conditions	11.8
	11.4.2 Remedial Alternatives	11.8
1	11.5 Source Area ST58	11.10
	11.5.1 Site Conditions	11.10
	11.5.2 Remedial Alternatives	11.10
12.0 1	Description of Alternatives-OU 5	12.1
1	12.1 Remedial Action Objectives	12.1
1	12.2 Site Conditions at Source Areas LF03/FT09	12.1
I	12.3 Remedial Alternatives	12.1
	12.3.1 Alternative 1-No Action	12.1
	12.3.2 Alternative 2-Institutional Controls and Groundwater Monitoring	12.2
	12.3.3 Alternative 3-Cover	12.3
	12.3.4 Alternative 4-Cap	12.3
13.0 \$	Summary of the Comparative Analysis of Alternatives	13.1
1	13.1 DP44 - Battery Leach Field	13.2
	13.1.1 Threshold Criteria	13.2
	13.1.2 Balancing Criteria	13.2
	13.1.3 Modifying Criteria	13.2
1	13.2 WP45/SS57 - Photo Laboratory/Fire Station Parking Lot	13.3
	13.2.1 Threshold Criteria	13.3
	13.2.2 Balancing Criteria	13.3
	13.2.3 Modifying Criteria	13.3
t	13.3 SS61 - Vehicle Maintenance Building 3213	13.4
•	13.3.1 Threshold Criteria	17.4
	13 3 2 Balancing Criteria	12.4
	rates and parameters and the second s	13.4

v

.- -

September 1995

		13.3.3 Modifying Criteria	13.4
	13.4	DP25 - E-6 Fuel Storage Tank Area	13.5
		13.4.1 Threshold Criteria	13.5
		13.4.2 Balancing Criteria	13.5
		13.4.3 Modifying Criteria	13.5
	13.5	SS35 - Asphalt Mixing and Drum Burial Area	13.6
		13.5.1 Threshold Criteria	13.6
		13.5.2 Balancing Criteria	13.6
		13.5.3 Modifying Criteria	13.6
	13.6	ST58 - Old Quartermaster Service Station	13.7
		13.6.1 Threshold Criteria	13.7
		13.6.2 Balancing Criteria	13.7
		13.6.3 Modifying Criteria	13.7
	13.7	LF03/FT09 - Inactive Base Landfill/Fire Training Area	13.8
		13.7.1 Threshold Criteria	13.8
		13.7.2 Balancing Critería	13.8
		13.7.3 Modifying Criteria	13.8
14.0	Select	ed Remedies	14.1
	14.1	No Further Action Sites	14.1
-	14.2	Recommended Limited Action Sites	14.2
		14.2.1 Source Area ST56 (OU 3)	14 4
		14.2.2 Source Area WP45/SS57 (OU 3)	14.4
		14.2.3 Source Area SS61 (OU 3)	14.4
		14.2.4 Source Area DP25 (OU 4)	14.5
	14 3	Recommended Treatment Action Sites	14.5
	14.2	14.3.1 Source Area DP44 (OIL3)	14.5
		14.3.2 Source Area \$\$35 (OII 4)	14.5
		14.3.3 Source Area ST58 (OU 4)	14.6
		14.3.4 Source Area J E03/ET09 (OU 5)	14.0
	14 4	Costs of the Selected Remedies	14.9
	14.4		14.0
15.0	Statut	ory Determinations	15.1
10.0	15 1	Protection of Human Health and the Environment	15 1
	15.2	Attainment of Applicable or Relevant and Appropriate Requirements of	10.1
	1.2.4	Freironmental I aus	15 1
		15.2.1 Apolicable or Relevant and Appropriate Requirements	15.1
		15.2.2 Chemical Specific ARARs	15.1
		15.2.3 Location-Specific ADARs	15.1
		15.2.4 Artion-Specific ADADs	15.2
		15.2.5 Criteria To Be Considered for Protectiveness	15.5
	15.2	Corr Effectiveness	15.4
	15.5	Use of Demonstry Solutions and Alternative Tractment Technologies to the	10.0
	15.4	Use of Permanent Solutions and Alternative Treatment Technologies to the	15.5
	15 5	Maximum Extent Practicable	15.5
	15.5	Preterence for freatment as a rrincipal Element	13.3
14.0	Ex-1-	nation of Simplifornt Differences	16 1
10.01			10.1
	10.1		10.L
	10.2		10.1
	10.3	Reason for Unange	10.1

.

17.0 References	17.1
Responsiveness Summary	RS.1
Appendix A	<b>A</b> .1

.-

# Figures

1.1	Map of Eielson Air Force Base	1.2
2.1	Sitewide Location Map for Operable Units 3, 4, and 5	2.2
6.1	DP44, Dichloroethylene in Soils, Depth = 0.06 to 1.2 m (August 1994)	6.10
6.2	DP44, Dichloroethylene in Soils, Depth = 1.2 to 1.8 m (August 1994)	6.11
6.3	DP44, Dichloroethylene in Soils, Depth = 2.4 to 3 m (August 1994)	6.12
6.4	DP44, Benzene in Groundwater (August 1992)	6.13
6.5	DP44, Benzene in Groundwater (August 1994)	6.14
6.6	DP44. Trichloroethane in Groundwater (August 1994)	6.15
67	DP44, Total Dichloroethylene in Groundwater (August 1994)	6.16
6.8	WP45. Trichloroethane in Groundwater (August 1992)	6.17
6.9	WP45, Total Dichloroethylene in Groundwater (August 1992)	6.18
6.10	WP45 Benzene in Groundwater (August 1992)	6.19
6 11	Map of ST56 with Well Locations and Water Table Elevations	6.20
6 12	SS61. Trichloroethane in Groundwater (August 1994)	6.21
6 13	SS61. Cis. 1.2-Dichloroethylene in Groundwater (August 1994)	6.22
71	DP25. Lead in Groundwater (1988)	7.15
72	DP25, Lead in Groundwater (April 1993)	7.16
7.3	DP25. Benzene in Groundwater (1988)	7.17
74	DP25. Toluene in Groundwater (1988)	7.18
7.5	DP25, Benzene in Groundwater (April 1993)	7.19
7.6	WP33. Site Man	7.20
77	SS35 Locations of Soil Sediment and Water Samples during August, 1992	7.21
78	SS35. DDT in Surface and Near-Surface Soil and Sediment Samples (1991 to 1992)	7.22
79	SS35. 4.4'-DDT in Soils and Sediments Depth = $0.00-0.15$ meters (August 1992)	7.23
7.10	SS36. Sampling Locations	7.24
7.11	SS39. Diesel/Kerosene in Soils and Sediments Depth = 0.00-0.15 m (July 1992)	7.25
7.12	SS39. Diesel/Kerosene in Soils and Sediments Depth = $0.9-1.2$ m (July 1992)	7.26
7.13	SS39. Diesel/Kerosene in Soils and Sediments Depth = 2,1-2,4 m (July 1992)	7.27
7.14	SS63. Diesel/Kerosene in Soils and Sediments Depth = 2,1-2.4 m (July 1992)	7.28
7.15	ST58, Benzene in Groundwater (April 1993)	7.29
7.16	ST58, Lead in Groundwater (April 1993)	7.30
8.1	LF02, Sampling Locations	8.6
8.2	LF03/FT09 Benzene in Groundwater, 1989	8.7
8.3	LF03/FT09 Benzene and Other Contaminants in Groundwater, 1994	8.8
8.4	LF03/FT09 Phenol and PC in Groundwater, 1994	8.9
8.5	LF03/FT09 Miscellaneous Contaminants in Groundwater, 1994	8.10
8.6	Map, LF06	8.11
9.1	Probability of Exceeding the Centers for Disease Control Blood Level Toxicity	
	Level at ST58	9.62
16.1	Simulated TCE plume centerline concentrations using input parameter values	
	minimizing the mean square error (MSE)	16.3
16.2	Simulated TCE plume centerline concentrations using input parameter values	
	minimizing the mean square error (MSE), but with no degradation ( $\lambda = 0/day$ )	16.4
16.3	Simulated TCE plume centerline concentrations using input parameter values	
	utilizing the calibrated model for source areas WP45/SS57, with the source	
	removal scenario at $T = 0$ years	16.5
	•	

16.4	5.4 Simulated TCE plume centerline concentrations using input parameter values	
	utilizing the calibrated model for source areas WP45/SS57, with the source	
	removal scenario at $T = 0$ years	16.6
16.5	BTEX Contours. Eielson AFB Source Areas WP45/SS57, May 1994	16.7
16.6	BTEX Contours, Eielson AFB Source Areas WP45/SS57, September 1994	16.8
16.7	TCE Contours, Eielson AFB Source Areas WP45/SS57, September 1994	16.9
16.8	tDCE Contours, Eielson AFB Source Areas WP45/SS57, September 1994	16.10
16.9	cDCE Contours, Eielson AFB Source Areas WP/SS57, September 1994	16.11
16.10	Eth (aq) Contours, Eielson AFB Source Areas WP45/SS57, September 1994	16.12

.

# Tables

5.1	DP44, Laboratory Analyses	5.2
5.2	WP45, Laboratory Analyses	5.3
5.3	SS57, Laboratory Analyses	5.3
5.4	ST56. Laboratory Analyses	5.4
5.5	SS61, Laboratory Analyses	5.5
5.6	DP25. Laboratory Analyses	5.6
5.7	ST27. Laboratory Analyses	5.7
5.8	WP33 Laboratory Analyses	5.8
5.9	SS35 Laboratory Analyses	5.9
5 10	SS36 Laboratory Analyses	5.9
5 11	\$\$37. Laboratory Analyses	5 10
5.12	SS39/SS63 Laboratory Analyses	5 10
5 13	ST58 Laboratory Analyses	5 11
5 14	SS64 Laboratory Analyses	5 11
5 15	LEG2 Laboratory Analyses	5 12
5.16	LE02/ET00 Laboratory Analyses	5 12
5.10	LEM Laboratory Analyses	5 12
5.18	LEOG Laboratory Analyses	5.15
5.10	Surface and Subsurface Soil Contaminants Greater Than Screening Levels DB44	5.14
0.1 ムウ	Groundument Contaminante Greater Than Screening Levels, Dr 44	6.2
63	Surface and Subsurface Soil Contaminants Greater Than Screening Levels, WPAS	6.4
0.5 6 A	Surface and Subsurface Soft Contaminants Ofcater Than Screening Levels, WF45	6.5
4 5	Groundunter Contaminants Greater Than Screening Levels, WF45	6.5
6.5	Groundwater Contaminants Greater Than Screening Levels, 5557	67
27	Soil Conteminante Greater Than Screening Levels, 5150	6.9
0.7	Surface and Subsurface Soil Contaminants Greater Than Screening Levels, DB25	0.0
7.1	Ded Floating Fuel Thickness DP25	7.2
73	Groundwater Contaminants Greater Than Screening Levels DD25	7.4
7.5	Groundwater Contaminants Greater Than Screening Levels, D125	7.4
7.5	Surface and Subsurface Soil Contaminants Greater Than Screening Levels, 5127	7.5
7.5	Groundunter Conteminante Greater Than Screening Levels, 5555	7.7
7.0	Surface and Subcurface Soil Contaminants Greater Than Screening Levels, SSSS	7.0
70	Surface and Subsurface Soil Contaminants Greater Than Screening Levels, 5559	7.10
7.0	Surface and Subsurface Son Containinants Orcater Than Screening Levels, 5158	7.12
7.10	Groundwater Contaminants Greater Than Screening Levels, 5158	7.15
7.10	Soil Contaminants Greater Than Screening Levels, 5504	7.14
0.1	Soli Contaminantis Orealer Than Scienting Levels, LF05/F109	0.3
ð.2	Groundwater Contaminants Greater Than Screening Levels, LF05/F109	0.3
9.1	Contaminants of Potential Concern, Media, and Concentration Ranges	9.10
9.2a	Cancer Risks for Source Areas in OUs 5, 4, and 5	9.31
9.20	INORCALCET FIGHTING Effects for Source Areas in OUS 5, 4, and 5	9.41
9,5	Summary of Exposure Factors	9.33
9.42	Cancel and Noncancer United Toxicity Factors for Major Containinants	9.33
9.4D	Cancer and Noncancer Toxicity Pactors for Uner Than Major Contaminants	9.06
9.5	Asses of Fielder Air Force Rose for Organic Chamical Contention and J and	0.00
• •	Areas at Eleison Air Force base for Organic Unemical Contaminants and Lead	9.60
9.0	ierrestrial riadital types and their rrimary wildlife Species on Eleison AFB	9.61

. . . <del>.</del> . . . . . . .

,

9.7	Actions Evaluated for Source Areas	9.61
10.1	Costs of Remedial Alternatives for Source Areas in OU3	10.3
10.2	Primary Components in Alternatives for Source Area DP44	10.5
10.3	Primary Components in Alternatives for Source Areas WP45/SS57	10.8
10.4	Primary Components in Alternatives for Source Area SS61	10.12
11.1	Costs of Remedial Alternatives for Source Areas in OU4	11.3
11.2	Primary Components in Alternatives for Source Area DP25	11.5
11.3	Primary Components of the Remedial Alternatives for Source Area SS35	11.8
11.4	Primary Components in Alternatives for Source Area ST58	11.10
12.1	Primary Components in Alternatives for Source Areas LF03/FT09	12.2
12.2	Costs of Remedial Alternatives for Source Areas in OU5	12.2
13.1	Comparison of Cleanup Alternatives for Source Area DP44 Using the Five	
	Balancing Criteria	13.9
13.2	Comparison of Cleanup Alternatives for Source Area WP45/SS57 Using the Five	
	Balancing Criteria	13.10
13.3	Comparison of Cleanup Alternatives for Source Area SS61 Using the Five	
	Balancing Criteria	13.11
13.4	Comparison of Cleanup Alternatives for Source Area DP25 Using the Five	
	Balancing Criteria	13.12
13.5	Comparison of Cleanup Alternatives for Source Area SS35 Using the Five	
	Balancing Criteria	13.13
13.6	Comparison of Cleanup Alternatives for Source Area ST58 Using the Five	
	Balancing Criteria	13.14
13.7	Comparison of Cleanup Alternatives for Source Area LF03/FT09 Using the Five	
	Balancing Criteria	13.15
14.1	Summary of Selected Alternatives	14.3
14.2	Summary of the Estimated Costs for Selected Remedies	14.8
15.1	Chemical-Specific ARARs for Contaminants of Potential Concern	15.3
15.2	Potential Action-Specific ARAR's	15.4
15.3	Alternative Soil Cleanup Levels Based on Leaching	15.5

FINAL

# Acronyms and Abbreviations

SDD	average daily dose
ADEC	Alaska Department of Environmental Conservation
AFB	Air Force Base
ARAR	applicable or relevant and appropriate requirement
AWQC	ambient water quality criteria
BAT	best available technology
BCT	best conventional technology
BLRA	baseline risk assessment
bls	below land surface
BTEX	benzene, toluene, ethylbenzene, and xylene
CDC	Centers for Disease Prevention and Control
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
CI	confidence interval
COE	Corps of Engineers
CRREL	Cold Region Research and Engineering Laboratory
DCE	dichloroethene
DNAPL	dense nonaqueous phase liquid
DO	dissolved oxygen
DP	disposal pit
EOD	explosive ordinance disposal
EPA	Environmental Protection Agency
FC	fecal coliform
FFA	Federal Facility Agreement

FNSB	Fairbanks North Star Borough
FS	feasibility study
FT	fire training area
GC/MS	gas chromatograph/mass spectrometer
Hasmat	hazardous material
HEAST	Health Effects Assessment Summary Table
HQ	hazard quotient
HS	headspace analysis
Hwy	highway
ICP	inductively coupled plasma
IEUBK	Integrated Exposure Uptake Kinetics
IRIS	Integrated Risk Information System
IRP	Installation Restoration Program
LF	landfill
LNAPL	light nonaqueous phase liquid
MCL	maximum containment level
MCLG	MCL goals
MFT	Membrane Filter Technique
MOGAS	motor gasoline
MPN	maximum probable number
MS	mass spectrometer
MSE	mean square error
MSL	mean sea level
NA	Not Applicable
NCP	National Contingency Plan

ND	Not Detected
NPL	National Priorities List
NST	no sample taken
NTU	nephelometric turbidity units
OSHA	Occupational Safety and Health Administration
OU	operable unit
OVM	organic vapor meter
PAH	polycyclic aromatic hydrocarbon
РСВ	polychlorinated biphenyl
PCE	polychioroethylene
PEL	permissible exposure limit
PID	photoionization detection
PNL	Pacific Northwest Laboratory
POL	petroleum, oil, and lubricant
ррт	parts per million
PRG	preliminary remediation goal
QA	quality assurance
RAB	Restoration Advisory Board
RAO	remedial action objective
RCRA	Resource Conservation and Recovery Act
RDX	cyclonite
RſD	reference dose
RI/FS	remedial investigation/feasibility study
RME	reasonable maximum exposure
ROD	Record of Decision

. -

SARA	Superfund Amendments and Reauthorization Act
SB	soil boring
SCS	Soil Conservation Service
SER	source evaluation report
SF	slope factor
SS	surface spill
ST	storage tank
SVE	soil vapor extraction
SVOC	semi-volatile organic compound
SWL	seasonal water level
ТАН	TBD
TBC	To Be Considered
TCA	trichloroethane
TCE	trichloroethylene
TCLP	Toxicity Characteristic Leaching Procedure
TDS	total dissolved solids
TRC	Technical Review Committee
TNT	trinitrotoulene
TOC	total organic carbon
тох	total organic halides
ТР	test pit
ТРН	total petroleum hydrocarbon
UBK	update biokenetic model
UWRL	Utah Water Research Laboratory
VOC	volatile organic compound
WP	waste pond

## Eielson Air Force Base Operable Units 3, 4, and 5 Record of Decision

## **Decision Summary**

## 1.0 Site Name, Location, and Description

Eielson Air Force Base (AFB) covers approximately 8000 hectares (19,700 acres), located along the Richardson Highway within the Fairbanks North Star Borough (FNSB) approximately 39 km (24 mi) southeast of Fairbanks and 16 km (10 mi) southeast of the city of North Pole, Alaska (Figure 1.1). Approximately 1477 hectares (3650 acres) are improved or partially improved and the remaining land encompasses forest, wetlands, lakes, and ponds. The base is bounded on the east and south by Fort Wainwright, a U.S. Army installation, and on the west and north by private and public land. The public and private land adjacent to the base is zoned for general use. The approximate population of the FNSB, Fairbanks, and North Pole is 82,000, 32,000, and 1600, respectively. Other communities near Eielson AFB include Moose Creek, which abuts the northern border of the base, and the Salcha area, which abuts the southern border of the base.

Eielson AFB is a major employer in the Fairbanks area. The base employs approximately 3400 military personnel and 500 civilians. The total residential population of Eielson AFB is 5132. Residential and occupational populations are primarily concentrated in the developed portion of the base. The area is active in ongoing base functions, including work, school, and recreational activities. The base contains three elementary schools, one junior-senior high school, a child care center, and one medical-dental clinic.

The base is located in the Tanana River Valley. Most of the base has been constructed on fill material. The developed portion of the base's topography is generally flat and somewhat featureless with elevations averaging about 168 m (550 ft) above mean sea level. The undeveloped east and northeast sides of the base are as high as 343 m (1125 ft) above mean sea level. Two-thirds of the base is covered with soils containing discontinuous permafrost.

Significant wildlife frequents Eielson AFB, and the base supports a variety of recreational and hunting opportunities. No threatened or endangered species live on the base.

The developed portion of the base is underlain by a shallow, unconfined aquifer comprised of 61 to 91 m (200 to 300 ft) of loose alluvial sands and gravel overlying relatively low-permeability bedrock. The aquifer is characterized by high transmissivities and relatively flat groundwater gradients. Groundwater at the base is generally encountered at approximately 2.4 m (8 ft) below grade with seasonal fluctuations up to 0.5 m (1.5 ft). The groundwater generally flows to the north-northwest,



S9310015.1a



with the direction of the flow locally influenced by surface water bodies (such as Garrison Slough and Hardfill Lake) and groundwater extraction from the base supply wells. Groundwater is the only source of potable water at the base and in the nearby communities. Potable water in the main base system is treated to remove iron and sulfide. Groundwater is the principal source for industrial, domestic, agricultural, and fire-fighting purposes.

## 2.0 Site History and Enforcement Activities

Eielson AFB was established in 1944, and military operations have continued to the present. During most of its history, Eielson AFB was a tanker base for midair refueling of strategic bombers. The current mission of the base is to train and equip personnel for close air support of ground troops in an arctic environment. Eielson AFB operations include industrial areas, aircraft maintenance and operations, an active runway and associated facilities, administrative offices, and residential and recreational facilities.

In carrying out its defense mission, the soils and groundwater at the base have been contaminated from the storage and handling of fuels and solvents, plus the operation of landfills. Initially, this contamination was evaluated under the U.S. Air Force Installation Restoration Program (IRP). The 4-phase IRP was initiated in 1982 with a Phase 1 records search to identify past disposal sites containing contaminants that may pose a hazard to human health or the environment. Under the IRP, the U.S. Air Force identified potential areas of contamination at Eielson AFB. Potential source areas included old landfills, storage and disposal areas, fueling system leaks, and spill areas.

Eielson AFB was listed on the National Priorities List (NPL) (54 FR 48184) on November 21, 1989 by the U.S. Environmental Protection Agency (EPA). This listing designated the facility as a federal Superfund site subject to the remedial response requirements of the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), as amended by the Superfund Amendments and Reauthorization Act (SARA).

In May 1991, the U.S. Air Force, the State of Alaska, and EPA entered into the Federal Facility Agreement (FFA) Under CERCLA Section 120 (EPA et al. 1990), which established the procedural framework and schedule for developing, implementing, and monitoring CERCLA response actions. An additional goal of the FFA was to integrate the U.S. Air Force's CERCLA response obligations and Resource Conservation and Recovery Act (RCRA) corrective action obligations. Under the FFA, potential source areas were placed in one of six operable units (OUs), based on similar contaminant and environmental characteristics, or were included for evaluation under a source evaluation report (SER). Source areas in OUs 3, 4, and 5 and selected SER sites are shown in Figure 2.1.



Figure 2.1.

2.2

# 3.0 Highlights of Community Participation

After the signing the FFA (EPA et al. 1990) with the State of Alaska and the EPA, and the listing of Eielson AFB on the NPL, the U.S. Air Force began its Superfund clean up program. As part of this program, in accordance with CERCLA Sections 113(k)(2)(B)(i-v) and 117, an extensive community relations program was initiated to involve the community in the decision-making process.

The community relations staff interviewed 40 local residents and community leaders to develop plans to inform residents about the clean up activity at Eielson AFB. Follow-up interviews and questionnaires of more than 100 residents helped revise the community relations plan. An environmental clean up newsletter was created and mailed to anyone who wished to be on the mailing list. Fact sheets were prepared on various topics related to the clean up operations. Several times a year, articles describing significant clean up events were released to the base newspaper *Goldpanner* and the *Fairbanks Daily News-Miner*. All of these efforts are designed to involve the community in the clean up process.

The remedial investigation/feasibility study (RI/FS) and baseline risk assessment (U.S. Air Force 1995a, 1995b, 1995c) and the Proposed Plan for Operable Units 3, 4, 5 and Other Areas of Eielson AFB (U.S. Air Force 1995d) were released to the public in May 1995. These documents were made available to the public in the administrative record and at an information repository maintained at the Elmer E. Rasmusen Library at the University of Alaska, Fairbanks. The selected remedies presented in this record of decision (ROD) are based on information contained in the Administrative Record.

The public comment period for the Proposed Plan was from May 18 to June 17, 1995. Comments received during this period are summarized in the Responsiveness Summary in an attachment at the end of this ROD. Five verbal comments were received during the public comment period. No written comments were received.

The public comment period, public meeting, and proposed plan for OUs 3, 4, and 5 were advertised four times in two local newspapers. The advertisements appeared in the *Fairbanks Daily News-Miner* on May 18 and 30, 1995 and in the *North Pole Independent* on May 19 and 26, 1995. In addition, more than 3500 copies of this notice were added as an insert in the base newspaper, *Goldpanner*, and delivered to every home in the Eielson AFB housing area on May 19. Proposed plans were mailed to more than 150 people on the clean up mailing list on May 16. Flyers announcing the public meeting were placed on store bulletin boards in the Moose Creek and North Pole communities.

A public meeting was held on May 31, 1995 in North Pole. Approximately 15 people attended the meeting, including representatives of the Air Force, EPA, ADEC, and the public.

## 4.0 Scope and Role of Operable Units

As with many Superfund sites, the problems at Eielson AFB are complex. Consequently, the FFA (EPA et al. 1990) divided the potential source areas at Eielson AFB into six OUs and three SER groups, based on common characteristics and contaminants. A final sitewide study is being conducted on human health and ecological risk.

The grouping of potential source areas into OUs was based on similar source characteristics or contaminants. The OUs are:

- OU 1 Petroleum, Oil, and Lubricant (POL) Contamination
- OU 2 POL Contamination
- OU 3 Solvent Contamination
- OU 4 Land Disposal of Fuel Tank Sludge, Drums, and Asphalt
- OU 5 Landfills and Fire Training Area
- OU 6 Ski Lodge Well Contamination.

An RI/FS was completed for OU 2 in November 1993 and for OU 1 and OU 6 in May 1994. A record of decision has been signed for each of these OUs. The sitewide OU is currently in the proposed plan stage.

This ROD addresses OUs 3, 4, and 5. Each of these OUs contains various source areas.

Five source areas with solvent contamination were designated under OU 3:

- DP44 Battery Shop Leach Field
- WP45 Photo Lab
- SS57 Fire Station Parking Lot
- ST56 Engineer Hill Spill Site
- SS61 Vehicle Maintenance Shop

Ten source areas that had land disposal of fuel tank sludge, drums, and asphalt were designated under OU 4:

- DP25 E-6 Fuel Storage Tank Area
- ST27 E-11 Fuel Storage Tank Area
- WP33 Effluent Infiltration Pond
- SS35 Asphalt Mixing and Drum Burial Area
- SS36 Drum Storage Area
- SS37 Drum Storage Area
- SS39 Asphalt Lake
- SS63 Asphalt Lake Spill Site
- ST58 Old Quartermaster Service Station
- SS64 Transportation Maintenance Drum Storage Site

Five source areas that are landfills were designated under OU 5:

- LF02 Old Base Landfill
- LF03/FT09 Inactive Base Landfill/Fire Training Area
- LF04 Old Army Landfill
- LF06 Old Landfill

The source areas in OUs 3, 4, and 5 contain soils contaminated with solvents, metals, and petroleum products at or near the source of contamination. Most of the contamination is in subsurface soils and the shallow groundwater. Much of the groundwater contamination is believed to originate from the sources in the soils through infiltration from precipitation. These 20 source areas pose various risks to human health because of the possibility for ingestion, inhalation, and dermal contact with contaminated soil and groundwater. The threat also exists for the environment with further migration of contaminants into the groundwater from contaminated soils and petroleum products floating on top of the water table.

The purpose of this ROD is to prevent current or future exposure to the contaminated groundwater and soils, to reduce further contaminant migration into the groundwater, and to remediate groundwater and soils.

## 4.1 Source Evaluation Areas

Through the source evaluation process (SER), 31 other source areas were evaluated. Based on the available information, these areas were believed to have a low probability of posing a significant risk to human health and the environment. Of those source areas, 21 were addressed in the OU 2 ROD and recorded for no further action. Seven of the source areas required further investigation; they were removed from the SER process and included in OUs 3, 4, or 5. The remaining three source areas that were included in this process do not pose an unacceptable risk to human health and the environment, and are, therefore, recorded in this ROD for no further action.

This group of SER sites was evaluated in a screening assessment to determine if each source poses a risk to human health or the environment. The screening of contaminants compared the maximum concentration of each contaminant detected in the source area to a risk-based concentration. This concentration was calculated using a conservative target risk that was based on EPA standard default exposure factors for a residential scenario. The target risks used for this conservative screening were chosen based on the lower end of the  $10^{-6}$  to  $10^{-6}$  risk range specified in the NCP. The assumption is: if no single sample exceeds a concentration representing a human risk concern, total exposure to the contaminant from the source area will not be of concern. Specifically, the area required no further action, if the maximum concentration detected was  $\leq 10^{-6}$  cancer risk for water,  $\leq 10^{-7}$  cancer risk for soil, and  $\leq 0.1$  hazard quotient.

In addition, soil contaminant concentrations were evaluated to determine the potential for contributing to groundwater contamination. Soil screening levels for the soil-to-groundwater pathway were determined, based on fate and transport modeling, in order to prevent exceeding the drinking water standards in the groundwater directly downgradient of the source area.

All of the sites were found to contain contaminants below screening levels (that is, maximum contaminant levels [MCLs], EPA Region 10 guidance) or the affected pathway was incomplete; therefore, no further action was required. These three sites are LF01, WP32, and DP55 (Figure 2.1).

The groundwater beneath Eielson AFB will continue to be monitored as part of the sitewide groundwater monitoring program. If it is determined that contaminant releases to the groundwater are originating from any areas recommended for no further action, the potential source of contamination will be reevaluated. This reevaluation may include additional sampling or source characterization.

A brief description and evaluation of each of the three no-action areas follows:

• LF01 (Original Base Landfill and Drum Storage Area). The site is located between the Richardson Highway and Piledriver Slough. LF01 includes an abandoned landfill and a drum disposal area. The landfill was used throughout the 1950s and received domestic and base operations wastes, including garbage, lumber, metal, construction debris, and empty cans. Solvent wastes, waste oils, and paint residue were also reportedly disposed in the landfill. The landfill was covered with a soil cap in 1960, but some refuse is visible. No historical record of use exists for the drum storage area. In 1992, approximately 2500 rusting, open, and generally empty drums were removed from the area and disposed in the borough landfill. Several drums found to contain liquids were removed in accordance with appropriate laws and regulations by base Hazmat (hazardous material) personnel.

The limited field investigation of this area included drilling two groundwater monitoring wells (one at the landfill and one at the drum storage area), digging seven soil pits down to groundwater at the drum storage area, sampling surface soils in a surface drainage that drains from the landfill to Piledriver Slough, and monitoring three existing wells near the landfill. Soil and groundwater samples were collected and analyzed for volatile and semivolatile organic compounds, pesticides, herbicides, and metals. Other than metals, none of the other compounds were detected in soil and groundwater samples. Metal concentrations did not exceed background levels. Site investigations and analyses of soil, groundwater, surface water, and sediment show no contamination that poses an unacceptable risk to human health or the environment.

• WP32 (Wastewater Plant Spill Ponds). This source area consists of two shallow earthen basins to which influent for the Wastewater Treatment Plant is diverted during wastewater incidents to avoid plant upsets or manage lagoon failures. The basins are normally dry. WP32 has been in use since 1970 for emergency disposal of wastewater. Major discharges include a one-time disposal of 5000 gallons of silver-nitrate-contaminated water. Water contaminated with an unknown industrial chemical or solvent was discharged in 1975. This area is located within an area used for treatment of base wastewater. Current operation of this area as a spill pond is regulated under a State of Alaska wastewater disposal permit.

The limited field investigation of this area included digging two soil pits to groundwater, sampling soil and groundwater at the bottom of the pits, and sampling existing groundwater monitoring wells in the area. The samples were analyzed for volatile and semivolatile organic compounds, pesticides, and metals. Of all of the analyses, only one water sample from the bottom of a test pit contained a pesticide (chlordane) in excess of screening levels. Because the sample contained significant quantities of sediment, this sample was not considered representative of groundwater in the area. Site investigations and analyses of standing water, basin sediments, and groundwater show no contamination that poses an unacceptable risk to human health or the environment. Future use of this area will be addressed within a wastewater disposal permit from the State of Alaska.

• DP55 (Birch Lake Recreation Area). This source area is a refuse pile adjacent to the maintenance yard at the Birch Lake Recreation Area administered by Eielson AFB. The refuse area has been used in the past to dispose of wastes generated in the maintenance and operation of the recreation area. This area has always been used to support the Birch Lake recreational mission. Minor amounts of fuel and vehicle maintenance supplies have been stored and disposed at the yard or in the surrounding undeveloped areas. Based on historical site use, materials used at the area might be small volumes of insecticides for mosquito control; solvents and degreasers for small motor maintenance and repair; paint, varnish, and wood preservatives; and small quantities of household refuse associated with the campground and picnic area. Based on the available groundwater data, site visits, and interviews, no contamination exists at DP55 that poses an unacceptable risk to human health or the environment.

## 5.0 Summary of OU 3, 4, and 5 Site Characteristics

Since 1988, contamination at the OU 3. 4, and 5 source areas has been investigated in detail. The sites were characterized as part of the Air Force IRP (SAIC 1989a, 1989b; HLA<sup>(a)</sup> 1989, 1990). A field sampling program was undertaken in 1992, designed to fill data gaps in the understanding of the 20 source areas based on the previous investigations. The 1992 field investigations were conducted, as described, in the Operable Units 3, 4, and 5 Management Plan (U.S. Air Force 1992). Additional investigations were carried out in 1993 and 1994 to further refine the conceptual models for selected source areas. Environmental samples were collected and analyzed, including soil, sediment, surface water, and groundwater samples. The analytes of interest list was comprehensive and included volatile organic compounds (VOCs), semi-volatile organic compounds (SVOCs), pesticides, polychlorinated biphenyls (PCBs), anions, and metals. The analytes and media sampled (groundwater, surface water, and soil) are summarized for each source area in the following text, and in Tables 5.1 through 5.15. In addition, a summary list of all soil sample analyses and most recent groundwater analyses is located in Appendix A.

The results of the ecological sampling are included in a Ecological Risk Assessment as part of the sitewide RI/FS program. The Operable Units 3, 4, and 5 Remedial Investigation document (U.S. Air Force 1995a) characterizes the source areas, so that risks to human health and the environment could be assessed in the Operable Units 3, 4, and 5 Baseline Risk Assessment (U.S. Air Force 1995b) and effective remedial alternatives could be developed in the Operable Units 3, 4, and 5 Feasibility Study (U.S. Air Force 1995c).

## 5.1 Contaminants of Concern Identified

The concentrations of the detected analytes were screened to assess their toxicological significance. Contaminants of potential concern were identified, based on the screening method suggested in the Supplemental Guidance for Superfund Risk Assessments in Region 10 (EPA 1991a). This method, called the *risk-based screening approach*, compares the maximum concentration levels detected at each source area to a risk-based screening concentration. The criteria for the screening, as given in the Region 10 supplemental guidance, are as follows:

- List the maximum concentration of each chemical in each medium for each source area.
- Compare the maximum concentration to risk-based screening concentration.
- · Eliminate the chemical if
  - the maximum detection for water  $\leq 10^{-6}$  cancer risk screening value and  $\leq 0.1$  Hazard Quotient (HQ) screening value and
  - the maximum detection for soil  $\leq 10^{-7}$  cancer risk screening value and  $\leq 0.1$  HQ screening value.
- Carry any chemicals not thus eliminated through the Baseline Risk Assessment.

<sup>(</sup>a) Harding Lawson Associates.

.

	· · · · · · · · · · · · · · · · · · ·	Groundwater				Soil				
Constituent	Method	1988(2)	1990 <sup>(b)</sup>	1992*	1994	1988 <sup>*(a)</sup>	1990 <sup>•(b)</sup>	1992*	1994	
Halogenated Volatile Organics	8010	x	x	X	x		x	x	x	
Aromatic Volatile Organics	8020	x	x	X	x		x	x	x	
Semivolatile Organic Compounds	8270			÷-		x				
Total Petroleum Hydrocarbons	E418.1				<b></b> `	x		x		
Arsenic	7060		х							
Lead	7241		X							
ICP Metals	6010		x	х						
Common Anions	E300		x							
(a) HLA (1989) (b) HLA (1990)	<ul> <li>(a) HLA (1989).</li> <li>(b) HLA (1990).</li> </ul>									

Table	5.1.	DP44,	Laboratory	Analyses
-------	------	-------	------------	----------

Data used in risk assessment.

X = analyzed.

..... رو

-- = not analyzed.

	[	Groundwater			Soil			
Constituent	Method	1988 <sup>(a)</sup>	1990 <sup>(b)</sup>	1992"	1988 <sup>•(a)</sup>	1990 <sup>*(b)</sup>	1992*	1994
Halogenated volatile organic compounds	8010	x	x	X		x	x	x
Aromatic volatile organic compounds	8020	X	X	X		x	x	x
Semivolatile organic compounds	8270				X			
Total petroleum hydrocarbons	E418.1				X			
Arsenic	7060	•	Х	X			X	
Lead	7241		Х	X			X	
ICP metals	6010		X	X				
Common anions	E300		X					
<ul> <li>(a) HLA (1989).</li> <li>(b) HLA (1990).</li> <li>* Data used in risk assessment.</li> <li>ICP = inductively coupled plasma.</li> <li>X = analyzed.</li> <li> = not analyzed.</li> </ul>								

#### Table 5.2. WP45, Laboratory Analyses

## Table 5.3. SS57, Laboratory Analyses

		Groundwater	Soil
Constituent	Method	1992	1992
Halogenated volatile organic compounds	8010	x	X
Aromatic volatile organic compounds	8020	x	x
Total petroleum hydrocarbons	E418.1	- 1	X
Lead	7241	x	x
X = analyzed. = not analyzed.			

			Groundwater				Soil			
Constituent	Method	1988	1990	1993	1994*	1988	1990	1993*	1994	
Halogenated volatile organic compounds	8010/8020	X <sup>(2)</sup>	X <sup>(a)</sup>	X	X			X		
Aromatic volatile organic compounds	8010/8020	X(a)	X <sup>(2)</sup>	х	X		•-	X	•-	
Semivolatile organic compounds	8270			Х				X		
Total petroleum hydrocarbons										
Arsenic	7060	X <sup>(1)</sup>	X <sup>(2)</sup>	X				X		
Lead	7421			X				X		
ICP metals	6010	X <sup>(a)</sup>	X <sup>(a)</sup>	X	X			X	X	
Common anions	······································	X <sup>(a)</sup>	X <sup>(2)</sup>							
(a) No method specified.	-mant				•••••••					

#### Table 5.4. ST56, Laboratory Analyses

\* Data used in risk assessment.

X = analyzed.

-- = not analyzed.

-..
1		
	m	
E	ā.,	
ŧ	÷.	
	ŏ	
1	Đ.	
ł	$\mathbf{r}$	
	4	
F	÷.	

		······	Grou	ndwater		<u>r</u>	Soil			
Constituent	Method	1989	1991	1993	1994	1989	1991	1993	1994*	
Purgeable halocarbons	8010		X <sup>(a)</sup>	x	X		X <sup>(a)</sup>	X	X	
Purgeable aromatics	8020	x	X(p)	x	X	x	X <sup>(b)</sup>	x	x	
Volatile organic compounds	8240		x				x		· · ·	
Semivolatile organic compounds	8270	x	x		x	x	x		x	
Pesticides and polychlorinated biphenyls	8080		x				x			
Herbicides	8150	**	x				x			
Total petroleum hydrocarbons	418.1 AK101/102	X <sup>(c)</sup>	unk. 	X 	 X	X <sup>(c)</sup>	unk.	X	 X	
Inductively coupled plasma metals scan	6010 6020				x		X			
Hnu		HS	HS			HS	HS			
Field gas chromatograph		HS				HS				
<ul> <li>(a) Method 601 used</li> <li>(b) Method 602 used</li> <li>(c) Method 8015M n</li> <li>* Data used in risk</li> <li>HS = headspace an</li> <li>X = analyzed.</li> <li>unk. = unknown.</li> <li> = not analyzed.</li> </ul>	1. J. used. assessment. alysis.									

## Table 5.5. SS61, Laboratory Analyses

сл СЛ

			Groundwater					
Constituent	Method	1988 <sup>(a)</sup>	1992	1993	1994 <sup>(b)</sup>	1988 <sup>(a)</sup>		
Halogenated volatile organic compounds	8010	x	x		X	X		
Aromatic volatile organic compounds	8020	X	X		X	x		
Volatile organic compounds	8240		ļ <del>.</del>	X	-			
Total petroleum hydrocarbons	E418.1					X		
Arsenic	7060	X	X		±=	X		
Lead	7241	X	X	· X	x	X		
ICP metals	6010	X	X			X		
Common anions	E300	X	X		*=			
<ul> <li>(a) SAIC (1988).</li> <li>(b) Nerney et al. (1994).</li> <li>* Data used in risk assessminic productively coupled point of the second second</li></ul>	lent. lasma.							

### Table 5.6. DP25, Laboratory Analyses

~

٠

	<u></u>		Groundwater					
Constituent	Method	1987 <sup>(a)</sup>	1988 <sup>(b)</sup>	1992*	1993	1988 <sup>(b)*</sup>		
Halogenated volatile organic compounds	8010	X	x	X		x		
Aromatic volatile organic compounds	8020	x	x	X		X		
Total petroleum hydrocarbons	E418.1					X		
Arsenic	7060		X	X		x		
Lead	7241	X	X	X	X	x		
ICP metals	6010		X	X		X		
Common anions	E300		X	X				
<ul> <li>(a) Method specified.</li> <li>(b) SAIC 1989b</li> <li>* Data used in risk assessme</li> <li>X = analyzed.</li> <li> = not analyzed.</li> </ul>	nt							

### Table 5.7. ST27, Laboratory Analyses

	Applytical		C	Fround Wa	ег		Surface Water		Soil
Parameter	Method	1985(**	1986 <sup>(b)</sup>	1988 <sup>(c)</sup>	1990 <sup>(d)</sup>	1994 <sup>(e)•</sup>	1988 <sup>(c)</sup>	1990 <sup>(d)</sup>	1990 <sup>(d)</sup>
Halogenated volatile	SW-8010	]	X	X	X	X	X	X	
organic compounds	······					·····			
Aromatic volatile	SW-8020		X	X	X	X	Х	X	
organic compounds						····			
Volatile organic	SW-8240	i				·	·······		x
compounds						······································			
Semivolatile organic	SW-8270				X				
compounds									
Pesticides/PCBs	SW-8080		i						х
Total petroleum	E418.1		X						
hydrocarbons	······································								
Arsenic	SW-7060				X				
Lead	SW-7241	X	X		X				···
Mercury	SW-7471				X				1
ICP Metals	SW-6010								X
TCLP	6010/7000								X
Total organic	415.1	X	х						
carbon									
Total organic	SW-9020	X							
halides									
Phenois	SW-9056	X							
Oil & Grease	413.1	X			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				
Anions	300					X			
Phosphate	365.2	X	X						
Nitrate/Nitrite	353.2	X	X	X			X	X	
$\overline{NA} = \overline{not} \text{ analyzed}.$									
(a) Dames & Moore 1985.									
(c) HLA 1991.									
(d) COE 1991.									
(c) USAF 1994b.									
* = Data used in risk ass	essment.								

## Table 5.8. WP33, Laboratory Analyses

		Gr	oundwater		Soil		
Constituent	Method	1988 <sup>(a)</sup>	1990 <sup>(b)</sup>	1992"	1988 <sup>(a)*</sup>	1990 <sup>(b)*</sup>	1992*
Halogenated volatile organics compounds	8010	x		x	x	·	x
Aromatic volatile organics compounds	8020	x	÷	x	x		x
Organochlorine pesticides and polychlorinated biphenyls	8080		x	x		x	x
Semivolatile organic compounds	8270					X	
Total petroleum hydrocarbons	E418.1	X		X	X		X
Arsenic	7060		X	X	x		••
Lead	7241		x	X	x		X
ICP metals	6010	X			x	· · · · · · · · · · · · · · · · · · ·	
<ul> <li>(a) HLA (1988).</li> <li>(b) HLA (1990).</li> <li>ICP = inductively coupled plasma.</li> <li>* = Data used in risk assessment.</li> <li>X = analyzed.</li> <li> = not analyzed.</li> </ul>							

Table 5	.9. 3	SS35,	Laboratory	Analyses
---------	-------	-------	------------	----------

Table 5.10	. SS36	, Laboratory	Analy	vses
------------	--------	--------------	-------	------

		Ground	water	Soil	
Constituent	Method	1988 <sup>(a)</sup>	1992*	1988 <sup>(a)*</sup>	1992*
Halogenated volatile organics compounds	8010	X	X	X	x
Aromatic volatile organics compounds	8020	x	x	X	x
Organochlorine pesticides and polychlorinated biphenyls	8080	x		X	
Total organic carbon	E415.1	x	1 1		
Oil and grease	E413.2	x			
Total petroleum hydrocarbons	E418.1	x		X	
Arsenic	7060	x	x	X	
Lead	7241	x	x	X	
ICP metals	6010	x	x	X	

(a) HLA (1989).

ICP = inductively coupled plasma.

\* = Data used in risk assessment.

X = analyzed.

-- = not analyzed.

	Groundwater			Soil		
Constituent	Method	1988 <sup>(a)</sup>	1992	1994*	1988(*)*	1992*
Halogenated volatile organic compounds	8010	x	X	x	X	x
Aromatic volatile organic compounds	8020	x	x	X	X	t
Semivolatile organic compounds	8270			x	X	
Lead	7241	x	X	x	X	
ICP/MS metals	6020	T		x		
ICP metals	6010	x	X		X	
<ul> <li>(a) SAIC (1989).</li> <li>ICP = inductively coupled plasma.</li> <li>MS = mass spectrometer.</li> <li>* = data used in risk assessment.</li> <li>X = analyzed.</li> <li> = not analyzed.</li> </ul>				-		

## Table 5.11. SS37, Laboratory Analyses

Table 5.12.	SS39/SS63,	Laboratory	Analyses
-------------	------------	------------	----------

		G	roundwater		Soil		
Constituent	Method	1988 <sup>(a)</sup>	1990 (b)	1992'	1988 <sup>(a)*</sup>	1990 <sup>(5)*</sup>	1992
Halogenated volatile organics compounds	8010	x	x	x	x	x	x
Aromatic volatile organics compounds	8020	<b>X</b> ·	x	x	x	x	x
Organochlorine pesticides and polychlorinated biphenyls	8080		x	x	x	x	x
Semivolatile organic compounds	8270		-	[ — ]	X	Х	X
Total petroleum hydrocarbons	E418.1			-	X	X	X
Arsenic	7060		X	X	X		
Lead	7241	X		X	X		
ICP metals	6010	X		X	X		
Common anions	E300			X			
<ul> <li>(a) HLA (1989).</li> <li>(b) HLA (1990).</li> <li>ICP = inductively coupled plasm.</li> <li>* = data used in risk assessment</li> <li>X = analyzed.</li> </ul>	a. nt.					<u> </u>	

-- = not analyzed.

.

	1		Groundwater					Soil			
Constituent	Method	1988	1992	1993*	1994	1988	1990	1993	1994		
Halogenated volatile organic compounds											
Aromatic volatile organic compounds	602/8240		x	x	X <sup>(2)</sup>						
Semivolatile organic compounds											
Total petroleum hydrocarbons	418.1		x						•-		
Arsenic											
Lead	7421	**		X	X <sup>(b)</sup>			+-			
ICP metals	200.7		X								
Common anions	300.0		X				*-				
<ul> <li>(a) Method 8020 used.</li> <li>(b) Method 239.2 used.</li> <li>* = data used in risk assessme</li> <li>X = analyzed.</li> <li> = not analyzed.</li> </ul>	nt.										

#### Table 5.13. ST58, Laboratory Analyses

## Table 5.14. SS64, Laboratory Analyses

			Ground	dwater		Soil			
Constituent	Method	1988	1990	1993	1994*	1988	1990	1993	1994
Halogenated volatile organic compounds	8010				X				
Aromatic volatile organic compounds	8020				X				
Semivolatile organic compounds	8270	·····			x				
Total petroleum hydrocarbons		1							
Arsenic	6020	,			X				
Lead	6020			-	X	-			
ICP metals	6010/6020				X				
Common anions									
* = data used in ri X = analyzed. = not analyzed.	sk assessment.								

FINAL

[			Ground	lwater			Sc	oil	
Constituent	Method	1988	1990	1993	1994	1988	1990	1993	1994
Halogenated volatile organic compounds	8010	X <sup>(2)</sup>	X <sup>(a)</sup>	X	X			x	
Aromatic volatile organic compounds	8020	X <sup>(a)</sup>	X <sup>(a)</sup>	Х	X <sup>(b)</sup>			X	
Semivolatile organic compounds	8270	X <sup>(a)</sup>			X		X <sup>(a)</sup>		
Total petroleum hydrocarbons	AK 102	X <sup>(2)</sup>			X		X <sup>(a)</sup>		
Arsenic	7060	X <sup>(a)</sup>		X	X <sup>(c)</sup>			X	
Lead	7421	X <sup>(a)</sup>		X	X <sup>(c)</sup>			X	
ICP metals	6010	X <sup>(a)</sup>		X	X <sup>(c)</sup>			X	
Common anions	300	X <sup>(a)</sup>			X	}			
NOTE: Surface water was sampled in 1988 and 1990. (a) No method specified. (b) Method 8010 used. (c) Method 3010/6020 used. * = data used in risk assessment. X = analyzed. = not analyzed.									

### Table 5.15. LF02, Laboratory Analyses

Table 5.16. LF03/FT09	J, Laboratory Analyses
-----------------------	------------------------

	T	Groundwater			[	Soil		
Constituent	Method	1988 <sup>(a)</sup>	1989 <sup>(b)</sup>	1992	1994	1988(*)*	1989 <sup>(b)*</sup>	1992
Halogenated volatile organic compounds	8010	X	X	X	X	X	X	X T
Aromatic volatile organic compounds	8020	X	X	X	X	X	X	X
Organochlorine pesticides and polychlorinated biphenyls	8080			-	X	X	-	
Semivolatile organic compounds	8270				X	X	X	X
Total petroleum hydrocarbons	E418.1			-		-	X	X
Nitrate	E300.0				X			
Total dissolved solids	E160.1			<b>[</b> –	X	<b>—</b>		7
Arsenic	7060		X	-		X	X	<b></b>
Lead	7241		X	X	X	X	X	( ·
ICP/MS metals	6020				X	- 1		
ICP metals	6010	-	X	X	-	X	X	1
Common anions	E300	-	X		X			7
<ul> <li>(a) HLA (1989).</li> <li>(b) HLA (1990).</li> <li>ICP = inductively coupled plasma.</li> <li>MS = mass spectrometer.</li> </ul>			<u>ka</u>	<b>1</b> # <u>1</u>	<u> </u>			<u>,</u>

\*

mass spectrometer.
 data used in risk assessment.
 analyzed.
 not analyzed.

Х

÷-

	1	G	Groundwater			Soil	
Constituent	Method	1988 <sup>(a)</sup>	1989 <sup>(b)</sup>	1992*	1988 <sup>(a)*</sup>	1989 <sup>(b)*</sup>	1992
Halogenated volatile organics	8010	X	X	X	X	X	X
Aromatic volatile organics	8020	X	X	X	Х	X	X
Organochlorine pesticides and polychlorinated biphenyls	8080				x	-	•-
Semivolatile organic compounds	8270	X	X		X	X	X
Total petroleum hydrocarbons	E418.1				X	X	
Total dissolved solids	E160.1	X	X				
Arsenic	7060	X	X		X	X	
Lead	7241		Х	X	Х	X	
ICP metals	6010	X	X	X	X	X	
Common anions	E300	X	X	X			
<ul> <li>(a) HLA (1989).</li> <li>(b) HLA (1990).</li> <li>* = data used in risk assessm</li> <li>X = analyzed.</li> <li> = not analyzed.</li> <li>ICP = inductively coupled plasm</li> <li>RDX = cyclonite.</li> <li>TNT = trinitrotoluene.</li> </ul>	ent. na.						

## Table 5.17. LF04, Laboratory Analyses

			Grour	ıdwater		So	oil	Surface Water
Constituent	Method	1988	1990	1993	1994"	1988	1990	1988
Purgeable halocarbons	8010	x	x	x	x			x
Purgeable aromatic compounds	8020	x	x	x	x			x
Volatile organic compounds	8240				+ <del>-</del>	x	x	
Semivolatile organic compounds	8270	x		x	X <sup>(2)</sup>	x	X <sup>(b)</sup>	x
Pesticides and polychlorinated biphenyls	8080				x			
Total petroleum hydrocarbon	E418.1	x			X <sup>(c)</sup>	x	x	x
Arsenic	7060	X <sup>(d)</sup>		X	X <sup>(c)</sup>			X <sup>(d)</sup>
Lead	7421	X <sup>(d)</sup>		Х	X <sup>(e)</sup>			X <sup>(d)</sup>
Mercury	7470	X <sup>(d)</sup>						X <sup>(d)</sup>
ICP metals scan	6010	x		X	X <sup>(f)</sup>			X
Total dissolved solids	E160.1	x			X <sup>(g)</sup>	NA	NA	x
Common anions	E300	Х			х	NA	NA	x
Nitrogen	E353.2	X			X <sup>(h)</sup>	NA	NA	x

Table 5.18. LFO6, Laboratory	Analyses
------------------------------	----------

(a) Method 8070 used.

(b) Soil boring samples only.

(c) Method AK102 used.

(d) Total and dissolved analysis performed.

(c) Method 6020 used.

(f) Method 6020/6010 used.

(g) Method 160.1 used.

(h) Method E3000 used.

\* = data used in risk assessment.

- X = Analyzed.
- -- = Not analyzed.
- ICP = inductively coupled plasma.

NA = not applicable.

Metals (such as arsenic, manganese, mercury, chromium, and others) were statistically compared to background metal concentrations established for the base (U.S. Air Force 1993a, 1993b). Metal concentrations were not found to be significantly different from background. Therefore, metals were not included in the contaminants of concern for the source areas. In the case of lead, any lead values exceeding the regulatory screening limit of 15  $\mu$ g/L in water were retained as a contaminant of concern. In addition, an effort was made to discriminate the contaminants of concern based on source. In particular, pesticides, such as DDT, chlordane, and dieldrin, are widespread in surface soil samples from OUs 3, 4, and 5. This widespread occurrence is believed to be the result of past sitewide spraying. Only at source area SS35 in OU 4 do soil concentrations clearly exceed those expected from spraying residues. Thus, except for SS35, consideration of pesticides has been referred to the Sitewide RI/FS.

Total petroleum hydrocarbons (TPHs) are discussed in the following sections and compared to Alaska State standards. No risk-based screening values are available for TPH, and, therefore, are not included in the contaminants of concern tables. The components of TPH (benzene, toluene, xylene, naphthalene, gasoline) are, however, included in the risk calculations. Concentrations of the contaminants of potential concern are summarized for each source area in the following text.

## 5.1.1 Meteorology

Eielson AFB is located in the continental climatic zone that covers interior Alaska. The climate is characterized by large diurnal and annual temperature variations, low precipitation, and low humidity.

Average summer temperatures range between 7 and 16°C (44.6 and 60.8°F). Average winter temperatures range between -26 and -13°C (-14.8 and -8.6°F). The extreme temperatures, since 1944, were 33°C (91.4°F) in June and -53°C (-63.4°F) in January.

Average annual precipitation is 36 cm (14.2 in.), which includes 180 cm (70.9 in.) of snow. Average monthly precipitation ranges from 1.3 to 6.4 cm (0.5 to 2.5 in.), and rainfall is generally highest in July and August. The average pan-evaporation rate is approximately 36 cm (14.2 in.)/year.

## 5.1.2 Soils

The developed portion of the base consists of three predominant soil types: sand and gravel fill, alluvium, and loess. Operable Units 3, 4, and 5 source areas generally consist of sand and gravel fill and alluvium. All OU 3, 4, 5 source areas are located on the flat (0- to 2-percent slope) alluvial plain. The soils are well-drained with moderate to high permeability (approximately  $10^{-1}$  to  $10^{-3}$  cm/s). A typical composite soil profile follows:

Upper 2-3 m (6.6 to 9.8 ft):

Sand to Sand with Gravel (SP-SW), brown to olive, moist, loose, some silt, 30 to 40 percent fine to coarse sand, 15 to 30 percent 3 to 5 cm (1.2 to 2 in.) gravel.

2-3 to 20+m (6.6-9.8 to 65.6+ft):

Sandy Gravel (GW-GP), brown to gray to black, wet, loose, 20 to 30 percent medium to coarse sand, gravel 5 + cm (2 + in.).

LF04 and ST56 are located on upland areas east of the developed portion of the base. Soils on these locations are thin loess overlying quartz-biotite schist bedrock that outcrops at many locations.

The organic content of the soils varies from one type to another. Total organic carbon (TOC) was analyzed to quantify the organic content. TOC values from subsurface soil samples were used to calculate average percentage of organic carbon in the soil. The average value obtained was 0.5 percent of the total weight of the soil. The wide variation in measured TOC values resulted in a standard deviation of 7 percent.

## 5.1.3 Background Soils

Background soil samples were also sampled and analyzed as part of the 1991 OU 2 effort. A summary of the soils encountered during the August-September 1991 background soil sampling effort is presented in the Operable Unit 2 Remedial Investigation Report (U.S. Air Force 1993a). Soil descriptions include those from the U.S. Department of Agriculture Soil Conservation Service (SCS) and from the Unified Soil Classification System.

On the basis of review of the soil survey conducted by the SCS, as well as the investigation of adjacent Eielson-area soils (Dames & Moore 1986), the three major soil groups sampled should be sufficient to describe the background chemical characteristics for the Eielson AFB vicinity. These groups include background data on several metals, naturally occurring organic materials, and possible human-manufactured compounds (for example, pesticides and PCB). Background soils have been shown to contain metals at concentrations that exceed screening levels. DDT and other pesticides have been widely used at Eielson and are generally found across the site.

## 5.1.4 Permafrost

Permafrost is discontinuous in the Fairbanks area, representative of the southern boundary of the permafrost in central Alaska. Near Fairbanks, permafrost extends up to 60 m (196.9 ft) below land surface (bls) and may act as an impermeable zone, causing groundwater to move around permafrost zones. The developed portions of Eielson AFB have little or no permafrost in the uppermost 9 m (29.5 ft) of the aquifer.

## 5.1.5 Groundwater

Only one aquifer is located within the main base. The unconfined aquifer consists of alluvial sand and gravel. It is 61- to 91.4-m (200- to 300-ft) thick and overlies crystalline bedrock (Birch Creek Schist). Within this unit, only the upper 18.3 to 27.4 m (60 to 90 ft) were characterized during this investigation. The aquifer was found to be relatively homogeneous between areas of investigation. The layering of materials indicates a greater horizontal than vertical permeability. All of the OUs 3, 4, and 5 source areas, except LF04 and ST56, are located in the flood plain of the Tanana River and are underlain by unconsolidated fluvial and glaciofluvial deposits that contain the uppermost unconfined aquifer. LF04 is located in an elevated area approximately 2 km (1.2 mi) east of the developed portion of the base. The aquifer underlying this source area is apparently contiguous with the aquifer in the lowland area. However, a groundwater discharge boundary may be present at French Creek, which separates LF04 from the developed portion of the base. ST56 is located on Engineer Hill, north of the main developed portion of the base. The uppermost aquifer at ST56 is composed of fracture zones in the schist bedrock. Since September 1991, water levels have been measured periodically in a sitewide monitoring well network. Measurements were made monthly from August 1992 through August 1993, and less frequently during other periods. Water table elevation contours are based on measurements made in September 1992.

Groundwater is the only source of potable water used at Eielson AFB. This water is supplied by three large-capacity wells of 1000 to 2000 gal/min capacity. The base water supply wells are completed at depths averaging approximately 30.5 m (100 ft). Seven wells are designated to provide water to fight fires on the base and are designed for emergency use only. These wells are plumbed to the water supply system. In addition to the base water supply wells, 41 private wells are within a 4.8-km (3-mi) radius of the base, most of which are located downgradient of the base (north-northwest of the base) in or near the community of Moose Creek (Figure 1.1) and in agricultural areas west of the base (HLA 1991). The city of North Pole is served by a small public water supply system, plus private wells.

The magnitude of the horizontal gradient was calculated for the main base. The average horizontal gradient is approximately 0.001 ft/ft. Data from a pumping test, slug tests, and grain size analyses were used to estimate a hydraulic conductivity of approximately 61 m (200 ft)/day. The direction of groundwater flow within the main base is north-northwest. Locally, groundwater flow is influenced by Garrison Slough, Hardfill Lake, and the pumping of base water supply wells. The direction of groundwater flow appears to be fairly constant year-round. LF04 is on a hillside about 2 km (1.2 mi) from the developed portion of the base. At LF04, the ground surface and the water table elevations are approximately 20 m (65.6 ft) higher than on the developed portion of the base in the French Creek lowlands. The hydraulic gradient at LF04 is much higher than the gradient found in the lowland portion of the base. However, the hydraulic conductivity of the silty sediments is probably at least two orders of magnitude lower than the hydraulic conductivity of the gravel and sand in the lowland area. Flow directions and gradients within the bedrock aquifer at ST56 are not known.

Water levels from nested wells were compared to provide information about vertical hydraulic gradients on the base. The shallow wells generally have a 6.1-m (20-ft) screen interval, beginning near the top of the aquifer, which is approximately 3 m (10 ft) below ground surface. The intermediate wells generally have a 3-m (10-ft) screen interval, beginning at approximately 9.1 m (30 ft) below ground surface. Pressure head differences between the shallow and intermediate wells were smaller than the potential error of the instruments. Therefore, the vertical gradient is negligible.

Seasonal changes in water levels were interpreted using a precipitation hydrography, snowpack data, and temperature data, primarily collected in 1991 and 1992. In general, the aquifer fluctuated uniformly across the site, indicating that similar hydrogeological conditions exist in the upper 30.5 m (100 ft) of the aquifer at all source areas. Typically, the water table reaches its minimum elevation in November. During this period, the discharge from the aquifer to the Tanana River and its tributaries exceeds recharge. In April, the water table typically rises about 0.9 to 1.8 m (1 to 2 ft) and a maximum is observed in the last week of May. This major recharge event coincides with the spring thaw, when runoff from the snowmelt is at a maximum. The water table drops relatively rapidly after the end of May.

## 5.1.6 Surface Water

Three source areas in OU 3, 4, and 5 are adjacent to surface water bodies. LF03, SS35, and SS37 are all adjacent to Garrison Slough. With the exception of a short period of time during spring runoff, the surface water elevation in the slough is lower than the groundwater elevations, indicating the slough is a *gaining* stream that receives recharge from the groundwater during most of the year.

# 6.0 Nature and Extent of Contamination at Operable Unit 3

OU 3 includes source areas DP44, WP45/SS57, ST56, and SS61. The source areas are primarily contaminated with solvents. The principal contaminants of concern for the OU 3 source areas include 1) TCE and its associated degradation product DCE, and 2) benzene, toluene, ethylbenzene, and xylene (BTEX). Other substances detected less frequently and at lower concentrations (such as 1,1,1-trichloroethane (1,1,1-TCA), tetrachloroethane, polycyclic aromatic hydrocarbons (PAHs), and lead) may be potential concerns. No vinyl chloride was detected in OU 3 source areas.

# 6.1 Source Area DP44

Source area DP44 is located near the Large Aircraft Maintenance Hangar. As originally defined, DP44 included wastewater disposal leach field from the battery shop (Building 1141) and the area around Building 1138 between the runway taxiway and Flightline Avenue west of the North Street intersection (see Figure 2.1). DP44 was identified as a source area because, in the past, the battery shop and Building 1138 may have discharged waste into a leach field system within the area. However, subsequent investigations have not confirmed the existence of this leach field. Most of the contamination in this source area is located south of the hangar, and is probably related to past jet-engine maintenance activities in the hangar. Identified contaminants of concern are fuel-related compounds and solvents in groundwater and soil.

## 6.1.1 Soil Contamination at DP44

Soil contaminants greater than EPA risk-based screening levels or background concentrations for DP44 are summarized in Table 6.1. Soil samples were collected and analyzed for the constituents shown in Table 5.1. A summary of soils data for source area DP44 can be found in Appendix A.

Soil contaminants listed in Table 6.1 are PAHs that were found at maximum concentrations in a surface soil sample collected during drilling of Well 44M03. This sample was collected in a gravel parking lot that contained fragments of asphalt. Because the parking lot is in close proximity to the runway, it was routinely maintained by spraying oil for dust suppression. The source of PAHs could be from exhaust from vehicles parked in the lot, exhaust from aircraft on the nearby runway, or asphalt residue. These contaminants are highly sorptive and immobile.

While solvent and benzene contamination was the reason for designating DP44 as a source area, the concentrations of these two constituents did not exceed screening levels. Subsurface concentrations of solvent and benzene were estimated to be sufficiently high to leach into the groundwater to yield concentrations that exceed groundwater screening levels. Therefore, even though solvent and benzene contamination in soils does not exceed screening levels based on direct exposure to the soil, it may be the source of groundwater contamination through the leaching pathway.

In August 1994, 13 soil borings were drilled in the vicinity of Weil 44M04 to determine the extent and concentration of chlorinated solvents in soils south of the large aircraft maintenance hangar. Locations for the borings were determined using a soil-gas survey. The soil-gas survey indicated that contaminated soils extended to the west under the aircraft parking ramp. The borings were completed through the vadose zone to the water table, located at approximately 3 m (10 ft) below land surface in this vicinity. Samples were taken at three depth intervals, 0.6 to 1.2 m (2 to 4 ft), 1.2 to 1.8 m (4 to

	Detection Limit	Analyzed/	Concentration Range	Location of
Chemical	(µg/kg)	Detected	(µg/kg)	Maximum
Anthracene	20	4/1	5500 - 5500	44M03
Benzo(a)anthracene	10	4/3	200 - 48,000	44M03
Benzo(a)pyrene	9	3/2	470 - 18,000	44M03
Benzo(b)fluoranthene	30	4/3	460 - 210,000	44M03
Benzo(g,h.i)perylene	40	4/3	280 - 14,000	44M03
Chrysene	70	4/2	280 - 21,000	44M03
Dibenzo(a,h)anthracene	50	4/1	6500 - 6500	44M03
Indeno(1,2,3-cd)pyrene	50	4/3	270 - 15,000	44M03

Table 6.1. Surface and Subsurface Soil Contaminants Greater Than Screening Levels, DP44

6 ft), and 2.4 to 3 m (8 to 10 ft) below land surface. The samples were analyzed for chlorinated solvents and BTEX compounds. TCE and total DCE results are illustrated in Figures 6.1 through 6.3. Low levels (less than screening level) of toluene were detected in some of the soil samples, as shown in Appendix A.

## 6.1.2 Groundwater Contamination

Groundwater contaminants in samples collected from monitoring wells that are greater than EPA risk-based screening levels or background concentrations for DP44 are summarized in Table 6.2. Groundwater samples were collected and analyzed for the constituents listed in Table 5.1. A summary of the sample concentrations can be found in Appendix A.

During field investigations at DP44 prior to 1994, benzene and TCE were found in the groundwater above their 5- $\mu$ g/L maximum contaminant levels (MCLs). Toluene, ethylbenzene, and xylene were also detected, but at concentrations below their MCLs. Benzene contamination above 5  $\mu$ g/L covered an area of approximately 3300 m<sup>2</sup> (3947 yd<sup>2</sup>), with Well 44M02 displaying the highest benzene concentration. Groundwater probe data collected in 1988 indicated that benzene concentrations up to 4000  $\mu$ g/L existed near the top of the water table. Benzene concentrations detected in 1990 had diminished fourfold since the 1988 sampling. By 1992 and 1994, benzene levels decreased to just above the MCL. For details on 1992 and 1994 benzene concentrations at DP44, refer to Figures 6.4 and 6.5. The 1992 data were used in the risk assessment and are reported in Table 6.2 and Appendix A.

Two areas of TCE contamination were detected in the groundwater above the 5- $\mu$ g/L MCL, including Well 44M03 and Well 44M04, with TCE concentrations above 100  $\mu$ g/L in Well 44M04. The two sites of TCE contamination appeared to be unrelated and relatively limited, based on groundwater probe results, which showed no detectable TCE between the sites or at adjacent probes or wells. The distribution of contamination near Well 44M04 indicated the source of TCE may be upgradient of DP44.

In August 1994, TCE and total DCE were still present in the vicinity of Well 44M04 in concentrations similar to previous years. These concentrations, as shown in Figures 6.6 and 6.7, were 109 ppb and 121 ppb, respectively. Results from the groundwater probe samples taken during the soil borings are also shown in Figures 6.6 and 6.7. They also show that chlorine solvent contamination extends away from Well 44M04 to the west under the aircraft parking ramp and to the north toward the hangar. The distributions for TCE and DCE are slightly different. Vinyl chloride has never been detected in any of the groundwater samples from DP44. No groundwater samples are available from underneath the hangar. Wells 44M03, 44M07, and 44M08, to the north of the hangar, show low levels of TCE and DCE contamination. The concentrations in all three wells are below MCLs. It is not known whether this contamination results from a second low-level source or is the leading edge of a plume located underneath the hangar. The total area of TCE-contaminated soils is approximately 6500 m<sup>2</sup> (69,000 ft<sup>2</sup>) with a volume of 20,000 m<sup>3</sup> (25,500 cu yd) containing an estimated 3.2 kg of TCE.

All contaminants detected at DP44 were in aqueous form. No free-phase solvent or fuel was encountered.

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range $(\mu g/L)$	Location of Maximum
Benzene	2	15/2	3.7 - 5.3	44M05
Trichloroethane	11	15/3	1.2 - 2500	44M04

Table 6.2.	Groundwater	Contaminants	Greater	Than	Screening	Leveis,	DP44
------------	-------------	--------------	---------	------	-----------	---------	------

## 6.2 Source Area WP45/SS57

The photo laboratory and dry well at Building 1183 were designated as Source Area WP45. Building 1183 is located near the main taxiway along the west side of Flightline Avenue (see Figure 2.1). The operational history of the dry well is not known. It was originally believed the dry well was the source of the solvent concentration found at WP45. However, in August 1992, two new wells were added upgradient from the dry well. Contaminant concentrations were higher on the two new wells than in the monitoring well downgradient of the dry well.

The dry well located at the west corner of Building 1183 has not been removed because removal would compromise the structure of the building. Standing groundwater in the well and sludge at the bottom of the well were sampled on April 1993, and the drain leading to the well was plugged with cement. Results indicated low levels of TCE in the water  $(3 \mu g/L)$  and low levels of chromium  $(1.2 \mu g/L)$  and silver  $(1.9 \mu g/L)$  in the sludge. Based on these results, the dry well appears to be a secondary source of groundwater contamination at WP45. The suspected primary source of contamination is currently believed to be a former maintenance shed that was located at the northwest corner of the fire station, Building 1206. No specific information explains the cause of the source of contaminants at WP45. The identified contaminants of concern are solvents in groundwater and soil.

Source Area SS57 is the area surrounding the fire station, Building 1206 (see Figure 2.1). SS57 is considered with Source Area WP45 because they are closely related and the groundwater contamination from the two sites overlap. Soils beneath the pavement in the parking lot of Building 1206 are contaminated with fuel. The primary contaminants of concern in SS57 are fuel-related compounds associated with spills of gasoline and jet propulsion fuel (JP-4) from fuel handling activities.

An independent study of natural attenuation by Utah State University (USU) was conducted concurrently with the remedial investigation at WP45/SS57. A meeting was held 6 July 95, during which USU presented their preliminary findings and modeling of site data collected at WP45/SS57. These findings are presented in Section 16.0, Explanation of Significant Differences.

## 6.2.1 Soil Contamination

Soil contaminants greater than EPA risk-based screening levels or background concentrations for WP45 are summarized in Table 6.3. No soil contaminants above EPA risk-based screening levels or background concentrations were identified for SS57. Soil samples were collected and analyzed for the constituents listed in Tables 5.2 and 5.3. A summary of the sample concentrations for WP45 and SS57 is found in Appendix A. TCE is the only contaminant listed in Table 6.3 that was detected in subsurface soils. The other contaminants, all PAHs, were found only in surface soils. This is consistent with the fact that the site is adjacent to the runway where jet exhaust containing PAHs is deposited.

Chemical	Detection Limit (µg/kg)	Analyzed/ Detected	Concentration Range (µg/kg)	Location of Maximum
Trichloroethane	1	7/2	3300 - 12,000	45SB08-B
Anthracene	1	17/4	88 - 921	45SS05
Benzo(a)anthracene	0.1	22/18	0.2 - 536	45S\$05
Benzo(a)pyrene	0.1	22/17	0.2 - 500	45M02
Benzo(b)fluoranthene	0.1	22/17	0.3 - 464	45SS05
Benzo(g,h,i)perylene	0.4	22/15	0.5 - 530	45M02
Dibenzo(a,h)anthracene	0.1	17/14	0.4 - 60	45SS05
Indeno(1,2,3-cd)pyrene	0.3	22/15	0.3 - 530	45M02

Table 6.3. Surface and Subsurface Soil Contaminants Greater Than Screening Levels, WP45

TCE and BTEX were detected only in soil boring 45SB08 (completed as monitoring well 45MW08) during 1992. The results indicated that TCE and BTEX concentrations in the soil increased with depth at WP45. Results from soil samples collected at SS57 during the same field season indicate that BTEX contamination is localized at soil boring 57SB02 and decreases with depth.

#### 6.2.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for WP45 are summarized in Table 6.4 and for SS57 in Table 6.5. Groundwater samples were collected and analyzed for the constituents listed in Tables 5.2 and 5.3. A summary of the sample concentrations for WP45 and SS57 is found in Appendix A.

TCE (7200  $\mu$ g/L) was found in groundwater samples at concentrations that exceed drinking water standards. The extent and concentration of TCE in groundwater is presented on Figure 6.8. TCE groundwater contamination covers an area of nearly 20,000 m<sup>2</sup> (almost 5 acres). The size of the TCE plume is relatively constant, but the maximum concentration of 7200  $\mu$ g/L in Well 45MW08 has never been confirmed with additional samples. TCE concentrations could be lower since the original analysis because snow removed from the tarmac was disposed on the area immediately west of the well. Infiltrating melt water may have dispersed and diluted the TCE in the groundwater. DCE, a decomposition product of TCE, has also been found at several locations within the TCE plume at concentrations up to 77  $\mu$ g/L (Figure 6.9). Traces of DCE (<1  $\mu$ g/L) have been found in the emergency fire well, Supply Well C, located approximately 45.5 m (150 ft) upgradient of the dry well. Benzene was detected in groundwater samples from WP45 in concentrations exceeding drinking water standards in 1988, 1989, and 1992. Based on the findings of the 1992 investigation, the benzene observed in the groundwater at Source Area WP45 may have migrated from Source Area SS57 and is not included in discussions of WP45. Vinyl chloride has not been detected in groundwater samples collected from either WP45 or SS57.

Groundwater samples from SS57 show concentrations of benzene (530  $\mu$ g/L) and toluene (1900  $\mu$ g/L) that exceed drinking water standards. The benzene is plotted in Figure 6.10. DCE was detected in three groundwater samples with a maximum concentration of 73  $\mu$ g/L.

All contaminants detected at WP45 and SS57 were in aqueous form. No free-phase solvent or fuel was encountered.

Chemical	Detection Limit	Analyzed/	Concentration	Location of
	(µg/L)	Detected	Range $(\mu g/L)$	Maximum
Trichloroethane	1	12/8	1.3 - 370	45M01

Table 6.4. Groundwater Contaminants Greater Than Screening Levels, WP45

Table 6.5.	Groundwater	Contaminants	Greater	Than	Screening	Levels,	SS57
------------	-------------	--------------	---------	------	-----------	---------	------

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
1,2-Dichloroethylene	0.5	3/1	5.3 - 5.3	57SB02
Benzene	2	3/2	5 - 530	57SB02
Toluene	2	3/1	1900 - 1900	57SB02

An independent study of natural attenuation by Utah Water Research Laboratory (UWRL), Utah State University, has been conducted concurrently with the remedial investigation at WP45/SS57. A meeting was held 6 July 1995, during which UWRL presented their preliminary findings and modeling of site data collected at WP45/SS57. The soil and groundwater contamination exists at this site in the form of low-level sorbed species and dissolved contaminant mass. Currently, no evidence of residual dense non-aqueous phase liquid (DNAPL) is present within a source area at the site; it also does not appear that any residual fuel material exists in the form of light nonaqueous phase liquid (LNAPL). The contamination is adsorbed and contained or in a dissolved phase and not accessible for source removal or treatment. UWRL focused its study on evaluating the current extent of the dissolved TCE plume, investigating evidence of TCE degradation existing throughout the site in the form of anaerobic dechlorination intermediate products, and evaluating the likelihood of biological mediated reactions, based on mass balance estimates and known stoichiometric relationships for these anaerobic transformation processes.

UWRL field data collection confirmed earlier findings reported by PNL. Low levels of soil contamination (<1 ppm TCE in all samples), an apparently contained groundwater plume (particularly benzene), no free product, and no vinyl chloride detected through DCE was present. New findings include further evidence of TCE anaerobic dechlorination with ethylene and large distribution of DCE product, significantly lower BTEX than previously reported, and rapid transportation of contaminants in the immediate vicinity of monitoring well 45MW08 (suspected source area).

The relative rate and extent of contaminant migration was evaluated through the use of a conventional 3-dimensional advective/dispersive groundwater model that incorporates groundwater flow, contaminant sorption, and contaminant degradation to describe the downgradient movement within the shallow aquifer over time. Model parameters that were not available or measured at the site were estimated using representative literature values.

The results of the UWRL study suggest that groundwater movement from this site is relatively slow (approximately 18 m/yr [59 ft/yr] pore water velocity, with approximately 6 m/yr [20 ft/yr] retarded TCE groundwater velocity based on measured field data). Additionally, with approximately 9 kg (20 lb) of TCE mass apparently lost in the aquifer over a 2-year monitoring period, it appears that TCE degradation is occurring at a first order degradation rate of approximately 0.00027\d (0.027%/d), yielding a TCE half life of approximately 7 years. With these values of contaminant velocity, apparent degradation rate, and an estimated source configuration based on model calibration, the remaining source of TCE contamination is predicted to be exhausted in another 7 years, with the subsequent groundwater plume generated from this source being attenuated within the aquifer to below regulatory limits of 5  $\mu$ g/L within 70 years, and within approximately 500 m (1640 ft) of the source.

## 6.3 Source Area ST56

ST56 (Engineer Hill Spill Site) is an active munitions storage and maintenance compound about 4.8 km (3.0 mi) north-northeast of the main part of the base (Figure 2.1). This compound is a secured area with a fence and guardhouse. Identified contaminants of concern are solvents, particularly tetrachloroethane (PCE), and fuel-related compounds in the ground within the hill. The original source of the contamination could not be found. Groundwater sampled in the lowland surrounding Engineer Hill and surface water and sediment in Lily Lake were not contaminated (see Figure 6.11).

The subsurface geology at ST56 consists of paleozoic quartz-mica schists, phyllites, and quartzite. The bedrock is characterized by a fracture pattern with a distinct orientation and low transmissivity. The supply well was pump-tested during the RI, giving a hydraulic conductivity value of 0.09 m/day. This number suggests an extremely slow transport velocity for any contaminant in the deep aquifer. Permafrost has been encountered at approximately 12 m (37 ft) bls.

#### 6.3.1 Soil Contamination

Two soil samples were collected near the wooden crib in which wastewater from Engineer Hill was discharged (see Figure 6.11). The samples were analyzed for VOCs, SVOCs, and total metals. No soil constituents exceeded EPA risk-based screening levels or background concentrations.

#### 6.3.2 Groundwater Contamination

The only groundwater contaminant greater than EPA risk-based screening levels or background concentrations for ST56 was PCE, as shown in Table 6.6. Groundwater samples were collected and analyzed for the constituents listed in Table 5.4. A summary of the sample concentrations can be found in Appendix A.

Inspection of the Engineer Hill compound, during June 1993, revealed no evidence for large use or release of solvents or petroleum products. No stressed vegetation, oily sheens, unusual odors, refuse, drums, or stained soil were observed at ST56. The septic-system leach field, at the bottom of the hill,

has eroded and is exposed. Water flows, at low discharge, into the road. The water has no distinctive odor or color. The septic-system leach field problem will be addressed with the State of Alaska under 18 AAC 72 Waste Water Disposal regulations.

Starting in 1986, groundwater samples were collected on a quarterly basis from the water supply wells at ST56. From 1986 on, a variety of organic compounds have been detected. The most consistently detected analyte in the wells has been PCE. No free-phase solvent contamination has been observed. In 1990, a new well was brought online and was similarly contaminated. Since 1991, ST56 has been supplied with drinking water, brought to the site and stored in tanks. One of the two supply wells has been shut down. Currently, groundwater point of use is restricted to toilets, boilers, and sinks with warning signs posted indicating the water is not for drinking.

INTE D.D. GIUDINAWALEI CUINAININANNA GIEALEI IIIAIN GEIEBININA GEVENA, GI	able 6.6.	6. Groundwater	Contaminants	Greater	Than	Screening	Levels,	ST5
---	-----------	----------------	--------------	---------	------	-----------	---------	-----

Chemical	Detection Limit	Analyzed/	Concentration Range	Location of
	(µg/L)	Detected	(µg/L)	Maximum
Tetrachloroethane (PCE)	0.5	5/2	13.8 - 25.1	WAD

## 6.4 Source Area SS61

SS61 is in the center of the developed portion of the base, just north of the water treatment plant pond on Garrison Slough, and is on the east and south sides of the Vehicle Maintenance Shop (Building 3213), as shown in Figure 2.1. The shop was originally built in 1954 and expanded in 1992. The shop has been used solely for vehicle maintenance. Waste oils, solvents, and water from maintenance activities were passed through an oil-water separator. The oil fraction was recovered for reuse. The wastewater was discharged to two dry wells located at the south end of the building. Identified contaminants of concern are fuel-related compounds and solvents in soil and groundwater. The source of the contamination appears to be one of the dry wells. During construction of the addition to Building 3213, both dry wells, along with surrounding contaminated soil, were removed, and the wastewater piping from the building was reconfigured to discharge to the sanitary waste system.

## 6.4.1 Soil Contamination

Currently, soils in the immediate vicinity of Well 61MW02 at SS61 are contaminated with TCE, cis-1,2-dichloroethylene, and BTEX. These soils provide some continuing source of groundwater contamination, but because wastes were discharged directly into the groundwater via the dry well, it is likely that most of the source for the plume is already in the groundwater.

No soil contaminants greater than EPA risk-based screening levels or background concentrations were identified for SS61. Soil samples were collected and analyzed for the constituents listed in Table 5.5. A summary of the sample concentrations for SS61 is found in Appendix A.

## 6.4.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for SS61 are summarized in Table 6.7. Groundwater samples were collected and analyzed for the constituents listed in Table 5.5. A summary of the sample concentrations for WP45 and SS57 is found in Appendix A.

The analytical results of groundwater samples collected from the three wells drilled near SS61 indicated that groundwater on the north side of the building addition is free of petroleum contamination, but groundwater near the eastern dry well (near Well 61MW01) is slightly contaminated with TCE (1  $\mu$ g/L) and petroleum constituents, such as benzene (2.8  $\mu$ g/L), toluene (6.8  $\mu$ g/L), ethylbenzene (3.6  $\mu$ g/L), xylene (26  $\mu$ g/L), and 1,2-dichlorobenzene (18  $\mu$ g/L). These concentrations are all less than their corresponding MCLs. These results are consistent with the fact the soil contained no or low concentrations of these same constituents. Groundwater near Well 61MW02 contained

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
Benzene	2	35/2	2.1 - 2.8	61MW01
Gasoline	120	32/4	400 - 2000	61-PS-3A
Trichloroethane (TCE)	1	35/15	1 - 1100	61-PS-3A

Table 6.7. Groundwater Contaminants Greater Than Screening Levels, SS61

significantly higher concentrations of TCE (78  $\mu g/L$ ), toluene (250  $\mu g/L$ ), and xylene (290  $\mu g/L$ ). TCE clearly exceeded the MCL of 5  $\mu g/L$ . Because of the dilution required to measure these concentrations, the reporting detection limit was increased from 0.5  $\mu g/L$  to 50  $\mu g/L$  for both benzene and tetrachloroethane. Concentrations of these two constituents was reported as < 50  $\mu g/L$ .

In 1994, lead was detected at concentrations above screening levels in water samples collected from wells 61MW01, 61MW02, and 61MW03. Concentrations in unfiltered samples ranged from 15.2  $\mu g/L$  to 40.4  $\mu g/L$ , as reported in Appendix A. It is believed that these lead concentrations are due to fine-grained sediment in the samples, because all unfiltered samples had a turbidity of >100 NTU. Only one of the filtered samples contained lead (22.3  $\mu g/L$ ) at greater than the screening level of 15  $\mu g/L$ . Lead in the other two samples decreased to 1.5  $\mu g/L$  and less than one  $\mu g/L$  after filtering. The lead detected at SS61 is believed to be associated with the soil and, therefore, immobile.

The data led to an additional investigation of the area, using the microwell technique (Nerney et al. 1994). In this investigation, 20 microwells were placed around the Vehicle Maintenance Shop and in the grass field across Division Street, north of the shop and downgradient of the dry wells.

The results indicate that groundwater just north of the original Vehicle Maintenance Shop is contaminated with TCE and cis-1,2-dichloroethylene at concentrations greater than their MCLs,  $5 \mu g/L$ and 70  $\mu g/L$ , respectively (Figures 6.12 and 6.13). Petroleum contamination is also indicated by the presence of TPH-G (Figure 6.10), but no specific BTEX compounds were detected in excess of their MCLs. Based on the assumption that sources for these contaminants are the dry wells on the south side of the building, a contaminant plume appears to extend from the dry well near Well 61MW02, beneath the building, to approximately Division Street. At that point, contaminant concentrations fall below MCLs, but they can still be detected for another several hundred meters south, extending beneath the grass field across Division Street. The lateral spread of the plume is limited to the footprint of the original vehicle maintenance building.

All contaminants detected at SS61 were in aqueous form. No free-phase solvent or fuel was encountered.

•





ì

s00

3, 4, and 5 Record of Decision





Figure 6.2. DP44 Dichloroethylene in Soils, Depth = 1.2 to 1.8 m (August 1994)

sno ω 4 and თ Record of Decision

Eielson AFB

September 1995

44M03

Trichloroethylene

in soil at 8-10 feet

August, 1994 (groundwater probes)

0





1

35

99

6

44MW111

19

204

252

1141

HANGAR

1140

44M08

0

44M07

Eielson AFB

FINAL



Figure 6.4. DP44, Benzene in Groundwater (August 1992)

OUs 3, 4, and 5

Record

of Decision

6.13

September 1995





Figure 6.5. DP44, Benzene in Groundwater (August 1994)

ł

September 1995

6.14

FINAL



ì



OUs 3, 4, and 5 Record of Decision

Eielson AFB

6.15

FINAL

September 1995



6.16

September 1995

FINAL

Record of Decision

0Us 3

4, and 5





Eielson AFB

Ν



0Us 3,

4, and 5 Record of Decision

FINAL

ţ

31860 ND 3131 ND Measured [31] Ø 45M08 concentration ND  $(\mu g/L)$ 3130 ND Not detected ND æ 9 Ø Groundwater പ്പാല monitoring WP45 45M04 0 45M03 445MW031 3229 Soil boring • ND 45MW00 (water sample)  $\mathcal{T}$ 3242 3240 1202 1183 45M0101 DRY VELL  $1-10 \ \mu g/L$ 1174 10-100 µg/L 045WW08 BASE TATES 0<sup>45M02</sup> 3241 ND .n >100 µg/L  $\overline{}$ 1185 1190 1176 206 Extent of contamination 45MW07 ND uncertain. 30 57SB02530 Benzene MAGNETIC DECLINATION MA 17 34° EAS in Groundwater 57SB01 5 575803 ND AVERAGE ANNUAL CHANGE - 21 WEST SS57 August, 1992 100 METERS

١.



OUs 3, 4, and 5 Record of Decision

Ì

6.19

September 1995

Eielson AFB





FINAL

1



Figure 6.11. Map of ST56 with Well Locations and Water Table Elevations

 $T_{\rm eff} = 1$ 

.

1





Figure 6.12. SS61, Trichloroethane in Groundwater (August 1994)

Eielson AFB

ì

6.21

September 1995

Legend: --- Inferred extent of contemination

FINAL



ì.



Eielson AFB

FINAL

Legend: --- Inferred extent of contamination

613480

# 7.0 Nature and Extent of Contamination at Operable Unit 4

This section discusses the nature and extent of groundwater and soil contamination identified at the Operable Unit 4 (OU 4) source areas DP25, ST27, WP33, SS35, SS36, SS37, SS39/SS63, ST58, and SS64.

# 7.1 Contaminants of Concern

OU 4 consists of a series of past-practices source areas that were characterized by drum disposal, asphalt-cement handling, road-oil mixing, and sludge disposal operations. Therefore, a variety of contaminants would be expected at these source areas. Because of the nature of the operations, most of the contaminant sources would be expected to be of relatively low volumes (such as drums). However, two of the reported sludge disposal areas, DP25 and ST27, are located within active fuel storage tank complexes, where large volumes of petroleum hydrocarbons are handled.

Any number of chemical components could have been stored in drums. Solvents and fuel hydrocarbons were sometimes used at asphalt-cement handling areas for cleaning and diluting the asphalt cement. Waste oils and fuels were used in road oiling operations. Chemicals that have been detected in previous work at these sites include pesticides (particularly DDT and its decomposition products DDD and DDE); lead and chromium (from paint); total petroleum hydrocarbons (TPHs) and polycyclic aromatic hydrocarbons (PAHs) from asphalt cement; benzene, toluene, ethylbenzene, and xylene (BTEX) from fuels and solvents; and chlorinated hydrocarbons from solvents. Compounds that might be expected, but were not detected in significant quantities, include polychlorinated biphenyls (PCBs) and ketones (paint solvents).

## 7.2 Source Area DP25

DP25, the E-6 Fuel Storage Tank Area, (shown in Figure 2.1) is located north of Quarry Road, adjacent to the E-11 Fuel Storage Tank Area, ST27. Six 4.78-million liter and two 800,000 L aboveground tanks are enclosed by a fence in a 210- by 340-m (229- by 372-yd) area. A single 16 millionliter above-ground tank is enclosed by a separate fence to the east of the main tank farm area. The fuel storage tanks were installed, with their underground piping and valves, in 1955 or 1956. Until 1992, the E-6 Tank Farm was used for storage of JP-4 jet fuel. Beginning in 1992, the tanks were converted to JP-8 fuel storage. The main fuel-product line feeding the tanks runs along Quarry Road. Weathered sludge from periodic cleaning of fuel tanks was buried in shallow trenches between the fuel storage tanks until 1980 (CH2M Hill 1982). The sludge consisted primarily of water, rust, dirt, and fuel. The trench or burial areas have never been located, despite a search of aerial photographs and old records in 1992. In 1987, a pipeline fuel spill of JP-4 reportedly occurred along Quarry Road adjacent to the source area.

The potential sources of contamination in DP25 are spills and leaks from the fuel storage tanks and their associated piping within the E-6 POL (petroleum, oil, and lubricants) Storage Area. The major contaminants of concern expected from fuel spills would be BTEX and lead. Additionally, the sludge from periodic cleaning of the tanks could be a source of lead contamination.

FINAL
## 7.2.1 Soil Contamination

Soil samples were collected from soil borings and analyzed for the constituents listed in Table 5.6. Soil contaminants greater than EPA risk-based screening levels or background concentrations for DP25 are summarized in Table 7.1. A summary of the sample concentrations can be found in Appendix A. Because DP25 is an active tank farm, the area is considered industrial. The 870,000  $\mu$ g/kg of lead (Table 7.1) at DP25 does not exceed the EPA industrial action level of 1,000,000  $\mu$ g/kg.

Chemical	Detection Limit (µg/kg)	Analyzed/ Detected	Concentration Range (µg/kg)	Location of Maximum
Dieldrin	1	15/1	8 - 8	25TP05
Heptachlor Epoxide	I I	15/2	1.4 - 250	25TP01
Lead	N/A	65/65	1800 - 870,000	25SD-1
PCB-1254 (Aroclor)	54	14/2	172 - 613	25TP01

Table 7.1. Surface and Subsurface Soil Contaminants Greater Than Screening Levels, DP25

BTEX constituents are curiously absent from soils at DP25, possibly because the water table is shallow at this site, often within two feet of land surface. Leaks in tanks or buried pipelines would have released fuel directly into the groundwater. Soils would be contaminated at the leak site or distributed in a layer over the area covered by the floating fuel. Because these latter soils are near land surface, volatile constituents such as the BTEX compounds could evaporate.

7.2.1.1 Floating Fuel. No floating fuel was detected at DP25 in 1986, but was detected in measurements made from 1988 through 1993. During this period of time, fuel thicknesses in the monitoring wells ranged from no floating fuel to 0.33 m (Table 7.2). The floating fuel samples from these wells were identified as JP-4. The greater thickness of product in Well B-15 in June 1992 may have been related to the increased recharge from spring snowmelt. No floating fuel has been detected

	Measurement (m)								
Monitoring Well Oct 1988 SAIC		Oct 1988 HLA	May-June 1992 USAF <sup>(a)</sup>	Aug-Sept 1992 PNL/CH2M Hill	April 1993 PNL				
B-1	ND	ND	0	ND	ND				
B-4	<0.006	0	ND	0.003	ND				
B-15	0.10	0.03	0.33	0.03	Sheen				
B-18	0.02	0	ND	0	0.003				
53M01	ND	0	ND	0	ND				
HLA = H ND = no PNL = P SAIC = So USAF = U (a) Inform	arding Lawson As ot determined. acific Northwest L cience Application .S. Air Force. <sup>(a)</sup> nal communicatio	ssociates. aboratory. n International Co n with Julie L. St	rporation. ringer, May-June 1	992.					

Table 7.2. JP-4 Floating Fuel Thickness, DP25

in measurements at Well 53M01, which is situated within the reported site of a 1987 JP-4 spill from a pipeline break along Quarry Road.

#### 7.2.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for DP25 are summarized in Table 7.3. A summary of the groundwater sample concentrations is presented in Appendix A. Groundwater samples were collected from soil borings and analyzed for the constituents listed in Table 5.6.

Significant lead concentrations were detected in 1988 groundwater samples from wells inside the main fence and ranged from 291 to  $362 \mu g/L$  (see Figure 7.1). Outside the main fenced area, lead concentrations that exceed the maximum contaminant level (MCL) of 15  $\mu g/L$  were detected in 1988 in three wells located west of the tank farm and in two wells east of the tank farm. Lead results varied from 1988 through 1993 somewhat erratically but were generally lower in 1993 (Figure 7.2) than in 1988. Lead also appeared to decrease in the downgradient wells. The lead distribution appears to have expanded between 1988 (Figure 7.1) and 1993 (Figure 7.2) only because of the additional monitoring locations in 1993. The source of the lead plume in the monitoring wells to the west of the facility is not known, as lead is expected to be less mobile in groundwater than BTEX. One of the unlocated – sludge pits could be the source of this contamination.

In 1988, BTEX was detected in groundwater at a number of the wells within the fenced areas of the tank farm. Significant benzene concentrations in groundwater at the wells in the main fenced area ranged from 46 to 290  $\mu$ g/L, with the concentration of benzene near the large tank to the east reading 7900  $\mu$ g/L. Toluene concentrations in three wells in the main tank farm area ranged from 1200 to 34,000  $\mu$ g/L; near the large tank, the concentration was 24,000  $\mu$ g/L. Based on these data, the apparent extent of the VOC plumes appeared to be limited to the fenced area, as shown in Figures 7.3 and 7.4.

BTEX was also detected at high concentrations (900 to 4700  $\mu$ g/L) in 1988 in groundwater at Well 53M01 near the reported 1987 pipeline fuel spill. A hydrocarbon identification test of the groundwater sample showed the source of the contamination to be motor gasoline, although the spill was reportedly JP-4 fuel. VOCs were fairly low (<10 mg/kg) in the adjacent soil, indicating that most of the product had already volatilized or migrated to the groundwater.

The results from 1992 show that BTEX contamination still existed at DP25 for the two monitoring wells sampled inside the bermed areas; concentrations have decreased by about a factor of 3 between 1988 and 1992. These observations, coupled with the essentially static amount of floating product since 1988, indicate the source of contamination at this site may have already ceased. Groundwater samples collected in April 1993 from downgradient wells indicate that contamination from DP25 has not migrated, in significant concentrations, beyond the bermed walls surrounding the tank farm (see Figure 7.5). Data from a microwell study in 1994, also indicate that BTEX contamination is still confined inside the E-6 tank farm berm (Nerney et al. 1994). No contamination was detected in any of the microwells north of DP25. No evidence exists that any contaminants, except lead, have spread beyond the bermed and fenced area of the facility.

7.3

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
Benzene	5.	38/11	20 - 1700	25B-18
Lead	5	38/18	5.7 - 60	25B-17
Toluene	5	38/9	36 - 8900	25B-18

Table 7.3. Groundwater Contaminants Greater Than	Screening Levels,	DP25
--	-------------------	------

# 7.3 Source Area ST27

ST27, the E-11 Fuel Storage Tank Area, (shown in Figure 2.1) is a fence-enclosed complex of five fuel tanks on the south side of Quarry Road approximately 600 m (654 yd) southeast of Hardfill Lake. At present, the E-11 Tank Farm is used for storage of jet fuel (JP-4 until 1992, now JP-8). The 8.72 million-liter above-ground tanks were installed in 1972 and are cleaned at 3- to 6-year intervals. Before 1980, the sludge from cleaning operations, composed of water, rust, dirt, and fuel, was buried in shallow trenches within the storage area. The trench burial areas have never been located, despite a search of aerial photographs and old records in 1992. Since 1980, the sludge has been drummed and shipped off-base for disposal.

The potential sources of contamination in ST27 are spills and leaks from the fuel storage tanks and their associated piping within the E-11 POL Storage Area. The major contaminants of concern expected from fuel spills would be BTEX and lead. Additionally, the sludge from periodic cleaning of the tanks could be a source of lead contamination.

### 7.3.1 Soil Contamination

Soil samples were collected and analyzed for the constituents listed in Table 5.7. No soil contaminants above EPA risk-based screening levels or background concentrations were identified for ST27. A summary of the soil sample concentrations is presented in Appendix A.

### 7.3.2 Groundwater Contamination and Floating Fuel

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for ST27 are summarized in Table 7.4. Groundwater samples were collected and analyzed for the constituents listed in Table 5.7. A summary of the groundwater sample concentrations can be found in Appendix A. No floating product was found at ST27 during surveys in 1987, 1988, and 1992. In addition, all of the 1992 VOC and BTEX analyses were below detection at all of the wells in and around ST27 (see Figures 7.3, 7.4, and 7.5.) None of the well samples, either within or downgradient of ST27, has shown any evidence of POL contamination. These tanks were installed in 1972 and have been maintained periodically since that time. It appears that these tanks have better integrity than those at ST10 and DP25, and no fuel leaks have occurred at this site.

A number of the wells within the source area boundary did show low levels of lead contamination. In June 1992, one well (B-13) showed a significant lead concentration (120  $\mu$ g/L). The high lead results for Well B-13 were not confirmed when the well was resampled on 1993 (Figure 7.2).

Chemical	Detection Limit	Analyzed/	Concentration Range	Location of
	(µg/L)	Detected	(µg/L)	Maximum
Lead	5	15/9	5.4-120	27B-13

Table 7.4.	Groundwater	Contaminants	Greater	Than	Screening	Levels,	ST27
------------	-------------	--------------	---------	------	-----------	---------	------

# 7.4 Source Area WP33

WP33, the effluent infiltration pond, is a 7.7-hectare (19-acre) unlined pond into which treated liquid effluent from the wastewater treatment plant is discharged. It is a major portion of the existing Wastewater Treatment Plant at Eielson AFB (Figure 2.1 and 7.6). The plant is on a separate access road from Central Avenue, about 0.5 km (0.3 mi) northeast of the main gate. The pond has been in use since 1979.

The wastewater treatment plant, built in 1953, currently treats most of the base domestic and operations wastewater. Some operations waste streams are treated by 12 distributed oil-water separators. The average daily flow through the plant in 1982 was 3400 m<sup>3</sup>/day (900,000 gal/day) (CH2M Hill 1982). Before 1973, primary treatment was effected through three clarifiers and two sludge digesters. The plant was expanded in 1973 to include secondary treatment at two aeration lagoons and a chlorination system. Until 1979, effluent was directly discharged to Garrison Slough under a National Pollution Discharge Elimination System permit (No. AK-002089-3). This permit requires routine monitoring of discharge waters for sanitary sewer parameters, chemical oxygen demand, and oil and grease. These parameters were frequently above normal, indicating unauthorized POL disposal or releases.

Wastewater entering the plant may have included wastes generated by spent solvents, deicers, and degreasers from shops and other repair facilities disposed of into storm drains, accidental discharges of these and other industrial chemicals, and photo shop and laboratory waste reagents from sinks and floor drains. The wastewater may also have included contaminated wash-down waters from runways, drive-ways, and roads; fire training pit discharges; dust suppressive oils; excess herbicides; pesticides; PCB; wood preservative containing solutions; spillage; and a variety of household and office supplies discharged through sanitary sewer outflows.

The infiltration pond, WP33, is engineered to discharge groundwater to the vadose zone, thereby disposing of treated wastewater effluent. According to conversations with Eielson AFB staff, the pond appears to be increasing in volume, evidenced by the increase in pond surface area over the last several years. This increased area would suggest that siltation of sediment or organic material has created a less permeable substrate. Groundwater monitoring data indicate that WP33 has not impacted the groundwater quality in the area. Nevertheless, historical data suggest the pond sediments may be contaminated from past practices. It is possible that future changes in the hydrology of the site or the operation of the ponds could expose pond sediments. No soil or groundwater contaminants greater than EPA risk-based screening levels or background concentrations were identified for WP33. A summary of the groundwater sample concentrations for WP33 is presented in Appendix A. Groundwater samples were collected and analyzed for the constinuents listed in Table 5.8.

# 7.5 Source Area SS35

SS35, the Asphalt Mixing and Drum Burial Area, (shown in Figure 2.1) is located in the central part of the base adjacent to Central Avenue, about 0.3 km (0.2 mi) south of the Water Treatment Plant.

SS35 was used as a mixing area from the early 1950s to the late 1960s. Asphalt cement was mixed in a tank and then used for road maintenance. Commingled waste oils and solvents were mixed with contaminated fuels and used for road oiling to control dust. Approximately 200 empty asphalt-cement drums were reportedly disposed of along the banks of Garrison Slough (CH2M Hill 1982). A ground tour of the site showed no evidence of the empty drums or the areas saturated with asphalt cement. It was speculated the drums might have been removed for proper disposal. The area was also apparently used for pesticide mixing and pesticide equipment cleaning operations.

The source area currently is inactive and covered with mowed grass. The only indications of past activities are several areas where asphalt cement and gravel are visible at the surface.

A number of potential sources of contamination at SS35 have been identified, including surface spills of asphalt cement, waste oils, solvents, contaminated fuels, and pesticides from the asphalt-cement mixing operations; pesticide mixing and cleaning operations; residual materials in buried drums; and metal, concrete rubble, and demolition debris remaining in the source area. Soil and groundwater samples were collected (see Figure 7.7) and analyzed for the constituents listed in Table 5.9.

Geophysical studies have been conducted at SS35 to identify potential locations of buried drums and other materials. During the 1988 geophysical investigation, two anomalies (anomaly areas A and B) indicating possible buried drums were identified (HLA 1989). The first anomaly (A) was a 23- by 69-m (25.2- by 75.5-yd) area identified in a location adjacent to Garrison Slough, roughly corresponding to an excavation area identified on historical aerial photographs. The area did not appear to have large numbers of buried drums, but does contain smaller areas concentrated with drums. The second anomaly (B), oriented northwest to southeast near Garrison Slough, may be an abandoned utility, such as a drainage culvert. Two small areas containing asphalt cement were present on the ground surface in this area, and aromatic and chlorinated hydrocarbons were detected near the center of the linear anomaly (B) in a soil gas survey.

It was concluded that metal debris and/or asphalt-cement drums had been disposed of in the trench (HLA 1989). The linear anomaly (B) is referred to as the trenched area.

A geophysical survey in 1990 identified a third anomaly (C) in the northern part of the source area extending about 38.1 m (41.7 yd) along the pond shoreline. Partially buried concrete rubble and demolition debris were visible in the brush and trees at the shoreline. It was concluded the area contained buried metal, concrete, and other demolition debris.

The contents of any drums buried at SS35 are generally unknown. In 1989, a number of drums were uncovered during the connection of Building 3460 to the utilidor. The burial site of the drums was located approximately 60 m (66 yd) east of anomaly area C, near the junction of the utilidor from Building 3460 with the Central Avenue utilidor. The contents of one drum uncovered at this time were analyzed for VOCs and semivolatile compounds (SVOCs) and metals. Its analysis is reported in HLA (1990). The compounds detected would be expected to be present in waste oils used in road oiling operations or waste solvent mixtures.

#### 7.5.1 Soil Contamination

Soil contaminants greater than EPA risk-based screening levels or background concentrations for SS35 are summarized in Table 7.5. Surface soils, subsurface soils, and sediments from Garrison Slough were collected and analyzed for the constituents listed in Table 5.9. A summary of the sample concentrations for SS35 is presented in Appendix A.

DDT and its derivatives were found in almost all surface and subsurface soil samples collected in the 1988 and 1990 investigations. The highest concentration observed was 396 mg/kg in a surface soil sample collected at 35SS03, located within anomaly area A near Garrison Slough. High DDT concentrations in surface soil were also observed at 35M01 (32.1 mg/kg), 35M02 (1.6 mg/kg) and two surface soil locations. In all cases, DDT concentrations were higher than the DDD and DDE concentrations. DDT was less than 1 mg/kg in all subsurface soil samples tested. Heptachlor was found in all soil samples tested in 1990 at concentrations less than 1 mg/kg. Chlordane was found in four of seven soil samples collected in 1990 at levels up to 260  $\mu$ g/kg.

One sediment sample from the slough (35S01) was tested for pesticides in 1988. It was located near the inlet of the pond Garrison Slough forms adjacent to the source area and showed 0.4 mg/kg of DDD. DDT concentrations were lower, at 0.097 mg/kg. In 1990, a sediment sample (62S04) was collected from the slough adjacent to SS35 just upstream of the pond. In that sample, DDT and DDD were 62.4 and 58.6 mg/kg, respectively.

Surface soil samples from SS35 in 1992 were analyzed for PCBs (such as Aroclor), pesticides, and lead. The results from the PCB analyses were at or below the detection limits reported by the analytical laboratory for soil samples. A summary of these sample concentrations is presented on Appendix A.

The surface soil samples from SS35 in 1992 contained detectable concentrations of 4,4'-DDT and derivative products 4,4'-DDD and 4,4'-DDE (Figures 7.8 and 7.9, respectively). The concentrations were highest for the soil samples taken from anomaly area B. The lowest concentrations of 4,4'-DDT, 4,4'-DDD, and 4,4'-DDE were measured in samples from anomaly area C along the pond shoreline. It is estimated that SS35 contains 765 m<sup>3</sup> (1000 cu yd) of contaminated soil covering a 1500 m<sup>2</sup> (16,000 ft<sup>2</sup>) area.

Chemical	Detection Limit (µg/kg)	Analyzed/ Detected	Concentration Range (µg/kg)	Location of Maximum
4,4'-DDD	1	23/18	0.4 - 58,500	355503
4,4'-DDE	N/A	23/21	0.09 - 19,000	35DIR05
4,4'-DDT	20	23/22	4 - 396,000	35SS03
Aldrin	1	4/1	6.2 - 6.2	35M01
Alpha-BHC	10	8/1	17 - 17	35DIR05
Chlordane	2	23/8	3 - 410	35DIR06
Heptachlor Epoxide	N/A	8/1	13 - 13	35DIR05

Table 7.5. Surface and Subsurface Soil Contaminants Greater Than Screening Levels, SS35

## 7.5.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for SS35 are summarized in Table 7.6. Groundwater samples were collected and analyzed for the constituents listed in Table 5.9. A summary of the sample concentrations for SS35 is presented in Appendix A. In 1992, the measured concentrations of PCBs and DDT and its derivative products 4.4'-DDD and 4.4'-DDE were at or below their reported limits of detection.

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
1,2-Dichloroethane	0.5	12/1	3.2 - 3.2	35GP03
4,4'-DDT	0.1	13/2	0.14 - 0.16	35GP02
Benzene	2	12/1	3.5 - 3.5	35GP03
Beta-BHC	0.05	13/1	0.05 - 0.05	35GP01

Table 7	.6.	Groundwater	Contaminants	Greater	Than	Screening	Levels.	SS35

- Groundwater samples from SS35 in 1992 were also analyzed for several other pesticide and organic contaminants. The concentrations of these compounds in all groundwater samples that were analyzed in August-September 1992 from SS35 were determined to be at or below the limits of detection. Groundwater probes installed in the area showed some BTEX, chlorinated solvent, and lead contamination. 1,1 dichloroethylene (1,1 DCE) was found in 17 of 33 groundwater probes, including 6 probes with concentrations exceeding the drinking water standard. 1,1 DCE was not detected in the deeper monitoring wells. Lead was detected at concentrations ranging up to 68  $\mu$ g/L, but this is likely due to high turbidity of the samples. None of the water samples collected in 1992 from the eight monitoring wells contained lead in excess of the 15  $\mu$ g/L screening level.

# 7.6 Source Area SS36

SS36, a drum storage site, is located in the central portion of the base, east of Industrial Drive and south of the base power plant (see Figure 2.1).

SS36 was used as a mixing area for asphalt cement and the road oiling operations from the late 1960s to the mid-1970s. In 1982, approximately 100 drums containing materials, such as waste oils, hydraulic fluid, diesel, JP-4, Stoddard solvent, and methyl ethyl ketone, were stored in this area. At that time, none of the drums were observed to be leaking; however, evidence of petroleum-contaminated soil and a pool of petroleum-contaminated water appeared near the mixing tank used for asphalt cement and waste oils. A number of laboratory analyses were performed on groundwater and soil samples from SS36 and the constituents are listed in Table 5.10. None of the available references indicate that drums may have been buried at SS36.

A paint spill in the northeast corner of the area was reported in 1989 (SAIC 1989a). The soil contaminated from the paint spill was removed in 1992. The paint spill was the source of the high lead and chromium values in this area. The drum storage location and paint spill area are identified on Figure 7.10.

#### 7.6.1 Soil Contamination

No soil contaminants above EPA risk-based screening levels or background concentrations were identified for SS36. A summary of the sample concentrations for SS36 is presented in Appendix A. One surface soil sample showed a lead concentration of 7800 mg/kg which is significantly above the EPA action level for lead in soil. However, this sample was collected directly from the paint spill area prior to removal of the soil.

#### 7.6.2 Groundwater Contamination

No groundwater contaminants above EPA risk-based screening levels or background concentrations were identified at SS36. Groundwater samples were analyzed for the constituents listed in Table 5.10. A summary of the sample concentrations for SS36 is presented in Appendix A.

BTEX and TCE were detected in a soil gas survey and in 14 groundwater probes installed at SS36. However, no volatile compounds were detected in the 1992 groundwater samples collected from three monitoring wells. One groundwater probe showed a lead concentration (44  $\mu$ g/L), above the EPA action level for lead in drinking water. However, the groundwater probe samples were used as site screening measurements, and are less accurate than the analysis of groundwater samples from the monitoring wells.

## 7.7 Source Area SS37

SS37, the Drum Storage/Asphalt Mixing Area, is located approximately 90 m (98.4 yd) east of Building 4333, just east of Flightline Avenue, between Quarry Road and Chena Street (as shown in Figure 2.1). The site was used as a mixing area for road oiling operations and a mixing area for asphalt cement from the mid-1970s to 1986. Some drums of miscellaneous liquid wastes were stored at the source area. Their contents included waste oils, diesel fuels, jet fuels (JP-4), and Stoddard solvent (PD-680). The drums have since been characterized, manifested, and disposed through the base hazardous waste handling facility. None of the drums were observed to be leaking during the IRP Phase I investigation; however, the ground surface appeared to be stained with petroleum, oil, and lubricant (POL) (HMTC 1986). None of the available references indicate that drums may have been buried at SS37. An area south of the storage area was used for fire-training exercises from 1976 to 1981. A number of laboratory analyses were performed on groundwater and soil samples from SS37.

Source area investigations in SS37 were conducted in 1984 (Dames & Moore 1985), in 1986, 1987, 1988 (SAIC 1989b), and in 1992 and 1994 (U.S. Air Force, 1995a,b,c). U.S. Air Force (1992, Tables 10.2 through 10.5) summarizes analytical results from all sampling events.

#### 7.7.1 Soil Contamination

No soil contaminants above EPA risk-based screening levels or background concentrations have been identified for SS37. Surface and subsurface soil samples were collected and analyzed for the constituents listed in Table 5.11. A summary of the sample concentrations for SS37 is presented in Appendix A.

#### 7.7.2 Groundwater Contamination

Benzene at a concentration above drinking water standards was detected in monitoring Well 37-2 in 1986 (26  $\mu$ g/L) and in Well 37-3 in 1987 (15  $\mu$ g/L) (SAIC 1989a). All four wells at source area SS37 were sampled in 1992 and benzene was detected in all of the wells, with two of the wells showing concentrations above the MCL (10 and 13  $\mu$ g/L). Traces of free-phase product were found in one well in February 1992, but were not confirmed in the summer of 1992.

By 1994, concentrations of all groundwater contaminants had decreased to below EPA risk-based screening levels or background concentrations at SS37. Groundwater samples were collected and analyzed for the constituents found in Table 5.11. A summary of the sample concentrations for SS37 can be found in Appendix A.

## 7.8 Source Areas SS39 and SS63

Asphalt Lake (SS39) and the adjacent Asphalt Lake Spill Site (SS63) are located approximately 2 km (1.2 mi) south of the Eielson AFB main gate. The two areas are separated by a gravel access road. Five groundwater monitoring wells (39M01 through 39M05) exist at SS39, and one groundwater monitoring well (53M02) lies approximately 100 m (109.4 yd) southeast (cross-gradient) of SS63.

Asphalt Lake was used as a disposal area for barrels and drums of asphalt cement after runway construction in the 1950s. As these containers deteriorated and leaked, a layer of asphalt cement up to 45 cm (17.7 in.) deep in places covered an area of approximately 4000 m<sup>2</sup> (1 acre). Besides the physical hazard imposed by the asphalt cement, the primary contaminants of concern were PAHs, which were a component of the asphalt cement. During May and June 1992, the asphalt cement, barrels, approximately 2500 drums, 91,750 m<sup>3</sup> (120,000 cu yd) of petroleum contaminated soil, and other debris were removed. A layer of fill dirt approximately 2 m (6.6 ft) deep was placed over the excavated area in June 1992. No record is shown for disposal activities at SS63.

#### 7.8.1 Soil Contamination

The only soil contaminant to exceed EPA risk-based screening levels or background concentrations at SS39 was DDT, as reported in Table 7.7. A summary of the sample concentrations for SS39 can be found in Appendix A. Soil samples were collected from pits, soil borings, and ground surface soil and analyzed for the constituents found in Table 5.12. Sample locations and diesel kerosene sampling results are presented on Figures 7.11 through 7.14.

Table 7.7.	Surface and	Subsurface	Soil	Contaminants (	Greater	Than :	Screening	Levels,	SS39
------------	-------------	------------	------	----------------	---------	--------	-----------	---------	------

Chemical	Detection Limit	Analyzed/	Concentration	Location of
	(µg/kg)	Detected	Range (µg/kg)	Maximum
DDT	1	26/15	1 - 437	39SB02

#### 7.8.2 Groundwater Contamination

No groundwater contaminants above EPA risk-based screening levels or background concentrations have been identified for SS39/63. Groundwater samples were collected and analyzed for the constituents listed in Table 5.12. A summary of the sample concentrations for SS39/63 can be found in Appendix A.

# 7.9 Source Area ST58

ST58, site of the old Quartermaster service station, is located on the northwest corner at the intersection of Division Street and Wabash Avenue (Figure 2.1). The service station covered approximately  $400 \text{ m}^2 (478 \text{ yd}^2)$ .

The Quartermaster service station was operated from 1970 to 1988. The service station was a source of petroleum products for private vehicles operated by Eielson AFB personnel and their dependents until 1975, and for Eielson AFB vehicles only after 1975. The service station used four 95-m<sup>3</sup> (25,000-gal) above-ground storage tanks, containing leaded and unleaded motor gasoline (MOGAS) and diesel. Two barrels of motor oil were stored at the service station for customer use.

Eielson AFB staff removed the above-ground storage tanks and above-ground piping in August 1988. Underground piping was left in place. During removal, workers noted evidence of product releases. No analytical work was performed. The surface was covered with a meter (a yard) of fill after the above-ground storage tanks and piping were removed (Liikala and Evans 1995). No spills have been reported at ST58. However, the pipeline that supplied fuel to ST58 was suspected of leaking at the intersection of Industrial Drive and Division Street, east of ST58. In 1993, approximately 532 cu m (700 cu yd) of fuel-contaminated soil was removed from the area most highly contaminated for a composting demonstration and replaced with clean fill material.

Motor gasoline (MOGAS) and diesel stored and used at the Quartermaster service station appears to have been spilled or leaked from the piping and diesel tanks. Some of the volatile components of petroleum products released at the surface may have evaporated. The less volatile components probably seeped into the soil. The less volatile contaminants may have adsorbed to the soils or been dissolved in surface infiltration and carried to the groundwater. Fuel from large spills, if any occurred, may have moved through the vadose zone and formed a floating layer on the water table. Because the primary potential source of contamination (fuel storage tanks) has been removed, the potential source is now any residual contamination in the soil that could be released to the groundwater. Additional volatilization could occur, if the area is excavated.

Potential contaminants of concern are fuel-related organic compounds (BTEX) and lead. A variety of laboratory analyses for geotechnical and chemical parameters have been performed using different methods. Investigations of the site were carried out in 1991 (Shannon & Wilson 1991), in 1992 (Shannon & Wilson 1992), in 1993 (U.S. Air Force 1994b), and in 1994 (U.S. Air Force, 1995a,b,c). Chemical analyses are summarized in Table 5.13.

### 7.9.1 Soil Contamination

Two soil-gas survey methods were used during June 1993 to identify heavily contaminated soil for use in an ex situ remediation demonstration. Soil samples were collected from 17 test pits, using a backhoe, and analyzed for VOCs, TPH-G. TPH-D. and lead.

The most contaminated soils were found in the central part of ST58 at the 1- to 1.7- and 2.7- to 3-m (4- to 5.5- and 9- to 10-ft) sampling intervals. Soils were contaminated with benzene and TPH (gasoline) in significant concentrations, as shown in Table 7.8. This soil was subsequently excavated for a composting demonstration. Significantly lower BTEX concentrations were found in the 0.1-m (0.5-ft) samples (for example, benzene,  $52 \mu g/kg$  to detection limit). No chlorinated hydrocarbons were detected in any of the samples (Liikala and Evans 1995).

Table 7.8. Surface and Subsurface Soil Contaminants Greater Than Screening Levels, ST58

Chemical	Detection Limit (µg/kg)	Analyzed/ Detected	Concentration Range (µg/kg)	Location of Maximum
Benzene	10	51/26	12 - >90,100	8E/5 ft
ТРН-G	30,000	60/9	26,000 - 19,900,000	7E/5 ft

Source: Liikala and Evans, 1995; p. 3.21.

## 7.9.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for ST58 are summarized in Table 7.9. Groundwater samples were collected and analyzed for the constituents listed in Table 5.13. A summary of the sample concentrations for ST58 can be found in Appendix A and in Liikala and Evans (1995).

Investigations conducted during the fall of 1991 and winter of 1992 indicated that benzene concentrations in the groundwater downgradient of source area ST58 were greater than the drinking water standard. Toluene, ethylbenzene, and xylene were also detected.

In April 1993, groundwater samples were collected from the same wells. Benzene concentrations above the drinking water standards were again found in several locations. Toluene and total xylenes were also detected. Gasoline-range petroleum hydrocarbons at a concentration of 200  $\mu$ g/L were detected in one well. Diesel-range petroleum hydrocarbons, ranging in concentrations from 0.1 to 99  $\mu$ g/L, were detected in 9 of 12 wells. Samples were analyzed for lead and it was found in concentrations above the EPA action level (15  $\mu$ g/L) in all of the source area wells.

A follow-on investigation was conducted in the fall of 1994. Lead and benzene concentrations in the groundwater remain above the drinking water standard. The extent of benzene and lead contamination is shown in Figures 7.15 and 7.16.

All contaminants detected at ST58 were in aqueous form. No floating product was encountered, but it is likely that past fuel releases resulted in transient product plumes that have since dissipated after the service station was closed and the buried supply pipeline removed from service.

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
Benzene	5	13/6	3.7 -180	58MW08
Gasoline	2000	14/1	261,000 - 261,000	58MW09
Lead	5	14/13	35 - 180	58MW12

Jable 7.9. Groundwater Contaminants Greater Than Screening Levels,	ST58
--	------

# 7.10 Source Area SS64

SS64, the Transportation Maintenance Drum Storage Area, is located in the center of the developed portion of the base, just north of the Water Treatment Plant pond on Garrison Slough, on the west side of the Vehicle Maintenance Shop (Building 3213) (Figure 2.1). SS64 and an area south of Building 3213 (officially part of SS61) were used for an unspecified number of years as a storage and staging area for drums containing hazardous materials and waste. Drums collected from routine base operations and cleanup were stored at this area until they were shipped for disposal.

In 1986, the EPA found 550 208-L (55-gal) drums labeled methanol, paint waste, lacquer, thinners, oils, acids, and asphalt at SS64. These drums were in poor condition and leaking. In 1987, the EPA reported that additional drums had been brought to SS64 since the 1986 survey. Approximately 160 drums were labeled as paint materials, and 300 drums were labeled as solvents including methyl ethyl ketone and cyclohexylamine. Six 322-L (85-gal) overpack drums were labeled as PD-680 solvent and paint thinner. Other inspection reports indicated that drums that were in other source areas during previous compliance inspections had been transferred to SS64. In 1989, the approximately 900 drums from SS64 were disposed of by base Hazmat personnel. At that time, stained soils were excavated and removed from SS64.

### 7.10.1 Soil Contamination

The drums stored at SS64 leaked and were the source of soil contamination. The locations of the leaking drums are not known, but they are assumed to have been located randomly, each producing a discontinuous spot of contamination. The contents of drums leaked onto the soil and then either evaporated, leached into the subsurface via precipitation, or remained on the surface soils as stains. In 1989, the last drums were removed, eliminating that source. At that time, surface spills were cleaned up, contaminated soil removed, and the area was graded and gravel added to prepare it as a parking lot. Because of the graveling and grading activities at SS64, surface soil sampling was not performed. Because contaminated soils were removed in 1989, subsurface soil sampling was not performed.

### 7.10.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for SS64 are summarized in Table 7.10. Samples were collected and analyzed for the constituents listed in Table 5.14. A summary of the sample concentrations for SS64 is presented in Appendix A.

The objective of the sampling was to check groundwater beneath the area where drums were stored for constituents that were contained in the drums. Groundwater was sampled to determine if drum contents reached the water table. The only contaminants of concern that were detected were TCE, tetrachloroethene, and trans-1.2-dichloroethene. These constituents were detected only in Well 64MW01, in the downgradient position, at concentrations less than MCLs, but greater than the risk-based screening levels. Because of the close proximity of SS61 where elevated concentrations of these constituents were detected, this measurement may indicate that groundwater in this area is impacted by SS61.

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
Tetrachloroethene (PCE)	0.5	4/2	0.82 - 1.7	64MW01
Trichloroethene (TCE)	0.5	4/2	0.6 - 2.7	64MW01

Table 7.10. Groundwater Contaminants Greater Than Screening Levels, SS64





OUs 3. 4, and 5 Record of Decision

Eielson AFB







Eielson AFB





s OO ω 4 and ι Ω Record of Decision

7.17

September 1995

Eielson AFB





September 1995

7.18

FINAL

Eielson AFB

OUs 3, 4, and 5 Record of Decision





FINAL

7.19

September 1995

ω 4 and сŋ Record of Decision

Eielson AFB

sno



















÷.

Eielson AFB

Measured

(mg/kg)

Not detected

0.1-1 mg/kg

uncertain.

100 METERS

sno ω 4 and

σ

Record of Decision

Eielson AFB





Figure 7.9. SS35, 4,4'-DDT in Soils and Sediments Depth = 0.00-0.15 meters (August 1992)

7.23





OUs 3, 4, and 5 Record of Decision



Figure 7.11. SS39, Diesel/Kerosene in Soils and Sediments Depth = 0.00-0.15 m (July 1992)

OUs 3. 4. and 5 Record of Decision

Eielson AFB

7.25

FINAL

September 1995



.

BOUNDARY OF CLEARED AREA SS39 FOR SS39 Well locations are surveyed. Locations of all other features are approximate. 0 0 Sample locations without flags have no detectable constituent. 39M03 θ ROAD 39M05 HAUL 0 θ Measured [31] concentration DIRT 63 (mg/kg)🚱 Зэмојі Groundwater 0 0 monitoring well 2.7 Soil/Sediment Θ sample 3.2 X 1.0-10 mg/kg 39M04 0 >10 mg/kg39M02 Extent of contamination GRAVEL ACCESS ROAD 0 uncertain. Diesel/Kerosene in Soils and Sediments MACHENC DECUMATION SE: EAST Depth = 0.9-1.2 meters BOUNDARY OF AVEHACE AND THE CHANGE & 28 WEST CLEARED AREA July, 1992 30 METERS FOR SS63



OUs 3, 4, and 5 Record of Decision

Eielson AF8

7.26



200

ω

4

and

თ

Record of Decision

7.27

September 1995

Figure 7.13. SS39, Diesel/Kerosene in Soils and Sediments Depth = 2.1-2.4 m (July 1992)

Eielson AFB



Figure 7.14. SS63, Diesel/Kerosene in Soils and Sediments Depth = 2.1-2.4 m (July 1992)

7.28

FINAL

OUs 3, 4, and 5 Record of Decision





OUs 3, 4, and 5 Record of Decision

Eielson AFB

7.29

September 1995







Figure 7.16. ST58, Lead in Groundwater (April 1993)

ĥ,

Eielson AFB

# 8.0 Nature and Extent of Contamination at Operable Unit 5

This section discusses the nature and extent of groundwater and soil contamination at source areas in Operable Unit 5 (OU 5). OU 5 includes LF02, LF03, LF04; and LF06, which are landfills. FT09, a former fire-training area, is located on LF03 and included in OU 5.

## 8.1 Contaminants of Concern

The contamination detected and characterized at the OU 5 source areas is primarily a result of land disposal practices. In addition, contamination caused by the burning of aviation fuels is present at FT09. Contaminants of concern in groundwater at LF03/FT09 include the volatile organic compounds (VOCs) benzene and vinyl chloride. Contaminants of concern in soil include total petroleum hydro-carbons (TPHs) and polycyclic aromatic hydrocarbons (PAHs) resulting from the incomplete combustion of fuel used in fire-training exercises. Contaminants of concern at LF02 and LF06 include metallic and nonmetallic elements.

# 8.2 Source Area LF02

LF02 is an abandoned, approximately 6-acre (2.4 hectares) landfill located about 0.8 km (250 ft) northwest of the intersection of Manchu Road and Gravel Haul Road on the banks of French Creek, a tributary of Moose Creek (Figure 2.1). LF02 is about 122 m (133.4 yd) west of Bear Lake, a 370-m<sup>2</sup> (3980-ft<sup>2</sup>) surface water body. A gravel road provides access from Gravel Haul Road. LF02 boundaries were located through a surface electromagnetic survey (HLA 1989). The minimum distance from the site to French Creek is about 6 m (6.6 yd) (Figure 8.1).

LF02 was used as the primary base landfill from 1960 to 1967 and received domestic and base operations waste. Refuse was burned from 1960 until 1964, when this practice was discontinued. Burial of refuse continued until 1967, when the landfill was closed and capped. Capping material included soil and fly ash from the base power plant. The cap was graded and has been maintained since closure; it appears to be in good condition in the center of the landfill. At the edges, debris is scattered on the surface, including miscellaneous household items, paper trash, metal and glass fragments, scrap lumber, and construction debris. The area has been used as a snow removal disposal area, and for at least 9 months of the year, a layer of scraped snow and ice overlies the cap. This snow layer is piled 6- to 9-m (20- to 30-ft) deep on the margins of the landfill and somewhat less near the center. During annual breakup, the snow melts and an unknown amount of water infiltrates the landfill.

#### 8.2.1 Soil Contamination

No soil contaminants were present in LF02 soils in excess of EPA risk-based screening levels or background concentrations. A summary of sample concentrations is presented in Appendix A. Soil samples were collected and analyzed for the constituents listed in Table 5.15.

## 8.2.2 Groundwater Contamination

No groundwater contaminants were present in excess of EPA risk-based screening levels or background concentrations. Groundwater samples have been collected and analyzed for the constituents listed in Table 5.15. A summary of sample concentrations is presented in Appendix A.

# 8.3 Source Area LF03/FT09

LF03 is located east of the south end of the runway and north of the refueling loop (see Figure 2.1). FT09 is located within the west-central part of LF03. LF03 and FT09 are approximately 39.5 hectares (98.8 acres). LF03 was used as the main base landfill from 1967 to 1987. The landfill received household garbage, scrap lumber and metal, construction debris, concrete slabs, empty cans and drums from flightline industrial shops, and possibly waste oils, spent solvents, and paint residues and thinners. The six trenches on the east side of the landfill received most of the waste after 1980. The landfill excavation reportedly extended below the water table. The landfill boundaries were established by geophysical surveys performed by HLA in 1988 and 1989. Potential sources of contamination include the leaching of landfill debris by groundwater, and subsurface soil and groundwater contamination by leaks from buried drums or cans. Because of the large number of potential contaminant sources on a landfill, samples were analyzed for a variety of constituents, as shown on Table 5.16.

Fire-training exercises at FT09 occurred from 1955 to 1989 and involved burning waste oils, contaminated fuels, and spent solvents. Procedures used during the most recent exercises involved saturating the ground with approximately 20,000 L of water, applying fuel, burning the fuel for 30 seconds, and extinguishing it with foam. A mixture of 2000 to 4000 L of clean JP-4 and up to 800 L of contaminated JP-4 was used in these exercises. Fire-training exercises were conducted at least twice per month. Expected contaminants from FT09 include fuel-related compounds (BTEX) and chlorinated solvents. The mock jet was removed in 1994. A new lined fire training facility is currently under construction.

## 8.3.1 Soil Contamination

Soil contaminants greater than EPA risk-based screening levels or background concentrations for LF03/FT09 are summarized in Table 8.1. A summary of sample concentrations for LF03/FT09 is presented in Appendix A. Soil samples were collected and analyzed for the constituents listed in Table 5.16.

Sample results delineated several areas of soil with TPH concentrations greater than 100 mg/kg. An area of surface contamination approximately 30 by 60 m (98.4 by 196.9 ft) was delineated near a mock-up jet fighter used for fire-training exercises. An area of subsurface contamination was identified west of the mock-up jet, at a depth of approximately 1 to 2 m (3.3 to 6.6 ft) bls, which is just above the water table. The subsurface TPH contamination may be the result of fire-training activities (such as spilled fuel) or leakage from landfill debris.

As a result of the TPH survey, samples were collected adjacent to the former site of the mock-up jet. PAHs were detected in all of these samples at concentrations ranging from 75 to 410 mg/kg. The greatest number of PAHs was detected in surface soil sample 03SYS05, at the west end of the mock-up jet. In addition, kerosene was detected in all but one sample. Naphthalene and 2-methyinaphthalene,

Chemical	Detection Limit (µg/kg)	Analyzed/ Detected	Concentration Range (µg/kg)	Location of Maximum
1.2 Dichloroethylene	100	10/1	300 - 300	03M02
Benzo(a)anthracene	660	9/2	84 - 181	03SYS05
Benzo(a)pyrene	660	9/3	79.1 - 200	03SYS05
Benzo(b)fluoranthene	660	9/3	83 - 250	03\$Y\$05
Benzo(ghi)perylene	660	9/1	200 - 200	03\$Y\$05
Indeno(1,2,3-cd)pyrene	660	9/1	200 - 200	03SYS05
Kerosene	N/A	10/10	48 - 1,590,000	03\$\$09

Table 8.1. Soil Contaminants Greater Than Screening Levels,	LF03/FT09
---	-----------

which are commonly found in JP-4 fuel, were detected in two samples. The presence of these compounds may be attributed to incomplete combustion of fuel used in fire-training exercises.

#### 8.3.2 Groundwater Contamination

Groundwater contaminants greater than EPA risk-based screening levels or background concentrations for LF03/FT09 are summarized in Table 8.2. A summary of the groundwater sample concentrations is presented in Appendix A.

Chemical	Detection Limit (µg/L)	Analyzed/ Detected	Concentration Range (µg/L)	Location of Maximum
1,4-Dichlorobenzene	10	79/2	64 - 82	03M08
Benzene	1	22/4	1.7 - 20	03M08
Tetrachloroethane (PCE)	0.5	22/1	53 - 53	03M08
Trichloroethane (TCE)	0.5	22/6	0.64 - 150	03M08
Vinyl chloride	0.5	22/5	0.54 - 17	03M08

Table 8.2.	Groundwater	Contaminants	Greater	Than Screening	Levels, LF03/FT09
------------	-------------	--------------	---------	----------------	-------------------

Sampling results, prior to 1994, indicated the presence of VOCs in groundwater at LF03/FT09 (see Figure 8.2). Leakage from a subsurface JP-4 fuel pipeline approximately 76 m (83 yd) upgradient of Well 03M13 had been cited as a possible source of benzene contamination; however, no toluene, ethylbenzene, or xylene were detected in samples with the highest benzene concentrations. No floating product was encountered. The presence of benzene without the other fuel-related VOCs is characteristic of the leading edge of a plume from an upgradient source because benzene is more mobile than the other constituents. Another possibility is the benzene plume may have originated from a source in the fire-training area. Solvents were also detected in groundwater at LF03, prior to 1994.

Groundwater sampling results in 1989 delineated plumes of trichloroethane (TCE) near Well 03M08 and vinyl chloride near Well 03M01. TCE was not detected in samples collected downgradient of Well 03M08 and, therefore, is probably caused by a localized leak from landfill debris. The plume appeared to have originated near Well 03M01 and extended to the north, toward Garrison Slough. These solvents were most likely derived from leaking containers of spent solvent in the landfill. Low concentrations of solvents were also detected in the 1992 groundwater samples.

By the time of the sampling event in August 1994, benzene and chlorinated solvents were below MCLs (and in many cases below detection limits) in all wells in LF03/FT09 and SS37, except for Well 03M08 (see Figures 8.2 through 8.5). This well contained relatively high concentrations of a number of contaminants (see Table 8.2), including 4-methyl phenol (p-cresol), dichlorodifluoromethane (Freon-11), and trichlorofluoromethane (Freon-12). Analyses were not reported for Well 03M08 in a previous investigation by HLA (1989) because of analytical difficulties. Well 03M08 was not sampled in 1992. This well is located in the eastern portion of LF03 in the area of the waste trenches, which were used after 1980.

Contaminated groundwater from LF03/FT09 has not migrated beyond the landfill boundaries.

## 8.4 Source Area LF04

LF04 is located approximately 5 km (3 mi) east-northeast of the south end of the runway (see Figure 2.1) and covers an area of greater than 100,000 m<sup>2</sup> (24.7 acres) (HLA 1989). LF04 reportedly received general refuse, small quantities of waste oil and spent solvents, and possibly small amounts of munitions and spent cartridges. A number of laboratory analyses were performed on groundwater and soil samples from LF04. The Army originally used the site to store ammunition in bunkers. Access to LF04 is currently restricted because of its designation as an emergency ordnance demolition area, where small munitions are incinerated in a burning kettle. Geophysical surveys conducted by HLA in 1988 and 1989 established the approximate boundaries of the landfill.

No significant surface or subsurface contamination has been detected at LF04. Small areas of TPH have been detected at concentrations of less than 250 mg/kg. No TNT or RDX compounds, resulting from ordnance activities, have been detected. Under a separate federal program, the Air Force has submitted a closure plan for the ordnance area at LF04 under RCRA, Section 3008(a).

### 8.5 Source Area LF06

LF06, the old landfill, is located near the central power plant just south of the power plant cooling pond on the eastern side of the main developed portion of Eielson AFB (Figure 2.1). The landfill is approximately 348 m (1000 ft) north of Hardfill Lake (Figure 8.6). From 1959 to 1963, LF06 was used as a secondary landfill to the original base landfill (LF01, used in the 1950s) and the old base landfill (LF02, used from 1960 to 1967). The landfill has been covered with a loose sand cover that is maintained by periodic grading.

LF06 reportedly received large construction and metal debris and smaller quantities of general refuse (such as wood, empty drums, and paint containers) from the flightline industrial shops (CH2M Hill 1982). Because most of the base refuse would have been disposed at other sites, only small quantities of waste paint, thinners, and spent solvents were assumed to have been present as drum residuals at LF06. Also, other liquid wastes, such as spent solvents, would likely have been disposed only in 1959, because beginning in 1960, refuse disposed of in LF02 was first burned and then buried. Therefore, only small quantities of these wastes are assumed to have been disposed in LF06.

#### 8.5.1 Soil Contamination

No contaminants were present in LF06 soils in excess of EPA risk-based screening levels or background concentrations. A list of analyses for samples collected at LF06 is presented in Table 5.18.

#### 8.5.2 Groundwater Contamination

The results of a 1994 investigation confirmed the findings of previous investigations; the landfill does not appear to be adversely impacting groundwater in the vicinity. No VOC, SVOC, pesticide, PCB, or TPH constituents were detected in groundwater samples from the four wells sampled.

No contaminants were present in LF06 groundwater in concentrations in excess of EPA risk-based screening levels or background concentrations. A list of analyses for groundwater samples collected at LF06 is presented in Table 5.18. A summary of sample concentrations is found in Appendix A.



Eictson AFB

OUs 3, 4, and 5 Record of Decision



١,

September 1995

8.6

FINAL

÷.



Figure 8.2. LF03/FT09 Benzene in Groundwater, 1989


Figure 8.3. LF03/FT09 Benzene and Other Contaminants in Groundwater, 1994



Figure 8.4. LF03/FT09 Phenol and PC in Groundwater, 1994



Figure 8.5. LF03/FT09 Miscellaneous Contaminants in Groundwater, 1994



Figure 8.6. Map, LF06

# 9.0 Summary of Operable Unit Risks

## 9.1 Human Health Risks

The objective of the baseline risk assessment (BLRA) was to characterize the current and potential threats to human health and the environment. The results helped establish remedial action objectives necessary to develop remedial alternatives in the feasibility studies (National Contingency Plan 40 CFR 300). The BLRA evaluated risks at Eielson now and into the future. Consistent with EPA guidance, the BLRA assumes that without cleanup measures, the identified source areas will remain in their present states of contamination.

The Alaska Department of Environmental Conservation (ADEC), U.S. Air Force, and the U.S. Environmental Protection Agency (EPA) have agreed to follow the guidelines for federal facilities under Comprehensive Environmental Response. Compensation, and Liability Act (CERCLA or Superfund), 40 CFR Part 300, Section 120. The health risk assessments are based on the following EPA guidance:

- Risk Assessment Guidance for Superfund, Volume 1, Human Health Evaluation Manual, Part A (EPA 1989a)
- Exposure Factors Handbook (EPA 1989b)
- EPA Region 10 Supplemental Risk Assessment Guidance for Superfund (EPA 1991a)
- Standard Default Exposure Factors (EPA 1991b)
- Guidance on Risk Characterization for Risk Managers and Risk Assessors (EPA 1992).

## 9.1.1 Source Evaluation Report Areas

Contamination within the Source Evaluation Report (SER) sites, LF01, WP32, and DP55, was analyzed by a conservative screening risk assessment that compared the maximum concentration of each contaminant detected at the source area to a conservative risk-based concentration using EPA standard default exposure factors for a residential scenario. The target risks used for the conservative screening were chosen based on the lower end of the  $10^{-4}$  to  $10^{-6}$  risk range specified in the NCP. This screening approach assumes that if no single sample exceeds a concentration representing a human health risk concern, total exposure to the contaminant from the source area will not be of concern. Based on this assumption, no further action is required for areas where maximum concentrations detected were  $\leq 10^{-6}$  cancer risk for water,  $\leq 10^{-7}$  cancer risk for soil, and  $\leq 0.1$  hazard quotient. No contamination at source areas LF01, WP32, or DP55 exceeded the screening levels; thus, further risk assessment was not necessary for these SER areas.

#### 9.1.2 Operable Unit 3, 4, and 5 Source Areas

As presented in the three previous sections of this ROD, the OUs were grouped according to their use and the chemical contaminants detected:

- OU 3 -- cleaning and maintenance of equipment (DP44, WP45, ST56, SS61), refueling (SS57)
- OU 4 -- pesticide storage and mixing (SS35), mixing and storage of asphalt (SS35, SS36, SS37, SS39/SS63, SS64), fuel tank farms (DP25, ST27), auto refueling (ST58), waste effluent ponds (WP33)
- OU 5 -- former landfill areas, including a fire-training area (LF02, LF03/FT09, LF04, LF06).

# 9.2 Site Data, Screening, and Identification of Contaminants of Potential Concern

The environmental data used in the BLRA was collected in the CERCLA Remedial Investigation and Feasibility Studies (RI/FSs) (SAIC 1988, 1989b; HLA 1989, 1990, 1991; U.S. Air Force 1993a,b,c). Fuel- and solvent-related organic chemical contamination of the groundwater and soil was found at specific source areas that resulted from the dispensing or use of

- petroleum, oil, or lubricants
- industrial (cleaning) solvents (electronic and other equipment repair)
- paint products
- · asphalt cement materials
- acids and bases (with storage batteries)
- other miscellaneous contaminants.

U.S. Air Force sampling results collected in 1991 and 1992 established background concentrations for inorganic chemicals in soil and water media. The elevated concentrations of inorganic chemicals in soil and water, including arsenic and manganese, are considered intrinsic in the Fairbanks region and are not considered to relate to base activities. Nevertheless, risks were determined for all measured chemicals that are not considered to be common laboratory contaminants (such as, acetone, chloroform, methylene chloride, and phthalates).

The BLRA used data collected in 1992 to 1994 for groundwater, and data from 1986 to 1994 for soils and other media. Groundwater data collected before 1992 is not considered to be representative of current conditions. Concentrations of contaminants in the soils are not believed to have changed significantly between 1986 and 1994.

For some source areas and for several media, it was necessary to estimate chemical concentrations using fate/transport modeling, either because samples were not collected or because concentrations at

points of human contact (exposure) were not available. For example, specific chemicals in garden vegetables were not measured, but have been estimated from appropriate soil concentration data.

Table 9.1 lists all contaminants of potential concern that were analyzed with the mean, maximum, and reasonable maximum exposure (RME) concentrations that were used in the assessment. Data provided in Table 9.1 include the values taken as half the detection limits. The RME value is either the upper 95th confidence interval (CI) of the mean or the maximum value, if the CI exceeded the maximum concentration value. If measured concentration data were not available, the RME values were estimated using modeling, as described in the BLRA. The RME value was used in the BLRA to estimate human intake of contaminants for risk assessment. The exposure point concentration (the upper 95th confidence interval on the mean or the maximum value, as appropriate) is the maximum concentration value that is expected for human contact.

All positively identified chemicals were screened to determine whether their concentrations and toxicity potentials exceeded EPA-established risk levels (EPA 1991b). Any chemicals with maximum concentrations that triggered cancer risk levels greater than one in  $10^{-6}$  for water or one in  $10^{-7}$  for soils, or noncancer hazard quotients (HQs) greater than 0.1 are presented in Tables 9.2a and 9.2b. Screening levels were set below toxicity thresholds to ensure the combined actions of chemicals are neither excluded in the calculations nor underestimated in the determination of net adverse health potentials in humans.

The chemicals of potential concern that result from the screening by source area and media are presented in Table 9.2a for carcinogens and Table 9.2b for noncarcinogens. The classes of chemicals detected in OUs 3, 4, and 5 were fuel-related [benzene, toluene, ethylbenzene, and xylene (BTEX), kerosene, and lead]; residuals from asphalt paving [kerosene, polycyclic aromatic hydrocarbons (PAHs)]; pesticides; chlorinated cleaning solvents; and paint-related compounds.

The screening process described previously allowed for identification of contaminants of concern above the EPA-established risk levels. This screening process was not used, however, to limit the number of contaminants carried through the quantified risk assessment; all of the contaminants listed in Table 9.1 as contaminants of potential concern were retained for further risk evaluation. All data was available in electronic form; therefore, it was a more straightforward process to run all data through the computerized spread sheets than to revise the database and remove contaminants that do not contribute significantly to the total risk.

## 9.3 Exposure Assessment

#### 9.3.1 Conceptual Site Model

An exposure assessment evaluates the potential for human contact with chemicals of concern present at, or migrating from, a source area. At Eielson, human exposures occur as a result of contact with organic chemicals (including pesticides, asphalt, and fuel mixtures) and inorganic chemicals.

The goal of the assessment is to appropriately combine the exposure point concentrations with land use/population scenarios to calculate chemical intake or dose due to human contact with contaminants. The calculated doses were then combined with toxicity data to characterize health risks.

FINAL

Chemicals may migrate along pathways from sources to points of human contact (such as exposure). Included as part of the pathway assessment are the contaminant sources, the environmental media (such as soils, groundwater, and air), the routes of contact (such as oral ingestion, dermal contact, and inhalation), and the contact itself. If contact is not made, exposure does not occur and the pathway is not complete. The completed pathways of concern at specific source areas within the OUs were evaluated quantitatively to provide numerical estimates of potential exposures, which were then used to estimate human health risks.

The exposure cases or scenarios evaluated for a given source area depend on the populations potentially exposed and on the current and potential land use at Eielson. In this assessment, a residential scenario is evaluated to consider potentially exposed future users who could spend 30 years on-site. This scenario, which includes children, who may be the most sensitive subpopulation to be exposed, is generally the most conservative case. The residential exposure scenario is evaluated for OUs  $3_{12}A$ , and 5 to determine potential risks in the unexpected event of base closure. The current land use as a military base is not expected to change in the foreseeable future. Other populations of concern are current military and civilian workers; and commercial workers who could be exposed in the future, if the base is converted to civilian use.

The conceptual model of pathways and exposures evaluated for current use, future use, and recreational scenarios is outlined in Figure 9.1.

#### 9.3.2 Exposure Factors

To ensure consistency in the risk assessment process, the EPA's Superfund program has developed standard default exposure factors for selected exposure pathways. Standard default factors are considered the most appropriate exposure parameter values for risk calculations, and EPA stipulates they should be used in BLRAs, unless alternate or site-specific values are clearly justified by supporting data (EPA 1991a, 1991b). The rationale for each standard default value is discussed in Human Health Evaluation Manual, Supplemental Guidance: Standard Default Exposure Factors (EPA 1991a), and the values are listed in Region X guidance (EPA 1991b). However, the Eielson location near the Arctic Circle required the use of site-specific factors to compensate for extended periods with snow cover and frozen ground. In addition, to accommodate a possible subsistence lifestyle, the assessment considered the ingestion of fish caught adjacent to two source areas (SS35 and SS37), vegetables potentially grown at the source areas evaluated for future residential use, and other site-specific parameters. Table 9.3 lists both the EPA default and the Eielson site-specific exposure factors used in the assessment. When the exposure factors are applied to standard risk equations, intake factors are calculated; the table includes these results. When the intake factor is multiplied by exposure point concentrations, a dose is calculated. Specific doses are used with toxicity factors (for each chemical) to estimate human health risk.

Chronic exposure periods (9-, 12-, 25-, or 30-year durations) that typify high-end recreational and average residential and worker exposures provide information for assessing human cancer risks and other chronic adverse effects. Cancer risks are considered for 70-year lifetimes. A 3-year exposure period is used for the current at-risk child in a recreational setting, and a 6-year exposure period is used for a child in a residential setting. These periods are consistent with the typical on-base stay of military personnel and dependents. At the levels of contamination existing in the source areas, acute poisoning and other short-term effects are considered unlikely.

The potential current- and future-use exposures assume above-average intake of contaminants that are used to calculate chemical (contaminant) intake by humans. Whether the contaminants are

measured from field sampling data or estimated using fate/transport modeling, the upper 95th confidence limit on mean (Gilbert 1987) RME concentrations represents conservative exposures expected for a site under current or future conditions. Non-detect values were assigned a concentration of onehalf the detection limit (EPA 1991b) and future-use exposures were based on the surface soil or the subsurface soil RME, whichever is larger.

## 9.4 Evaluation of Lead Contamination

Lead contamination was identified at source areas DP25, ST27, and ST58. Source area DP25 is the E-6 Fuel Storage Tank area, ST27 is a fuel-storage tank farm, and ST58 is the site of the old Quartermaster Service Station, which has been removed.

Exposure to inorganic lead is treated separately from other contaminants for purposes of determining protective levels. In July 1994, EPA issued OSWER Directive # 9355.4-12, <u>Revised Interim</u> <u>Soil Lead Guidance for CERCLA Sites and RCRA Corrective Action Facilities</u> to establish an approach to determine protective levels for lead in soil. This directive identifies the Integrated Exposure Uptake Biokinetics (IEUBK) model as the most appropriate and applicable method for assessing and managing risks from lead in soils (Bennett 1990; EPA 1991c, 1994a).

The EPA guidance set residential and industrial screening levels for lead at 400 mg/kg and 1000 mg/kg, respectively. The screening level serves as an indicator that additional study may be appropriate. With the exception of one soil sample in 1986 of 870 mg/kg at DP25, the E-6 Fuel Storage Area, soil lead levels found at all sites reviewed at Eielson AFB are below the 400 mg/kg screening level. The E-6 Fuel Storage Area is an industrial area and lead levels do not exceed the industrial screening level. No further study is necessary, given that special circumstances are absent.

Presently, no MCL exists for lead. In lieu of an MCL, EPA has established a lead action level of 15  $\mu$ g/L for water. Groundwater lead levels found at the sites reviewed at Eielson AFB were compared to this action level. Based on this screening, lead was identified as a contaminant of concern in groundwater for source areas DP25, ST27, and ST58.

## 9.5 Toxicity Assessment

Where available, the EPA-authorized chemical-specific toxicity factors are the reference values used to express cancer risk levels and noncancer effects. The available factors, listed in Integrated Risk Information System (IRIS) and Health Effects Assessment Summary Tables (HEAST), have been corrected for exposure times, animal-to-man extrapolations, and others. Potential cancer risks are quantified through the use of dose-response slope factors. The cancer potency or risk is characterized as an upper-bound estimate, meaning the true risk to humans is not likely to exceed the estimate and may even be lower (EPA 1989a). The reference dose (RfD) is used to evaluate toxic effects from noncarcinogens and estimates the maximum daily exposure to human populations that is not likely to result in an appreciable risk of adverse effects. The critical toxicity values used for the major 20 contaminants of concern are shown in Table 9.4a. Table 9.4b shows the additional toxicity factors that were needed in the risk assessment for screening of chemicals and risk quantification.

## 9.6 Risk Characterization

The carcinogenic risk from exposure to a chemical is described in terms of the probability that an exposed individual will develop cancer over a lifetime. This value is a function of the estimated chronic daily intake (dose) and the slope factor for the chemical. The slope factor converts the estimated dose, averaged over a 70-year lifetime of exposure, to a risk for an individual (EPA 1989a):

Cancer risk = Dose x SF 
$$(9.1)$$

where

risk = n excess probability (such as  $2 \times 10^{-3}$ ) of an individual developing cancer as a consequence of chronic exposure

Dose = chronic daily intake averaged over 70 years, in  $mg/kg \cdot day$ 

 $SF = slope factor, in mg/kg \cdot day$ 

The estimated carcinogenic risks from each contaminant and pathway are added to determine total additional risk resulting from site-specific contamination. This additional risk is above the rate of cancer in the United States from all causes which is estimated to be as high as 1 in 4. The National Contingency Plan (40 CFR 300) defines acceptable risk from Superfund site as additional cancers due to site-specific contamination in the range of 1 chance in 10,000 (1 x  $10^{-4}$ ) to 1 chance in 1,000,000 (1 x  $10^{-6}$ ).

The risk of noncarcinogenic effects from contaminant exposure is expressed in terms of the hazard quotient (HQ). The HQ is the ratio of the estimated average daily dose (ADD) (for an appropriate period of exposure) to the RfD. The HQ for chronic effects is expressed by the following equation:

$$HQ = ADD/RfD$$
(9.2)

where

 $ADD = average daily dose (in mg/kg \cdot day)$ 

 $RfD = reference dose for chronic exposure (in mg/kg \cdot day).$ 

An HQ that is greater than 1.0 indicates a potential for adverse health effects. Although the incidence or severity of those effects is likely to increase as the HQ increases, the dose-response rates can differ among contaminants and health effects. Thus, an HQ value of 1.0 does not define a sharp distinction between no effects and adverse effects, but rather a transition to the potential for adverse effects.

Exposures to a single contaminant from different pathways, or exposures to multiple contaminants, can act cumulatively to produce adverse health effects, even if all individual exposures are below RfD values. Therefore, a hazard index is calculated by summing HQs across contaminants and pathways. A hazard index that exceeds 1.0 indicates a potential for adverse health effects, under the assumption that risks are additive across chemicals and pathways.

A summary of cancer risks and noncancer effects is found in Table 9.5. For each source area, the summary table shows the probable cancer risks that are expected to be greater than a rate of 1 in 1,000,000 or noncarcinogenic effects with a HI greater than 0.1. In these tables, organic and inorganic chemical risks have been summed, according to EPA guidance (EPA 1989a).

A breakdown of the routes of contaminant intake, the environmental media, the chemical, and the corresponding risk or effect level for all positively identified chemicals is shown in Tables 9.2a and 9.2b. These tables identify the contaminants of concern for each pathway along with its corresponding risks. The cumulative risks presented in Table 9.5 are somewhat higher than the cumulative risks presented in Tables 9.2a and 9.2b, which include only positively identified contaminants, because all contaminants analyzed are included in Table 9.5, regardless of whether they were detected. In Table 9.5, nondetect values were assigned a concentration of one-half the detection limit (EPA 1991b) and future use exposures were based on the surface soil or the subsurface soil RME, whichever is larger.

Cancer risks: In OU 3, the cancer risks at source area DP44 shown in Table 9.2a result from soil exposure to benzo(b)fluoranthene, benzo(a)anthracene, and dibenzo(a,h)anthracene, while groundwater exposure results from contact with trichloroethane. Benzene in the groundwater is the major contributor at SS57. At SS61, pentachlorophenol, trichloroethane, and gasoline are the risk contributors.

In OU 4, exposure to the groundwater via ingestion and inhalation arise from contact with benzene. At source area SS35, dermal exposure to DDT drives risks. The inadvertent ingestion and inhalation of traces of gasoline in the groundwater at ST58 provides elevated cancer risk probability.

At OU 5, 1,1-dichloroethylene, vinyl chloride, 1,4-dichlorobenzene, and trichloroethane may result in cancer risk from exposure to soil and groundwater media at LF03.

Noncancer risks: Table 9.2b shows noncancer risks that metals contribute are most of the noncancer risks at Eielson AFB. However, because they are assumed to be at background levels and not from Eielson sources, inorganic chemical are not considered as risk drivers. With the exception of DDT at source area SS35, all of the following noncancer OU 3, 4, and 5 exposures result from use of the groundwater.

In OU 3 at DP44, trichloroethane may cause adverse effects from drinking the groundwater. At WP45, ingestion of trichloroethane in the groundwater may cause ill effects. At SS57, toluene is the major contributor to noncancer effects. At SS61, trichloroethane contributes the bulk of the contaminant load.

For OU 4, the DDT levels at SS35 approach a HQ of 1 to become the major risk source.

At the OU 5 source area LF03, the major organics, trichloroethane and tetrachloroethylene, sum to less than 1.

## 9.7 Uncertainty Evaluation for the Human Health Risk Assessment

The calculated exposures and risks are based on numerous assumptions and parameter estimates that are themselves uncertain. These uncertainties affect both exposure estimates and toxicity values. Overall, the calculated exposures and upper-bound risks for the defined adverse effects are unlikely to result in underestimates; however, true risks could be lower than those calculated. Some assumptions and uncertainty factors associated with the BLRA include the following:

- The numerical estimates, based on standard EPA default and site-specific exposure factors for potential exposure and risk development for a quantitative risk assessment, cannot be precise values. Therefore, the uncertainty in the risk estimates is an important consideration, particularly for the Eielson sub-Arctic climate. Some risks, based on standard defaults, may be overestimated, despite compensation for location-specific factors.
- The baseline risk assessment presumes media concentration will prevail over time at their current levels. This assumption does not account for any natural biodegradation of contaminants with time. At LF03, most recent contaminant concentrations are considerably lower than the previously measured values. In this instance, it is likely that groundwater risks have been overestimated. Conversely, if the groundwater contaminant concentrations increase in the future due to additional leaching from a remaining source, the groundwater risks may be underestimated.
- At WP45, the maximum TCE concentration in groundwater has not been confirmed on resampling. This could be due to the disposal of snow on the area, resulting in the dispersal and dilution of groundwater TCE. Therefore, this value was not included in the risk assessment. If the value was included, the risk at WP45 due to TCE in groundwater would be 8E-4, doubling the total risk at WP45/SS57.
- Several of the basic assumptions used to develop appropriate exposure scenarios at Eielson AFB have little or no uncertainty associated with them. The existence of elevated concentrations of some contaminants within the various source areas is not in question, because these contaminants have consistently appeared in sampling studies, including data collected since 1986. Also, little uncertainty exists that specific source areas are contaminated by mixtures of aviation and motor fuels, cleaning solvents, and pesticides, even if the full extent of the contributions of each chemical is not completely defined. The primary land use in the source areas is military/commercial, a fact not expected to change in the foreseeable future (unless the base is closed). As a result, a worker exposure scenario is considered highly appropriate for current and future risk characterization at Eielson AFB.
- Exposures and risks to individuals (at a given source area) will differ because of different behavioral patterns, or genetic differences and sensitivity among individuals. Consequently, in addition to the uncertainty associated with a given estimate of risk for a particular population group, an underlying distribution will be present (on the risk level) that reflects alternative potentially exposed populations.
- Exposure point concentrations are reasonably certain for measured data. However, concentration data for sediments, surface water, and fish (for locations and chemicals not sampled in 1994) were modeled. The modeled data may be uncertain; for the Eielson region, it is difficult without site-specific fate/transport data to judge whether the resulting exposure point concentrations overestimate or underestimate risks.
- The background sampling for metals is thought sufficient to show that, with the exception of lead, these metals did not result from Air Force activities. For example, considerable information exists to verify that arsenic, which occurs at elevated concentrations throughout the region, is responsible for elevated risk rates near Fairbanks from the ingestion of groundwater. Sources of lead have been identified on the base; unfiltered lead concentrations were used to for comparison to current action levels.
- Toxicity factors associated with the 20 risk-driver chemicals shown in Table 9.6 are based on scientific information with measurable uncertainty. In contrast, many of the other toxicity factors used in the screening process have an unknown degree of uncertainty. Most of the carcinogenic

chemicals considered to be risk drivers are class A (sufficient evidence of carcinogenicity in humans), B (probable evidence), or C (limited evidence for humans). The uncertainty associated with the risk driver chemicals is not considered to either overestimate or underestimate risks.

• Some uncertainty may exist in the determination of the risks associated with human exposure to fuels. Although total petroleum hydrocarbons (TPHs) were measured, their risk was not estimated because they are fuel mixtures and not pure chemicals. Consequently, they lack the toxicity factors needed for risk analyses. Usually the BTEX chemicals (benzene, toluene, ethylbenzene, and xylene) are considered surrogates for TPHs, thus allowing for the estimation of fuel-related risks. The EPA has estimated preliminary factors for the gasoline, diesel, kerosene, and jet fuel mixtures, but these factors may not be applicable to the weathered fuel frequently found at Eielson AFB. Even though these preliminary factors have been withdrawn, they were used in the assessment. Surrogate and preliminary data that do not necessarily apply to the contamination found at Eielson Air Force Base were used to calculate fuel-related risks; this may result in overestimated or underestimated risks.

## 9.8 Environmental Risks

Table 9.6 summarizes the terrestrial habitat types and their primary wildlife species on Eielson - AFB.

No endangered or threatened species are resident to Eielson AFB. The American peregrine falcon (federally endangered) breeds within 50 miles of the base and the Arctic peregrine falcon (federally threatened) migrates to within 50 miles of the base. Bald eagles (federally threatened) are occasionally sighted on Eielson AFB.

No acute ecological risks were identified for source areas in Operable Units 3, 4, or 5. With the exception of SS35, the Asphalt Mixing and Drum Burial area, these areas do not appear to be acting as sources of surface water or sediment contamination. PCBs and pesticides, particularly DDT, were found in soil at source area SS35, which is located adjacent to Garrison Slough. Elevated body burdens of DDT were found in fish caught near SS35. Cumulative ecological risks at Eielson AFB are currently being evaluated under the Sitewide program. Preliminary conclusions indicate that SS35 may present reproductive risks to birds and mammals from ingestion exposure to PCBs and DDT. The Sitewide biological risk assessment addresses ecological risks from all areas on base.

## 9.9 Summary of Source Area Disposition

The nature and extent of contamination and its corresponding risk, as well as the potential future use of a given source area, were evaluated to determine which source areas required development of cleanup alternatives in the Feasibility Study. A summary of the dispositions for source areas is presented in Table 9.7.

[	T		T	Ī	ļ	Maximum	Reasonable
Source			Matrix		Average	Value	Maximum
Area	Analyte Measured		Code			Detected	Exposure
Operable l	Jnit 3						
	Groundwater			·····	<u></u>	<b></b>	·····
DP44	Banum	7440-39-3	Water	[ug/L	133.3	200	200
DP44	Benzene	71-43-2	Water	ug/L	1.5	5.3	2.12
DP44	Calcium	7440-70-2	Water	ug/L	38350	59000	59000
DP44	Chloride	16887-00-6	Water	ug/L	1567	2300	2300
DP44	Cis-1.2-dichloroethylene	156-59-2	Water	ug/L	28.24	260	61.8
DP44	Fluoride	7782-41-4	Water	ug/L	83.33	100	100
DP44	h-Chlorofluorobenzene	PPP-PP-P	Water	ug/L	10.06	12	10.74
DP44	Iron	7439-89-6	Water	ug/L	4137	6400	6400
DP44	Magnesium	7439-95-4	Water	ug/L	8017	12000	12000
DP44	Manganese	7439-96-5	Water	ug/L	2635	4600	4600
DP44	Nitrate	14797-55-8	Water	ug/L	300	700	700
DP44	p-Chiorofluorobenzene	000-00-0	Water	ug/L	9,413	10	9.764
DP44	Pocassium	7440-09-7	Water	ug/L	2283	3400	3400
DP44	Sodium	7440-23-5	Water	ug/L	3617	5600	5600
DP44	Sulfate	12808-79-8	Water	ug/L	9417	16000	16000
DP44	Tetrachioroethylene	127-18-4	Water	ug/L	0.2821	0.7	0.3391
DP44	Trans-DCE	156-60-5	Water	ug/L	1.429	5.4	2.081
DP44	Trichloroethene	79-01-6	Water	ug/L	182.5	2500	498.3
	Surface Soil					<b></b>	······································
DP44	2-Methylnaphthalence	91-57-6	Soil	ug/kg	193.3	550	550
DP44	Acenaphthene	83-32-9	Soil	ug/kg	970	2900	2900
DP44	Acenaphthylene	208-96-8	Soil	ug/kg	163.3	480	480
DP44	Aluminum	7429-90-5	Soil	ug/kg	8.47e+06	1.39e+07	1.39e+07
DP44	Anthracene	120-12-7	Soii	ug/kg	1840	5500	5500
DP44	Antimony	7440-36-0	Soil	ug/kg	4167	8000	8000
DP44	Barium	7440-39-3	Soii	ug/kg	1.24e+05	1.63e+05	1.63c+05
DP44	Benzo(a)anthracene	56-55-3	Soil	ug/kg	16080	48000	48000
DP44	Benzo(a)pyrene	50-32-8	Soil	ug/kg	6158	18000	18000
DP44	Benzo(b)fluoranthenene	205-99-2	Soil	ug/kg	70240	2.10e+05	2.10e+05
DP44	Benzo(g,h,i)perylene	191-24-2	Sail	ug/kg	4883	14000	14000
DP44	Cadmium	7440-43-9	Soil	ug/kg	827	906	906
DP44	Calcium	7440-70-2	Soil	ug/kg	5.61e+06	1.10e+07	1.10c+07
DP44	Chioroform	67-66-3	Soil	ug/kg	16.02	170	41.24
DP44	Chromium	7440-47-3	Soil	ug/kg	21700	27400	27400
DP44	Chrysene	218-01-9	Soil	ug/kg	7023	21000	21000
DP44	Cobalt	7440-48-4	Soil	ug/kg	8413	13700	13700
DP44	Copper	7440-50-8	Soil	ug/kg	21970	36500	36500
DP44	Dibenzo(a,h)anthracene	53-70-3	Soil	ug/kg	2183	6500	6500
DP44	Dibenzofuran	132-64-9	Soil	ug/kg	770	2300	2300
DP44	Fluoranthene	206-44-0	Soil	ug/kg	16350	49000	49000
DP44	Fluorenc	86-73-7	Soil	ug/kg	1277	3800	3800
DP44	Indeno(1,2,3-cd)pyrene	193-39-5	Soil	ug/kg	5228	15000	15000
DP44	Iron	7439-89-6	Soil	ug/kg	1.53e+07	2.58e+07	2.58e+07
DP44	Lead	7439-92-1	Soil	ug/kg	28800	47900	41240
	r .	,				1	

Table 9.1.	Contaminants o	f Potential	Concern.	Media.	and (	Concentration	Ranges
	- Containing into G	in rotonition	001100111,	mound,	Q1104 1	oon contration.	nungea

Ν.

·	T	T	1	l	1	Matumum	Reasonable
Source	1	ļ	Matrix		Average	Value	Maximum
Area	Analyze Measured	CAS Number	Code	Unics	Value	Detected	Exposure
DP44	Magnesium	7439-95-4	Soil	ug/kg	4.27e+06	7.12e+06	7.12e+06
DP44	Manganese	7439-96-5	Soil	ug/kg	2.62e+05	4.45c+05	4.45e+05
DP44	Methylenechloride	75-09-2	Soil	ug/kg	1244	11000	2884
DP44	Naphthalene	91-20-3	Soil	ug/kg	296.3	880	880
DP44	Nickel	7440-02-0	Sol	ug/kg	19070	34000	34000
DP44	Phenanthrene	85-01-8	Soil	ug/kg	12340	37000	37000
DP44	Potassium	7440-09-7	Soil	ug/kg	7.90e+05	9.40e+05	9.40e+05
DP44	Pyrene	129-00-0	Soil	ug/kg	10690	32000	32000
DP44	Sodium	7440-23-5	Soil	ug/kg	2.93e+05	4.10e+05	4.10e+05
DP44	Tetrachloroethylene	127-18-4	Soil	ug/kg	0.4222	i.8	0.7426
DP44	Toluene	108-88-3	Soil	ug/kg	5.611	34	12.26
DP44	ТРН	TPH	Soil	ug/kg	7.40e+05	7.00e+06	1.24e+06
DP44	Vanadium	7440-62-2	Soil	ug/kg	32630	54100	54100
DP44	Zinc	7440-66-6	Soü	ug/kg	79600	1.46e+05	1.46e+05
	Subsurface Soil						
DP44	2-Methylnaphthalenee	91-57-6	Soil	ug/kg	30	30	30
DP44	Acetophenone	98-86-2	Soil	ug/kg	18.86	100	25.51
DP44	Aluminum	7429-90-5	Soil	ug/kg	6.65e+06	6.65e+06	6.65e+06
DP44	Banum	7440-39-3	Soil	ug/kg	1.13e+05	1.13e+05	1.13e+05
DP44	Benzo(a)anthracene	56-55-3	Soil	ug/kg	200	200	200
DP44	Benzo(a)pyrene	50-32-8	Soil	ug/kg	390	390	390
DP44	Benzo(b)fluoranthenene	205-99-2	Soii	ug/kg	460	460	460
DP44	Benzo(g,h,i)perylene	191-24-2	Soil	ug/kg	280	280	280
DP44	Bis(2-ethylhexyl)phthalate	117-81-7	Soil	ug/kg	27.95	90	33.04
DP44	Buryl benzyl phthlate	85-68-7	Soil	ug/kg	17.05	60	20.57
DP44	Calcium	7440-70-2	Soil	ug/kg	4.48c+06	4.48c+06	4.48e+06
DP44	Chromium	7440-47-3	Soil	ug/kg	11200	11200	11200
DP44	Chrysene	218-01-9	Soil	ug/kg	280	280	280
DP44	Cobalt	7440-48-4	Soil	ug/kg	6490	6490	6490
DP44	Copper	7440-50-8	Soil	ug/kg	13900	13900	13900
DP44	Fluoranthene	206-44-0	Soil	ug/kg	770	770	770
DP44	Indeno(1,2,3-cd)pyrene	193-39-5	Soil	ug/kg	270	- 270	270
DP44	Iron	7439-89-6	Soil	ug/kg	1.15e+07	1.15e+07	1.15e+07
DP44	Kerosene	8008-20-6	Soil	ug/kg	335.6	510	412.1
DP44	Lead	7439-92-1	Soil	ug/kg	8978	46700	12140
DP44	Magnesium	7439-95-4	Soil	ug/kg .	3.56e+06	3.56e+06	3.56e+06
DP44	Manganese	7439-96-5	Soil	ug/kg	2.15e+05	2.15e+05	2.15e+05
DP44	Methylene chloride	75-09-2	Soil	ug/kg	1700	1700	1700
DP44	Naphthalene	91-20-3	Soil	ug/kg	17.37	80	22.32
DP44	Nickel	7440-02-0	Soil	ug/kg	14300	14300	14300
DP44	Phenanthrene	85-01-8	Soil	ug/kg	530	530	530
DP44	Porassium	7440-09-7	Soil	ug/kg	6.05e+05	6.05e+05	6.05e+05
DP44	Pyrene	129-00-0	Soil	ug/kg	400	400	400
DP44	Sodium	7440-23-5	Soil	ug/kg	2.55e+05	2.55e+05	2.55e+05
DP44	Total dissolved solids	TDS	Soil	ug/kg	94.49	97.5	97.06
DP44	Total petroleum hydrocarbons	трн	Soil	ug/kg	1.99c+05	7.20c+06	4.53c+05

Т	abl	e	9.	1	. (	(con	t'd}
---	-----	---	----	---	-----	------	------

	[		1		1	Махітит	Reasonable
Source	Í		Matrix		Average	Value	Maximum
Алеа	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
DP44	Vanadium	7440-62-2	Sou	ug/kg	24400	24400	24400
DP44	Zinc	7440-66-6	Soil	ug/kg	28700	28700	28700
	Groundwater			<b></b>			
WP45	I.I.I.Trichloroethane	71-55-6	Water	ug/L	0.2708	0.5	0.3083
WP45	1.2-Dichloroethane	107-06-2	Water	ug/L	0.2792	0,6	0.3316
WP45	1.4-Dichlorobenzene	108-88-3	Water	ug/L	0.9232	1	1
WP45	Barium	7440-39-3	Water	ug/L	89.5	140	140
WP45	Calcium	7440-70-2	Water	ug/L	46500	53000	53000
WP45	Chioride	16887-00-6	Water	ug/L	4550	6200	6200
WP45	Cis-1.2-dichlomethene	156-59-2	Water	ug/L	11.52	47	19.92
WP45	Ethylbenzene	67-66-4	Water	ug/L	0.00385	0.00385	0.00385
WP45	Fluoride	7782-41-4	Water	ug/L	150	200	200
WP45	h-Chlorofluorobenzene	PPP-PP-P	Water	ug/L	9.35	11	10.22
WP45	Iron	7439-89-6	Water	ug/L	215	420	420
WP45	Magnesium	7439-95-4	Water	ug/L	8800	11000	11000
WP45	Manganese	7439-96-5	Water	ug/L	1202	2400	2400
WP45	Nitrate	14797-55-8	Water	ug/L	4050	6600	6600
WP45	p-Chlorofluorobenzene	000-00-0	Water	ug/L	9.283	10	9.93
WP45	Potassium	7440-09-7	Water	ug/L	2950	3000	3000
WP45	Sodium	7440-23-5	Water	ug/L	4500	4800	4800
WP45	Sulfate	12808-79-8	Water	ug/L	11150	14000	14000
WP45	Trans-DCE	156-60-5	Water	ug/L	8.4	39	14.73
WP45	Trichlomethene	79-01-6	Water	ug/L	71.77	370	140.9
WP45	Vanadium	7440-62-2	Water	ug/L	102.5	190	190
	Surface Soil		*				<u></u>
WP45	Acenaphthene, soil	83-32-9	Soil	ug/kg	367.1	452	398.4
WP45	Aluminum	7429-90-5	Soil	ug/kg	7.74c+06	8.83e+06	8.83c+06
WP45	Anthracene, soil	120-12-7	Soil	ug/kg	109.1	921	219.6
WP45	Anumony	7440-36-0	Soil	ug/kg	4825	8900	8608
WP45	Barium	7440-39-3	Soil	ug/kg	1.06e+05	1.50e+05	1.42e+05
WP45	Benzo(a)anthracene	56-55-3	Soil	ug/kg	75.3	536	130.9
WP45	Benzo(a)pyrene	50-32-8	Soil	ug/kg	98.45	500	161.3
WP45	Benzo(b)fluoranthene	205-99-2	Soil	ug/kg	96.61	464	151.8
WP45	Benzo(g,h,i)perylene	191-24-2	Soil	ug/kg	111.1	530	176.8
WP45	Benzo(k)fluoranthene	207-08-9	Soil	ug/kg	42.74	286	75.71
WP45	Bis(2-ethylhexyl)phthalate	117-81-7	Soil	ug/kg	200	650	553
WP45	Cadmium	7440-43-9	Soil	ug/kg	1126	2810	2479
WP45	Calcium	7440-70-2	Soil	ug/kg	5.11e+06	7.16e+06	6.96c+06
WP45	Chromium	7440-47-3	Soil	ug/kg	13220	14900	14650
WP45	Chrysene	218-01-9	Soil	ug/kg	104.3	627	169.9
WP45	Cobalt	7440-48-4	Soil	ug/kg	8055	9730	9573
WP45	Copper	7440-50-8	Soil	ug/kg	16720	18300	18170
WP45	Dibenzo(a,h)anthracene	53-70-3	Soil	ug/kg	15.78	60	24.99
WP45	Diethylphthalate	84-66-2	Soil	ug/kg	75	210	180.9
WP45	Fluoranthene	206-44-0	Soil	ug/kg	251.7	1810	420.6
WP45	Fluorene	86-73-7	Soil	ug/kg	7.765	60	13.46

Table 3.1. (Contru	Т	abi	le	9.1	1.	(cont	d)
--------------------	---	-----	----	-----	----	-------	----

•,

	1		T	1	J	Maximum	Reasonable
Source		ļ	Matrix	ļ	Average	Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
WP45	Indeno(1,2,3-cd)pyrene	193-39-5	Soil	ug/kg	81.3	530	140.7
WP45	Iron	7439-89-6	Soil	ug/kg	1.35e+07	1.45e+07	1.45e+07
WP45	Lead	7439-92-1	Soil	ug/kg	10250	18900	14670
WP45	Magnesium	7439-95-4	Soil	ug/kg	4.24e+06	5.02e+06	4.910+06
WP45	Manganese	7439-96-5	Soil	ug/kg	2.65e+05	4.03e+05	3.75e+05
WP45	Mercury	7439-97-6	Soil	ug/kg	207.2	3172	464.4
WP45	Nickel	7440-02-0	Soit	ug/kg	16400	18500	18500
WP45	Phenanthrene	85-01-8	Soil	ug/kg	246.5	1290	385.7
WP45	Potassium	7440-09-7	Soil	ug/kg	6.86e+05	8.07c+05	8.07e+05
WP45	Pyrene	129-00-0	Soil	ug/kg	487.7	3770	843.9
WP45	Sodium	7440-23-5	Soil	ug/kg	3.01e+05	3.35e+05	3.30e+05
WP45	Tetrachloroethylene	127-18-4	Soil	ug/kg	206.8	620	620
WP45	Trichloroethene	79-01-6	Soil	ug/kg	1100	3300	3300
WP45	Trichlorofluoromethane	75-69-4	Soil	ug/kg	160	320	177.5
WP45	Vanadium	7440-62-2	Soil	ug/kg	28750	32900	32530
WP45	Xylenes (total)	1330-20-7	Soil	ug/kg	1102	3300	3300
WP45	Zinc	7440-66-6	Soil	ug/kg	38580	46800	45630
	Subsurface Soil						
WP45	Aluminum	7429-90-5	Soil	ug/kg	5.86e+06	5.86e +06	5.86e+06
WP45	Barium	7440-39-3	Soil	ug/kg	64100	64100	64100
WP45	Cadmium	7440-43-9	Soil	ug/kg	3280	3280	3280
WP45	Calcium	7440-70-2	Soil	ug/kg	3.95e+06	3.95e+06	3.95c+06
WP45	Chromium	7440-47-3	Soil	ug/kg	11600	11600	11600
WP45	Cobalt	7440-48-4	Soil	ug/kg	6780	6780	6780
WP45	Copper	7440-50-8	Soil	ug/kg	16800	16800	16800
WP45	Ethylbenzene	100-41-4	Soil	ug/kg	193.2	770	645.6
WP45	fron	7439-89-6	Soil	ug/kg	1.27e+07	1.27e+07	1.27e+07
WP45	Lead	7439-92-1	Soil	ug/kg	5437	7810	7810
WP45	Magnesium	7439-95-4	Soil	ug/kg	3.82e+06	3.82e+06	3.82e+06
WP45	Manganese	7439-96-5	Soil	ug/kg	2.15e+05	2.15e+05	2.15e+05
WP45	Nickel	7440-02-0	Soil	ug/kg	15400	15400	15400
WP45	Potassium	7440-09-7	Soil	ug/kg	6.14e+05	6.14e+05	6.14e+05
WP45	Sodium	7440-23-5	Soil	ug/kg	2.65e+05	2.65e+05	2.65e+05
WP45	Tetrachioroethylene	127-18-4	Soll	ug/kg	62.69	250	209.6
WP45	loluenc	108-88-3	ISOU	ug/kg	275.8	1100	922.2
WP45	Inchlorochene	79-01-0	Sou	ug/Kg	3000	12000	10060
WP45		7440-62-2	Sou	ug/kg	29400	29400	29400
WP45	Aylenes (spal)	1330-20-7	Sou	ug/kg	1152	4000	5856
WF45		1/440-00-0	200	ug/Kg	39200	39200	39200
		B 400 00 6	hy6		1092	1470	
5156		7429-90-5	water	ug/L	1086	1470	1470
5150		1/440-38-2	water		10.05	14.7	14.7
2120		1440-39-3	water		582.3	400	400
130 CT64	Cabali	7940-70-2	WAET		1.036+05	1.136+03	1.136+05
31.20	Conser	1440-40-4	WIEI	108/L	11.0	13.7	13.7
2120	Copper	1/440-30-8	I WERE L	ng/L	20.2	27	27

Т	abi	e	9.	1		(cont'd)
---	-----	---	----	---	--	----------

Source Area	Analyte Measured	CAS Number	Matrix Code	Units	Average Value	Maximum Value Detected	Reasonable Maximum Exposure
ST56	līon	7439-89-6	Water	ug/L	6200	8890	8890
ST56	Lead	7439-92-1	Water	ug/L	7.1	10.5	10.5
ST56	Magnesium	7439-95-4	Water	ug/L	29600	31300	31300
IST56	Manganese	7439-96-5	Water	ug/L	1710	2290	2290
ST56	Nickel	7440-02-0	Water	wg/L	18.65	20.4	20.4
ST56	Porassium	7440-09-7	Water	ug/L	8560	10300	10300
ST56	Sodium	7440-23-5	Water	ug/L	18300	22600	22600
ST56	Tetrachloroethylene	127-18-4	Water	ug/L	4.767	13.8	13.8
ST56	Trichloroethene	79-01-6	Water	ug/L	0.3	0.4	0.4
ST56	Vanadium	7440-62-2	Water	ug/L	13.3		19
IST56	Zinc	7440-66-6	Water	ug/L	38.75	46.2	46.2
∦	Groundwater					·	
5557	1.2-Dichloroethane	107-06-2	Water	ug/L	1.933	5.3	5,3
ISS57	Велиеле	71-43-2	Water	ug/L	178.7	530	530
SS57	Cis-1,2-dichloroethylene	156-59-2	Water	ug/L	25.37	73	73
SS57	Ethvibenzene	100-41-4	Water	ug/L	60.67	180	180
SS57	Lead	7439-92-1	Water	ug/L	5.867	9.3	9.3
S\$57	Tetrachloroethylene	127-18-4	Water	ug/L	0.5	1	1
SS57	Toluene	108-88-3	Water	ug/L	634	1900	1900
SS57	Xvienes (total)	1330-20-7	Water	ug/L	335	1000	1000
<b>}</b>	Surface Soil		L				
SS57	Benzene	71-43-2	Soil	ue/ke	1.4	2.2	2.2
SS57	Ethylbenzene	100-41-4	Soil	ug/kg	8669	26000	26000
SS57	Lead	7439-92-1	Soil	ug/kg	4200	4200	4200
SS57	Тоциеле	108-88-3	Soil	ug/kg	12670	38000	38000
ISS57	Xylenes (total)	1330-20-7	Soil	ue/ke	50000	1.50e+05	1.50e + 05
	Subsurface Soil		<u>.                                    </u>				
SS57	Ethvibenzene	100-41-4	Soil	ug/kg	800.7	2400	2400
5557	Lead	7439-92-1	Soil	ug/kg	3900	3900	3900
SS57	Toluene	108-88-3	Soil	ug/kg	539	1600	1600
SS57	Xylenes (mai)	11330-20-7	Soil	ug/kg	2768	8300	8300
	Groundwater				·····		
SS61	1.2-Dichlorobenzene	95-50-1	Water	ue/L	15.82	50	30,54
SS61	2-Methylnaphthalene	91-57-6	Water	ug/L		16	16
SS61	4-Methylphenol	106-44-5	Water	ug/L	8.667	16	16
SS61	Aluminum	7429-90-5	Water	ue/L	13200	24200	24200
S\$61	Arsenic	7440-38-2	Water	ue/L	44.13	81.2	81.2
ISS61	Barium	7440-39-3	Water	ug/L	705.7	1340	1340
SS61	Benzene	71-43-2	Water	ue/L	2.606	50	5.667
SS61	Beryllium	7440-41-7	Water	NE/L	0.9333	1.8	1.8
SS61	Bis(2-ethylhexyl)phthalan	117-81-7	Water	ue/L	6.7	10.1	10.1
ISS61	Butylbenzylohihalase	85-68-7	Water	ug/L	4.033	5	5
SS61	Cadmium	7440-43-9	Water	ug/L	0.7	1.1	1.1
SS61	Calcium	7440-70-2	Water		88130	96800	96800
SS61	Chromium	7440-47-3	Water	ug/L	31.6	56.1	56.1
SS61	Cis-1.2-Dichloroethene	156-59-2	Water	ug/L	342.2	3200	619.3

Table 9.1.	(cont'd)
------------	----------

	T		1	[	( <sup></sup>	Maximum	Beeconchie
Source		)	Matrix	i	Average	Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
ISS61	Cobalt	7440-48-4	Water	ug/L	41.87	84.8	84.8
SS61	Соррег	7440-50-8	Water	ug/L	60.2	69.8	69.8
5561	Di-n-butylphthalate	84-74-2	Water	ug/L	4.267	5	5
5561	Di-n-octylphthalate	117-84-0	Water	ug/L	2.533	5	5
SS61	Ethylbenzene	100-41-4	Water	ug/L	3.247	50	6.281
SS61	Gasoline	8006-61-9	Water	ug/L	213.4	2000	364.6
ISS61	Iron	7439-89-6	Water	ug/L	65770	1.24e+05	1.24e+05
SS61	Isophorone	78-59-1	Water	ug/L	5.567	6.7	6.7
SS61	Lead	7439-92-1	Water	ug/L	28.8	40.4	40.4
SS61	M.p-xyiene	MPXYLENES	Water	ug/L	102.5	290	290
\$\$61	Magnesium	7439-95-4	Water	ug/L	23300	26400	26400
S\$61	Manganese	7439-96-5	Water	ug/L	5727	8820	8820
SS61	n-Nitrosodiphenylamine	86-30-6	Water	ug/L	4.233	5	5
\$561	Naphthalene	91-20-3	Water	ug/L	18.33	38	38
SS61	Nickel	7440-02-0	Water	ug/L	81.83	153	153
SS61	o-Xylene	95-47-6	Water	ug/L	19.77	50	50
SS61	Pentachloropheno!	87-86-5	Water	ug/L	17.43	25	25
SS61	Phenanthrane	85-01-8	Water	ug/L	3.9	5	5
SS61	Porassium	7440-09-7	Water	ug/L	8517	9740	9740
SS61	Sodium	7440-23-5	Waser	ug/L	16470	23600	23600
SS61	Toluene	108-88-3	Water	ug/L	8.947	250	24.5
SS61	Trans-1.2-dichloroethene	156-60-5	Water	ug/L	16.68	140	26.46
SS61	Trichloroethene (TCE)	79-01-6	Water	ug/L	72.87	1100	160.6
SS61	Vanadium	7440-62-2	Water	ug/L	98.2	165	165
SS61	Xylenes (total)	1330-20-7	Water	ug/L	5.972	35	8.66
SS61	Zinc	7440-66-6	Water	ug/L	196	340	340
	Surface Soil						
SS61	Aluminum	7429-90-5	Soil	ug/kg	2.93e+06	2.93e+06	2.93e+06
\$561	Arsenic	7440-38-2	Soil	ug/kg	2700	2700	2700
SS61	Barium	7440-39-3	Soil	ug/kg	38600	38600	38600
SS61	Calcium	7440-70-2	Soil	ug/kg	1.97c+06	1.97e+06	1.97c+06
SS61	Chromium	7440-47-3	Soil	ug/kg	8100	8100	8100
SS61	Cobait	7440-48-4	Soil	ug/kg	4100	4100	4100
SS61	Copper	7440-50-8	Soil	ug/kg	12400	12400	12400
SS61	Iron	7439-89-6	Soil	ug/kg	5.20e+06	5.20e+06	5.20e+06
SS61	Lead	7439-92-1	Soil	ug/kg	5700	5700	5700
SS61	Magnesium	7439-95-4	Soil	ug/kg	1.65e+06	1.65e+06	1.65e+06
5561	Manganese	7439-96-5	Soil	ug/kg	1.05e+05	1.05e+05	1.05e+05
5561	Nickel	7440-02-0	Soil	ug/kg	10300	10300	10300
SS61	Vanadium	7440-62-2	ISoil	ug/kg	15000	15000	15000
	Subsurface Soil						
5561	1.1.1-Trichloroethane	71-55-6	Soil	ug/kg	1.023	4.3	1.673
ISS61	1,1-Dichloroethane	75-34-3	Soil	ug/kg	0.8045	2.7	I.173
5561	II.I-Dichloroethene	75-35-4	Soil	ug/kg	0.3227	0.76	0.4042
SS61	11,2-Dichiorobenzene	95-50-1	Sou	ug/kg	474.2	6600	1027
5561	1.3-Dichiorobenzene	541-73-1	Soil	ug/kg	473.4	6600	1027

Table	9.1.	(cont'd)
-------	------	----------

					}	Maximum	Reasonable
Source	Analyze Measured	CAS Number	Matrix Code	Units	Average Value	Detected	
CC61		106-46-7	ISoil	l. hug/kg	4737	6600	1027
19201	2. Methyloanbrhalent	91-57-6	Soil	uro/ko	2001	13000	4251
5501	Abminum	7429-90-5	Sail	ue/ke	5 530+06	8 480+06	6 540 + 06
5561		7440-38-7	Soil	ug/kg	4372	11900	6412
001	Pa-may	7440-30-2	Soil	wg/kg	1 650 + 05	7 050+05	2 1104 05
5501		71 42 2	Soil .	ug/ ng	5 626	1.050+05	12 22
10001	Benzene		1300	ug/kg	1 07- + 06	30	12.23
5501	Calcium	100.00.7	1500	ug/kg	3.0/0+00	3.43e+00	3.800 + 00
5561	Chiorobenzene	108-90-7	15011	ug/kg	C81.7	100	15.72
SS61	Chloroform	67-66-3	Sou	ug/Kg	0.7636	2.4	1.08
ISS61	Chromuum	7440-47-3	Soil	ug/kg	10260	14400	11750
SS61	Cobait	7440-48-4	Soil	ug/kg	3718	4600	4247
SS61	Copper	7440-50-8	Soil	ug/kg	20670	45900	27700
SS61	Ethylbenzene	100-41-4	Soil	ug/kg	2.041	6.2	3.275
SS61	Fluorene	86-73-7	Soil	ug/kg	954.8	6600	2003
SS61	Iron	7439-89-6	Soil	ug/kg	8.48c+06	1.03e+07	1.03e+07
SS61	Isophorone	78-59-1	Soil	ug/kg	1728	10000	3555
S\$61	Lead	7439-92-1	Soil	ug/kg	7500	19500	10820
\$\$61	M,p-xylene	MPXYLENES	Soil	ug/kg	15.45	1 10	34,77
SS61	Magnesium	7439-95-4	Soil	ug/kg	3.05e+06	5.28e+06	3.72c+06
IS\$61	Manganese	7439-96-5	Soil	ug/kg	1.35e+05	1.96e+05	1.60e+05
S\$61	Methylene chloride	75-09-2	Soil	ug/kg	0.8909	2.3	1.217
ISS61	Naphthalene	91-20-3	Soil	ug/kg	4003	35000	9716
SS61	Nickel	7440-02-0	Soil	ue/ke	10060	12400	11300
5561	o-Xviene	95-47-6	Soil	ue/ke	1.073	3.5	1 179
10001	Phenoprhyme	85-01-8	Soil	uo/ka	1173	6600	2328
10001	Termchlomethylene (PCE)	127-18-4	ISou	us/ka	29.03	140	52 72
10001	Toluene	109-99-3	Sail		14.24	100	31.75
0001	Trans. ( 2 disblamethere	156 40 5	Soll	uging	0 8219	100	1 226
3301	Trisble and and (DCE)		1304 Coll	ug/sg	60.02	2.7	1.230
3301		79-01-0	1001	UB/KB	0.62	2.50	93.78
5301	1 richloronuoromeutane	73-09-4	1500		17040	2.3	10100
5201	Vanadium	/440-62-2	Son	ug/kg	1/050	22200	19100
5561		[/440-00-0	501	ug/kg	23420	43900	29050
	Groundwater						,
SS61-sp	Cis-1,2-dichloroethene	156-59-2	Water	ug/L	1285	3200	2376
SS61-sp	Ethylbenzene	100-41-4	Water	ug/L	2.114	5.8	3.308
SS61-sp	Gasoline	8006-61-9	Water	ug/L	647.1	2000	1288
SS61-sp	Trans-1.2-dichlomethylene	156-60-5	Water	ug/L	.21.9	56	38.69
SS61-sp	Trichloroethene	79-01-6	Water	ug/L	310.4	1100	664.5
SS61-sp	Xylenes (total)	1330-20-7	Water	ug/L	12.57	35	23.7
Operable	Unit 4						
[	Groundwater						
DP25	4-Methyl-2-pentanone	108-10-1	Water	ug/L	27.39	78	30.7
DP25	Acetone	67-64-1	Water	ug/L	43.64	100	57.98
DP25	Acetone		Water	ug/L	12.66	100	17,66
DP25	Arsenic	7440-38-2	Water	ug/L	16.75	31	31
DP25	Barium	7440-39-3	Water	ug/L	150.3	210	210
4				· ~			

Т	abl	e	9.	1		(cont'd)	ł
		_			-		

		<u> </u>	T	l		Maximum	Ressonable
δουπε	ł	1	Matrix	1	Average	Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
DP25	Benzene	71-43-2	Water	ug/L	171.7	1700	325
DP25	Calcium	7440-70-2	Water	ug/L	63670	77000	77000
DP25	Chloride	16887-00-6	Water	ug/L	1550	1700	1700
DP25	Cis-1.2-dichloroethene	156-59-2	Water	ug/L	0.5667	1.1	0.6907
DP25	Copper	7440-50-8	Water	ug/L	21	28	28
DP25	Ethylbenzene	100-41-4	Water	ug/L	21.44	150	51.72
DP25	Fluoride	7782-41-4	Water	ug/L	300	400	400
DP25	h-Chlorofluorobenzene	PPP-PP-P	Water	ug/L	11	11	11
DP25	Iron	7439-89-6	Water	ug/L	7820	16000	16000
DP25	Lead	7439-92-1	Water	ug/L	12.88	60	18.25
DP25	Magnesium	7439-95-4	Water	ug/L	14000	17000	17000
DP25	Manganese	7439-96-5	Water	ug/L	3167	6600	6600
DP25	Methylenechloride	75-09-2	Water	ug/L	7,958	25	13.08
DP25	Nitrate	14797-55-8	Water	ug/L	300	400	400
DP25	p-Chlorofluorobenzene	000-00-0	Water	ug/L	11	11	11
DP25	Potassium	7440-09-7	Water	ug/L	3900	4500	4500
DP25	Sodium	7440-23-5	Water	ug/L	5267	7000	7000
DP25	Sulfate	12808-79-8	Water	ug/L	11550	22000	22000
DP25	Toluene	108-88-3	Water	ug/L	527.2	8900	1214
DP25	Total organic carb	TTOC	Water	ug/L	1000	1000	1000
DP25	TPH-diesel	TPH-D	Water	ug/L	2000	11000	3871
DP25	TPH-gasoline	TPH-G	Water	ug/L	2893	19000	5335
DP25	Xylenes (mtal)	1330-20-7	Water	ug/L	431	3400	748
DP25	Zinc	7440-66-6	Water	ug/L	42.67	86	86
	Surface Soil				·····	L	
DP25	Benzene	71-43-2	Soil	ug/kg	410	410	410
DP25	DDT, pp	50-29-3	Soil	ug/kg	1,533	5.7	3.244
DP25	Dieldrin	60-57-1	Soil	ug/kg	1.75	8	4.269
DP25	Endosulfan, a	115-29-7	Soil	ug/kg	2.833	9	5.908
DP25	Ethylbenzene	100-41-4	Soil	ug/kg	615.9	3000	1278
DP25	Heptachlor epoxide	1024-57-3	Soil	ug/kg	0.65	1.4	0.9523
DP25	Lead	7439-92-1	Soil	ug/kg	5098	25300	5936
DP25	PCB-1254 (aroclor)	11097-69-1	Soil	ug/kg	56	172	117.8
DP25	Toluene	108-88-3	Soil	ug/¥g	1054	3100	3100
DP25	ТРН	ТРН	Soil	ug/kg	1.07c+05	5.89e+05	3.01e+05
DP25	Xylenes (total)	1330-20-7	Soil	ug/¥2	5887	15000	13710
<u></u>	Subsurface Soil		I				·
DP25	DDE, pp'	72-55-9	Soil	ug/kg	0.6667	2	0.9767
DP25	DDT, pp'	50-29-3	Soil	ugi¥g	2.667	17	6.023
DP25	Dicthyl ether	60-29-7	Soil	ug/kg	144.4	500	227.1
DP25	Ethylbenzene	100-41-4	Soil	ug/kg	56.67	190	87.67
DP25	Heprachlot epoxide	1024-57-3	Soil	ug/kg	28.22	250	79.79
DP25	Lead	7439-92-1	Soil	ug/kg	1.31e+05	8.70e+05	3.71e+05
D <b>P25</b>	PCB-1254 (Aroclor)	11097-69-1	Soil	ug/icz	92.11	613	213.2
DP25	ТРН	ТРН	Soil	ug/kg	66570	2.89c+05	1.40e+05
DP25	Xylenes (total)	7440-66-6	Soil	ug/kg	47.78	110	62.24

T	able	9	1 (	(cont'd)
	avie	<b>J</b> .		COLL U/

C	······································	······	r#	ý	(		
Source		l	Matrix	ļ	Average	Maximum Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
( <u> </u>	Ground water		<u></u>				
ST27	Arsenic	7440-38-2	Water	ug/L	14.77	30	20.36
ST27	Barium	7440-39-3	Water	ug/L	279.8	590	381
ST27	(Calcium	7440-70-2	Water	ug/L	46750	56000	50880
ST27	Chloride	16887-00-6	Water	ug/L	1233	1400	1400
ST27	Chromium	7440-47-3	Water	ug/L	11.38	21	13.98
ST27	Cobali	7440-48-4	Water	ug/L	12,75	32	17.96
ST27	Copper	7440-50-8	Water	ug/L	96	430	188.8
ST27	Fluonde	7782-41-4	Water	ug/L	166.7	200	200
ST27	h-Chlorofiuorobenzene	PP <b>P-PP</b> -P	Water	ug/L	9.4	11	10.04
ST27	Iron	7439-89-6	Water	ug/L	9012	23000	13730
ST27	Lead	7439-92-1	Water	ug/L	16.62	120	31.09
ST27	Magnesium	7439-95-4	Water	ug/L	11140	15000	12770
ST27	Manganese	7439-96-5	Water	ug/L	3385	12000	5807
ST27	Nickel	7440-02-0	Water	ug/L	25.5	82	41.31
ST27	Nitrate	14797-55-8	Water	ug/Ľ	533.3	1100	1100
ST27	p-Chlorofluorobenzene	QQQ-QQ-Q	Water	ug/L	9.757	10	9,985
ST27	Porassium	7440-09-7	Water	ug/L	3038	3900	3343
ST27	Sodium	7440-23-5	Water	ug/L	4538	5600	4997
IST27	Sulfate	12808-79-8	Water	ug/L	11270	15000	15000
ST27	Total organic carb.	TOC	Water	ug/L	5000	5000	5000
ST27	Vanadium	7440-62-2	Water	ug/L	24.5	91	42.5
ST27	Zinc	7440-66-6	Water	ug/L	106.2	400	188.7
	Surface Soil						
ST27	Benzene	71-43-2	Soil	ug/kg	6	6	6
ST27	Lead	7439-92-1	Soil	ug/kg	6590	37600	9288
ST27	Toluene	108-88-3	Soil	ug/kg	17	17	17
	Subsurface Soil						······
ST27	Lead	7439-92-1	Soil	ug/kg	7500	8000	8000
	Groundwater						
WP33	Aluminum	7429-90-5	Water	ug/L	593.3	969	965
WP33	Arsenic	7440-38-2	Water	ug/L	14.67	18.2	18.2
WP33	Barium	7440-39-3	Water	ug/L	230.7	335	335
WP33	Calcium	7440-70-2	Water	ug/L	<del>69</del> 870	94400	94400
WP33	Copper	7440-50-8	Water	ug/L	12.77	15	15
WP33	lron	7439 <b>-89-6</b>	Water	ug/L	3640	4360	4360
WP33	Lead	7439-92-1	Water	ug/L	3.85	3.9	3.9
WP33	Magnesium	7439-95-4	Water	ug/L	12820	16000	16000
WP33	Manganese	7439-96-5	Water	ug/L	6297	8850	8850
WP33	Nickel	7440-02-0	Water	ug/L	20.9	20.9	20.9
WP33	Potassium	7440-09-7	Water	ug/L	11250	14000	14000
WP33	Sodium	7440-23-5	Water	ug/L	34970	36900	36900
WP33	Zinc	7440-66-6	Water	ug/L	34,23	40.9	40.9
[[	Groundwater						
SS35	1.2-Dichloroethane	107-06-2	Water	ug/L	0.4958	3.2	0.9374
ISS35	4.4'-DDE	72-55-9	Water	ug/L	0.03231	0,12	0.04533

#### Table 9.1. (cont'd)

.

N

. مىر

(~	<u>۲</u>		<u></u>		1	Maximum	Besconshiel
Source	1		Matrix		Average	Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
SS35	4,4'-DDT	50-29-3	Water	ug/L	0.06538	0.16	0.08405
SS35	Benzene	71-43-2	Water	ug/L	1.208	3.5	1.582
5535	Bena-BHC	319-85-7	Water	ug/L	0.02692	0.05	0.03035
SS35	Cis-1.2-dichloroeshylene	156-59-2	Water	ug/L	0.55	1.1	0.6398
SS35	Gamma-BHC (lindane)	58-89-9	Water	ug/L	0.02846	0.07	0.03463
SS35	Lead	7439-92-1	Water	ug/L	11.62	68	21.14
SS35	Toluene	108-88-3	Water	ug/L	1.192	3.3	1.536
SS35	Xylenes (total)	1330-20-7	Water	ug/L	3.075	9.4	4.108
	Surface Soil					<b></b>	······································
SS35	2-Methylnaphthalene	91-57-6	Soil	ug/kg	5270	21000	17610
SS35	4,4'-DDD	72-54-8	Soü	ug/kg	437.8	5100	1076
SS35	4.4'-DDE	72-55-9	Soil	ug/kg	2731	19000	5711
SS35	4,4'-DDT	50-29-3	Soil	ug/kg	4304	49000	10430
SS35	Alpha-BHC	319-84-6	Soil	ug/kg	6.5	17	9.342
SS35	Aluminum (sed)	7429-90-5	Soil	ug/kg	7.06c+06	9.25c+06	9.25c+06
SS35	Barium (sed)	7440-39-3	Soil	ug/kg	87000	1.10e+05	1.10e+05
S\$35	Beryllium (sed)	7440-41-7	Soil	ug/kg	607	1010	1010
SS35	Beta-BHC	319-85-7	Sail	ug/kg	4.227	10	5.765
\$\$35	Cadmium (sed)	7440-43-9	Soil	ug/kg	969.8	2040	2040
SS35	Calcium (sed)	7440-70-2	Soil	ug/kg	3.99e+06	5.66e+06	5.66e+06
SS35	Chlordane	57-74-9	Soil	ug/kg	60.46	410	119
SS35	Chromium (sed)	7440-47-3	Soil	ug/kg	12680	16400	16400
SS35	Cobait (sed)	7440-48-4	Soil	ug/kg	6837	8170	8170
SS35	Copper (sed)	7440-50-8	Soil	ug/kg	18830	25600	25600
\$\$35	Endosulfan I	115-29-7	Soii	ug/kg	8.25	31	14.41
SS35	Endrin	72-20-8	Soil	ug/kg	6.875	20	10.43
SS35	Gamma-BHC (lindane)	58-89-9	Soil	ug/kg	141.9	1100	401.3
SS35	Hepmchlor	76-44-8	Soü	ug/kg	3.791	5	4.922
\$\$35	Heptachlor epoxide	1024-57-3	Soil	ug/kg	27.88	30	30
SS35	Iron (sed)	7439-89-6	Soil	ug/kg	1.40e+07	1.82e+07	1.82e+07
SS35	Lesd	7439-92-1	Soil	ug/kg	28950	1.20e+05	47210
SS35	Magnesium (sed)	7439-95-4	Soil	ug/kg	4.17e+06	5.38e+06	5.38e+06
SS35	Manganese (sed)	7439-96-5	Soil	ug/kg	2.63e+05	3.38c+05	3.38e+05
5535	Naphihalene	91-20-3	Soil	ug/kg	911.2	3600	3020
SS35	Nickel (sed)	7440-02-0	Sou	ug/kg	15930	19900	19900
5535	Ponskun (scd)	/440-09-7	2011	ug/kg	0.420+03	7.08e+05	7.080+05
2232	Sognin (sea)	/440-23-3	201	ug/kg	2.896+03	3.080+US	3.080+05
5535	ftph Martin (and	1177	Sou	ug/kg	22380	3.800+05	47240
5535	Vanadrum (SEG)	/440-62-2	1201	ug/kg	26530	35500	35500
2222	Zinc (Seg)	/440-00-0	501	ug/kg	41400	54100	54100
00016	Subsurface Soll	101 67 C	5			-	
3533 6676	12-Meunyinaphutalene	0-/c-14	30U	ug/xg	31	70	54.21
CC26		309-00-2	301	ug/kg	1.925	0.2	3.278
5555	Auminum (sea)	/429-90-5	150U	ug/kg	9.33e+06	1.19e+07	1.190+07
2222	Bartum (Sed)	/440-39-3	501	ug/kg	1.85e+05	3.886+05	3.540+05
2272	(Sed)	[/ <del>440-4</del> ]-7	301	ug/kg	750.5	1030	1030

Tabl	le !	9.1	I. (	cont'	d)
IQN				CONC	чı.

Source			Maren	ļ	Avemae	Maximum	Reasonable
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
ISS35	IBHC. beta	1319-85-7	I ISoil	lug/kg	0.825	1.8	1.59
\$\$35	Caicium (sed)	7440-70-2	Soil	ug/kg	6.79e+06	1.14e+07	1.13e+07
5535	Chlordane	57-74-9	Soil	ug/kg	76.94	269	147.8
SS35	Chromium (sed)	7440-47-3	Soil	ug/kg	15880	21300	20630
SS35	Cobalt (sed)	7440-48-4	Soil	ug/kg	8270	11200	10590
SS35	Copper (sed)	7440-50-8	Soil	ug/kg	24550	33000	33000
SS35	DDD. pp	72-54-8	Soil	ug/kg	6819	58500	18840
SS35	DDE, pp	72-55-9	Soil	ug/kg	2196	9710	4272
SS35	DDT. pp`	50-29-3	Soil	ug/kg	64680	3.96e+05	1.46e+05
SS35	Fluoranthene	206-44-0	Soil	ug/kg	33	70	\$7.45
SS35	Heptachior	76-44-8	Soil	ug/kg	0.52	0.6	0.5626
SS35	Iron (sed)	7439-89-6	Soil	lug/kg	1.71e+07	2.40e+07	2.30e+07
\$\$35	Lead (sed)	7439-92-1	Soil	ug/kg	11370	45100	19760
SS35	Magnesium (sed)	7439-95-4	Soil	ug/kg	4.93e+06	6.42e+06	6.24e+06
SS35	Manganese (sed)	7439-96-5	Soil	ug/kg	4.25e+05	9.00e+05	8.02e+05
SS35	Nickel (sed)	7440-02-0	Soil	ug/kg	19750	24900	24900
\$\$35	Phenanthrene	85-01-8	Soil	ug/kg	18	30	24.4
\$\$35	Potassium (sed)	7440-09-7	Soit	ug/kg	7.57e+05	9.34e+05	9.22e+05
SS35	Ругеле	129-00-0	Soil	ug/kg	26	70	49.45
SS35	Sodium (sed)	7440-23-5	Soil	ug/kg	3.47e+05	4.65e+05	4.48c+05
SS35	Трн	ТРН	Soil	ug/kg	90310	7.93e+05	1.66e+05
SS35	Vanadium (sed)	7440-62-2	Soil	ug/kg	33220	38500	38500
SS35	Zinc (sed)	7440-66-6	Soil	ug/kg	50650	73400	69780
	Groundwater	,			····	,	
SS36	Barium	7440-39-3	Water	ug/L	210	330	330
SS36	Calcium	7440-70-2	Water	ug/L	1.01e+05	1.60e+05	1.60e+05
SS36	licon	7439-89-6	Water	ug/L	10230	19000	19000
SS36	Magnesium	7439-95-4	Water	ug/L	19330	26000	26000
SS36	Manganese	7439-96-5	Water	ug/L	2600	4400	4400
SS36	Potassium	7440-09-7	Water	wg/L	4333	4900	4900
SS36	Sodium	7440-23-5	Water	ug/L	7067	8300	8300
SS36	Zinc	7440-66-6	Water	ug/L	7.333	12	12
	Surface Soil	·	A	······	A	J	
SS36	Lead	7439-92-1	Soil	ug/kg	5273	7000	5824
	Subsurface Soil	,			<u></u>		
SS36	Arsenic	7440-38-2	Soil	ug/kg	24000	24000	24000
SS36	Barium	7440-39-3	Soil	ug/kg	48000	62000	62000
SS36	Beryllium	7440-41-7	Soil	ug/kg	566.7	1400	1400
5\$36	Calcium	7440-70-2	Soil	ug/kg	2.35e+06	2.40e+06	2.40e+06
SS36	Chromium	7440-47-3	Soil	ug/kg	5.71e+05	1.70c+06	1.70c+06
\$\$36	Cobalt	7440-48-4	Soil	ug/kg	4000	4000	4000
SS36	Copper	7440-50-8	Soil	ug/kg	33670	83000	83000
SS36	DDD, pp'	72-54-8	Soil	ug/kg	15	15	15
S\$36	DDE, pp'	72-55-9	Soil	ug/kg	80	80	80
<u>SS36</u>	DDT. pp'	50-29-3	Soil	lug/kg	240	240	240
SS36	lron	7439-89-6	Soil	ug/kg	6 35+06	7.60 + 06	7.60 + 06

ISDIE 9.1. ICONT	a)
------------------	----

	T		<u></u>	<del> </del>		Maximum	Bestonable
Source			Манля	[	Average	Value	Maximum
Агеа	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
SS36	Lead	7439-92-1	Soil	ug/kg	2.61e+06	7.80e+06	7.80e+06
SS36	Magnesium	7439-95-4	Soil	ug/kg	2.05e+06	2.30e+06	2.30e+06
\$S36	Manganese	7439-96-5	Soil	ug/kg	99000	1.20 <del>c</del> +05	1.20e+05
SS36	Nickel	7440-02-0	Soil	ug/kg	20670	45000	45000
SS36	Potassium	7440-09-7	Soil	ug/kg	3.80c+05	4.40c+05	4.40e+05
SS36	Selenium	7782-49-2	Soil	ug/kg	600	600	600
SS36	Sodium	7440-23-5	Soil	ug/kg	2.70e+05	2.90e+05	2.90c+05
SS36	Vanadium	7440-62-2	Soil	ug/kg	11000	13000	13000
SS36	Zinc	7440-66-6	Soil	ug/kg	40670	96000	96000
[	Groundwater						
SS37	Aluminum	7429-90-5	Water	ug/Ĺ,	141.6	387	334.5
SS37	Алзепіс	7440-38-2	Water	ug/L	25.98	56.7	50.12
5537	Barium	7440-39-3	Water	ug/L	162.8	207	207
SS37	Bis(2-ethylhexyl)phthalate	117-81-7	Water	ug/L	3.86	5	5
S\$37	Calcium	7440-70-2	Water	ug/L	48280	50400	50400
SS37	Chromium	7440-47-3	Water	ug/L	2.25	2.7	2.7
S\$37	Cobait	7440-48-4	Water	ug/L	0.8	1.1	1.1
SS37	Copper	7440-50-8	Water	ug/L	3.475	9	7.835
SS37	ໄໝ	7439-89-6	Water	ug/L	9405	17000	16000
S\$37	Lead	7439-92-1	Water	ug/L	1.225	3.4	2.931
SS37	Magnesium	7439-95-4	Water	ug/L	11020	11400	11400
SS37	Manganese	7439-96-5	Water	ug/L	1782	2590	2590
SS37	Nickel	7440-02-0	Water	ug/L	2.05	3.3	3.069
SS37	Pomssium	7440-09-7	Water	ug/L	3498	3620	3620
SS37	Sodium	7440-23-5	Water	ug/L	5720	7950	7471
SS37	Ta	7440-31-5	Water	ug/L	1.85	2.2	2.2
SS37	Total dissolved solids	ρ	Water	ug/L	60320	2.40e+05	1.04e+05
SS37	Vanadium	7440-62-2	Water	ug/L	1.675	4.6	3.993
SS37	Zinc	7440-66-6	Water	ug/L	8.475	15.3	13.95
<u> </u>	Surface Soil						
SS37	2-Butanone	78-93-3	Soil	ug/kg	245.3	1300	588.7
SS37	2-Methylnaphthalene	91-57-6	Soil	ug/kg	4795	30000	9059
SS37	Anthracene	120-12-7	Soil	ug/kg	240	240	240
SS37	Benzoic acid	65-85-0	Soil	ug/kg	66	66	66
SS37	Chrysene	218-01-9	Soil	ug/kg	430	430	430
SS37	Dibenzofuran	132-64-9	Soil	ug/kg	323.3	440	440
SS37	Ethylbenzene	100-41-4	Soil	ug/kg	827.5	4200	2198
SS37	Fluoranthene	206-44-0	Soil	ug/kg	570	570	570
SS37	Fluorene	86-73-7	Soil	ug/kg	902.5	3100	2631
SS37	Lead	7439-92-1	Soil	ug/kg	11010	2.50c+05	19200
SS37	Naphthalene	91-20-3	Soil	ug/kg	3567	12000	6948
SS37	0-Xylene	95-47-6	Soil	ug/kg	38.67	42	42
SS37	Phenanthrene	85-01-8	Soil	ug/kg	1188	2300	2300
SS37	Phenoi	108-95-2	Soil	ug/kg	30.5	35	35
SS37	Pyreae	129-00-0	Soil	ug/kg	420	640	640
5537	Tolucne	108-88-3	Soil	ug/kg	426.1	3500	884.1

Т	abi	e	9		1		(con	ť¢	(t
---	-----	---	---	--	---	--	------	----	----

Source			Matrix	1	Ачегаде	Maximum Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
SS37	Trichloroethene	79-01-6	Soil	ug/kg	2	3	2.674
SS37	Xylenes (total)	1330-20-7	Soil	ug/kg	3158	24000	8078
	Subsurface Soil					······································	••
SS37	1.1.1-Trichloroethane	71-55-6	Soil	ug/kg	72	72	72
SS37	2-Butanone (mek)	78-93-3	Soil	ug/kg	17	17	Ī7
SS37	2-Methyinaphthalene	91-57-6	Soil	ug/kg	8595	26000	22390
SS37	4.4'-DDT	50-29-3	Soil	ug/kg	70	70	70
SS37	Benzo(a)anthracene	56-55-3	Soil	ug/kg	210	210	210
SS37	Bea-BHC	319-85-7	Soil	ug/kg	5	5	5
SS37	Chrysene	218-01-9	Soil	ug/kg	290	290	290
SS37	Dibenzofuran	132-64-9	Soil	ug/kg	544.5	1000	1000
SS37	Ethylbenzene	100-41-4	Soil	ug/kg	11640	43000	29460
SS37	Fluoranthrene	206-44-0	Soil	ug/kg	98.5	290	250.9
SS37	Fluorene	86-73-7	Soil	ug/kg	633.5	1200	1200
SS37	Lead	7439-92-1	Soil	ug/kg	26510	1.80e+05	43520
SS37	Naphthalene	91-20-3	Soil	ug/kg	5633	12000	12000
S\$37	Phenanthrene	85-01-8	Soil	ug/kg	153.8	410	303.8
SS37	Pyrene	129-00-0	Soil	ug/kg	116.5	360	308.7
SS37	Toluene	108-88-3	Soil	ug/kg	16690	79000	49950
SS37	Trichloroethene	79-01-6	Soil	ug/kg	2.667	3	3
SS37	Xylenes (total)	1330-20-7	Soil	ug/kg	51790	2.10e+05	1.24e+05
	Groundwater						
SS39/63	Barium	7440-39-3	Water	ug/L	138.3	210	195.4
SS39/63	Calcium	7440-70-2	Water	ug/L	41010	51000	51000
SS39/63	Copper	7440-50-8	Water	ug/L	14,33	36	23.07
S\$39/63	lron	7439-89-6	Water	ug/L	3835	9100	6648
SS39/63	Magnesium	7439-95-4	Water	ug/L	8475	11000	11000
SS39/63	Manganese	7439-96-5	Water	ug/L	1768	3300	2661
SS39/63	Potassium	7440-09-7	Water	ug/L	2942	3800	3800
SS39/63	Sodium	7440-23-5	Water	ug/L	3492	4600	4600
S\$39/63	Total organic carb.	тос	Water	ug/L	5000	5000	5000
SS39/63	Zinc	7440-66-6	Water	ug/L	12.67	32	21.38
-	Surface Soil						
SS39/63	2-Methylnaphthalene	91-57-6	Soil	ug/kg	85.59	1200	207.2
S\$39/63	Aluminum	7429-90-5	Soil	ug/kg	4.59c+06	7.26e+06	6.82e+06
SS39/63	Banum	7440-39-3	Soil	ug/kg	69280	92000	92000
SS39/63	Beryllium	7440-41-7	Soil	ug/kg	408.8	750	750
SS39/63	BHC, beta	319-85-7	Soil	ug/kg	0.875	2	1.757
SS39/63	Cadmium	7440-43-9	Soil	ug/kg	692.9	1080	1080
SS39/63	Calcium	7440-70-2	Soil	ug/kg	2.06e+06	3.72e+06	3.39e+06
SS39/63	Chlordane	57-74-9	Sail	ug/kg	0.875	2	1.757
SS39/63	Chromium	7440-47-3	Soil	ug/kg	13450	32300	28340
SS39/63	Cobalt	7440-48-4	Soil	ug/kg	5380	8250	7797
SS39/63	Copper	7440-50-8	Soil	ug/kg	15500	22100	22100
SS39/63	DDD, pp'	72-54-8	Soil	ug/kg	37.74	328	67.33
SS39/63	DDE, pp	72-55-9	Soit	ue/ke	10.19	127	20.61

Table 9.1.	(cont'	d)
------------	--------	----

		1	1	T			-
5			Marris		Avenar	Maximum	Reasonable
Апта	Analyse Measured	CAS Number	Code	Units	Value	Detected	Exposure
ISS39/63	DDT. pp	50-29-3	lSoit	ug/kg	28.5	437	64.26
S\$39/63	Endosulfan, a	115-29-7	Soil	ug/kg	0.875	2	1.757
SS39/63	Endrin aldehyde	7421-93-4	Soil	ug/kg	0.875	2	1.757
SS39/63	Fluorene	86-73-7	Soil	ug/kg	93.53	680	186.2
SS39/63	liron	7439-89-6	Soil	ug/kg	1.00e+07	1.53e+07	1.43e+07
SS39/63	Isophorane	78-59-1	Soil	ug/kg	779.7	13000	2113
SS39/63	Kerosene	8008-20-6	Soil	ug/kg	570.4	3900	1132
SS39/63	Lead	7439-92-1	Soil	ug/kg	7870	11500	11500
S\$39/63	Magnesium	7439-95-4	Soil	ug/kg	2.80e+06	4.34e+06	4.11e+06
SS39/63	Manganese	7439-96-5	Soil	ug/kg	2.00c+05	2.86e+05	2.73e+05
SS39/63	Naphthalene	91-20-3	Soil	ug/kg	69.71	930	163.6
SS39/63	Nickel	7440-02-0	Soil	ug/kg	11120	16700	15540
SS39/63	Pomssium	7440-09-7	Soil	ug/kg	4,36e+05	5.44e+05	5.44e+05
SS39/63	Sodium	7440-23-5	Soil	ug/kg	1.88c+05	2.86e+05	2.66e+05
\$\$39/63	Total petroleum hydrocarbons	ТРН	Soil	ug/kg	8.74e+05	1.78e+07	1.52e+06
SS39/63	Vanadium	7440-62-2	Soil	ug/kg	17950	26900	25450
SS39/63	Zinc	7440-66-6	Sou	ug/kg	32700	45200	45200
ļ	Subsurface Soil		L				
SS39/63	Atuminum	7429-90-5	Soil	ug/kg	8.99e+06	1.04c+07	1.04e+07
SS39/63	Arsenic	7440-38-2	Soil	ug/kg	5093	8890	8890
SS39/63	Barium	7440-39-3	Soil	ug/kg	89170	99200	99200
SS39/63	Beryilium	7440-41-7	Soil	ug/kg	987.3	1450	1450
SS39/63	Cadmium	7440-43-9	Soil	ug/kg	753.5	1290	1290
SS39/63	Calcium	7440-70-2	Soil	ug/kg	3.60e+06	4.28c+06	4.28c+06
SS39/63	Chlordane	57-74-9	Soil	ug/kg	4	11	11
SS39/63	Chromium	7440-47-3	Soil	ug/kg	16300	18700	18700
SS39/63	Cobalt	7440-48-4	Soil	ug/kg	8287	9390	9390
SS39/63	Copper	7440-50-8	Soil	ug/kg	18120	21900	21900
SS39/63	DDD. pp'	72-54-8	Soil	ug/kg	58.4	262	167.4
SS39/63	DDE, pp'	72-55-9	Soil	ug/kg	12.9	56	36.03
SS39/63	DDT, pp'	50-29-3	Soil	ug/kg	97,4	356	244.1
S\$39/63	Endosulfan, a	115-29-7	Soil	ug/kg	5.667	16	16
SS39/63	Endrin aldehyde	7421-93-4	Soil	ug/kg	17	50	50
SS39/63	lron	7439-89-6	Soii	ug/kg	1.78e+07	2.06e+07	2.06c+07
SS39/63	Kerosene	8008-20-6	Soil	ug/kg	12890	82000	21690
SS39/63	Lead	7439-92-1	Soil	ug/kg	11640	14000	14000
SS39/63	Magnestum	7439-95-4	Soil	ug/kg	4.83c+06	5.45e+06	5.45e+06
SS39/63	Manganese	7439-96-5	Soil	ug/kg	2.31e+05	2.84e+05	2.84e+05
SS39/63	Nickel	7440-02-0	Soil	ug/kg	18000	20100	20100
SS39/63	Potassium	7440-09-7	Soil	ug/kg	6.54e+05	7.05e+05	7.05e+05
SS39/63	Sodium	7440-23-5	Soil	ug/kg	3.35e+05	3.89c+05	3.89e+05
SS39/63	Total petroleum hydrocarbons	ТРН	Soil	ug/kg	6.45e+06	4.10c+08	1.67e+07
SS39/63	Vanadium	7440-62-2	Soil	ug/kg	33000	37800	37800
SS39/63	Zinc	7440-66-6	Soil	ug/kg	47130	51800	51800

Т	abi	е	9.	. 1		(cont'd)	
---	-----	---	----	-----	--	----------	--

Source Area	Analyte Measured	CAS Number	Matrix Code	Units	Average Value	Maximum Value Detected	Reasonable Maximum Exposure
	Groundwater		<u></u>	<u></u>	<u></u>		
ST58	4-Methyl-2-pentanone	108-10-1	Water	ug/L	29	77	36.13
ST58	Benzene	71-43-2	Water	ug/L	29.35	180	57.81
ST58	Diesei	68334-30-5	Water	ug/L	8312	99000	21810
ST58	Gasoline	8006-61-9	Water	ug/L	20920	2.60e+05	56430
ST58	Lead	7439-92-1	Water	ug/L	76.5	180	103.8
ST58	Methylene chloride	75-09-2	Water	ug/L	2.7	9	3.641
ST58	Toluene	108-88-3	Water	ug/L	13.1	140	31.94
ST58	(Xyienes (total)	1330-20-7	Water	ug/L	71.46	830	184.3
	Groundwater		L.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	المحمد ويشبها	l	L	<b></b>
SS64	Aluminum	7429-90-5	Water	ug/L	1230	1460	1460
SS64	Arsenic	7440-38-2	Water	ug/L	10.1	13.1	13.1
<u>SS64</u>	Barium	7440-39-3	Water	ug/L	213	230	230
SS64	Bis(2-ethylhexyl)phthalate	117-81-7	Water	ug/L	3.967	5	5
SS64	Calcium	7440-70-2	Water	ug/L	54770	55100	55100
5564	Cobalt	7440-48-4	Water	ug/L	7.5	7.5	7.5
SS64	Copper	7440-50-8	Water	ug/L	24.3	36.1	36.1
SS64	Di-n-butylphthalate	84-74-2	Water	ug/L	3.133	3.6	3.6
SS64	liron	7439-89-6	Water	ug/L	5727	7770	7770
S\$64	Lead	7439-92-1	Water	ug/L	8.467	12.8	12.8
SS64	Magnesium	7439-95-4	Water	ug/L	13200	13300	13300
SS64	Manganese	7439-96-5	Water	ug/L	3347	5770	5770
S\$64	Nickel	7440-02-0	Water	ug/L	9.9	11.8	11.8
\$564	Sodium	7440-23-5	Water	ug/L	5437	5780	5780
S\$64	Tetrachloroethylene (PCE)	127-18-4	Water	ug/L	0.9233	1.7	1.7
S\$64	Trans-1.2-dichloroethene	156-60-5	Water	ug/L	0.8667	1.6	1.6
SS64	Trichloroethene (TCE)	79-01-6	Water	ug/L	1.067	2.7	2.7
S\$64	Vanadium	7440-62-2	Water	ug/L	7.65	7.71	7.7
SS64	Zinc	7440-66-6	Water	ug/L	23.3	23.4	23.4
Operable I	Jnit 5			A	h	·	
1	Groundwater			·····	······	·····	
LF02	Aluminum	7429-90-5	Water	ug/L	5245	8570	8570
LF02	Antunony	7440-36-0	Water	ug/L	5.325	19.8	16.68
LF02	Arsenic	7440-38-2	Water	ug/L	120.1	295	257.6
LF02	Barium	7440-39-3	Water	ug/L	238.6	392	392
LF02	Calcium	7440-70-2	Water	ug/L	49520	56400	56400
LF02	Chloride	Chloride	Water	ug/L	8265	16000	16000
LF02	Chromium	7440-47-3	Water	ug/L	14.9	19.9	19.9
LF02	Copper	7440-50-8	Water	ug/L	38.48	55.9	53.71
LF02	Di-n-butyiphthalate	84-74-2	Water	ug/L	1.75	27	2
LF02	Iron	7439-89-6	Water	ug/L	31600	60600	54450
LF02	Lead	7439-92-1	Water	ug/L	11.57	18.3	18.14
LF02	Magnesium	7439-95-4	Water	ug/L	17000	23200	22960
LF02	Manganese	7439-96-5	Water	ug/L	1345	1460	1460
LF02	Methylene chloride	75-09-2	Water	ug/L	0.7.	1.7	1.103
LF02	Nickel	7440-02-0	Water	ug/L	22.9	22.9	22.9

#### Table 9.1. (cont'd)

ſ		l	Υ```	l	1	Marimum	Ressonable
Source			Matrix		Average	Value	Maximum
Area	Analyze Measured	CAS Number	Code	Units	Value	Detected	Exposure
LF02	Potassiumi	7440-09-7	Water	ug/L	9250	11400	11400
LF02	Sodium	7440-23-5	Water	ug/L	13690	22400	21150
LF02	Sulfate	Sulfate	Water	ug/L	11450	39000	33200
LF02	Toluene	108-88-3	Water	ug/L	0.7833	2.2	1.354
LF02	Total dissolved solids	TDS	Water	ug/L	2.38e+05	3.10e+05	3.10e+05
LF02	Trans-1.2-dichloroethene	156-60-5	Water	ug/L	0.65	1.4	0.9523
LF02	Vanadium	7440-62-2	Water	ug/L	21.17	35.2	33.93
LF02	Zinc	7440-66-6	Water	ug/L	90.6	126	118.6
	Groundwater	· · · · · · · · ·			·····		
LF03/FT09	1.1.1-Trichloroethane	71-55-6	Water	ug/L	0.775	5	1.169
LF03/FT09	1.1-Dichloroethane	75-34-3	Water	ug/L	2.125	33	4.935
LF03/FT09	1.2-Dichlorobenzene	95-50-1	Water	ug/L	9.462	250	21.86
LF03/FT09	1.4-Dichlorobenzene	106-46-7	Water	ug/L	10.91	250	23.82
LF03/FT09	4.4'-DDD	72-54-8	Water	ug/L	0.058	0.21	0.07183
LF03/FT09	4-Methylphenol	106-44-5	Water	ug/L	314.8	6200	850.3
LF03/FT09	Aluminum .	7429-90-5	Water	ug/L	8136	37800	12340
LF03/FT09	Arsenic	7440-38-2	Water	ug/L	34.05	100	44.46
LF03/FT09	Barium	7440-39-3	Water	ug/L	358.9	1070	449.1
LF03/FT09	Benzene	71-43-2	Water	ug/L	1.695	20	3.374
LF03/FT09	Bis(2-ethylhexyl)phthalate	117-81-7	Water	ug/L	20.83	250	42.61
LF03/FT09	Bromide	Bromide	Water	ug/L	80	480	117.8
LF03/FT09	Cadmium	7440-43-9	Water	ug/L	1.276	11.7	2.424
LF03/FT09	Calcium	7440-70-2	Water	ug/L	90040	5.84e+05	1.35e+05
LF03/FT09	Chloride	Chloride	Water	ug/L	11270	99000	20300
LF03/FT09	Chromum	7440-47-3	Water	ug/L	20.34	31.8	24.2
LF03/FT09	Copper	7440-50-8	Water	ug/L	62.77	252	88.05
LF03/FT09	Di-n-butyiphthalate	84-74-2	Water	ug/L	17.09	250	38.29
LF03/F109	Dichlorodifluoromethane	75-71-8	Water	ug/L	33.44	650	89.54
LF03/F109	Dethyiphthalate	84-00-2	Witer	ug/L	17.06	250	38.26
LF03/F109	Eury Idenzene	100-41-4	water		4.460	36	5.72
LF03/F109	Linon	7439-89-0	Water	ug/L	08390	5.47c+05	1.300+03
LF03/F109		/*37-72+1	Walter	ug/L	23.03	01.9	31.37
1 503/5100	Magaacium	7430-05-4	Whee	ug/L	2.605	53700	24090
1 F03/F709	Manage	7430-06.4	Water	ugris ua/I	2652	7420	24300
1 603/ETD0	Methylene chloride	75.00.2	Water	up/1	12 98	250	34 54
LEO3/ETDO	Naphthelene	91-20-3	Water -	and t	12.76	250	38.26
LF03/FT09	Nickel	7440-02-0	Water	10g/J	38.28		40 04
LF03/FT09	o-Xviene	95-47-6	Water	ue/]	2.995	47	7 005
LF03/FT09	Phenol	108-95-2	Water	ur/L	17.25	250	38.43
LF03/FT09	Popassium	7440-09-7	Water	ue/L	10060	46600	15610
LF03/FT09	Sodium	7440-23-5	Water	ng/L	16440	91500	24630
LF03/FT09	Sulfate	Sulfate	Water	ug/L	9792	22000	11720
LF03/FT09	Terrachioroethylene (PCE)	127-18-4	Water	ug/L	2.888	53	7,448
LF03/FT09	Toluene	108-88-3	Water	ug/L	23.58	460	63.29
LF03/FT09	Total dissolved solids	TDS	Water	ug/L	5.02e+05	4.60e+06	8.77e+05

т	abl	e 9.	.1	. (	cont	d)
---	-----	------	----	-----	------	----

....

...

Southe		[	Marrix		Average	Maxumum Value	Reasonable
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposure
1.E03 ET09	Trichlomethene (TCE)	79-01-6	Water	ug/L	7.993	150	20.92
LF03/FT09	Trichlorofluoromethane	75-69-4	Water	ug/L	1.125	13	2.206
LF03/FT09	Vanadium	7440-62-2	Water	ug/L	20.79	43	29.89
LF03/FT09	Vinvi chloride	75-01-4	Water	ug/L	1.209	17	2.65
LF03/FT09	Zinc	7440-66-6	Water	ug/L	142.8	1030	241.3
	Surface Soil	J			£	<u>.</u>	
LF03/FT09	1.1.J-Trichloroethane	71-55-6	Soil	ug/kg	50.5	240	89.11
LF03/FT09	1,1-Dichloroethylene	75-35-4	Soil	ug/kg	75	300	120.8
LF03/FT09	2-Methylnaphthalene	91-57-6	Soil	ug/kg	748.9	4100	1528
LF03/FT09	4-Methylphenol	106-44-5	Soil	ug/kg	205.8	600	281.6
LF03/FT09	Acenaphthene	83-32-9	Soil	ug/kg	164.5	330	229.3
LF03/FT09	Aluminum	7429-90-5	Soil	ug/kg	6.94e+06	2.66e+07	1.08c+07
LF03/FT09	Arsenic	7440-38-2	Soil	ug/kg	4207	12300	6088
LF03/FT09	Banun:	7440-39-3	Soil	ug/kg	3.26e+05	2.70e+06	7.60e+05
LF03/FT09	Benzene	71-43-2	Soil	ug/kg	26.3	65	34.18
LF03/FT09	Benzo(a)anthracene	56-55-3	Soil	ug/kg	216.8	1500	354.6
LF03/FT09	Benzo(a)pyrene	50-32-8	Soil	ug/kg	216.3	1700	370.7
LF03/FT09	Benzo(b)fluoranthene	205-99-2	Soil	ug/kg	271.4	330	330
LF03/FT09	Benzo(ghi)perylene	191-24-2	Soil	ug/kg	232.6	1400	360
LF03/FT09	Benzo(k)fluoranthene	207-08-9	Soil	ug/kg	242.9	2100	430.5
LF03/FT09	Beryllium	7440-41-7	Soil	ug/kg	757.3	1725	1015
LF03/FT09	Bis(2-ethylhexyl) phthalate	117-81-7	Soil	ug/kg	209.4	1700	316
LF03/FT09	Burylbenzylphthalate	85-68-7	Soil	ug/kg	181.4	330	245.4
LF03/FT09	Cadmium	7440-43-9	Soil	ug/kg	344.4	1280	562
LF03/FT09	Calcium	7440-70-2	Soil	ug/kg	1.09e+07	5.65e+07	2.00e+07
LF03/FT09	Chromium	7440-47-3	Soil	ug/kg	17680	35700	22780
LF03/FT09	Chrysene	218-01-9	Soil	ug/kg	252.5	2300	456.3
LF03/FT09	Cobalt	7440-48-4	Soil	ug/kg	10310	30700	14340
LF03/FT09	Copper	7440-50-8	Soil	ug/kg	28090	85100	40910
LF03/FT09	Di-n-butylphthalate	84-74-2	Soil	ug/kg	175	330	240.5
LF03/FT09	Di-n-octylphthalate	117-84-0	Soil	ug/kg	206.1	330	257.1
LF03/FT09	Dibenz[a,h]anthracene	53-70-3	Soil	ug/kg	189.2	400	253.2
LF03/FT09	Diethyl phthalate	84-66-2	Soil	ug/kg	181.7	330	246.1
LF03/FT09	Ethylbenzene	100-41-4	Soil	ug/kg	46	100	57
LF03/FT09	Fluoranthene	206-44-0	Soil	ug/kg	380.8	4500	781.7
LF03/FT09	Indeno(1,2,3-cd)pyrene	193-39-5	Soil	ug/kg	240.3	1500	375.2
LF03/FT09	Iron	7439-89-6	Soil	ug/kg	1.72e+07	4.20e+07	2.23e+07
LF03/FT09	Kerosene	8008-20-6	Soil	ug/kg	16970	1.10e+05	39720
LF03/FT09	Lead	7439-92-1	Soil	ug/kg	12850	1.32e+05	21900
LF03/FT09	Magnesium	7439-95-4	Soil	ug/kg	5.40e+06	1.48c+07	7.28e+06
L,F03/FT09	Manganese	7439-96-5	Soil	ug/kg	3.01e+05	6.25e+05	3.77e+05
LF03/FT09	Mercury	7439-97-6	Soil	ug/kg	46.76	160.6	83.93
LF03/FT09	Methylene chloride	75-09-2	Soil	ug/kg	760	4400	1632
LF03/FT09	Molybdenum	7439-98-7	Soil	ug/kg	2137	7250	3422
LF03/FT09	Naphthalene	91-20-3	Soil	ug/kg	431.6	1500	685.5
LF03/FT09	Nickel	7440-02-0	Soil	ug/kg	22590	64300	30890

Table 9.1. (	(cont'd)
--------------	----------

<u> </u>		[		[		Maximum	Reasonable
Source Area	Analyte Measured	CAS Number	Code	Units	Average Value	Detected	Exposure
1 F03/FT09	Phenanthrene	185-01-8	ISoil	lug/kg	214.8	2200	412.5
LF03/FT09	Potassium	7440-09-7	Soil	ug/kg	9.80e+05	3.45e+06	1.44e+06
LF03/FT09	Ругеле	129-00-0	Soil	ug/kg	263.2	2400	475.9
LF03/FT09	Sodium	7440-23-5	Soil	ug/kg	6.04e+05	2.71e+06	1.01e+06
LF03/FT09	Thallium	7440-28-0	Soil	ug/kg	13130	29400	18290
LF03/FT09	Toluene	108-88-3	Soil	ug/kg	119	470	190.5
LF03/FT09	ТРН	Трн	Soil	ug/kg	1.97e+05	4.90e+06	4.16e+05
LF03/FT09	Trichloroethene	79-01-6	Soil	ug/kg	34	160	60.94
LF03/FT09	Trichlorofluoromethane	75-69-4	Soil	ug/kg	121.5	540	206.7
LF03/FT09	Vanadium	7440-62-2	Soil	ug/kg	42890	1.55e+05	64870
LF03/FT09	Zinc	7440-66-6	Soil	ug/kg	47570	1.68e+05	70740
	Subsurface Soil	····	h	·····	······································	····	
LF03/FT09	Acenaphthene	83-32-9	Sotl	ug/kg	14.17	60	32.64
LF03/FT09	Aluminum	7429-90-5	Soil	ug/kg	8.95c+06	2.41e+07	1.33e+07
LF03/FT09	Arsenic	7440-38-2	Soil	ug/kg	4088	8550	5886
LF03/FT09	Barium	7440-39-3	Soil	ug/kg	2.74e+05	1.31e+06	5.56e+05
LF03/FT09	Beryllium	7440-41-7	Soil	ug/kg	694.1	1220	951.4
LF03/FT09	Cadmium	7440-43-9	Soil	ug/kg	479.9	788	711.4
LF03/FT09	Calcium	7440-70-2	Soil	ug/kg	1.57e+07	6.54e+07	2.94e+07
LF03/FT09	Chromium	7440-47-3	Soil	ug/kg	21620	47800	29190
LF03/FT09	Cobalt	7440-48-4	Soil	ug/kg	11100	17300	13170
LF03/FT09	Copper	7440-50-8	Soil	ug/kg	33460	61400	42500
LF03/F109	DDE, pp'	72-55-9	Soil	ug/kg	1	2	2
LF03/F1091	DDT, pp	50-29-3	Soll	ug/kg	2	C	5
LF03/F T09	Iron	7439-89-0	Soll	ug/kg	1.79e+07	2.58e+07	2.06e+07
LF03/F109	Kerosene	8008-20-0	Sou	ug/kg	2.51e+05	1.396+00	5.81e+05
LFUS/FIVY	Lead	7439-92-1	Sou	ug/ug	230	11000	8/4/
LP03/E 109	Magnesium	1439-92-9	Sou	ug/kg	2.020+05	8.020+00	0./80+00
LF03/F107	Manganese	7439-70-3	Sou Coil	ug/kg	2.920-03	9.2367.03	3.000 - 03
LE03/6105	McBy che autrus	7430.08-7	Sou	ugrag	1665	1990	2602
EF03/1 109	Molybecken	7440-02-0	Sou	10/kg	24650	44100	30640
LE03/ET09	Don cilim	7440-09-7	Soil	10/20	1 22++06	1 74+ + 06	1 92+06
LF03/FT09	Sadium	7440-23-5	Sail	110/kg	4.48+05	6.50e+05	5.15e+05
LF03/FT09	Total dissolved solids	TDS	Soil	ug/kg	90.87	97.4	93.14
LF03/FT09	Total perroleum hydrocarbons	ТРН	Soil	ug/kg	4.66e+05	1.50e+07	1.20e+06
LF03/FT09	Vanadium	7440-62-2	Soil	ug/kg	53800	1.42e+05	78640
LF03/FT09	Zinc	7440-66-6	Soil	ug/kg	46600	68200	58510
	Groundwater			ليتستيها		d	
LF03- <b>sp</b>	1,1-Dichloroethane	75-34-3	Water	ug/L	33	33	33
LF03-sp	1,2-Dichlorobenzene	95-50-1	Water	ug/L	137	250	250
LF03-sp	1,4-Dichlorobenzene	106-46-7	Water	ug/L	166	250	250
LF03-sp	4-Methylphenol	106-44-5	Water	ug/L	6200	6200	6200
LF03-sp	Aluminum	7429-90-5	Water	ug/L	333	333	333
LF03-sp	Arsenic	7440-38-2	Water	ug/L	5.4	5.4	5.4
LF03-sp	Barium	7440-39-3	Water	ug/L	375	375	375

т	able.	9.1	1. (	(cont'd)
		~ -		

(	T		1	<u>1</u>	/*************************************	Maximum	Reasonable
Source		1	Matrix	1	Average	Value	Maximum
Area	Analyte Measured	CAS Number	Code	Units	Value	Detected	Exposute
LF03-sp	Benzene	71-43-2	Water	ug/L	20	20	20
LF03-sp	Cadmium	7440-43-9	Water	ug/L	1.9	1.9	1.9
LF03-sp	Calcium	7440-70-2	Water	ug/L	5.84e+05	5.84c+05	5.84e+05
LF03-sp	Chloride	Chloride	Water	ug/L	99000	99000	99000
LF03-sp	Chromium	7440-47-3	Water	ug/L	20.7	20.7	20,7
LF03-sp	Copper	7440-50-8	Water	ug/L	20.2	20.2	20.2
LF03-sp	Dichlorodifluoromethane	75-71-8	Water	ug/L	650	650	650
LF03-sp	Ethylbenzene	100-41-4	Water	ug/L	38	38	38
LF03-sp	līron	7439-89-6	Water	ug/L	5.45e+05	5.45e+05	5.45e+05
LF03-sp	Lead	7439-92-1	Water	ug/L	11	11	[]
LF03-sp	M.p-xylene	MPXYLENES	Water	ug/L	44	44	44
LF03-sp	Мадпезіцт	7439-95-4	Water	ug/L	49900	49900	49900
LF03-sp	Manganese	7439-96-5	Water	ug/L	7420	7420	7420
LF03-sp	Methylene chloride	75-09-2	Water	ug/L	250	250	250
LF03-sp	Nickel	7440-02-0	Water	ug/L	27.2	27.2	27.2
LF03-sp	o-Xylene	95-47-6	Water	ug/L	47	47	47
LF03-sp	Phenol	108-95-2	Water	ug/L	250	250	250
LF03-sp	Porassium	7440-09-7	Water	ug/L	46600	46600	46600
LF03-sp	Sodium	7440-23-5	Water	ug/L	91500	91500	91500
LF03-sp	Tetrachloroethylene (PCE)	127-18-4	Water	ug/L	53	53	53
LF03-sp	Toluene	108-88-3	Water	ug/L	460	460	460
LF03-sp	Total dissolved solids	TDS	Water	ug/L	4.60e+06	4.60e+06	4.60e+06
LF03-sp	Trichlomethene (TCE)	79-01-6	Water	ug/L	150	150	150
LF03-sp	Trichlorofiuoromethane	75-69-4	Water	ug/L	13	13	13
LF03-sp	Vinyl chloride	75-01-4	Water	ug/L	17	17	17
LF03-sp	Zinc	7440-66-6	Waer	ug/L	249	249	249
	Groundwater	l			L		
LF04	Arsenic	17440-38-2	Water	ug/L	120	120	120
LF04	Barium	7440-39-3	Water	ug/L	1227	1900	1900
LF04	Beryllium	7440-41-7	Water	ug/L	2.6	4.8	4.8
LF04		7440-70-2	Water	ug/L	1.83e+05	2.40e+05	2.40e+05
LF04	Chloride	16887-00-6	Water	ug/L	16000	16000	16000
LF04	Chromium	7440-47-3	Water	ug/L	109	200	200
LF04	Cobalt	7440-48-4	Water	ug/L	69.33	120	120
LF04	Copper	7440-50-8	Water	ug/L	143.3	260	260
LF04	Fluoride	7782-41-4	Water	ug/L	1700	1700	1700
LF04	h-Chloroftuorobenzene	PPP-PP-P	Water	ug/L	9	9	9
LF04	Iron	7439-89-6	Water	ug/L	1.72e+05	2.70e+05	2.70e+05
LF04	Lead	7439-92-1	Water	ug/L	25	25	25
LF04	Magnesium	7439-95-4	Water	ug/L	96670	1.40e+05	1.40e+05
LF04	Manganese	7439-96-5	Water	ug/L	3800	5500	5500
LF04	Nickel	7440-02-0	Water	ug/L	186.7	330	330
LF04	p-Chlorofluorobenzene	000-00-0	Water	ug/L	8.8	8.8	8.8
LF04	Potassium	7440-09-7	Water	ug/L	11700	16000	16000
LF04	Sodium	7440-23-5	Water	ug/L	46330	67000	67000
LF04	Sulfare	12808-79-8	Water	ug/L	2700	2700	2700
	L		L	L	t	L	

Т	abl	e	9.	1		(cont	'd)
---	-----	---	----	---	--	-------	-----

÷

[	1	Τ	1	Τ	T T	Maximum	Reasonable
Source	A State Management	Oto Number	Marrix		Average	Value	Maximum
Агеа	Analyte Measured		Code	Unics	Value	Detected	Exposure
LF04	Total dissolved solids		Water	ug/L	820	1000	1000
LF04	Vanadium	7440-62-2	Water	ug/L	141.7	350	350
LF04	Zinc	7440-66-6	Water	ug/L	276.7	510	510
	Surface Soil						
LF04	2-Methylnaphthalene	91-57-6	Soil	ug/kg	63.75	210	178.5
LF04	Barium	7440-39-3	Soil	ug/kg	4.09e+05	1.60e+06	7.89c+05
LF04	Beryllium	7440-41-7	Soil	ug/kg	306.2	1400	602.3
LF04	Bis(2-ethylhexyl)phthalate	117-81-7	Soil	ug/kg	36.25	70	62.72
LF04	Calcium	7440-70-2	Soil	ug/kg	7.31e+06	1.10e+07	8.74c+06
LF04	Chromium	7440-47-3	Soil	ug/kg	18880	23000	21240
LF04	Cobali	7440-48-4	Soil	ug/kg	7250	9000	8109
LF04	Copper	7440-50-8	Soil	ug/kg	30120	44000	35540
LF04	lron	7439-89-6	Soil	ug/kg	1.60e+07	1.80e+07	1.72c+07
LF04	Magnesium	7439-95-4	Soil	ug/kg	4.35e+06	5.60e+06	4.87c+06
LF04	Мапеалезе	7439-96-5	Soil	ug/kg	3.06e+05	3.70c+05	3.30e+05
LF04	Nickel	7440-02-0	Soil	ug/kg	18250	26000	21040
LF04	Potassium	7440-09-7	Soil	we/kg	1.79e+06	2.10c+06	1.95e+06
LF04	Sodium	17440-23-5	Soil	ug/kg	4 85e+05	5.80e+05	5 37e+05
LF04	Трн	Трн	Soil	ue/kg	59390	3 70e+05	92110
1.F04	Wanaduum	7440-62-2	Soil	ug/kg	35120	46000	39800
T F04	7 inc	17440-66-6	Soil	hig/kg	59880	1 00+05	71850
	Subsurface Soil		<u> </u>		L	L	
L F04	17 4-Dinitrotoluene	1121-14-2	Isoil	her/kg	77.5	120	120
LI FOA	Ris(7-erhylhexyl)phthalate	117-81-7	ISnil	hiø/kg	46.43	190	72.36
T FOA	DischeryInhthalan		Isoil	nø/kg	503.1	6300	1127
1 F04	In Nitrosodiohenvlamine	86-30-6	Soil	hø/kg	140	530	445,9
I FOA	Трц	-тры	Soil	no/kg	60970	5 14e+05	08420
	Groundwater		<u> </u>	<u>م. ، 6. ر</u>		P.1	
I FOA	A huminum	17479-90-5	Ru <sub>ater</sub>	h.a/1.	6075	9360	9360
LFOG	Authinan	7440-38-2	Tilater	148/2 ; hug/[	31.6	38.3	38 3
LT-00	Arsenic Ig'a	7440-20-2	Tuher	U872-	285.2	184	177 1
I Enk	Barron Biel3 sehulberuthehibalate		Tuhner		444.0		
LTOG	1015(2+eutymexytyphia	11/-01	TULL			Įį	0.0101
LFUG			Water		65420	73600	71600
LIVU			Waw.		2000	4700	4464
LFUG		7440 47.2	Water		18	20.2	20.2
LFVG		7440-41-3	Watts		<0 65	20,2 P2 1	92.1
LIVO			Water			<u> </u>	4 670
LIV	Di-n-outyiphuaiae		Water		4.2	ļ	9.330
LIVO		84-00-2	WAICI		4.73		. 767
LFVO	Eunyibenzene		Water	Ug/L	V.0/3		1./3/
LFOO	lion	7439-89-0	Water	ug/L	10000	21500	21280
LFU6	Lead	7439-92-1	Water		17.43	23.7	23.7
LF06	Magnesium	7439-95-4	Water	ug/L	16820	21200	21200
LF06	Manganese	7439-96-5	Water	ug/L	2018	2790	2655
LF06	Nicket	7440-02-0	Water	ug/L	30.9	31.8	31.8
LF06	Potassium	7440-09-7	Water	ug/L	5440	5440	5440

T	ab	le	9.	1.	(cont'd)	
---	----	----	----	----	----------	--

Source Area	Analyte Measured	CAS Number	Matrix Code	Units	Average Value	Maximum Value Detected	Reasonable Maximum Exposure
LF06	Sodium	7440-23-5	Water	ug/L	8875	9520	9520
LF06	Sulfate	Sulface	Water	ug/L	11080	18000	17640
LF06	Total dissolved solids	TDS	Water	ug/L	2.50e+05	2.70e+05	2.70e+05
LF06	Vanadium	7440-62-2	Water	ug/L	22.92	30.1	30.1
LF06	Zinc	7440-66-6	Water	ug/L	68.62	84.6	84.6

Table 9.1. (cont'd)

ىدەرىد. بەرمى - ---

	-								
Hazard Quoti	ent <u>à</u> 0.100 w	ith rounding	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child	
ble Unit 3							[		
Ingestion	Soil	Chromium 6	<	<	<	<			
		Manganese	<	<	0.135	<			
		Anthracene	<	<	<	<			
		Inorganic Sum	0.0185	0.0369	0.144	0.0185			
Inhalation	Soil	Chromium 6	0.108	0.108	0.157	0.108			
(		Manganese	<	<	<.	<		•-	
		Inorganic Sum	0.117	0.117	0.17	0.117			
Ingestion	Groundwater	Manganese	9	13.9	25.2	<	-		
		Trichlorgethene	0.812	1.25	2.27	<			
	[	Organic Sum	0.812	1.25	2.27				
		Inorganic Sum	9	13.9	25.2	-	_	_	
Demial	Groundwater	Manganese	0.184	0.19	0.257	<			
i	·····	Trichloroethene	<	<	<	<			
	·····	Loorganic Sum	0.184	0.19	0.257				
Ingestion	Soil	Chromium 6	<	<	<	< .	_		
	L	Manganese	<	<	0.114	<			
		Vanadium	<	<	<	<			
		Anthracene	< ·	<	<	<			
_		Trichtoroethene	<	<	<	<			
		Inorganic Sum	0.016	0.0319	0.124	0.016			
Inhalation	Soil	Chromium 6	<	<	0.0696	<			
		Manganese	<	<	<	< .	_		
		Inorganic Sum	0.0542	0.0542	0.0789	0.0542			
Ingestion	Groundwater	Manganese	4.69	7.25	13.2	. <			
-	······	Vanadium	0.206	0.319	0.578	<	_		
	1	Trichloroethene	0 23	0.355	0.644	<			
		Organic Sum	0.23	0.355	0.644			<u> </u>	
-		Inorganic Sum	4.9	7.57	13.7	_			
Dermal	Groundwater	Manganese	0.0959	0.0991	0.134	<.			
		Vanadium	<	<	<	<			
		Trichloroethene	<	<	<	<		 	
		Inorganic Sum	0.127	0   31	0177				
	Hazard Quoti ble Unit 3 Ingestion Inhafation Ingestion Ingestion Ingestion	Hazard Quotient à 0,100 w ble Unit 3 Ingestion Soil Ingestion Soil Inhalation Soil Dermal Groundwater Ingestion Soil Ingestion Soil	Hazard Quotient & 0.100 with rounding         ble Unit 3         Ingestion       Soil         Ingestion       Soil         Ingestion       Soil         Ingestion       Anthracene         Inhalation       Soil         Inhalation       Soil         Inhalation       Groundwater         Manganese       Inorganic Sum         Ingestion       Groundwater         Manganese       Trichloroethene         Organic Sum       Inorganic Sum         Ingestion       Groundwater         Manganese       Trichloroethene         Demial       Groundwater         Manganese       Inorganic Sum         Ingestion       Soil       Chromium 6         Ingestion       Groundwater       Manganese         Inhalation       Soil       Chromium 6         Ingestion       Groundwater	Hazard Quotient $\frac{1}{4}$ 0.100 with rounding       Future Worker         ble Unit F	Hazard Quotient & 0.100 with rounding         Future Worker         Future Resident Worker           Ingestion         Soil         Chromium 6         <	Hazard Quotient $\dot{a}$ 0.100 with roundingFuture VorkerFuture Resident Resident Resident Residentbe Unit 3Imagenesis<	Hazard Quotient & 0.100 with rounding         Future Worker         Future President         Future Resident         Future Worker           Ingestion         Soil         Chromium h         <	Hazard Quotient $\frac{3}{2}$ 0.100 with rounding         Future Variable Values         Future Variable Values         Future Variable Values         Current Values<	

Table 9.2a	Cancer	Risks for	Source	Areas in	OUs 3.	. 4	, and	5
------------	--------	-----------	--------	----------	--------	-----	-------	---

	Hazard Quoti	ent ĝ. 0.100 w	ith rounding	Future Work <b>e</b> r	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
ST56	T56 Ingestion Groundwater		Arsenic	0.479	0.74	1.34	<		
			Manganese	4.48	6.92	12.5	<		
i			Tetrachloroethene	<	<	<	<		
			Inorganic Sum	4.96	7.66	13.9		_	
	Dermal	Groundwater	Arsenic	0.0568	0.0587	0.0795	<		_
			Manganese	0.0915	0.0945	0.128	<		-
[	. {		Tetrachloroethene	<	<	<	<		-
<u>.</u>			Inorganic Sum	0.148	0.153	0.208			
SS57	Ingestion	Groundwater	Toluene	0 0929	0.143	0.26	<		
			Organic Sum	0.0929	0.143	0.26			
	Inhalation	Groundwater	Toluene	0.465	0.511	0.651	<	-	_
			Organic Sum	0.465	0.511	0.651		-	-
SS61	Ingestion	Soil	Arsenic	<	<	<	<	_	
			Barium	<	<	<	<	_	-
			Chromium 6	<	<	<	<		-
			Manganese	<	<	<	<		-
			Vanadium	<	<	<	<		-
			Trichloroethene	<	<	<	<	_ 1	-
			Inorganic Sum	0.0122	0.0244	0.0947	0.00663	-	
	Inhalation	Soit	Chromium 6	0.101	0.101	0.147	0.0695	-	-
			Manganese	<	<	<	<		-
			Inorganic Sum	0.108	0.108	0.158	0.0742	+	-
ssõi	Ingestion	Groundwater	Arsenic	2.65	4.09	7.42	<		-
· ···			Barium	0.187	0.2 <b>89</b>	0.525	<	-	-
			Beryllium	<	<	<	<	-	-
			Cadmium	<	<	0.0603	<	-	_
<u>`</u>			Chromium 6	0.11	0.169	0.307	<		-
			Manganese	17.3	26.6	48.3	<	-	
			Vanadium	0.179	0.277	0.502	<	_	-
			Pentachlorophenol	<	<	<	<		-
			Gasoline	<	<	<	<		-
			Trichloroethene	0.262	0.405	0.735	<	-	_
Ì			Organic Sum	0.288	0.445	0.808	-	-	-

Table 9.2a. (cont'd)
	Hazard Quoti	ent ā 0 100 w	ath rounding	Future Work <b>e</b> r	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Inorganic Sum	20.4	31.5	57.2			•-
	Dermal	Groundwater	Arsenic	0.314	0.324	0.439	<		
			Barium	<	<	<	<	-+	
			Beryllium	<	<	<	<		_
			Cadmium	<	<	<	<	-	
•			Chromium 6	0.074	0.0764	0.104	<		
1			Manganese	0.352	0.3 <del>6</del> 4	0.493	<	-	
			Vanadium	<	<	<	<		
· · · · · · · · · · · · · · · · · · ·			Pentachiorophenoi	0.302	0.203	0.422	<		
	1		Trichloroethene	<	<	< .	<		-
			Organic Sum	0.302	0.203	0.423		-	
			Inorganic Sum	0.773	0.798	1.08	••		
SS61-sp	Ingestion	Groundwater	Gasoline	0.0631	0.0974	0.177	<		
			Trichloroethene	1.08	1.67	3.03	<	-	
			Organic Sum	1.15	1.77	3.21			
Operat	le Unit 4								
DP25	Ingestion	Groundwater	Arsenic	1.01	1.56	2.83	<		_
			Manganese	12.9	19.9	36.2	<	_	
			Toluene	0.0592	0.0914	0.166	<		
			Organic Sum	0.0592	0.0914	0.166			
			Inorganic Sum	13.9	21.5	39	-	_	-
	Inhalation	Groundwater	Tolucne	0.296	0.325	0.414	<		
	}		Organic Sum	0.296	0.325	0.414	-	· -	-
	Dermai	Groundwater	Arsenic	0.12	0.124	0.168	<	-	
			Manganese	0.264	0.272	0.369	<	-	_
			Toluene	<	<	<	<		
			Inorganic Sum	0.383	0.396	0.537			
	Ingestion	Vegetables	Heptachlor epoxide	<	<	0.0672	<	_	-
			Dieldrin	<	<	<	<	-	_
			Toluene	<	<	<	<		
			Organic Sum	_		0.0677	<u> </u>	-	
ST27	Ingestion	Groundwater	Arsenic	0.665	1.03	1.86	<		-
		ļ	Chromium 6	<		0.0767			

Table 9.2a. (cont'd)

	Hazard Quoti	ent <u>a</u> 0.100 w	ith rounding	Future Worker	Future Typical Resident	Future Resid <del>e</del> nt	Current Worker	Current Playing Child	Future Playing Child
			Copper	<	0.0771	0.14	<		
			Manganese	11.4	17.5	31.8	<		
			Inorganic Sum	12.1	18.7	33.9			
	Dermal	Groundwater	Arsenic	0.0788	0.0814	0.11	<		
			Chromium 6	<	<	<	<	-	
		}	Соррег	<	<	<	<		
			Manganese	0.232	0.24	0.325	<	-	
· .			Inorganic Sum	0.33	0.341	0.462	-		
WP33	Ingestion	Groundwater	Arsenic	0.593	0.916	1.66	~	1	
			Manganese	17,3	26.7	48.5	<	-	
			Inorganic Sum	17.9	27.6	50.2		-	_
WP33	Dermal	Groundwater	Arsenic	0.0703	0.0726	0.0984	<	-	
			Manganese	0.354	0.365	0.495	· <		-
			Inorganic Sum	0.424	0.438	0.593	-		
SS35	Ingestion	Soil	Beryllium	<	<	<	<		_
			Chromium 6	<	<	<	<		-
			Manganese	<	0.0627	0.244	<		
			Chlordane	<	<	<	<	-	_
			DDT	0.0572	0.114	0.444	<		-
_		]	Heptachlor epoxide	<	< .	<	<		-
			Aldrin	<	<	<	<		
ا يوني			Organic Sum	0.0582	0.116	0.451	0.00492		_
			Inorganic Sum	0.0323	0.0644	0.25	0.0139	-	-
·- ـ ـ بې	Inhalation	Soil	Chromium 6	0.081	0.081	0.118	0.0644	_	-
			Manganese	<	<	<	<	-	-
			Inorganic Sum	0.0974	0.0974	0.142	0.0713	-	-
	Dermal	Soîl	Beryllium	<	<	<	<		-
			Chromium 6	<	<	<	. <		-
			Manganese	<	<	0.0529	<	-	-
			Chlordanc	<	< .	<	<		
			DDT	0.356	0.499	0.964	<	_	
		]	Heptachlor epoxide	<	<	<	<	-	-
			Aldrin	<	<	<	<		-

Table 9.2a. (con	t'd)	t'd)
------------------	------	------

	Hazard Quotic	ent <i>&amp; 0</i> 100 w	ith rounding	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Organic Sum	0.362	0.508	0.98	0 0306		
			Inorganic Sum	0 0201	0 0282	0.0544	0.00867		
L	ingestion	Vegetables	Beryllium	<	<	<	<		
			Manganese	<	<	<	<		
			Chlordane	<	<	<	<		
			DÐT	<	<	<	<		
			Heptachlor epoxide	<	<	<	<	·	
			Aldrin	<	<	<	<		
	·		Organic Sum			0.0582			
	Ingestion	Sheilfish	Beryllium	<	<	<	<		
			Manganese	<	<	< 1	<		
			Chilordane	<	<	<	<	•-	
			DDT	<	<	<	<		-
	1		Heptachior epoxide	<	<	<	<		-
			Aldrin	<	<	<	<		
			Organic Sum				- 1	0,104	0.0814
 SS36	Ingestion	Soil	Arseniç	<	<	0.122	<	-	
			Beryilium	<	<	<	<	-	
			Chromium 6	0.0666	0.133	0.517	<		
		····	Manganese	<	<	<	<		
			DDT	<	<	<	<		
			Inorganic Sum	0.0871	0.174	0.675			
	Inhalation	Soil	Chromium 6	L.79	1.79	2.6	<	-	
			Manganese	<	<	<	<	-	-
		[	Inorganic Sum	1.79	1.79	2.6	I	_	
	Dermal	Soil	Arsenic	<	<	<	<	-	
			Beryllium	<	<	<	<		_
,_,			Chromium 6	<	0.0581	0.112	<		
<del></del>			Manganese	<	<	<	<		
			DDT	<	<	<	< ,		
			Inorganic Sam	0.0542	0.076	0.147			-
	Ingestion	Groundwater	Manganese	8.61	13.3	24.1	<		
				-				-	

Table 9.2a. (cont'd)

-...

	Hazard Quoti	ent G 0.100 w	ith rounding	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Ptaying Child
SS36	Dermal	Groundwater	Manganese	0176	0.182	0.246	<		
			Inorganic Sum	0176	0.182	0.246	i		
			Organic Sum					0.104	0.0814
SS37	Ingestion	Groundwater	Агѕеліс	1.63	2.52	4.58	<		
			Chromium 6	<	<	<	<	•-	
	1		Manganese	5.07	7.82	14.2	<		
			Inorganic Sum	6.7	10.4	18.8			
	Dermai	Groundwater	Arsenic	0.194	0.2	0.271	<		
		[	Chromium 6	<	<	<	<	·	
	·····		Мапралезе	0.103	0.107	0.145	<		-
			Inorganic Sum	0.301	0.311	0.421			
SS39/63	Ingestion	Soil	Arsenic	<	<	<	<		-
			Beryllium	<	<	<	<		-
			Chromium 6	<	<	<	<		
			Ma⊓ganes¢	<	<	0.0863	<		
			DDT	<	<	<	<		
			Inorganic Sum	0.0181	0.0361	0.14	0.0118		
	Inhalation	Soil	Chromium 6	0.111	0.111	0.162	0.111	••	
		L	Manganese	<	<	<	<	·	_
			Inorganic Sum	0.117	0.117	0.17	0.117		
	Ingestion	Groundwater	Manganese	5.2	8.03	14.6	<	-	-
	1		Inorganic Sum	5.2	8.03	14.6	-		
. هم ون 	Dermai	Groundwater	Manganese	0.106	0.11	0.149	<		-
			Inorganic Sum	0.106	0.11	0.149	-	L	
ST58	Ingestion	Groundwater	Gasoline	2.76	4.26	7.73	· <	_	· _
			Organic Sum	2.76	4.26	7.73			
SS64	Ingestion	Groundwater	Arsenic	0.427	0.659	1.2	<		-
			Manganese	11.3	17.4	31.6	<		
			Tetrachloroethene	<	<	<	<		_
			Trichloroethene	<	<	<	<		
			Inorganic Sum	11.7	18.1	32.8	i		-
	Dermal	Groundwater	Arsenic	0.0506	0.0523	0.0708	<		
	<u> </u>	<u> </u>	Manganese	0.231	0.238	0.323	<		-

Table 9,2a.	(cont'd)
-------------	----------

					Future			Ситеот	Future
) (	azard Quote	ent <i>à 1</i> 0100 w	ith rounding	Future Worker	Typical Resident	Future Resident	Current Worker	Playing Child	Playing Child
			Tetrachloroethene	<	<	<	<		
		l	Trichloroethene	<	<	<	<		
			Inorganic Sum	0.281	0.29	0.393		-	
Operable	: Unit 5								
LF02	Ingestion	Groundwater	Arsenic	841	13	23.6	<		
			Chronitum 6	<	0.0601	0.109	<		
	[		Manganese	2.86	4.41	8	<		
	[	<b> </b>	tnorganic Sum	11.3	17.5	31.7		<b></b> .	
	Dermai	Groundwater	Arsenic	0.997	1.03	1.39	<		
	1		Chromium 6	<	<	<	<		
[			Manganese	0.0583	0.0603	0.0816	<	•-	
			Inorganic Sum	1.08	1.12	1.51			
LF03/FT09	Ingestion	Soil	Arsenic	<	<	<	<	-	
	<b>†</b>	· · · · · · · · · · · · · · · · · · ·	Barium	<	<	<	<		
		· · · · · · · · · · · · · · · · · · ·	Beryllium	<	<	<	<		
	[		Chromium 6	<	<	<	<		-
			Manganese	<	<	0.115	<		
<b></b> ,	[		1.1.1-Trichloroethane	<	<	<	<		
			Kerosene	<	<	<	<		-
			1.1 Dichloroethylene	<	· <	<	<		
		· ·	Trichloroethene	<	<	<	<		
	<b>†-</b>		Inorganic Sum	0.0221	0.044	0.171	0.0218		-
LF03/FT09	Inhalation	Soil	Chromium 6	0.268	0.268	0.39	0.209	=	
			Manganese	<	<	<	<		
	1		Inorganic Sum	0.286	0.286	0.416	0.227	_	
	Dermal	Soit	Arsenic	<	<	<	<	_	
	<b>1</b>	······································	Barium	<	<	<	<		
	[		Beryllium	<	<	<	<		
	ţ		Chromium 6	<	<	<	<		
			Manganese	<	<	<	<		
	<u>+</u>		1.1.1-Trichloroethane	<	<	<	<		
	<u> </u>		Kerosene	<	<	0.0959	<		
			1.1 Dichloroethylene	<	<	<	<		

Table 9.2a. (cont'd)

Table	9.2a.	(cont	d)
-------	-------	-------	----

			Concerned with the second of the second s			·····			
E Fi	lazard Quoti	ent @ 0.100 w	ith rounding	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Trichloroethene	<	<	<	<	_	
·			Organic Sum	0.0355	0.0497	0.0959	0.00245		
	Ingestion	Groundwater	Arsenic	1.45	2.24	4.06	<		
[			Barium	0.0627	0.0969	0.176	<		
			Chromium 6	<	0.0731	0.133	<	-	-
[		Í	Мапganese	6.42	9.91	18	<		
·			1.1.1-Trichloroethane	<	<	<	<	-	-
			1.4-Dichlorobenzene	<	<	<	<		-
i.			Tetrachloroethene	<	<	<	<		-
			Trichloroethene	<	0.0526	0.0954	<	-	_
[ ]		[	Organic Sum	0.0418	0.0645	0.117		-	
		-	Inorganic Sum	7.98	12.3	22.3	_	_	-
	Dermal	Groundwater	Arsenic	0.172	0.178	0.241	<	••	-
[			Barium	<	<	<	<		1
[			Chromium 6	<	<	<	<	-	_
l			Manganese	0.131	0.135	0.183	<		-
			I,I.I-Trichloroethane	<	<	<	<		-
			1,4-Dichlorobenzene	<	<	<	<	-	-
Ĺ			Tetrachloroethene	<	<	<	<		
/			Trichloroethene	<	<	<	<	•	-
			Inorganic Sum	0.336	0.347	0.47		-	
	Ingestion	Vegetables	Beryllium	<	<	<	<	-	_
ंभ			Manganese	<	<	<	<	-	
			1,1,1-Trichloroethane	<	<	<	<	-	-
			Kerosene	<	<	0.0688	<		-
[			1.1 Dichloroethylene	<	<	<	<		
			Trichloroethene	<	<	<	<		-
	1		Organic Sum		-	0.0695	-	_	-
LF03-sp	Ingestion	Groundwater	Arsenic	0.176	0.272	0.493	<		-
			Cadmium	<	0.0574	0.104	<		
			Chromium 6	<	0.0625	0.113	<		
			Manganese	14.5	22.4	40.7	<		
			1,4-Dichlorobenzene	<	<	<	<		_

	Hazard Ouse	ent ā 11400 u	ith rounding	Future	Future Typical Resident	Future	Current	Current Playing Child	Future Playing
			Tetrachloroethene	0.0518	() DR	0 145			
			Trichloroethene	0.245	0.378	0.685	   <		
				0.799	0.462	0.839			
			Inorganic Sum	14.8	27.8	414			 
	Dermal	Groundwater	Arsenic	<	<	<	 <	 	
			Cadmium	<	<	<	<		
			Chromium 6	<	<	<	. <		
			Manganese	0.197	0.306	0.415	<		
			1.4-Dichlorobenzene	<	<	<	<		
			Tetrachloroethene	<	<	<	<		
		<u> </u>	Trichloroethene	<	<	<	<		
			Inorganic Sum	0.348	0.359	0.487			
			Organic Sum		-	0.0695			
.F04	Ingestion	Soil	Barium	<	<	<	<		 
			Bervillium	<	<	<	<		
			Chromium 6	<	<	<`	~ ~		
			Manganese	<	<	0.1	<		
			Nickel	< .	<	<	<		
			Vanadium	<	<	<	<		
<b></b>		Ì	Inorganic Sum	0.0171	0.0341	0.132	0.0171		
	Initialation	Soil	Chromium 6	0.111	0.111	0.162	0.111	   •-	
			Manganese	<	<	<	<		
			Inonzanic Sum	0.12	0.12	0.175	0.12		
	Ingestion	Groundwater	Arsenic	3.91	6.04	11	<		-
	-, <b>,</b>		Barium	0.265	0.41	0.744	<		
			Beryllium	<	<	<	<		
		<u>}</u>	Chromium 6	0.391	0.604	1.1	<		
	-1		Manganese	10.8	16.6	30.1	<		
		<b>*</b>	Nickel	0.161	0.249	0.452	<		
		<u> </u>	Vanadium	0.38	0.587	L.07	<		
		·	Inorganic Sum	15.9	24.5	44.5			
	Dermai	Groundwater	Arsenic	0.464	0.479	0.649	<	_	
			Barium	< 1	<	<	<		

Table 9.2a. (cont'd)

	Hazard Quoti	ent ä 0.100 w	ith rounding	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Bery Ilium	<	<	<	<		
		[	Chromium 6	0.264	0.272	0.369	<		
			Manganese	0.22	0.227	0.308	<		
	1		Nickel	<	<	<	<		
		1	Vanadium	0.0567	0.0586	0.0794	<		
		]	tnorganie Sum	1.01	1.05	1.42			-
LF06	Ingestion	Groundwater	Arsenic	1.25	1.93	3.5	<		
<b>1</b>			Chromam 6	<	0.061	0.111	< 1		
			Manganese	5.2	8.03	14.6	<	_	
1			Inorganic Sun:	6.49	10	18.2			
`   `	Dermal	Groundwater	Arsenic	0.148	0.153	0.207	<		
	[		Chromium 6	<	<	<	<	+=	
		[	Manganese	0.106	0.11	0.149	<		-
		[	Inorganic Sum	0.281	0.29	0.393		±	-

Table 9.2a. (cont'd)

Note: the < symbol denotes values less than 0.1 with round-off.

the -- symbol means that the computation does not apply.

A "future typical resident" exposure is for 275 days per year and a "future resident" exposure is for 350 days per year

neta Unine Maria

Can¢	er Risk ä <sub>.</sub> te	406 in Water (j rounding	i te-0° in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
Opera	ble Unit 3	[			ļ				
DP44	Ingestion	Soil	Benzotaipyrene	9.18e-07	6.61e-07	8.58e-06	9.18e-07	<	<
			Benzo(b)fluoranthene	1.07e-05	7.71e-06	1000.0	1.07e-05	<	· <
			Benzo(g.h.i)pervlene	7.14e-0*	5.14e-07	6.67c-06	7.14e-07	<	<
			Anthracene	<	<	3.66e-07	<	<	<
			Chrysene	<	<	1.00c-07	<	<	<
			Dibenz(a,b)anthracene	3.32 <b>e-</b> 06	2.39e-06	3.10e-05	3.32e-06	<	<
_			Indeno(1.2.3-cd)pyrene	7.65e-07	5.51e-07	7. <b>15e-06</b>	7.65c-07	<	<
		)	Benz(a)anthracene	2.45e-06	1.76e-06	2.29e-05	2.45e-06	<	<
			Organic Sum	1. <b>89e-0</b> 5	1.36e-05	0.00018	1.89e-05	-	
	Inhalation	Soil	Chromium 6	9.00e-07	3 25e-07	1.58e-06	9.00e-07	<	<
		-	Anthracene	<	<	<	<	<	<
			Inorganic Sum	9.00e-07	3.25e-07	1.58e-06	9.00 <del>c-0</del> 7		
	Dermai	Soil	Benzo(a)pyrene	5.73 <b>e-</b> 06	2.89e-06	1.85e-05	5.73 <b>e-</b> 06	<	<
			Benzo(b)fluoranthene	6.68e-05	3.37e-05	0.00022	6.68e-05	<	<
	1		Benzo(g.h.i)perylene	4.46e-06	2.25 <b>e-0</b> 6	1.44e-05	4.46e-06	<	<
		ľ	Anthracene	2.45 <b>e-0</b> 7	1.23c-07	7.91e-07	2.45e-07	<	<
	1		Chrysene	6.68e-08	<	2.16e-07	6.68c-08	<	<
			Dibenz(a,h)anıhracene	2.07e-05	1.04c-05	6.69e-05	2.07e-05	<	<
	- <u> </u>		Indeno(1,2,3-cd)pyrene	4.77e-06	2.41 <b>e-06</b>	1.54e-05	4.77 <b>e-</b> 06	<	<
	1		Benz(a)anthracene	1.53e-05	7.71e-06	4.94e-05	1.53e-05	<	<
			Or <u>e</u> anic Sum	0.00012	5.96e-05	0.00038	0.00012		
	Ingestion	Groundwater	Вслдепе	<	<	7.19 <b>e-</b> 07	<	<	<
	1		Trichloroethene	1.91c-05	1.06e-05	6.41c-05	<	•<	<
		[	Отдаліс Sum	1.93e-05	1.070-05	6.48e-05	-		
	Inhalation	Groundwater	Велгене	2.15e-06	8.51e-07	3.60e-06	<	<	<
			Trichloroethene	0.0001	4.14c-05	0.00018	< .	<	<
		]	Organic Sum	0.00011	4.22e-05	81000.0			
	Ingestion	Vegetables	Benzo(a)pyrene	<	<	7.11c-07	<	<	<
		1	Benzo(b)fluoranthene	<	<	8.29e-06	<	<	<
<del></del>			Benzo(g.h.i)perviene	<	<	2.77 <b>∈</b> -07	<	<	<
	_[	[	Anthracene	<	<	2.60e-07	<	<	
	1	1	Chrysene	< -	<	<	<	<	<
······································		I	ka	<b>L</b>	I	اس مستحد الم			·

#### Table 9.2b. Noncancer Health Effects for Source Areas in OUs 3, 4, and 5

Салс	er Risk ä Te	-06 in Water a rounding	le-07 in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Dibenz(a,h)anthracene	<	<	4.68e-06	<	<	<
			Indeno(1,2,3-ed)pyrene	<	<	3.30e-07	<	<	<
		 	Benz(a)anihracene	<	<	3.50e-06	<	<	<
	_		Organic Sum			1.81e-05			
WP45	Ingestion	Soil	Benzo(b)fluoranthene	<	<	7.25e-08	<	<	<
 		· · · · · · · · · · · · · · · · · · ·	Benzo(g.h.i)perylene	<	<	8.4 <b>∔e-0</b> 8	<	<	<
 		i i	Anthracene	<	<	<	<	<	<
··-			Dibenz(a,h)anthracene	<	× _	1.19e-07	<	<	<
			Indeno(1,2,3-cd)pyrene	<	<	6.72e-08	<	<	<
		 	Benz(a)anthracene	<	<	6.24e-08	<	<	<
		[	Benzo(a)pyrene	<	<	7.67e-08	<	<	<
· 			Trichloroethene	<	<	7.25e-08		<	<
			Organic Sum	6.10 <b>e-08</b>	4.39e-08	5.70e-07	5.57e-08		
	Inhalation	Soil	Trichloro <b>ethe</b> ne	8.75e-08	<	1.53e-07	8.75 <b>c-0</b> 8	<	<
			Organic Sum	8.75c-08	3.16e-08	1.53e-07	8.75e-08		
l 	Inhalation	Soil	Chromium 6	4.00e-07	1.44e-07	6.99e-07	4.00e-07	<	<
		 	Anthracene	<	<	<	<	<	<
 			Trichloroethene	<	. <	<	<	< 1	<
		ļ.,,	Inorganic Sum	4.00e-07	1.44c-07	6.99e-07	4.00c-07		
	Dermal	Soil	Benzo(b)fluoranthene		<	1.56e-07	<	<	<
<b></b>			Benzo(g.h,i)perylene	5.63e-08	<	1.82c-07	5.63e-08	<	<
· ···			Anthracene	<	<	<	<	<	< 1
WP45			Dibenz(a,h)anthracene	7.96c-08	<	2.57e-07	7.9 <del>6c</del> -08	<	<
* 			Indeno(1.2.3-cd)pyrene	<	<	1.45c-07	<	<	<
			Benz(a)anthracene	<	<	1.35e-07	<	<	<
			Benzo(a)pyrene	5.12c-08	<	1.66e-07	5.12c-08	<	<
		<u> </u>	Trichloroethene	<	<	1.57e-07	<	<	<
		L	Organic Sum	3.80e-07	1.92e-07	1.23e-06	3.48e-07		-
	Ingestion	Groundwater	Trichloroethene	5.41c-06	3.01 <b>c-0</b> 6	1.81c-05	<	<	<
			1.2 Dichloroethane	<	<	<	<	<	<
		L	Organic Sum	5.52e-06	3.07e-06	1.85e-05		-	
	Inhalation	Groundwater	Trichioroethene	2.96e-05	1.17e-05	4.95c-05	<	<	<
			1.2 Dichloroethane	1.0 <b>6c-</b> 06	<	1.77c-06	<ul> <li></li> </ul>	<	

Table 9.2b. (cont'd)

Салс	er Risk älle-	un Water a reunding	) le-07 in Soil with	Foture Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
·	1	Ţ	Organie Sum	3.06e-05	1 21e-05	5.13e-05			
	Ingestion	Vegetables	Benzo(b)fluoranthene	<	<	<	<	<	<
	-		Benzo(g.h.i)perylene	<	<	<	<	<	<
		=	Anthracene	<	<	<	<	<	<
		i	Dibenz(a.h)anthracene	<	<	<	<	<	<
		<u></u>	Indeno(1.2.3-cd)pyrene	<	<	<	<	<	<
		••••••••••••••••••••••••••••••••••••••	Benz(a)anthracene	<	<	<	<	<	<
		<b></b>	Benzo(a)pyrene	<	<	<	<.	<	<
		}	Trichloroethene	<	<	8.07e-07	<	<	<
			Organic Sum			8. <b>64e-</b> 07		_	
ST56	Ingestion	Groundwater	Tetrachioroethene	2.50e-06	1.39e-06	8.40c-06	<	<	<
			Organic Sum	2.50 <b>e-</b> 06	1.39 <b>e-0</b> 6	8.40 <del>c</del> -06	_		
	Inhalation	Groundwater	Tetrachioroethene	9.65e-07	<	1.61e-06	<	<	<
		   	Organic Sum	9.65e-07	3.82e-07	1.61e-06	·	_	
SS57	Ingestion	Groundwater	1.2 Dichloroethane	1.68c-06	9.36e-07	5.64c-06	<	<	<
	1	]	Benzene	5.36e-05	2.98e-05	0.00018	<	<	< 1
			Organic Sum	5.53e-05	3.08e-05	0.00019			
	Inhalation	Groundwater	1.2 Dichloroethane	1.69e-05	6.68e-06	2.82e-05	<	<	<
			Benzene	0.00054	0.00021	0.0009	<	<	<
	-		Organic Sum	0.00055	0.00022	0.00093		-	-
	Dermal	Groundwater	1.2 Dichloroethane	<	<	<	<	_ <	<
			Benzene	2.15e-06	5.20e-07	3.62e-06	< .	. <	<
	1	f	Organic Sum	2.16c-06	5.21e-07	3.63c-06	_		-
SS61	Inhalation	Soil	Arsenic	1.68c-07	6.08c-08	2.95e-07	7.10e-08	<	<
		1	Chromium 6	8.46e-07	3.05e-07	1.48e-06	5.81c-07	<	<
			Trichloroethene	<	~ ~	<	<	<	
			Benzene	<	<	<	<	<	<
······································		Î.	Inorganic Sum	1.01 <b>c-06</b>	3.66e-07	1.78e-06	6.52e-07	_	
	Ingestion	Groundwater	Beryllium	2.70e-05	1.50e-05	9.06e-05	<	<	<
-		f	Pentachiorophenol	1.05e-05	5.82e-06	3.51e-05	<	<	<
		1	Gasoline	2.17e-06	1.20e-06	7.26e-06	<	<	<
1			Trichlorocthene	6.18c-06	3.44e-06	2.07e-05	<	<	<
		1	Benzene	5.74c-07	~	1.92e-06	<	<	<

Table 9.2b. (cont'd)

. -

				1			·····		
Cance	r Risk ij Te	-06 in Water ( rounding	te-07 m Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
		[	Organic Sum	1 942-05	1.0 <b>8e-0</b> 5	6.50e-05			
			Inorganic Sum	2.70e-05	1.50e-05	9.06e-05			+-
	Inhalation	Groundwater	Gasoline	2.17e-05	8.62e-06	3.64e-05	<	<	<
			Trichloroethene	3 380-05	1.34e-05	5.63e-05	<	<	<
			Benzene	5.74e-06	2.27e-06	9.60e-06	<	<	v
			Organic Sum	6.12e-05	2.43e-05	0.0001		-	
SS61	Dermal	Groundwater	Beryllium	9 <b>39e-</b> 07	<	1.58e-06	<	<	<
t.			Pentachiorophenol	0.00039	9. <b>38e-0</b> 5	0.00065	<	<	<
-			Trichloroethene	<	<	<	<	<	<
			Benzene	<	<	<	<	<	<
			Organic Sum	0.00039	9.39 <b>e-</b> 05	0.00065			
			Inorganic Sum	9.39e-07	1.98e-07	1.58e-06	+-	-	
SS61-sp	Ingestion	Groundwater	Gasoline	7.65e-06	4.25e-06	2.57 <b>e-</b> 05	<	<	<
			Trichloroethene	2.55e-05	1.42c-05	8.55e-05	<	<	<
			Organic Sum	3.31e-05	1.84e-05	0.00011		-	
	Inhalation	Groundwater	Gasoline	7.66e-05	3.04 <b>c-</b> 05	0.00013	<	<	<
. <u>.</u>			Trichloroethene	0.00014	5.52e-05	0.00023	<	<	<
			Organic Sum	0.00022	8.56e-05	0.00036			
Operab	le Unit 4								
DP25	Ingestion	Soil	Atoclor 1254 (PCB)	t.15e-07	8.25e-08	1.07 <b>e-0</b> 6	6.35e-08	<	<
			Heptachlor epoxide	5.08e-08	<	4.74e-07	<	<	<
18-			Dieldrin	<	< .	<	<	<	<
t ar ign			Benzene	<	<	<	<	<	<
			Organic Sum	1.71e-07	1.23 <b>c-</b> 07	1.60c-06	6.97e-08		
	Inhalation	Soil	Benzene	<	<	6.20e-08	<	<	<
			Organic Sum	3.54e-08	1.28e-08	6.20e-08	3.54e-08	-	-
	Dermal	Soil	Arocior 1254 (PCB)	7.15e-07	3.61 <b>c-0</b> 7	2.31e-06	3.96 <b>e-</b> 07	<	<
			Heptachior epoxide	3.17e-07	1.60c-07	1.02 <b>e-06</b>	<	<	<
			Dieldrin	<	<	9.63 <b>c-08</b>	<	<	<
			Benzene	<	<	<	<	<	<
			Organic Sum	1.07e-06	5.38c-07	3.45e-06	4.35e-07		-
	Ingestion	Groundwater	Benzene	3.29e-05	1.83e-05	0.00011	<	<	<
			Organic Sum	3.29e-03	1.83e-05	0.00011		-	

Table 9.2b. (cont'd)

Cane	er Risk är le	-06 in Water a rounding	i Tean in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Futura Playin Child
	Inhalation	Groundwater	Benzene	0.00033	0 00013	0.00035	<	<	<
		[ 	Organic Sum	0 00033	0.00013	0.00055			
	Dermal	Groundwater	Benzene	1.32e-06	<	2.22e-06	<	<	
		t 	Organic Sum	1.32e-06	3.19e-07	2.22e-06			
	Ingestion	Vegetables	Aroelor 1254 (PCB)	<	<	9.10e-08	<	<	
			Heptachlor epoxide	<	<	3.40e-06	<	<	
			Dieldrin	<	<	1.12e-07	<	<	
			Benzene	<	<	1.22e-07	<	<	
			Organic Sum			3.73e-06		-	
SS35	Ingestion	Soil	Beryllium	3.10e-07	2.23e-07	2.89e-06	3.04e-07	<	•
 			Beta-BHC	<	<	<	<	<	
			Chlordane	<	<	1.26e-07	<	<	
			Alpha-BHC	<	<	<	<	<	
		[	DDD	3.15e-07	2.27e-07	2.95e-06	. <	<	
	1		DDĘ	1.36e-07	9.77e-08	1.27e-06	1.36e-07	<	
			DDT	3.47e-06	2.50e-06	3.24e-05	2.47e-07	<	
			Heptachlor epoxide	<	<	1.78c-07	<	<	
			Aldrin	<	<	5.86 <del>c</del> -08	<	<	
		<b></b>	Organic Sum	3.96e-06	2.85c-06	3.70e-05	4.36e-07		<u>,,,, _</u>
	-		Inorganic Sum	3.10e-07	2.23e-07	2. <b>89e-06</b>	3.04e-07		
 SS35	inhalation	Soil	Beryllium	<	<	<	<	<	·····
		<b>_</b>	Chromium 6	6.77e-07	2.44e-07	1. <b>19e-0</b> 6	5.38e-07		
		<u> </u>	Beta-BHC	<	<	<	<	· <	
		······	Chlordane	<	<	<	<	<	
		1	Alpha-BHC	<	<	<	<	<	
<u> </u>			DDT	<	<	6.96e-08		~	
			Heptachlor cpoxide			<	<	<	
<b></b>			Aldrin	~	<	<	<	~ ~	·····
			Organic Sum	4.03e-08	1.45e-08	7.05e-08	3.24c-09		
		{		6.84e-07	2.47e-07	1.20e-06	5.45e-07		
	Derma <sup>1</sup>	Soil	Beryllium	1 9307	9 7402	6 24-07	1.8007		
	1.00011100	1000	least ment	1	2.1.14-00				
		1	Reta-BHC		-	-			

Table 9.2b. (cont'd)

---.

Cancer F	Cancer Risk $\bar{g}$ (1e-06 in Water $\bar{g}$ (1e-07 in Soil with rounding				Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Alpha-BHC	<	<	8.30e-08	<	<	<
			DDD	1.97e-06	9.93e-07	6.36e-06	1.13e-07	<	<
			DQE	8.46e-07	4.27e-07	2.74e-06	8.46e-07	<	<
			DDT	2.16e-05	1.09e-05	7.00e-05	1.54e-06	<	<
			Heptachlor epoxide	1.19e-07	6.01 <b>c-</b> 08	3.85e-07	1.19 <b>e-</b> 07	<	<
			Aldrin	<	<	1.27e-07	<	<	<
			Organic Sum	2.47 <b>e-</b> 05	1.25e-05	8.00e-05	2.72 <b>c</b> -06		
			Inorganic Sum	1.93e-07	9 74e-08	6.24e-07	1. <b>89c-</b> 07		
l l	Ingestion	Groundwater	Beta-BHC	<	<	6. <b>38e-0</b> 7	<	<	<
			DDE	<	<	<	<	<	. <
			DDT	<	<	<	<	<	<
			Benzene	<	<	5.36e-07	<	<	<
			1.2 Dichloroethane	<	<	9.98c-07	<	<	<
			Organic Sum	8.01c-07	4.45e-07	2.69e-06			
[[	Inhalation	Groundwater	Benzene	1.60 <b>e-</b> 06	6. <b>35e-</b> 07	2.68e-06	<	<	<
			1.2 Dichloroethane	2.98e-06	1.18e-06	4.98 <del>c</del> -06	<	<	<
			Organic Sum	4.58e-06	1 81c-06	7.66e-06			
1	Dermal	Groundwater	Beta-BHC	<	<	<	<	<	<
			DD£	1.17 <del>c</del> -06	<	1,96¢-06	<	<	<
			DDT	<	<	7.48e-07	<	<	<
			Benzene	<	<	<	<	<	<
		i 	1.2 Dichloroethane	<	<	<	<	<	<
			Organic Sum	1.62e-06	3.91e-07	2.72e-06		_	-
	Ingestion	Vegetables	Beryllium	<	<	<	<	<	<
			Beta-BHC	<	<	<	<	<	<
			Chlordan <del>e</del>	<	<	5.72e-08	<	<	<
			Alpha-BHC	<	<	5.65e-08	<	<	<
			DDD	<	<	2.03e-07	<	<	<
			DDE	<	<	<	<	<	<
			DDT	<	<	2.26e-06	<	<	<
			Heptachlor epoxide	<	<	1.28e-06	<	<	<
			Atdrin	<	<	<	<	<	<
			Organic Sum		-	3.92e-06		_	

Table 9.25. (cont'd)

į.

÷.,

			and the second se			***			
Canc	Cancer Risk $\dot{q}$ Te-96 in Water $\bar{q}$ Te-07 in Soil with rounding			Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
\$\$35	Ingestion	Surfacewater	Beryllum	<	<	<	<	<	<
			Beta-BHC	< _	<	<	<	<	<
			Chlordane	<	<	<	<	<	<
			Alpha-BHC	<	<	<	<	<	<
			DDD	<	<	<	<	<	<
		}	DDE	<	<	<	<	<	<
			DDT	<	<	<	<	<	<
			Heptachlor epoxide	<	<	<	<	<	<
			Aldrin	<	<	<	<	<	<
			Inorganic Sum					3.15e-06	2.50e-06
	Ingestion	Shellfish	Beryllium	<	<	<	<	<	<
		<u> </u>	Beta-BHC	<	<	<	<	<	<
			Chlordane	<	<	<	<	<	<
			Alpha-BHC	<	<	<	<	<	<
			DDD	<	<	<	<	<	<
			DDE	<	<	<	<	<	<
		<u> </u>	тад	<	<	<	<	<	<
			Heptachlor epoxide	<	<	<	<	<	<
			Aldrin	<	<	<	<	<	<
			Organic Sum					7.69e-06	2.40c-05
			Inorganic Sum		_			7.71e-07	2.45e-06
SS36	Ingestion	Soit	Beryllium	4.21e-07	3.03 <b>c-</b> 07	3.93e-06	<	<	<
			DDT	<	<	5.33e-08	<	<	<
			Organic Sum	5.70e-09	4.10 <b>c-0</b> 9	5.33e-08			
			Inorganic Sum	4.21e-07	3.03e-07	3.93e-06		-	-
	Inhalation	Soil	Arsenic	7.69c-08	<	1.35e-07	<	<	<
			Beryllium	<	<	< .	<	<	<
			Chromium 6	1.49e-05	5.38e-06	2.61e-05	<	<	<
			DDT	<	<	<	<	<	<
			Inorganic Sum	1.50e-05	5.41e-06	2.63e-05			
	Dermal	Soil	Beryllium	2.62e-07	1.32e-07	8.49e-07	<	<	<
			DDT	<	<	1.15e-07	<	<	<
			Organic Sum	3.56e-08	1.80e-08	1.15e-07		-	

Table 9.2b. (cont'd)

Cance	er Risk ĝi le-	06 in Water <u>a</u> rounding	le-07 in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
······	1		Inorganic Sum	2.62e-07	1.32e-07	8.49e-07			
	Ingestion	Vegetables	Bery Ilium	<	<	5.04e-08	<	<	<
			TDD	<	<	<	<	<	<
	ļ		Inorganic Sum			5.040.08			
			Inorganic Sum				•-	3.15e-06	2.50e-06
			Organic Sum					7.69e-06	2.40c-05
			Inorganic Sum					7.71e-07	2.45e-06
SS37	Ingestion	Soil	Benz(a)antitracene	<	<	t .00e-07	<	<	<
			Organic Sum	1.07e-08	7.71e-09	1.00e-07	-		-
	Dermal	Soil	Benz(a)anthracene	6 68e-08	<	2. <b>16e-0</b> 7	<	<	<
_			Organic Sum	6.68e-08	3.37e-08	2.16e-07			_
SS39/63	Ingestion	Soil	Beryilium	4.36e-07	3.14e-07	4.07⊂-06	2.25e-07	<	<
	]	1 	DDT	<	<	5.42e-08	<	<	<
_			DDD	<	<	<	<	<	<
			Organic Sum	8.60c-09	6.19e-09	8.03c-08	2.66c-09		
			Inorganic Sum	4.36e-07	3.14e-07	4.07 <b>e-0</b> 6	2.25e-07		-
	Inhalation	Soil	Arsenic	1.07e-07	<	1.87e-07	<	<	<
			Beryllium	. <	<	<	<	<	<
		[	Chromium 6	9.30e-07	3.35e-07	1.63e-06	9.30e-07	<	<
			DDT	<	<	<	<	<	<
			Inorganic Sum	1.05e-06	3.77e-07	1.83e-06	9.35e-07		-
SS39/63	Dermal	Soil	Beryllium	2.72e-07	1.37e-07	8.79e-07	1.41c-07	<	<
i i			DDT	<	<	1.17e-07	<	<	<
			DDD	<	<	5.65e-08	<	<	<
		 	Organic Sum	5.36e-08	2.71e-08	1.73e-07	1.66c-08	-	-
			Inorganic Sum	2.72e-07	1.37e-07	8.79e-07	1.41e-07		
	Ingestion	Vegetables	Beryllium	<	<	5.22e-08	<	<	<
			DDT	<	<	<	<	<	<
			DDD	<	<	<	<	<	<
			Inorganic Sum			5.22e-08			
ST58	Ingestion	Groundwater	Benzene	5.85e-06	3.25e-06	1.96e-05	<	<	<
		 	Gasoline	0.00034	0.00019	0.00112	<	<	<
		]	Organic Suin	0.00034	0.00019	0.00114			-

Table 9.2b. (cont'd)

Cancer	Risk $\hat{a}$ te-	06 in Water <u>d</u> rounding	le-07 in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
	Inhalation	Groundwater	Benzene	5.86e-05	2.32e-05	9.81e-05	<	<	<
			Gasoline	0.00335	0.00133	0.00561	<	<	<
	L		Organic Sum	0.00341	0.00135	0.00571			
SS64	Ingestion	Groundwater	Tetrachioroethene	<	<	1.03e-06	<	<	<
L			Trichloroethene	<	<	<	<	<	<
			Organic Sum	4.12e-07	2.29 <b>e-</b> 07	1.38e-06			
	Inhalation	Groundwater	Tetrachloroethene	<	<	<	<	. <	<
			Trichloroethene	5.66c-07	<	9.48c-07	<	<	<
			Organic Sum	6.85e-07	2.71e-07	1.15e-06			
Operable	e Unit 5								
LF03/FT09	Ingestion	Soil	Beryllium	3 07e-07	2.21 <b>e-</b> 07	2.86e-06	3.07 <b>e-</b> 07	<	<
			Benzo(g.h.i)perylene	<	<	1.72e-07	<	<	<
			Benzo(k)fluoranthene	<	<	<	<	<	<
			Benz(a)anthracene	< <sup>1</sup>	<	1.69e-07	<	<	<
			Benzo(a)pyrene	<	<	1.77e-07	<	~	<
			Dibenz(a,h)anthracene	t.29e-07	9.29c-08	l.21c-06	1.29¢-07	<	<
			Indeno(1.2,3-cd)pyrene	<	<	1.79e-07	<	<	<
_			Benzo(b)fluoranthene	<	<	1.57c-07	<	<	<
			1.1.1-Trichloroethane	<	<	<	~	<	<
			Benzene	<	<	<	<	<	<
i			1.1 Dichloroethylene	<	<	<	<	<	<
			Trichloroethene	<	<	<	<	<	<
			Organic Sum	2.28e-07	1.64e-07	2.13e-06	2.28e-07	_	. <del>.</del> .
			Inorganic Sum	3.07e-07	2.21e-07	2.86e-06	3.07¢-07	-	_
	Inhalation	Soil	Benzene	5.10e-07	1.84c-07	8.92e-07	5.10e-07	<	<
			1.1 Dichloroethylene	0.00253	0.00091	0.00443	0.00253	<	<
			Trichloroethene	2.20e-07	7.94c-08	3.85e-07	2.20e-07	<	<
			Organic Sum	0.00253	0.00091	0.00443	0.00253	_	-
	Inhalation	Soil	Arsenic	1.71e-07	6.17e-08	2.9 <b>9e-</b> 07	1.71e-07	<	<
		[	Beryllium	<	<	<	<	<	<
			Chromium 6	2.24 <b>c-0</b> 6	8.07 <del>c</del> -07	3.92e-06	1.7 <b>4c-0</b> 6	<	<
			Велгенс	<	<	<	<	<	<
			1.1 Dichloroethylene	<	<	<	<	<	<
		· · · · · · · · · · · · · · · · · · ·							

Table 9.2b. (cont'd)

Cancer	Risk @ Le	06 in Water <u>a</u> rounding	le-07 in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Trichloroethene	<	<	<	<	<	<
			Inorganic Sum	2.43e-06	8 75e-07	4.24e-06	1.93e-06		
	Dermal	Soil	Beryllium	1.91e-07	9.65 <b>c-</b> 08	6.18e-07	1.91e-07	<	<
			Benzo(g.h.i)perylene	1.15e-07	5.78e-08	3.71e-07	1.15e-07	<	<
	Ĺ		Benzo(k)fluoranthene	<	<	<	<	<	<
			Benz(alanthracene	1.13 <b>e-</b> 07	5.70e-08	3.65e-07	1.13e-07	<	<
		 	Benzo(a)pyrene	1.18e-07	5.96e-08	3.82e-07	1.18e-07	<	<
	[		Dibenz(a.h)anthracene	8.05e-07	4.0 <del>6c</del> -07	2.60e-06	8.05e-07	<	<
· •··	) 		Indeno(1.2.3-cd)pyrene	1.19e-07	6.02e-08	3. <b>86e-</b> 07	1.19e-07	<	<
<b></b>			Benzo(b)fluoranthene	1.05 <b>e-</b> 07	5.30e-08	3.40e-07	1.05e-07	<	<
LF03/FT09			1.1.1.1.Trichloroethane	<	<	<	<	<	<
	   		Benzene	<	<	<	<	<	<
	,,,,_,_,_,,_,,,,,,,,,,,,,,,		1.1 Dichloroethylene	< -	<	1.02e-07	<	<	<
			Trichtoroethene	<	<	<	<	<	<
<u>.                                    </u>			Organic Sum	1.42e-06	7.18 <b>c-</b> 07	4.60e-06	1.42e-06		
		······································	Inorganic Sum	1.91e-07	9.65e-08	6.18e-07	1.91e-07		
	Ingestion	Groundwater	i.1.i-Trichloroethane	<	<	7.84c-07	<	<	<
	 	 	1.4-Dichlorobenzene	1.99e-06	1.11e-06	6. <b>68e-</b> 06	<	<	<
		ļ	Benzene	<	<	1.14e-06	<	<	<
			Tetrachloroethene	1.35e-06	7.52e-07	4.53e-06	<	<	· · ·
	 		Trichloroethene	8.02e-07	<	2.6 <del>9c</del> -06	<	<	<
			Vinyl chloride	1.76e-05	9.77c-06	5. <b>89e-</b> 05	<	<	<
			Organic Sum	2.23e-05	1.24e-05	7.47e-05	-	-	
	Inhalation	Groundwater	Benzene	3.41e-06	1.35e-06	5.70e-06	<	<	<
			Tetrachloroethene	5.20e-07	<	8.70e-07	<	<	<
	 		Trichloroethene	4.36e-06	1.73e-06	7.30e-06	<	<	<
L			Vinyl chloride	2. <b>79e-0</b> 5	1.11e-05	4.67e-05	<	<	<
			Organic Sum	3.62e-05	1.43e-05	6.06c-05	-		-
	Dermal	Groundwater	1.1,1-Trichloroethane	<	<	<	<	<	<
			1.4-Dichlorobenzene	3.93e-06	8.85e-07	6. <b>60e-0</b> 6	<	<	<
		L	Benzene	<	<	<	<	<	<
	ļ	 	Tetrachioroethene	<	<	<	<	<	<
			Trichloroethene		<	<	<	<	<

Table 9.2b.	(cont'd)
-------------	----------

Cance	er Risk ä te	-06 m Water g rounding	e le-07 in Soil with	Future Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Child
			Vinyl chloride	<	<	<	<	<	<
	<u> </u>	ļ 	Organic Sum	4 (i4e-t)6	9.12e-07	6.78e-06			
	Ingestion	Vegetables	Beryllium	<	<	<	<	<	<
			Bouzo(g.h.i)perylene	<	<	<	<	<	<
			Benzo(k)fluoranthene	<	<	<	<	< <	• <
		1	Benz(a)anthracene	<	<	<	<	<	<
	1	1	Benzo(a)pyrene	<	<	<	<	<	<
			Dibenz(a,h)anthracene	<	<	1.82e-07	<	<	<
	-	1	Indeno(1.2.3-cd)pyrene	<	<	<	<	<	<
<u> </u>	1		Benzo(b)fluoranthene	<	<	<	<	<	<
	1	}	1.1.1-Trichloroethane	<	<	<	<	<	<
			Benzene	<	<	<	<	<	~
			1.1 Dichloroethylene	<	<	1.08c-06	<	<	<
			Trichloroethene	<	<	<	<	<	<
	1		Organic Sum			1.38e-06			
			Organic Sum	2.28e-07	1.64 <b>c-</b> 07	2.13c-06	2.28e-07		·
			Inorganic Sum	3.07e-07	2.21 <b>e-0</b> 7	2.86e-06	3.07e-07		
	1	]	Organic Sum	0.00253	0.00091	0.00443	0.00253		_
	1		Inorganic Sum	2.43e-06	8.75e-07	4.24e-06	1.93 <b>c-06</b>		
			Organic Sum	1.42e-06	7.18e-07	4.60 <del>c</del> -06	1.42e-06		
			Inorganic Sum	1.91 <b>c</b> -07	9.65 <del>c</del> -08	6.18c-07	1.91e-07		
LF03-sp	Ingestion	Groundwater	Benzene	2.02e-06	1.13c-06	6.79e-06	<	<	<
	1		1.4-Dichlorobenzene	2.09e-05	1.16e-05	7.02e-05	<	<	<
	T		Tetrachloroethene	9.62e-06	5.35e-06	3.22e-05	<	<	<
			Trichloroethene	5.76e-06	3.20e-06	1.93e-05	<	<	<
	[		Vinyl chloride	0.00011	6.27c-05	0.00038	<	<	<
			Organic Sum	0.00015	8.40e-05	0.00051			
	Inhalation	Groundwater	Benzenc	2.03e-05	8.03 <b>c-</b> 06	3.39e-05	<	<	<
			Tetrachloroethene	3.70e-06	1.47c-06	6.20 <del>c</del> -06	<	<	<
			Trichloroethene	3.15e-05	1.25e-05	5.27e-05	<	<	<
			Vinyl chloride	0.00018	7.06e-05	0.0003	<	<	<
			Organic Sum	0.00023	9.26 <b>c-</b> 05	0.00039		_	-
F03-sn	Demal	Groundwater	Benzene	<	<	<	<	<	. <

able 9.2b.	(cont'd)
------------	----------

.....

Сапс	cer Risk ij te-	-06 in Water of rounding	r Le-07 m Soil with	Futore Worker	Future Typical Resident	Future Resident	Current Worker	Current Playing Child	Future Playing Chiid
			1.4-Dichlorobenzene	4 12e-05	9 30e-06	6.93e-05	<	<	<
			Tetrachloroethene	<	<	7.67e-07	<	<	<
			Trichloroethene	<	<	<	<	<	<
			Vinyl chlotide	<	<	<	<	<	<
			Organic Sum	4.19e-05	9.46e-06	7.04e-05			
			Organic Sum			1.38e-06			
LF04	Ingestion	Soit	Beryllium	1.81e-07	i 30e-07	1.69e-06	1.81c-07	<	<
			Inorganic Sum	1.81e-07	1.30e-07	1.69e-06	1.81c-07		
	Inhalation	Soil	Beryllium	<	_ < _	<	<	<	<
			Chromium 6	9.29e-07	3.35e-07	1.63e-06	9.29e-07	<	<
•			Inorganic Sum	9.34c-07	3.37e-07	1.63e-06	9.34e-07		
	Dermal	Soil	Beryllium	1.13e-07	5.69e-08	3.65e-07	1.13e-07	<	<
			Inorganic Sum	1.13e-07	5.69e-08	3.65e-07	1.13e-07		
	Ingestion	Groundwater	Beryllium	7.20e-05	4.00c-05	0.00024	<	<	<
		-	Inorganic Sum	7.20e-05	4.00e-05	0.00024			
	Dermai	Groundwater	Beryllium	2.50e-06	5.28e-07	4.21e-06	<	<	<
v=			Inorganic Sum	2.50e-06	5.28e-07	4.21e-06			

Table 9.20. (C	ont'	(D
----------------	------	----

Note: the < symbol denotes values less than 1E-06 with round-off.

the -- symbol means that the computation does not apply.

A "future typical resident" exposure is for 275 days per year and a "future resident" exposure is for 350 days per year

September 1995

1

•

Exposure		Ave.Time	Body		Frequency	Duration	Surface	Intake F	actors <sup>(b)</sup>
Route	Scenario	(or /0 (yr)	(kg)	Contact Rate	(day/yr)	(yr)	Area (cm²)	Cancer	Non-cancer
1	Current Worker	25	70	50 mg/day	100	25		6.99 x 10"	1.96 x 10 <sup>-7</sup>
Soil/Dust	Future Worker	25	70	50 mg/day	100	25	AF A (b)	6.99 x 10 <sup>a</sup>	1.96 x 10 <sup>.1</sup>
Ingestion	Future Resident	30	15/70	200/100 mg/day	146	6/24	NA.	6.53 x 10 <sup>.7</sup>	1.52 x 10 <sup>6</sup>
	Future Typ. Res.	9	70	100 mg/day	100	9		5.03 x 10 <sup>e</sup>	3.91 x 10 <sup>7</sup>
Sediment	Current Recreation	3	17.6	200 mg/day	30	3	NA	4.00 x 10 <sup>+</sup>	9.34 x 10 <sup>.7</sup>
Ingestion	Future Recreation	12	22.6	200 mg/day	30	12		1.25 x 10 <sup>.7</sup>	7.27 x 10'
Country	Future Worker	25	70	1 L/day	250	25		0.0035	0.0098
Groundwater	Future Resident	30	70	2 L/day	350	30	NA	0.0117	0.0274
mgendou	Future Typ. Res.	9	70	1.4 L/day	275	9		0.0019	0.0151
Surface-Water	Current Recreation	3	17.6	0.5 L/day	60	3		0.0002	0.0047
Ingestion	Future Recreation	12	22.6	0.5 L/day	60	12	NA	0.0006	0.0036
Vegetable Ingestion	Future Resident	30	70	17.7 g/day	60	30	NA	1.78 x 10 <sup>-5</sup>	4.16 x 10 <sup>-3</sup>
Fish	Current Recreation	3	17.6	300 g/day	30	3		6.00 x 10 <sup>-5</sup>	0.0014
Ingention	Future Recreation	12	22.6	300 g/day	30	12		1.87 x 10 <sup>-4</sup>	0.0011
Dermal Contact w/	Future Worker	25	70	0.17 hr/day	250	25	20,000	0,119	0.0333
Groundwater	Future Resident	30	15/70	0.17 hr/day	350	6/24	20,000	0.020	0.0466
(bathing)	Future Typ. Res.	9	70	0.12 hr/day	275	9	20,000	0.0033	0.0258
Dermal Contact w/	Current Recreation	3	17.6	2.6 mg/cm <sup>2</sup>	30	3	2,750	0,0014	0.0334
Surface Water	Future Recreation	12	22.6	2.6 mg/cm <sup>2</sup>	30	12	2,750	0,0045	0.0260
	Current Worker	25	70	1 mg/cm <sup>2</sup>	100	25	3,120	4.36 x 10 <sup>-6</sup>	1.22 x 10 <sup>-5</sup>
Dermal Contact w/	Future Worker	25	70	l mg/cm <sup>2</sup>	100	25	3,120	4.36 x 10 <sup>-6</sup>	1.22 x 10 <sup>-5</sup>
Soil	Future Resident	30	15/70	t mg/cm <sup>2</sup>	146	30	5,000	1.41 x 10 <sup>-5</sup>	3.30 x 10 <sup>5</sup>
	Future Typ. Res.	9	70	0.6 mg/cm <sup>2</sup>	146	9	5,000	2.20 x 10 <sup>-6</sup>	1.71 x 10 <sup>5</sup>

# Table 9.3. Summary of Exposure Factors

#### Table 9.3. (contd)

 $\sum_{i=1}^{n} (i + i) = \sum_{i=1}^{n} (i + i)$ 

Froste		Ave.Time	Body		Frequency	Duration	Surface	Intake F	actors <sup>(b)</sup>
Route	Scenario	(or /0 (yr)	(kg) Contact Rate		(day/yr)	(yr)	(cm <sup>2</sup> )	Cancer	Non-eancer
Dermal Contact w/	Current Recreation	3	17.6	1.5 mg/cm <sup>2</sup>	30	3	4,800	2.11 x 10 <sup>5</sup>	4.93 x 10 4
Sediments	Future Recreation	12	22.6	1.5 mg/cm <sup>2</sup>	30	12	4,800	6.58 x 10 <sup>-5</sup>	3.84 x 10 <sup>-4</sup>
Inhalation	Future Worker	25	70	20 m³/day	250	25		0.0699	0.196
Daily <sup>(d)</sup> Contact w/	Future Resident	30	70	20 m³/day	350	30	NA	0.117	0.215
Groundwater	Future Typ. Res.	9	70	20 m³/day	275	9		0.0277	0.274
Volatiles									
	Current Worker	25	70	20 m <sup>3</sup> /day	100	25		0.028	0.0783
Inhalation Contact w/	Future Worker	25	70	20 m³/day	100	25	NA	0.028	0.0783
Soil Particles	Future Resident	30	15/70	20 m³/day	146	30	"^	0.049	0.114
	Future Typ. Res.	9	70	20 m³/day	146	9		0.0101	0.0783

Sources: EPA 1991b, this study.

(a) The value 70 is used to calculate the intake term for the 70-year cancer case; other values are used for the noncancer exposures.

(b) Intake factors multiplied times exposure point concentration terms yield dose for risk characterization. Units for intake factors are kg/kg day (soils, sediments, soil particles, vegetables and fish), L/kg day (groundwater and surface water), and m<sup>3</sup>/kg day (volatile chemicals from the groundwater and surface water).

1

(c) NA = not applicable.

(d) The inhalation of groundwater volatiles is daily, not only for bathing or showering.

÷,

Note: A chemical-specific permeability factor is needed to calculate intake for dermal contact with surface water and groundwater.

		Cancer Risk		Nonc	ancer
		Oral	Inhalation	Oral	Inhalation
Chemical	CAS No.	(kg-day/mg)	(kg-day/mg)	(mg/kg-day)	(mg/kg-day)
Volatile Organic Compounds		·····	······		
Benzene	71-43-2	2.90E-02(2)	2.90E-02 <sup>(a)</sup>	<u> </u>	
1,4-Dichlorobenzene	106-46-7	2.4E-02 <sup>(b)</sup>			2.3E-01(2)
1.2-Dichloroethane	107-06-2	9.10E-02 <sup>(a)</sup>	9.10E-02 <sup>(a)</sup>		2.86E-03 <sup>(c)</sup>
Dichlorodifluoromethane	75-71-8			2.00E-01(a)	5.7E-02(2)
Tetrachloroethylene	127-18-4	5.2E-02 <sup>(c)</sup>	2E-03 <sup>(c)</sup>	$1.00E-02^{(a)} w^{(d)}$	
Toluene	108-88-3			2.00E-01 <sup>(a)</sup>	1.14E-01 <sup>(a)</sup>
Trichloroethene	79-01-6	1.10E-02 <sup>(d)</sup>	6E-03 <sup>(c)</sup>	6E-03 <sup>(c)</sup>	
Vinyl chloride	75-01-4	1.90E+00 <sup>(b)</sup>	3E-01 <sup>(b)</sup>		
Xylenes (total)	1330-20-7		-	$2.00E \pm 00^{(a)}$	
Semivolatile Organic Compo	unds				
Benzo(b)fluoranthene	205-99-2	7.3E-01 <sup>(c)</sup>	6.1E-01 <sup>(c)</sup>		
Benzo(a)pyrene	50-32-8	7.30E+00 <sup>(a)</sup>			
4-Methylphenol	106-44-5			5.0E-03 <sup>(b)</sup>	
Pesticides					
Beta-BHC	319-85-7	1.80E+00 <sup>(a)</sup>	1.80E+00 <sup>(a)</sup>	3.00E-04 <sup>(b)</sup>	
DDD	72-54-8	2.40E-01(a)			
DDT	50-29-3	3.40E-01 <sup>(a)</sup>	3.40E-01 <sup>(a)</sup>	5.00E-04 <sup>(a)</sup>	
Inorganics					
Arsenic	7440-38-2	1.75E+00 <sup>(b)</sup>	1.5E+01 <sup>(a)</sup>	3.00E-04 <sup>(a)</sup>	
Barium	7440-39-3			7.00E-02 <sup>(a)</sup>	1.40E-04 <sup>(b)</sup>
Chromium 6	7440-47-3		4.10E+01 <sup>(a)</sup>	5.00E-03 <sup>(a)</sup>	
Copper	7440-50-8			3.70E-02 <sup>(b)</sup>	
Manganese	7439-96-5			5.00E-03 <sup>(a)</sup>	1.40E-05 <sup>(a)</sup>

### Table 9.4a. Cancer and Noncancer Critical Toxicity Factors for Major Contaminants

NOTE: Spaces indicate that risk factors are not available.

In general, the risk numbers are listed with higher precision than is given in IRIS or HEAST.

Toxicity values without references were estimated from available values.

(a) EPA 1993c, 1993d, 1994b, 1995.

(b) HEAST (EPA 1994a, 1994b).

(c) EPA 1995a.

(d) Even though the toxicity factor has been withdrawn from IRIS, it has been used in the risk assessment per EPA Region X recommendation.

	<u> </u>	FPA	Cancer Risk				Noncancer			
Chemical (February 15, 1995)	CAS No.	Weight of Evidence	Orai (kg-day/mg)	Ref	Inhalation (kg-day/mg)	Ref	Oral (mg/kg-day)	Ref	Inhatation (mg/kg-day)	Ref
1,1,1-Trichloroethane	71-55-6	D		(******			Withdrawa		Under review	$\square$
1.1.2.2-Tetrachloroethane	79-34-5	с	2.00e-01		2.6e-02	.	Under review		Етрсу	
1.1.2-Trichloroethane	79-00-5	С	5.70e-02		5.7e-02		4.00e-03		Under review	
1.1-Dichlomethane	75-34-3	С	Inadequate		Inadequate		1.00e-01	(b)	Under review	
1.1-Dichloroethene	75-35-4	C	6.00e-01		1.2c+00		9.00e-03		Under review	
1.2-Dichloroethene (Total)	540-59-0						9.00e-03	(b)		
1.2-Dichlomethene, cis	156-59-2	D	Inadequate		Inadequate		Under Review		Unavailable	
1.2-Dichloroethene, trans	156-60-5		Empty		Empty		2.00e-02		Unavailable	
1.2-Dichloropropane	78-87-5	B2	6.80e-02	(b)	Empty		1.30e-02	(b)	4.00e-03	
2-Вигалопе	78-93-3	D	Inadequate		Іладедчане	1	6.00e-01		1.00e+00	
2-Hexañone	591-78-6		Under review		Under review		Етрту		Empty	
4-Methyl-2-pentanone	108-10-1		Empty		Επριγ		Withdrawn		Under review	$\square$
Acetone	67-64-1	D	inadequate		Inadequate		1.00e-01		Empty	
Bromodichloromethane	75-27-4	<b>B</b> 2	6.20e-02		Empty		2.00e-02		Empty	
កាល់ណូស	75-25-2	82	7.90e-03		3.9e-03		2.00e-02		Inadeq.	
Bromomethane	74-83-9	D	Inadequate		Inadequate		1.40e-03		5.00e-03	
Carbon Disulfide	75-15-0		Empty		Empty		1.00e-01		Under review	
Carbon Tetrachloride	56-23-5	B2	1.30e-01		5.3e-02		7.00e-04		Етру	
Chlombenzene	108-90-7	D	Inadequate		Inadequate		2.00e-02		Under review	
Chloroform	67-66-3	B2	6.10e-03		8.1c-02		1.00e-02		Under review	
Chioromethane	74-87-3	С	1.30e-02	(b)	6.3e-03	(b)	Under review		Under review	$\square$
cis-1,2-Dichloroethene	156-59-2	D	inadequate		Inadequate		1.00e-02	(b)	Етру	
cis-1,3-Dichloropropene	542-75-6	B2	Inadequate		Inadequate		3.00e-04		2.00e-02	
Dibromochloromethane	124-48-1	С	Inadequate		Empty		2.00e-02		Empty	
Ethylbenzene	100-41-4	D	Empty		Етргу		1.00e-01		1.00c+00	
Methylene Chloride	75-09-2	82	7.50e-03		1. <del>6c</del> -03		6.00e-02		Under review	
Styrene ****	100-42-5		U. Rev		U. Rev		2.00e-01		1.00e+00	
1,2,4-Trichlorobenzene	120-82-1	D	Етргу				1.00e-02		Under review	
1.2-Dichlorobenzene	95-50-1						9.00e-02			
1.3-Dichlorobenzene	541-73-1	D	Empty				Under review		Empty	
2.4.5-Trichlorophenol	95-95-4		Under teview				1.00e-01		Insdequase	
2.4.6-Trichlorophenol	88-06-2	B2	1.10e-02		1.0e-02		empty		Insdequate	
2.4-Dichlorophenol	120-83-2		Empty				3.00e-03		Empty	
2.4-Dimethylphenol	105-67-9		Empty				2.00e-02			
2.4-Dinitrophenol	51-28-5		Етру	L			2.00e-03		insdequale	
2.4-Dinitropluene	121-14-2		Етргу				2.00c-03		Insdequase	$\Box$
2,6-Dinitrotoluene	606-20-2		Empty				1.00e-03	(b)	Етрту	
2-Chloronaphthalene	91-57-8		Empty		Empty		8.00c-02		Етргу	

### Table 9.4b. Cancer and Noncancer Toxicity Factors for Other Than Major Contaminants

		ED	] (	Cancer	Risk		Noncancer				
Chemical (February 15, 1995)	CAS No.	Weight of Evidence	Oral (kg-day/mg)	Ref	Inhalation (kg-day/mg)	Ref	Oral (mg/kg-day)	Oral Inhatation (mg/kg-day) Ref (mg/kg-day)			
2-Chlorophenol	95-57-8		Етру				5.00e-03 Empty		Empty		
2-Methylphenoi	95-48-7	с	Empty				5.00e-02		Inadequate		
2-Nitroaniline	88-74-4		Empty		·		Under review		Етпрту		
3,3'-Dichlorobenzidine	91-94-1	B2	4.50e-01		· · · · · · · · · · · · · · · · · · ·		Empry	1	Inadequate	╏─┥	
4,6-Dinitro-2-methylphenol	534-52-1								Under Rev.		
4-Bromophenyl-phenylether	101-55-3						Inadequate		Inadequate		
4-Chloro-3-methylphenol	59-50-7						2.00e+00	(b)			
4-Chloroaniline	106-47-8	<u> </u>	Empty				4.00e-03		Етреу		
4-Nitroaniline	100-01-6				······································		Inadequate		Inadequate	Η	
4-Nitrophenol	100-02-7		Empty				Under review		Inadequate	$\square$	
Acenaphthene	83-32-9	D	Under review				6.00e-02		Empry		
Anthracene	120-12-7	D	Inadequate				3.00e-01		Under review		
Benzo(a)anthracene	56-55-3	B2	7.3e-01	(b)			Empty		Under review		
Benzo(g,h.i)perylene	191-24-2	D	Empty		Empty		Empry		Empty		
Benzo(k)fluoranthene	207-08-9	<b>B</b> 2	7.3e-02	(b)	Етргу	,	Етргу		Empry		
Benzoic Acid	65-85-0	D	Empty	···			4.00e+00		Empty		
Benzyl alcohol	100-51-6						3.00e-01	(ბ)			
bis(2-Chloroethoxy)methane	111-91-1	D	Inadequate				Empty		Empty		
bis(2-chloroisopropyl)ether	39638-32-9	С	7.00e-02	(b)	3.5e-02	<b>(b)</b>	4.00e-02	(b)	Етргу		
Bis(2-Chloroethyl)ether	111-44-4	С			1.1e+00	(b)					
bis(2-ethylhexyl)phthaiate	117-81-7	<b>B</b> 2	1.40e-02		Empty		2.00e-02		Empry		
Butylbenzylphthalate	85-68-7	с	Іладецият				2.00e-01		Empty		
Chrysene	218-01-9	B2	7.3e-03	(b)			Inadequate		Empty		
Di-n-butylphthalate	84-74-2	D	Inadequate				1.00e-01		Inadequate		
Di-n-octylphthalate	117-84-0		Empty		Empty		2.00e-02	(b)	Етргу		
Dibenz(a.h)anthracene	53-70-3	B2	7.3e+00	<b>(b)</b>	Етру		Empty		Етргу	Π	
Dibenzofuran	132-64-9	D	Empty				Inadequate	-	Under review		
Diethylphthalate	84-66-2	D	Empty				8.00e-01		Empty		
Dimethylphthalate	131-11-3						1.00e+01	(b)			
Fluoranthene	206-44-0	D	Inadequate				4.00e-02		Under review		
Fluorene	86-73-7	D	Етру				4.00c-02		Empty		
Hexachlorobenzene	118-74-1	82	1.60c+00		1.6e+00		Under review		Inadequate		
Hexachlorobutadiene	87-68-3	С	7.80e-02		7.7e-02		Withdrawn Empty		Етргу		
Hexachlorocyclopentadiene	77-47-4	D	Empty				7.00e-03 Empty		Empty		
Hexachloroethane	67-72-1	С	1.40e-02		1.4e-02		1.00e-03		Under review		

Table 9.4b. (cont'd)

		EPA	(	r Risk		Noncan	cer			
Chemical (February 15, 1995)	CAS No.	Weight of Evidence	Oral (kg-day/mg)	Ref	Inhatation (kg-day/mg)	Ref	Oral (mg/kg-day)	Ref	Inhalation (mg/kg-day)	Ref
Indeno(1.2.3-cd)pyrene	193-39-5	B3	7.3e-01	(b)	Empty		Empty		Empty	Ī
Isophorone	78-59-1	С	Empty				2.00e-01		Inadequate	
N-Nitroso-Di-n-propylamine	621-64-7	B3	7.00e+00				Етргу		Empty	
N-Nitroso-Dimethylamine	62-75-9	B2	5.10e+01		5.1e+01		Етрту		Under review	
N-Nitrosodiphenylamine	86-30-6	82	4.90e-03	i	Етру		Empty		Empty	
Nitrobenzene	98-95-3	D	Επιριγ				5,00e-04		Under review	
Pentachlorophenol	87-86-5	B2	1.20e-01		Етрту		3.00e-02		Under review	
Phenanthrene	85-01-8	D	Inadequate				Empty		Under review	
Phenol	108-95-2	D	Empty		Empty		6.00e-01		Inadequate	
Pyrene	129-00-0	D	Inadequate				3.00e-02	·····	Under review	
trans-1,2-Dichloroethene	156-60-5		Етру		Empty		2.00e-02		Empry	
trans-1,3-Dichloropropene	542-75-6	B2	Inadequate		Inadequate		3.00e-04		2.00e-02	
Viny! Acetate	108-05-4		Under review		Under review		1.00e+00	(b)	2.00e-01	
4,4'-DDE	72-55-9	B2	3.40e-01		Empty		Empty	 	Empty	
Aldrin	309-00-2	82	1.70e+01		1.7e+01		3.00e-05		Empty	
Alpha-BHC	319-84-6	B2	6.30e+00		6.3e+00		Empty		Етрту	
Chlordane	57-74-9	<b>B</b> 2	1.30e+00		1.3e+00		6.00e-05		Under review	
Detta-BHC	319-86-8	D	Empty		Етргу		Empty		Empty	
Dieldrin	60-57-1	B2	1.60e+01	İ	1.6e+01		5.00e-05		Empty	
Endosulfan I	115-29-7		,				6.00e-03		Empty	
Endrin	72-20-8	D					3.00e-04		Empty	
Gamma-BHC	319-89-9	·	Empty		Етргу		3,00e-04		Under review	
Heptachlor	76-44-8	B2	4.50e+00		4.5c+00		5.00e-04		Empty	
Heptachlor Epoxide	1024-57-3	<b>B</b> 2	9.10e+00		9.1e+00		1.30e-05		Етру	
Methoxychlor	72-43-5	D	Empty		Етргу		5.00e-03		Inadequate	
PCB-1016 (Aroclor 1016)	12674-11-2		Етарку		Empty		7,00e-05		Empty	
PCB-1221	1336-36-3	B2	7.70e+00		Етру					
PCB-1248 (Aroclor 1248)	12672-29-6		Empty		Етргу		Inadequate		Empty	
PCB-1254 (Aroclor 1254)	11097-69-1		Етргу		Empty		2.00e-05		Empty	
2,4,5-T	93-76-5		Етру		Empty		1.00e-02		Empty	
2.4.5-TP(Silvex)	93-72-1	D	Empty		Empty		8.00e-03		Етарту	
2,4-D	94-75-7		Етрту		Empty		1.00e-02		Етргу	
2.4-DB	94-82-6		Empty		Етру		8.00e-03		Empty	

Table 9.4b. (c	:ont'd)
----------------	---------

		EPA	Cancer Risk				Noncancer				
Chemical (February 15, 1995)	CAS No.	Weight of Evidence	Oral (kg-day/mg)	Ref	Inhalation (kg-day/mg)	Ref	Oral (mg/kg-day)	Ref	inhalation (mg/kg-day)	Ref	
Dalapon	75-99-0		Empty*		Empty		3.00e-02		Етргу	$\square$	
Dicamba	1918-00-9		Empty		Empry		3.00e-02		Empty		
Dichloroprop	120-36-5		Empty		Empty		Under review		Empty		
Dinoseb	88-85-7	D	Empty	·····	Empty		1.00e-03		Empty	1-1	
мсра	94-74-6	<u> </u>	Empty		Етрту		5.00e-04		Етрту		
мсрр	93-65-2		Empty		Empty		1.00e-03		Émpty	Π	
Toxaphene	8001-35-2	B2	t.10e+00		1.1e+00		Empty		Empty		
Aluminum	7429-90-5		Empty		Empty		Under review		Empty		
Апцтолу	7440-36-0		Empty		Empty		4.00e-04		Empty		
Beryllium	7440-41-7	<b>B</b> 2	4.30e+00		8.4e+00		5.00e-03		Етргу		
Cadmium	7440-43-9	<b>B</b> 1	Етру		6.3e+00		5.00e-04		Under review		
Cobalt	7440-48-4		Empty	·····	Empty		Under review		Empty		
Lead	7439-92-1	B2	Inadequate		Inadequate		No threshold		Empty		
Мексигу	7439-97-6	Đ	Inadequate		Inadequate		3.00e-04	(b)	Under review		
Nickel	7440-02-0		Not evaluated		Not evaluated		2.00e-02		Under review		
Selenium	7482-49-2	D	Inadequate.		Inadequate		5.00e-03		Empty		
Silver	7440-72-4	D	Inadequate,		Inadequate		5.00e-03		Етргу		
Thallium (acetate)	563-68-8	D	Inadequate		Inadequate 1		9.00e-05		Empry		
Vanadium	7440-62-2		Empty		Empty		Under review		Empty		
Zinc	7440-66-6	D	Inadequate		Inadequate		3.00e-01		Empty		
Diesel (as Kerosene)	68334-30-5		Empty		Етру		2.00c-02		5.00e-03		
Gasoline	8006-61-9	С	1.70e-03		1.7e-03		2.00e-01		Етру		
JP-4	JP-4						8.00¢-02				

Table 9.4b. (cont'd)

Note: Unless referenced otherwise, the factors and information are from IRIS2, February 1995.
(a) HEAST 1994.
(b) EPA Region III Risk-Based Concentrations: R.L. Smith (01/31/05).
Note: \* Empty is an IRIS term that suggests scientific data are lacking for determination of animal or human risk.

...

<u> </u>	Worker Playing Child Residential												
Source	Curre Risk	ent IS	Future I	Risks	Curre Risk	ent S	Future I	Risks	Future I	Risks	Future Ty Risk	/pical s	
Area	Cancer	HI*	Cancer	HI	Cancer	HI	Cancer	HI	Cancer	HI	Cancer	HI	
Operable U	nit 3												
DP44	1E-04	<	3E-04	0.8	< .	<	<	<	8E-04	2.3	1E-04	1.3	
WP45	<	<	4E-05	0.2	<	<	<	<	7E-05	0.6	2E-05	0.4	
ST56	<	<	3E-06	<	<	<	_ <	<	1E-05	<	2E-06	<	
SS57	<	<	6E-04	0.6	<	<	<	<	1E-03	1.0	3E-04	0.7	
SS61	<	<	5E-04	1.0	<	<	<	<	8E-04	1.8	1E-04	1.1	
Operable U	nit 4			<u> </u>		····		<u></u>				·····	
DP25	5E-07	<	5E-04	0.4	<	<	<	<	8E-04	0.7	2E-04	0.4	
ST27	<	<	<	<	<	<	<	<	<	<	<	<	
WP33	<	<	<	<	<	<	<	<	<	<	<	<	
SS35	3E-06	<	4E-05	0.4	<	<	5E-06	0.2	1E-04	1.5	2E-05	0.6	
SS36	<	<	<	<	<	<	5E-06	0.2	<	<	<	<	
SS37	<	<	<	<	<	<	<	<	<	<	<	<	
SS39/63	<	<	<	<	<	<	<	<	<	Ċ,	<	<	
ST58	<	<	4E-03	2.8	<	<	<	<	7E-03	7.7	2E-03	4.3	
SS64	<	<	1E-06	<	<	<	<	<	3E-06	<	5E-07	<	
Operable U	nit 5												
LF02	<	<	<	<	<	<	<	<	<	<	<	<	
LF03/FT09	3E-03	<	3E-03	0.1	<	<	<	<	5E-03	0.3	9E-04	0.1	
LF04	<	<	<	<	<	<	<	<	<	<	<	<	
LF06	<	<	<	<	<	<	<	<	<	<	<	<	
HI* = Haza Note: - "Futu frequ Note: < m	HI* Hazard Index. Note: Hazard Index (HI) is the sum of the noncancer effects. Note: -"Future Typical Risks" are calculated the same as "Future Risks" except that the contact rate, frequency and duration of exposure, and intake factors are lower for the future typical resident. Note: < means either that the cancer risk is less than 5E-7 or less than 0.1 for noncancer effects.												

Table 9.5.	Estimated Potential Reasonable Maximum Exposures in OU 3, 4, and 5 Source Areas
	at Eleison Air Force Base for Organic Chemical Contaminants and Lead

Note: < means either that the cancer risk is less than 5E-7 or less than 0.1 for noncancer effects.

٩,

Habitat Type/Community Type	Area (acres)	Wildlife Species	
Black spruce forest	8,678	Black bear, marten, moose, red squirrel, and spruce grouse	
White spruce forest	1,457	Black bear, marten, red squirrel, and spruce grouse	
Birch forest (dbh > 2", 15-65' tall	3,062	Black bear, raptors, red squirrel, and ruffed grouse	
Birch (dbh 0-1", 8-15' tall)	81	Moose and snowshoe hare	
Balsam poplar forest	1,286	Beaver, moose, raptors, and snowshoe hare	
Willow shrub	517	Moose and snowshoe hare	
Old burn (small white spruce, birch and willows)	295	Moose and snowshoe hare	
Marsh (some ponded water)	131	Moose and waterfowl	
Grassland/mown	Unquantified	Canada geese, hare, voles	
12 lakes <sup>(b)</sup>	315	Beaver, moose, and waterfowl	
50 ponds <sup>(b)</sup>	246	Beaver, moose, and waterfowl	
Designated wetlands <sup>(c)</sup>	10,202	Beaver, moose, and waterfowl	
Streams	25 miles	Веачег	
(a) Table adapted from U.S. Air Force	e (1993f), except c.		

Table 9.6.	Terrestrial Habitat	Types and	Their Primary	Wildlife Species	on Eielson AFB <sup>(a)</sup>
------------	---------------------	-----------	---------------	------------------	-------------------------------

(b) One lake and 6 ponds are natural; the remaining lakes and ponds are borrow pits (HLA 1990). (c) Taken from HLA (1990).

Action	Source Evaluation Sites	Operable Unit 3 Source Areas	Operable Unit 4 Source Areas	Operable Unit 5 Source Areas
No further cleanup action /long-term Monitoring	LF01 WP32 DP55	none	ST27 WP33 SS36 SS37 SS39/SS63 SS64	LF02 LF04 LF06
Limited Action	none	WP45/SS57 ST56 SS61	none	none
Cleanup Required	none	DP44	DP25 SS35 ST58	LF03/FT09

Table 9.7. Actions Evaluated for Source Areas



 $\mathbf{z}_{i}$ 

Figure 9.1. Conceptual Model of Pathways Common to all Source Areas

)

September

1995

FINAL

Eielson AFB

ļ

# 10.0 Description of Alternatives – OU 3

This section provides a concise description of the remedial action objectives and remedial alternatives evaluated for each source area in OU 3. The feasibility study (FS) provides information on the assumptions and calculations used in the development of the alternatives and the cost estimates for the alternatives.

# 10.1 Remedial Action Objectives

Remedial action objectives (RAOs) were developed to specify actions and contaminant levels necessary to protect human health and the environment. RAOs define the contaminants of concern, exposure routes and receptors, and remediation levels, which are defined as acceptable contaminant levels for each exposure route. The primary RAO is protection of groundwater.

The RAOs for the source areas within OU 3 are:

- At DP44, prevent the continued migration of TCE into the groundwater at concentrations that present a risk to potential future groundwater users.
- At WP45/SS57, prevent the continued migration of TCE and benzene into the groundwater at concentrations that present a risk to future groundwater users. Subsequent sampling indicates that little contamination remains in the unsaturated zone (see Section 16).
- At ST56, supply drinking water, apply wellhead treatment, as applicable, prevent use of groundwater that exceeds state or federal drinking water standards.
- At SS61, determine if an additional source of contaminants exists on the north side of the building and if so, prevent the continued migration of TCE into the groundwater at concentrations that present a risk to future groundwater users.
- At all source areas, prevent human exposure to groundwater contaminated above the drinking water standards and restore the beneficial uses of the aquifer.

The goal of the Superfund approach is to return usable groundwaters to their beneficial uses within a timeframe that is reasonable, given the particular circumstances of the site. Reasonable restoration time periods may range from very rapid (one to five years) to relatively extended (several decades). Location, proximity to population, anticipated future land use, and mobility of the contaminant plume are factors considered when determining an appropriate restoration timeframe.

The use of: (1) natural attenuation with institutional controls, (2) source reduction through treating soil contamination to prevent additional contaminant leaching into the groundwater, and (3) ground-water pumping and treating were considered viable options for addressing groundwater contamination at Eielson AFB. For source areas within OUs 3, 4, and 5, the following site specific conditions were considered when determining reasonable restoration timeframes:

 Contaminant plumes in this relatively homogeneous aquifer do not appear to be spreading or are decreasing in size.

- Biodegradation, dispersion, dilution, or adsorption appear to be effectively containing or reducing the size of the contaminant plume.
- Areas impacted by the contamination are relatively small with little likelihood of extended exposure to groundwater anticipated.
- Future land use as a military installation is not anticipated to change in the foreseeable future.

# **10.2 Remedial Action Components**

The sources areas comprising OU 3 (DP44, WP45/SS57, ST56, and SS61) contain groundwater and soils contaminated with VOCs, SVOCs, and metals. Subsurface treatment of the groundwater at ST56 was not evaluated in the FS because of the complex hydrogeology of the fractured bedrock; the limited extent of the contamination in a remote, restricted area of the base; and the reliability of available-institutional controls to restrict the use of the contaminated water. The alternatives to address the remaining source areas are assembled from one or more of the following remedial action components:

No Action

Components to Address Groundwater Contamination

- Institutional Control and Groundwater Monitoring
- Groundwater Extraction and Treatment

Components to Address Soil Contamination

- Soil Vapor Extraction (SVE)/Bioventing
- Soil Excavation

The no action alternative does not address the RAOs established for each source area. It represents the baseline risk without institutional controls or active remediation. The institutional controls and groundwater monitoring alternative prevents the use of contaminated groundwater, but does not include active remediation. The other components provide a range of control with varying timeframes to achieve the RAOs. Because it is difficult to remove all of the residual contamination and to reduce groundwater contaminant concentrations below state and federal water quality standards, even with groundwater extraction and treatment, it is expected that all of the alternatives would require long-term management of the area.

The five remedial action components are described in the following paragraphs. Following this discussion, the components are assembled into remedial action alternatives for each of the source areas. Cost for the alternative for each source area is included in Table 10.1.

# 10.2.1 No Action

Under this approach, no action is taken to remove contaminants from the soils or the groundwater. Evaluation of the no action approach is required by the NCP to provide a baseline against which other alternatives can be compared. Under the no action alternative, no remedial measures are implemented

Source Area	Alternative Description	Costs		
	·	Capital	30-yr O&M	Total
DP44	No Action	\$0	\$0	\$0
	Groundwater Monitoring/Institutional Controls	\$5,300	\$134,700	\$140,000
	Soil Vapor Extraction/Groundwater Monitoring/Institutional Controls	\$1,300,000	\$300,000	\$1,600,000
	Soil Vapor Extraction/Groundwater Extraction and Treatment/ Institutional Controls	\$2,100,000	\$1,500,000	\$3,600,000
WP45/SS57	No Action	\$0	<b>\$</b> 0	\$0
	Groundwater Monitoring/Institutional Controls	\$5,300	\$174,700	\$180,000
	Soil Vapor Extraction/Bioventing Groundwater Monitoring/Institutional Controls	\$660,000	\$540,000	\$1,200,000
	Bioventing/Excavation/Groundwater Monitoring/Institutional Controls	\$11,000,000	\$0	\$11,000,000
	Soil Vapor Extraction/Bioventing Groundwater Extraction/Institutional Controls	\$1,900,000	\$3,700,000	\$5,600,000
	Bioventing/Excavation/Groundwater Extraction/Institutional Controls	\$12,000,000	\$4,000,000	\$16,000,000
<b>SS</b> 61	No Action	\$0	<b>\$</b> 0	\$0
	Groundwater Monitoring/Institutional Controls	\$5,300	\$154,700	\$160,000
	Soil Excavation/Groundwater Monitoring/Institutional Controls	\$540,000	\$160,000	\$700,000
	Groundwater Extraction and Treatment/Groundwater Monitoring/Institutional Controls	\$1,300,000	\$1,800,000	\$3,100,000

### Table 10.1. Costs of Remedial Alternatives for Source Areas In OU 3

with no long-term review, monitoring, or controls. Contaminants would degrade and disperse through natural attenuation. No direct costs are associated with the no action alternative.

# 10.2.2 Institutional Controls and Groundwater Monitoring

Institutional controls are used to prevent current and future human exposure to contamination remaining at the source areas at concentrations above health-based risk levels and federal and state standards. Specific controls include restrictions limiting access to the source areas and administrative controls to limit groundwater and future land use. Access restrictions, including such measures as permanent markers, are used to prevent direct human exposure to contaminants. Groundwater restrictions are implemented by placing written notification in base policies prohibiting the use of contaminated groundwater. In addition, all existing and any new wells located in or near the contaminated portion of the aquifer shall be locked to prevent unauthorized use.

The administrative controls for limiting future land use include placing written notification of these remedial actions in the base land use master plan. The notification shall prohibit any activity that disrupts aspects of the engineered controls. A copy of the notification is provided to any prospective transferees of the property and is included in any transfer documents, including deeds, in the event that the Air Force released control of the affected property. The Air Force shall provide EPA and the state with written verification that notification(s) have been implemented.

# 10.2.3 Groundwater Extraction and Treatment

Groundwater extraction was evaluated for source areas DP44, WP45/SS57, and SS61 as a way to potentially accelerate aquifer restoration. The number and location of extraction wells, as well as estimated times to achieve cleanup, are discussed in Sections 10.3, 10.4, and 10.5.

VOCs and SVOCs in the extracted groundwater would be removed using physical/chemical treatment, such as air stripping and activated carbon. Due to the high metals levels naturally found in the groundwater at Eielson AFB, metals removal may be necessary prior to treating the water for sitespecific contamination. The need for air emission controls would be evaluated and implemented in accordance with the Clean Air Act (42 USC 7401). Treated groundwater would be discharged to Garrison Slough in accordance with federal ambient water quality criteria, 18 AAC 70, and 18 AAC 72.

#### 10.2.4 Soil Vapor Extraction/Bioventing

A combination of SVE and/or bioventing was evaluated for source areas DP44 and WP45/SS57. This alternative enhances bioremediation and volatilization of contamination in the vadose zone for SVE/bioventing and in the smear zone for bioventing. For WP45/SS57, the system could be operated in the air injection or air withdrawal mode with wells screened across the water table. Although air emission controls would be installed in accordance with the Clean Air Act, if needed, the system would be designed and operated to minimize the need for air emission controls.

Based on assumed contaminant concentrations and expected removal rates, it is estimated the SVE system would operate from 1 to 3 years (6 months/year) to meet soil RAOs for TCE and related compounds in the vadose zone. SVE may be enhanced with air injection or capping. For bioventing, estimates of treatment times are provided in specific alternatives incorporating this component.

# 10.2.5 Soil Excavation

Excavation of contaminated soil from source areas WP45/SS57 and SS61 and out-of-state disposal was evaluated. Soil excavation would be conducted during one summer construction season. Contaminated soils would be removed from the ground surface to the top of the saturated zone. Using a field screening technique, clean soils would be segregated from contaminated soils. The excavated area would then be backfilled with clean material.

# 10.3 Source Area DP44

Site conditions of source area DP44 followed by the remedial action alternatives are discussed in this section.

# 10.3.1 Site Conditions

The primary concern at source area DP44 is TCE in the groundwater. However, tetrachloroethylene, benzene, and cis-1,2-DCE (a TCE breakdown product) were also found in the groundwater. In addition, TCE was found in the subsurface soil along the edge of and beneath the tarmac south of the Hangar (Building 1140).

The primary potential exposure pathway for source area DP44 is from the future unrestricted use of contaminated groundwater. Therefore, contaminants in groundwater and the risk from continued leaching of contaminants to groundwater are of primary concern.

# 10.3.2 Remedial Alternatives

Four alternatives have been developed to address the contaminated soil and groundwater at source area DP44. The four alternatives are listed in Table 10.2.

A description has been developed for each of the four alternatives identified for source area DP44. The descriptions identify the technologies, describe the representative process options, and present the assumptions that provide the basis for the analyses of the alternatives.

F	Alternative	Soils Components	Groundwater Components
1.	No Action	None	None
2.	Institutional Controls /Monitoring	None	Natural attenuation, groundwater monitoring, institutional controls
3.	SVE	SVE with carbon adsorption, optional extension of cap	Natural attenuation, groundwater monitoring, institutional controls
4.	SVE/Extraction of Groundwater	SVE, optional extension of cap	High-rate extraction, metals precipitation, air stripping, discharge to Garrison Slough, institutional controls, groundwater monitoring

Table 10.2. Primary Components in Alternatives for Source Area DP44

#### Alternative 1-No Action

The no action alternative was evaluated as discussed in Section 10.2.1.

#### Alternative 2-Institutional Controls/Groundwater Monitoring

This alternative is a limited action alternative consisting of two components.

Based on currently observed concentrations in the soil and groundwater, it was estimated that TCE may continue to leach into groundwater for more than 10 years. Trichloroethane-contaminated soils could continue to act as a contaminant source to groundwater, especially during periods of snow melt.

Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because of this continuing source of release and the probability that groundwater concentrations will remain above federal and state standards for an extended period of time, a long-term groundwater monitoring program and institutional controls are included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

#### Alternative 3-SVE

The SVE alternative includes the following remedial action components.

Under this alternative, SVE would be used to reduce the concentrations of TCE in the subsurface soils along the edge of the runway on the southeast side of Hanger 1140. This area is the suspected source of the TCE contamination found in groundwater at nearby Well 44M04. It is assumed that approximately six vertical extraction wells on a 18.3-m (60-ft) spacing would be installed to remove contaminated soil vapor from the assumed 76- to 84-m (250- by 275-ft) target area.

Seven passive air injection wells were assumed to be installed below the surface of the existing tarmac to provide improved air flow through the vadose zone, and to prevent short-circuiting from the southern edge of the target area. Due to the relatively short distance from the surface to the contaminated vadose zone, and the narrow zone of contamination, a portion of the unpaved target area would be capped as part of the operation.

The groundwater would not be actively remediated. The contaminants would be allowed to disperse and degrade naturally. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. If a significant volume of TCE is remaining in the vadose zone, the natural attenuation of the TCE in groundwater would benefit from the remedial activities designed to address the contaminant source.

The TCE in the vadose zone would be removed through SVE, and the chlorinated VOCs in groundwater would be allowed to attenuate through natural processes. The continuing source from the vadose zone soil will be treated by SVE in 1 to 3 years. Because it is probable that groundwater concentrations will remain above federal and state standards for an extended period of time, a long-term groundwater monitoring program and institutional controls was included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

#### Alternative 4 - SVE/Extract

The SVE/Extract alternative includes the following remedial action components:

SVE. Soil vapor extraction (SVE) would be implemented as discussed in Alternative 3.
Groundwater Extraction and Treatment. These actions would be implemented in an attempt to accelerate aquifer restoration. A single groundwater extraction well would be operated at 30 gpm to increase the speed of aquifer restoration. Extracted groundwater will be treated as discussed in Section 10.2.3.

Based on modeling projections, the groundwater extraction and treatment system proposed under this alternative would need to operate for at least 10 years. Year-round operation is assumed.

The TCE in the vadose zone would be removed through SVE, and the chlorinated VOCs in groundwater would be reduced through extraction.

The continuing source from the vadose zone soil will be treated by SVE in 1 to 3 years. Because it is probable that groundwater concentrations will remain above federal and state standards for an extended period, a groundwater monitoring program and institutional controls was included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2 and Alternative 3, to restrict groundwater use, prevent human exposure to contaminants at concentrations above federal and state standards, and maintain an accurate definition of the area of contamination.

# 10.4 Source Area WP45/SS57

# 10.4.1 Site Conditions

The primary concern at source area WP45/SS57 is the presence in groundwater of TCE and benzene above drinking water standards. However, a breakdown product of TCE, *cis*-1,2-DCE, and tetrachloroethylene (PCE) were also found in the groundwater. TCE was found in the subsurface soils at Well 45M08.

The primary potential exposure pathway for source area WP45/SS57 is from the future unrestricted use of contaminated groundwater. Therefore, contaminants in groundwater and the risk from continued leaching of contaminants to groundwater are of primary concern.

# 10.4.2 Remedial Alternatives

Six alternatives have been developed to address both the contaminated soil and groundwater at source area WP45/SS57. The six alternatives are listed in Table 10.3.

A description has been developed for each of the six alternatives identified for source areas WP45/SS57. The descriptions identify the technologies, describe the representative process options, and present assumptions that provide the basis for analyses of the alternatives.

# Alternative 1-No Action

The no action alternative is evaluated as discussed in Section 10.2.1.

Alternative	Soils Components	Groundwater Components
1. No action	None	None
<ol> <li>Institutional Controls / Groundwater Monitoring</li> </ol>	None	Natural attenuation, groundwater monitoring, institutional controls
3. In situ	SVE with carbon adsorption, bioventing	Natural attenuation, groundwater monitoring, institutional controls
4. Remove	Removal of TCE-contaminated soil, bioventing	Natural attenuation, groundwater monitoring, institutional controls
5. In situ/Extract	SVE with carbon adsorption, bioventing	Extraction, metals precipitation, air stripping, discharge to Garrison Slough, institutional controls, monitoring
6. Remove/Extract	Removal of TCE-contaminated soil with offsite disposal, bioventing	Extraction, metals precipitation, air stripping, discharge to Garrison Slough, institutional controls, monitoring

#### Table 10.3. Primary Components in Alternatives for Source Areas WP45/SS57

#### Alternative 2-Institutional Controls and Groundwater Monitoring

This alternative is a limited action alternative. No action will be taken for subsurface soil. The primary components are briefly outlined in the following paragraphs.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above federal and state standards, and maintain an accurate definition of the area of contamination. Fire well C is located in the middle of the TCE plume, but only trace concentrations of DCE have been detected in the well. The well is connected to the potable water supply on base and a slight chance exists that it could introduce solvent contamination into the water supply. Institutional controls also apply to use of this well to prevent using the contaminated groundwater in a manner that would pose an unacceptable risk to human health and the environment.

Results of a recent Utah Water Research Laboratory (UWRL) study at WP45/SS57 concluded that the remaining source of TCE contamination is predicted to be exhausted in approximately 7 years. No evidence is present of residual dense non-aqueous phase liquid (DNAPL) and it also does not appear that any residual fuel material exists in the form of light non-aqueous phase liquid (LNAPL). The contamination is adsorbed and contained or is in a dissolved phase and not readily accessible for source removal or treatment. The data indicate that natural processes are expected to achieve remediation within the same relative timeframe as that predicted for active remediation.

Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because of the continuing source of release and because groundwater concentrations are currently above state and federal standards and are expected to remain at those levels for an extended period of time, a long-term groundwater monitoring program and institutional controls would be implemented under this alternative.

### Alternative 3-In Situ

The in situ alternative includes the following remedial action components:

SVE. SVE would attempt to reduce the concentrations of TCE in the subsurface soils, where accessible, in the area of the old maintenance shop off the northeast corner of the fire station. This area is the suspected source of the TCE contamination found in the wells downgradient from this area.

The extraction and treatment system is discussed in Section 10.2:4. It was assumed the SVE system would use five vertical extraction wells on an approximately 18.3-m (60-ft) spacing to remove contaminated soil vapor from the assumed 37- by 37-m (120- by 120-ft) target area. The area is currently planted in grass and would not be capped as part of the operation.

Based on assumed contaminant concentrations and expected contaminant removal rates, it is estimated the SVE system would have to operate in this area for 1 to 3 years (6 months per year) to meet the soil cleanup levels for TCE and related compounds in the vadose zone.

**Bioventing**. Bioventing would be used to treat any residual BTEX-contaminated soils beneath the fire station parking lot. This area is the suspected source of the benzene and toluene found in the wells downgradient from this area.

Conceptually, the bioventing system was assumed to include 20 vertical injection wells on approximately 15-m (50-ft) centers to provide oxygen to the assumed 61-m by 76-m (200- by 250-ft) contaminated zone to enhance in situ degradation. The air injection rate would be based on achieving an optimal number of pore-volume exchanges per week to stimulate microbial degradation of the BTEX compounds. No extraction wells would be used as part of the bioventing system, and no attempt would be made to capture the injected air for treatment. The system would operate year-round.

It was previously estimated the bioventing system would have to operate year-round in this area for 2 to 4 years to meet the soil cleanup levels for BTEX compounds. Based on recent UWRL findings, it appears that levels of BTEX are significantly lower than previously reported and that little residual BTEX contamination exists in the vadose zone.

As described in Section 10.2.4, the groundwater would not be actively remediated. The TCE and BTEX plumes in the shallow groundwater would continue to disperse and degrade naturally. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination.

Any residual continuing source from the vadose zone soil would be treated by SVE in an estimated 1 to 4 years and by bioventing in 2 to 4 years. Because it is probable that groundwater concentrations may remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls were included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

### Alternative 4 – Remove

The remove alternative includes the following remedial action components:

Excavation of Soil. Under this alternative, approximately 2800 m<sup>3</sup> (3700 cu yd) of VOCcontaminated soil would be excavated from the TCE- and PCE-contaminated maintenance shop and other source locations at source area WP45/SS57 and transported for disposal out-of-state as hazardous waste. Excavation is expected to take one summer to complete. The excavation component is discussed in further detail in Section 10.2.5.

**Bioventing**. The BTEX-contaminated soils from beneath the fire station parking lot would be treated with bioventing as described Alternative 3.

Under this alternative, the groundwater would not be actively remediated. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. The TCE and PCE in the vadose zone would be removed through excavation and the BTEX in the vadose zone would be removed through bioventing. The chlorinated VOCs and BTEX in groundwater would be allowed to attenuate through natural processes.

The continuing source from the vadose zone soil would be removed in 1 year and, by bioventing, in 0 to 4 years. Because it is probable that groundwater concentrations will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls were included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

### Alternative 5-In Situ/Extract

The in situ/extract alternative includes the following remedial action components:

SVE. SVE would be implemented, as discussed in Alternative 3, to treat TCE-contaminated soils located near the old maintenance shop on the northeast corner of fire station (Building 1206).

Bioventing. Bioventing would be implemented, as discussed in Alternative 3, to treat BTEXcontaminated soils beneath the fire station parking lot.

Groundwater Extraction and Treatment. This component would be implemented to remove and treat VOC contaminated groundwater. The extraction network for this alternative is assumed to consist of two extraction wells with a combined extraction rate of approximately 60 gpm. Extracted groundwater will be treated as discussed in Section 10.2.3.

Based on modeling projections, the groundwater extraction and treatment system proposed under this alternative would need to operate for at least 30 years. Year-round operation is assumed. Institutional controls would be used to prevent exposure to groundwater until state and federal standards are achieved.

The TCE and PCE in the vadose zone would be removed through SVE and the BTEX in the vadose zone would be removed through bioventing. The chlorinated VOCs and BTEX in groundwater would be reduced through extraction.

The continuing source from the vadose zone soil will be treated by SVE in an estimated 1 to 4 years and by bioventing in 0 to 4 years. Because it is probable that groundwater concentrations will

remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls was included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

#### Alternative 6-Remove/Extract

The remove/extract alternative includes the following remedial action components:

Excavation of Soil. Excavation of VOC-contaminated soil would be implemented as discussed for Alternative 4.

Bioventing. Bioventing would be implemented as discussed for Alternative 3.

Groundwater. The groundwater component is the same as described in Alternative 5.

The TCE and PCE in the vadose zone would be removed through excavation and the BTEX in the vadose zone would be removed through bioventing. The chlorinated VOCs and BTEX in groundwater would be reduced through extraction.

Based on a batch flush groundwater model, extraction of the TCE that is already in groundwater could take 34 to 68 years for a plume in a highly permeable aquifer. The continuing source from the vadose zone soil will be removed in 1 year and, by bioventing, in 0 to 4 years. Because it is probable that groundwater concentrations will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls was included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

# 10.5 Source Area SS61

# 10.5.1 Site Conditions

The primary concern at source area SS61 is TCE in the groundwater. However, a breakdown product of TCE, cis-1,2-DCE, was also found in the groundwater, and low-level TCE contamination was found in the subsurface soil adjacent to the old dry well at the Vehicle Maintenance Building (Building 3213). This dry well, along with surrounding soils, was removed during construction of the new building.

The primary potential exposure pathway for source area SS61 is from the future unrestricted use of contaminated groundwater. Therefore, contaminants in groundwater and the risk from continued leaching of contaminants to groundwater are of primary concern. The groundwater contamination is currently limited to the area directly beneath the building.

# 10.5.2 Remedial Alternatives

Based on the remedial action objectives, a range of remedial response actions has been developed to address residual contaminated soil, if any, and groundwater at source area SS61. The alternatives range from no action to removal for ex situ treatment and disposal. Four comprehensive alternatives have been developed and are listed in Table 10.4.

	Alternative	Soil Components	Groundwater Components			
1.	No action	None	None			
2.	Institutional Controls / Groundwater Monitoring	None	Natural attenuation, groundwater monitoring, institutional controls			
3.	Remove	Excavate contaminated soil for offsite treatment and disposal	Natural attenuation, groundwater monitoring, institutional controls			
4.	Extract	No action	Extraction, precipitation, air stripping, discharge to Garrison Slough, institutional controls, groundwater monitoring			

Table	10.4.	Primary	Components	în	Alternatives	for	Source	Area	SS61
-------	-------	---------	------------	----	--------------	-----	--------	------	------

A description has been developed for each of the four alternatives identified for source area SS61. The descriptions identify the technologies, describe the representative process options, and present the assumptions that provide the basis for the individual and comparative analyses.

### **Alternative 1-No Action**

The no action alternative would be implemented as discussed in Section 10.2.1.

### Alternative 2-Institutional Controls and Groundwater Monitoring

This alternative is a limited action alternative. It focuses on the following components.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above federal and state standards, and maintain an accurate definition of the area of contamination.

No evidence of residual dense non-aqueous phase liquid (DNAPL) is found at SS61. The contamination is adsorbed and contained or is in a dissolved phase and not readily accessible for source removal or treatment. The data indicate that natural processes are expected to achieve remediation within the same relative timeframe as those predicted for active remediation.

This alternative would achieve the RAOs for source area SS61. It does not appear that a continuing source of release is present in the vadose zone. However, if TCE-contaminated soils are present, they would continue to act as a contaminant source to groundwater. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because of the potential for a continuing source and the probability that

groundwater concentrations will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls would be implemented under this alternative

### Alternative 3-Remove

The remove alternative includes the following remedial action components:

**Excavation**. If a significant source of residual contamination could be located, up to  $115 \text{ m}^3$  (150 cu yd) of contaminated soil would be excavated from the vadose zone on the north side of the building at source area SS61 and placed in roll-off bins. Close observation of the building foundation would occur during excavation. Pressure grouting may be necessary to facilitate the excavation. The excavation and offsite treatment through incineration of the contaminated soil at source area SS61 is expected to take one month to complete. Excavation is discussed in further detail in Section 10.2.5. This component may not be applicable unless it is shown that a source of TCE is present in soil on the north side of the building.

No action would be taken to treat groundwater. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. The natural dispersion of the TCE in groundwater may benefit from the remedial activities designed to address the contaminant source if any was found.

If a TCE source is located in the vadose zone soils on the north side of the building, it would be removed by excavation. The chlorinated VOCs in groundwater would be allowed to attenuate through natural processes.

If located, any continuing source from the soil could be removed in 1 year. Because it is probable that groundwater concentrations will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls would be implemented under this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above federal and state standards, and maintain an accurate definition of the area of contamination.

#### Alternative 4-Extract

In the extract alternative, only groundwater is treated; any residual contamination found in subsurface soil is not treated. This alternative includes the following components:

Groundwater Extraction and Treatment. Would be used to capture the contaminant plume. One extraction well would be installed at the downgradient edge of the plume, directly adjacent to the building to extract groundwater from beneath the building. Extracted groundwater will be treated as discussed in Section 10.2.3.

Based on modeling projections, the groundwater extraction and treatment system proposed under this alternative would need to operate for at least 8 years. Year-round operation is assumed. Institutional controls would be used to prevent exposure until state and federal standards are achieved.

FINAL

Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. But, because it is probable that groundwater concentrations will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls would be implemented under this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 10.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above federal and state standards, and maintain an accurate definition of the area of contamination.

1.4

# 11.0 Description of Alternatives - OU 4

This section provides a concise description of the remedial action objectives and remedial alternatives for each source area in OU 4. The feasibility study provides information on the assumptions and calculations used in the development of the alternatives and the cost estimates for the alternatives.

# **11.1 Remedial Action Objectives**

Remedial action objectives (RAOs) were developed to specify actions and contaminant levels necessary to protect human health and the environment. RAOs define the contaminants of concern, exposure routes and receptors, and remediation levels, which are defined as acceptable contaminant levels for each exposure route.

The specific RAOs for the source areas within OU 4 are as follows:

- At DP25, prevent the continued migration of contaminants (BTEX) into the groundwater from the floating product and smear zone.
- At ST58, prevent the continued migration of benzene into the groundwater at a concentration that presents an unacceptable risk to future groundwater users.
- At DP25 and ST58, prevent human exposure to groundwater contaminated above state and federal standards.
- At SS35, clarify the extent of DDT contamination in surface soil, prevent migration of DDT from the surface soils into Garrison Slough, and remove any drums, if practical, that may present a threat of future release of hazardous materials to the environment.

The goal of the Superfund approach is to return usable groundwaters to their beneficial uses within a timeframe that is reasonable, given the particular circumstances of the site. Reasonable restoration time periods may range from very rapid (one to five years) to relatively extended (several decades). Factors, such as location, proximity to population, anticipated future land use, and mobility of the contaminant plume are considered when determining an appropriate restoration timeframe.

The use of: (1) natural attenuation with institutional controls. (2) source reduction through treating soil contamination to prevent additional contaminant leaching into the groundwater, and (3) groundwater pumping and treating were considered viable options for addressing groundwater contamination at Eielson AFB. For source areas within OUs 3, 4, and 5, the following site specific conditions were considered when determining reasonable restoration timeframes:

- Contaminant plumes in this relatively homogeneous aquifer do not appear to be spreading or are decreasing in size.
- Biodegradation, dispersion, dilution, or adsorption appear to be effectively containing or reducing the size of the contaminant plume.

- Areas impacted by the contamination are relatively small with little likelihood anticipated of extended exposure to groundwater.
- Future land use as a military installation is not anticipated to change in the foreseeable future.

# 11.2 Remedial Action Components

Two source areas within OU 4 (DP25 and ST58) contain groundwater and soils contaminated with petroleum hydrocarbons. BTEX: lead and DDT have been found in surface soils only at SS35. The individual alternatives to address these three source areas are assembled from one or more of the following remedial action components:

No Action

Components to Address Groundwater Contamination

- · Institutional Controls and Groundwater Monitoring
- Groundwater Extraction and Treatment

Components to Address Soil Contamination

- Excavation/Removal
- Cover for Containment in Place
- Bioventing

The no action alternative does not address the RAOs established for each source area. It represents the baseline risk without institutional controls or active remediation. The institutional controls and groundwater monitoring alternative prevents the use of contaminated groundwater, but does not include active remediation. The other components provide a range of control with varying timeframes to achieve the RAOs. Because it is difficult to remove all of the residual contamination and to reduce groundwater contaminant concentrations below state and federal water quality standards, even with groundwater extraction and treatment, it is expected that all of the alternatives would require long-term management of the area.

The six remedial action components are described in the following paragraphs. Following this discussion, the components are assembled into remedial action alternatives for each of the source areas. Costs for the alternative for each source area is included in Table 11.1.

### 11.2.1 No Action

Under this approach, no action is taken to remove contaminants from the soils or the groundwater. It represents the baseline risk without institutional controls or active remediation. Evaluation of the no action approach is required by the NCP to provide a baseline against which other alternatives can be compared. Under the no action alternative, no remedial measures are implemented, including longterm review, monitoring, or controls. No direct costs are associated with the no action alternative.

			Costs	
Source Area	· Alternative Description	Capital	30-yr O&M	Total
DP25	No Action	<b>\$</b> 0	<b>\$</b> 0	\$0
	Groundwater Monitoring/Institutional Controls	\$5,300	\$204,700	\$210,000
	Limited Soil Excavation/Groundwater Monitoring/Institutional Controls	\$3,800,000	\$200,000	\$4,000,000
	Groundwater Extraction and Treatment/ Groundwater Monitoring/Institutional Controls	\$1,500,000	\$4,000,000	\$5,500,000
SS35	No Action	\$0	<b>\$</b> 0	<b>\$</b> 0
	Institutional Controls	<b>\$</b> 0	\$0	\$0
	Soil Cover	\$40,000	<b>\$</b> 0	\$40,000
	Soil Excavation/Drum Removal (a) Onsite Disposal (b) Offsite Disposal (c) Offsite Disposal and Treatment	\$410,000 \$490,000 \$2,00,000	\$0 \$0 \$0	\$410,000 \$490,000 \$2,100,000
SS58	No Action	<b>\$</b> 0	<b>\$</b> 0	\$0
	Groundwater Monitoring/Institutional Controls	\$5,300	\$134,700	\$140,000
	Bioventing/Groundwater Monitoring/ Institutional Controls	\$170,000	\$180,000	\$350,000
	Bioventing/Groundwater Extraction and Treatment/Groundwater Monitoring/Institutional Controls	\$1,300,000	\$1,500,000	\$2,800,000

#### Table 11.1. Costs of Remedial Alternatives for Source Areas in OU 4.

# 11.2.2 Institutional Controls and Groundwater Monitoring

Institutional controls are used to prevent current and future human exposure to contamination remaining at the source areas at concentrations above health-based risk levels and federal and state standards. Specific controls include restrictions limiting access to the source areas and administrative controls to limit groundwater and future land use. Access restrictions, including such measures as permanent markers are used to prevent direct human exposure to contaminants. Groundwater restrictions are implemented by placing written notification in base policies prohibiting the use of contaminated groundwater. In addition, all existing and any new wells located in or near the contaminated portion of the aquifer shall be locked to prevent unauthorized use.

The administrative controls for limiting future land use include placing written notification of these remedial actions in the base land use master plan. The notification shall prohibit any activity that

disrupts aspects of the engineered controls. A copy of the notification is provided to any prospective transferees of the property and is included in any transfer documents, including deeds, in the event that the Air Force released control of the affected property. The Air Force shall provide EPA and the State with written verification that notification(s) have been implemented.

# 11.2.3 Groundwater Extraction and Treatment

Groundwater extraction was evaluated for source areas DP25 and SS58 as a way to potentially accelerate aquifer restoration. Groundwater extraction focuses on reduction of contaminant mass through removing the dissolved constituents in groundwater. It involves removing contaminated groundwater for aboveground physical/chemical treatment. The estimated number and location of extraction wells, as well as estimated timeframes to achieve cleanup, are discussed in Sections 11.3 and 11.5.

Due to the high metals levels naturally found in the groundwater at Eielson AFB, metals removal may be necessary prior to treating the water for site-specific contamination. Treatability studies are recommended to evaluate iron fouling problems and determine if the sludge generated by dissolved metals would be a hazardous waste. In addition, no proven method is known for removing lead from groundwater at a reasonable cost in a reasonable amount of time. However, a treatability test is being performed at another site at Eielson AFB to determine the fate and transport of lead and the most viable option for extraction and treatment, if warranted. Results from this test will be used to further evaluate lead remediation at ST58.

The need for air emission controls would be evaluated and implemented in accordance with the Clean Air Act (42 USC 7401). Treated groundwater would be discharged to surface water in accordance with federal ambient water quality criteria. and state 18 AAC 70, and 18 AAC 72.

# 11.2.4 Excavation/Removal

The excavation/removal component is a source control measure involving the excavation and removal of contaminated materials for offsite or onsite treatment or disposal. For source area DP25, contaminated soils would be excavated. For source area SS35, drums would be removed and contaminated soil would be excavated. Three disposal options, onsite, offsite, or offsite with treatment, were evaluated for disposal of contaminated material from SS35. The excavated area would then be back-filled with clean material.

# 11.2.5 Cover

The cover component was evaluated for SS35 and involves placement of a soil cover over the surface of the source area eliminating direct contact with surface contamination and runoff into nearby Garrison Slough. The type of cover material is dependent on the specific remedial action objectives for the source area.

# 11.2.6 Bioventing

Bioventing was evaluated for source area SS58. The bioventing component involves enhancing the destruction of fuel-derived organic constituents in soil by injecting air into the soil to sustain naturally occurring microorganisms that break down the fuel. The number of injection wells and estimates of treatment times are provided in Section 11.5.

# 11.3 Source Area DP25

## 11.3.1 Site Conditions

The primary concerns at source area DP25 are the presence of a layer of fuel-saturated soils at the water table, BTEX-contaminated soils in the smear zone, and BTEX compounds and lead in the groundwater. The primary potential exposure pathway for source area DP25 is from the future unrestricted use of contaminated groundwater. Therefore, contaminants in groundwater and the risk from continued leaching of contaminants to groundwater are of primary concern.

A source control program to identify and replace leaking tanks or pipes and line the berms will be implemented as part of the base tank program. This plan would include locating and repairing leaking tanks and piping, removing 0.3 to 0.6 m (1 to 2 ft) of gravel within the berms, installing an impermeable liner, and replacing the gravel. The alternatives developed for source area DP25 assume that no fuel continues to leak into the soil.

# 11.3.2 Remedial Alternatives

In accordance with NCP guidance, a range of potential remedial alternatives have been developed. The alternatives range from no action to removal for ex situ treatment and disposal. Four comprehensive alternatives have been developed to address the contaminated soil, floating fuel, and contaminated groundwater at source area DP25. The alternatives are summarized in Table 11.2.

Alternative		Soils	Fuel	Groundwater
1.	No action	None	None	None
2.	Institutional Controls/ Groundwater Monitoring	None	None	Natural attenuation, groundwater monitoring, institutional controls
3.	Limited Removal	Remove contaminated soils during berm liner installation, landfarming or composting	None	Natural attenuation, groundwater monitoring, institutional controls
4.	Extract	None	Active skimming	High-rate groundwater extraction, treatment, discharge to surface water, institutional controls, and groundwater monitoring

Table 11.2. Prin	ary Components	in Alternatives	for	Source /	Area DP25	j.
------------------	----------------	-----------------	-----	----------	-----------	----

The following subsections describe the remedial alternatives developed for source area DP25. The descriptions identify the technologies, describe the process options, and present the primary assumptions.

### Alternative 1 - No Action

The no action alternative was evaluated as discussed in Section 11.2.1.

#### Alternative 2 - Institutional Controls and Groundwater Monitoring

This alternative is a limited action alternative consisting of the following components.

Under this alternative, floating fuel and BTEX-contaminated soils beneath the storage tanks would continue to act as a contaminant source to groundwater and the groundwater would not be actively remediated. Remediation time in this case is a function of natural processes in groundwater including sorption/desorption, biodegradation, and contaminant dispersion resulting from groundwater flow through the aquifer.

Based on currently observed concentrations in the soil and groundwater, it was estimated that BTEX constituents may continue to leach into groundwater. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively degrading the contamination and containing it within the bermed areas around the tanks. Because of this continuing source of release and the probability that groundwater concentrations will remain above federal and state standards for an extended period of time, a long-term groundwater monitoring program and institutional controls are included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 11.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

### Alternative 3-Limited Removal

The limited removal alternative focuses on reducing the long-term source of contamination in soils at source area DP25 to the extent possible without interfering with the integrity of the tank berms. The removal alternative includes the following components.

It is assumed that approximately 21,000 m<sup>3</sup> (28,000 cu yd) of BTEX-contaminated soil would be excavated for treatment onsite using composting or landfarming. The excavated soil would include the top 30.5 cm (12 in.) of soil from the entire area within the berm, approximately 153 m<sup>3</sup> (200 cu yd) of BTEX-contaminated soil from the area of the fuel spill near monitoring well 53M01, and soils to the top of the water table at approximately 1.2 m (4 ft) bls from the areas where floating fuel has been identified. The excavation of soil from these areas could be completed in one construction season.

Because some of the source would be removed. RAOs may be achieved slightly faster under this alternative than without source control. The groundwater would not be actively remediated. The contaminants would be allowed to disperse and degrade naturally. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because some soil and smear zone contamination will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls are included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented as discussed in Section 11.2.2 to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

### Alternative 4-Extract

The extract alternative focuses on reducing the contaminant mass in groundwater. It includes short-term active skimming of floating product in conjunction with high-rate extraction of groundwater with treatment. The three components of the extract alternative are briefly outlined as follows.

Active skimming and pumping of approximately seven dual-phase extraction wells would be used to address the floating fuel layer at source area DP25. The active skimming wells, equipped with groundwater extraction pumps and product skimmer pumps, would be installed within the bermed area as part of this alternative. During the active skimming phase of operation, the groundwater extraction rate would be optimized to enhance product recovery, while minimizing the chance of distributing the product over a larger area. The alternative assumes the system will operate year-round for 3 years and recover a small fraction of the product present (150,000 L [40,000 gal]). The recovered fuel would be sent to the power plant to be burned.

When no more fuel can be recovered through active skimming and to minimize the potential for any further distribution of residual undissolved contamination, full-scale groundwater extraction would be implemented. It is assumed that two extraction wells would be installed downgradient of the two areas with residual floating fuel and would be pumped at a rate to remove as much contaminated groundwater as possible. The combined extraction rate of both wells is estimated to be 80 gpm.

The extracted groundwater would be treated using an air stripper to remove the VOCs and using activated carbon to remove SVOCs from the extracted water as described in Section 11.2.3. The effluent from the treatment system would be discharged to Hardfill Lake. The need for air emission controls would be evaluated and implemented in accordance with the Clean Air Act (42 USC 7401). Treated groundwater would be discharged to surface water in accordance with federal ambient water quality criteria. and state 18 AAC 70. and 18 AAC 72.

Based on modeling projections, the high-rate groundwater extraction and treatment system proposed under this alternative would need to operate for more than 30 years after the active skimming operation has ceased. Year-round operation is assumed, although permafrost present in the area will affect yearly maintenance requirements.

The BTEX in the residual fuel and soils will continue to act as a source of contaminants to groundwater for an estimated 60 years. Removal of some of the contaminant mass through groundwater extraction may achieve the RAOs slightly faster. In addition, contaminants not treated would be allowed to disperse and degrade naturally. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because the continuing source of contamination will remain above state and federal standards for an extended period of time, a long-term groundwater monitoring program and institutional controls are included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented as discussed in Section 11.2.2 to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

# 11.4 Source Area SS35

# 11.4.1 Site Conditions

The primary concern for source area SS35 is from DDT found in the surface soils in an area adjacent to Garrison Slough. DDT has also been found in the sediment samples from Garrison Slough in the vicinity of source area SS35. In addition, drums that may contain hazardous materials may be buried in three small disposal areas within the site. Groundwater in this area does not pose an unacceptable risk to human health and the environment.

Based on the homogeneous nature of the site and the limited data available, the extent of the contamination and, therefore, the volume of contaminated soil, cannot be well defined. It is not known whether these concentrations represent hot spots associated with drum burial areas where the samples were taken or whether DDT contamination could be more widespread. Based on the 1994 data, 6.5 kg (13.3 lb) of DDT was estimated to be spread over 459 m<sup>3</sup> (600 cu yd) of DDT-contaminated soil.

SS35 appears to be acting as a source of surface water or sediment contamination. PCBs and pesticides, particularly DDT, were found in soil at source area SS35, which is located adjacent to Garrison Slough. Elevated body burdens of DDT were found in fish caught near SS35. Cumulative ecological risks at Eielson AFB are currently being evaluated under the Sitewide program. Preliminary conclusions indicate that SS35 may present reproductive risks to birds and mammals from ingestion exposure to PCBs and DDT. The Sitewide biological risk assessment addresses ecological risks from all areas on base.

# 11.4.2 Remedial Alternatives

Based on the remedial objectives, four remedial alternatives were developed for source area SS35. The alternatives are listed in Table 11.3. The following subsections describe the remedial alternatives developed for source area SS35. The descriptions provide details about the technologies, describe process options, and present key assumptions.

Alternative	Soil Components
1. No action	None
2. Control	Institutional control to prevent human access
3. Cover	Surface cover to prevent contaminant transport to Garrison Slough and to eliminate surface soil exposure
4. Removal	Remove the drums and excavate hot spots of contaminated soil

**\_\_Table 11.3.** Primary Components of the Remedial Alternatives for Source Area SS35

### Alternative 1-No Action

The no action alternative was evaluated as discussed in Section 11.2.1.

### Alternative 2-Control

The control alternative is a limited action alternative that includes institutional controls and groundwater monitoring. It focuses on the use of institutional controls to prevent human intrusion into the drum disposal areas. The alternative includes base policies restricting access to the area as described in Section 11.2.2.

The control alternative would not achieve RAOs for source area SS35. The DDT in surface soil would continue to be a potential exposure hazard to human and ecological receptors until it degraded. Institutional controls would also not prevent surface water runoff into Garrison Slough.

#### Alternative 3 - Cover

The cover alternative consists of the installation of a cover as described in Section 11.2.5. The soil cover serves to prevent the migration of contaminants into Garrison Slough by overland runoff and to prevent direct soil contact and ingestion of contaminated surface soils by the base personnel and ecological receptors. The cover is proposed for those areas where DDT has been detected above risk-based levels in the surface soil. In this alternative, the contamination would be contained until natural processes eventually degraded the DDT.

The soil cover would be designed to coordinate with the existing contours to facilitate runoff and minimize the need for maintenance. The cover would extend beyond the areas where risk-based levels are exceeded and would be tapered, as appropriate, to the existing grade. It is assumed the cover consists of 15.2 cm (6 in.) of native top soil overlying 30.5 cm (12 in.) of native sand and gravel. Both layers would be placed in horizontal lifts and compacted. After the cover is placed, the surface would be planted with grass to stabilize the soil cover. Construction of the soil cover could be completed within one summer season.

#### Alternative 4 – Removal

The removal alternative consists solely of the excavate/remove component discussed in Section 11.2.4. It focuses on removing and disposing of surface soil with DDT concentrations above riskbased levels. Removal of this soil would prevent migration of contaminants into Garrison Slough and prevent exposure through direct contact and ingestion. In addition, buried drums, if found, would be excavated and residual contents would be consolidated, characterized, and incinerated, if necessary. A description of the primary components follows.

Soil in areas where DDT exceeds risk-based levels would be excavated from the site. It is estimated the extent of contaminated surface soil is less than 0.3 m (1 ft) deep and covers an area of approximately 279 m<sup>2</sup> (3000 ft<sup>2</sup>) which would produce approximately 459 m<sup>3</sup> (600 cu yd) of contaminated soil. The soil would be disposed in one of three ways: (1) by offsite transportation and incineration, (2) by offsite transportation and landfill disposal in a county landfill, or (3) by disposal in an on-base landfill. Although the soil is not expected to be a RCRA waste, the concentrations are higher than the universal treatment standard, as published in the land disposal restrictions. Therefore, the county landfill may not accept this material. An alternate disposal option for this small amount of material would be to place it in an on-base landfill. This action would remove the contaminated soil from the vicinity of Garrison Slough and place it in a managed landfill.

Three areas of the site that exhibited magnetic anomalies are presumed to contain up to 200 buried drums. Part of this alternative consists of excavating soil to remove, consolidate, characterize, and

dispose of the drums. It is estimated that up to  $8410 \text{ m}^3$  (11,000 cu yd) of soil would have to be excavated to remove all of the drums. One fourth of the drums are assumed to contain residual material that must be incinerated. The soil surrounding the drums is assumed to be uncontaminated and suitable for backfilling the excavation. After excavation, the drums will be opened, characterized, and consolidated for offsite transportation and incineration.

# 11.5 Source Area ST58

# 11.5.1 Site Conditions

The primary concern for source area ST58 is benzene and lead in the groundwater. Petroleum hydrocarbons were also found in the subsurface soils.

The primary potential exposure pathway for source area ST58 is from the future unrestricted use of contaminated groundwater. Therefore, contaminants in groundwater and the risk from continued leaching of contaminants to groundwater are of primary concern.

# 11.5.2 Remedial Alternatives

Four alternatives were developed for remediation of ST58 and are listed in Table 11.4. Descriptions for each of the four alternatives is presented in the following.

	Alternative	Soil Components	Groundwater Components			
1.	No action	None	None			
2.	Institutional Control/ Groundwater Monitoring	None	Natural attenuation, groundwater monitoring, institutional controls			
3.	In situ	Bioventing	Natural attenuation, groundwater monitoring, institutional controls			
4.	In situ/Extract	Bioventing	Extraction, metals precipitation, air stripping, discharge to Garrison Slough, institutional controls, groundwater monitoring			

Table 11.4. Primary Components in Alternatives for Source Area ST58

# Alternative 1-No Action

The no action alternative was evaluated as discussed in Section 11.2.1.

# Alternative 2-Institutional Controls and Groundwater Monitoring

Based on currently observed concentrations in the soil and groundwater, it was estimated that BTEX constituents may continue to leach into groundwater. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination. Because of this continuing source of release and the probability that groundwater concentrations will remain above federal and state standards for an extended period of time, a longterm groundwater monitoring program and institutional controls are included as part of this alternative.

Institutional controls and groundwater monitoring would be implemented, as discussed in Section 11.2.2, to restrict groundwater use, prevent human exposure to contaminants at concentrations above state and federal standards, and maintain an accurate definition of the area of contamination.

#### Alternative 3-In Situ

The in situ alternative consists of the following remedial action components:

Bioventing is implemented, as described in Section 11.2.6, to reduce the BTEX concentrations in soils. BTEX in the groundwater would be allowed to attenuate through natural processes. It has been assumed that hot spots have been removed and bioventing will achieve long-term bioremediation of residual contaminants in the soil. Bioventing employs a target air flow rate through the vadose zone to optimize oxygen availability for microbial growth.

It is assumed the bioventing system would use nine vertical injection wells on a 15.2-m (50-ft) spacing. The injection wells would be used to introduce oxygen or withdraw soil vapor from an assumed 45.7- to 91.4-m (150- by 300-ft) target area. The air injection rate would be based on achieving an optimal number of pore-volume exchanges per week for contaminant degradation. BTEX in vadose zone soils is expected to be removed by bioventing after a period of 1 year.

Based on currently observed concentrations in the soil and groundwater, it was estimated that BTEX constituents may continue to leach into groundwater. Because it is probable the groundwater concentrations may remain above state and federal standards for an extended period of time, institutional controls and groundwater monitoring will be implemented, as described in Alternative 2.

### Alternative 4-In Situ/Extract

The in situ/extract alternative consists of the following remedial action components:

Bioventing will be implemented as described in Alternative 3 to remove vadose zone BTEX constituents.

Groundwater extraction and treatment will be implemented, as described in Section 11.2.3, to expedite groundwater restoration. A single extraction well removing approximately 40 gpm is assumed for this alternative. The extracted groundwater would be treated using an air stripper to remove the VOCs and using activated carbon to remove other petroleum contaminants from the water. The treated effluent would be discharged to Garrison Slough. The need for air emission controls would be evaluated and implemented in accordance with the Clean Air Act (42 USC 7401). Treated groundwater would be discharged to surface water in accordance with federal ambient water quality criteria, 18 AAC 70, and 18 AAC 72.

Because it is probable the groundwater concentrations may remain above state and federal standards for an extended period of time, institutional controls and groundwater monitoring will be implemented as described in Alternative 2.

# 12.0 Description of Alternatives-OU 5

This section provides a concise description of the remedial action objectives and remedial alternatives for source areas LF03/FT09 in OU 5. Detailed information on the assumptions and calculations used in the development of the alternatives are presented in the FS.

# 12.1 Remedial Action Objectives

The primary contaminants of concern at source areas LF03/FT09 are TCE, PCE, vinyl chloride, and 1.4-dichlorobenzene in the groundwater directly beneath the landfill. The specific objectives are, therefore, to

- prevent direct human contact with landfill contents
- minimize the migration of chlorinated VOCs into the groundwater
- · control surface water runoff and erosion
- continue to comply with state and federal standards at the boundary of the waste management area.

# 12.2 Site Conditions at Source Areas LF03/FT09

The main part of the landfill was excavated to a depth below the groundwater table. During the time the landfill was active, wastes were reportedly dumped into the standing water. The shallow trenches on the eastern side of the landfill may be above the water table. This theory will be confirmed in the remedial design phase. The trenches were active during the early 1980s and received waste from the base, including industrial wastes and solvents from the flightline shops. The fire training area (source area FT09) was located on top of the fill in the west-central portion of the landfill. During fire training exercises, JP-4 and other liquids were dumped into an unlined pit and ignited. The soils and waste in the fire training area are contaminated with fuel and solvents.

# 12.3 Remedial Alternatives

Four alternatives have been developed to address the landfill and fire training area at LF03/FT09. The alternatives are presented in Table 12.1 and are described in the following subsections. Costs are presented in Table 12.2.

# 12.3.1 Alternative 1 - No Action

Under this alternative, no action would be taken to remove contaminants from or prevent exposure to the landfill or groundwater in source area LF03/FT09. It represents the baseline risk without institutional controls or active remediation. Evaluation of the no action alternative is required by the NCP to provide a baseline against which other alternatives can be compared. The no action alternative will not protect human health or the environment or achieve RAOs. No risk reduction or reduction in the toxicity, mobility, or volume of contaminants will occur. No direct cost is associated with the no action alternative.

	Alternative	Landfill Components	Groundwater Components			
1.	No action	None	None			
2.	Institutional controls Groundwater Monitoring	Restrictions to prevent exposure to landfill contents and contaminated groundwater	Natural attenuation, groundwater monitoring, institutional controls			
3.	Cover	Soil cover to promote drainage and prevent direct contact	Natural attenuation, groundwater monitoring, institutional controls			
4.	Сар	Geosynthetic landfill cap over trenches, soil cover over remaining landfill	Natural attenuation, groundwater monitoring, institutional controls			

#### Table 12.1. Primary Components in Alternatives for Source Areas LF03/FT09

## Table 12.2. Costs of Remedial Alternatives for Source Areas In OU 5

Source Area	Alternative Description	Costs				
f		Capital	30-yr O&M	Total		
LF03/FT09	No Action	\$0	<b>\$</b> 0	\$0		
	Groundwater Monitoring/Institutional Controls	\$11,000	\$219,000	\$230,000		
	Soil Cover/Groundwater Monitoring/ Institutional Controls	\$4,000.000	\$300,000	\$4,300,000		
	Soil Cover/Composite Cover/ Groundwater Monitoring/Institutional Controls	\$7,100,000	\$400,000	\$7,600,000		

# 12.3.2 Alternative 2 – Institutional Controls and Groundwater Monitoring

Institutional controls are implemented to achieve one or more of the following:

- · prevent human contact with the landfill waste.
- prevent exposure to groundwater contaminants at concentrations above state and federal standards.
- · restrict access or development on the landfill.

Institutional controls are implemented to reduce potential risks from intrusion into the landfill or the use of contaminated groundwater from beneath the landfill.

This alternative is a limited action alternative. It consists of the following components:

Groundwater monitoring is conducted at the edge of the waste management area to monitor any migration of contaminants from the source areas LF03/FT09 landfill beyond the boundary of the waste management area and Garrison Slough. Monitoring will also be performed at the edge of the waste management area to confirm continuing compliance with state and federal standards.

This alternative would not achieve the remedial action objectives for source areas LF03/FT09. Without proper closure, the landfill debris, which may include buried drums of organic contaminants and the residual soil contamination at the fire training area, is likely to pose a tisk from direct contact and uncontrolled surface runoff.

# 12.3.3 Alternative 3 - Cover

The cover alternative includes the following components:

A soil cover would be placed over the landfill area in accordance with applicable or relevant and appropriate requirements of RCRA Part 264. The cover would minimize the long-term migration of liquids from the landfill and would prevent direct contact with landfill debris and contaminated soil.

Conceptually, the covered area would be less than 40.5 hectares (100 acres) in size. The cover would consist of native soil (sand and gravel) excavated from the base and placed and graded to provide a minimum 1 percent slope. Fine soil, if available, would be more suitable material for the cap, because the permeability of the cap would be reduced. Because the cover would promote surface water drainage, a drainage system (perimeter ditches) would be included to direct and manage surface runoff.

The cover alternative would reduce or eliminate the threat of direct contact of personnel with buried landfill debris. It would also help control surface water runoff and erosion and, subsequently, will reduce the leaching of contaminants to groundwater.

Currently, no state or federal standards are exceeded at the edge of the waste management area, but institutional controls and groundwater monitoring will be implemented to achieve the following:

- · prevent human contact with the landfill waste.
- prevent exposure to groundwater contaminants at concentrations above state and federal standards.
- restrict access or development on the landfill.

Institutional controls are implemented to reduce potential risks from intrusion into the landfill or from use of contaminated groundwater from beneath the landfill. Groundwater monitoring is conducted at the edge of the waste management area to monitor any migration of contaminants from the source areas LF03/FT09 landfill beyond the boundary of the waste management area and Garrison Slough. Monitoring will also be performed to confirm continuing compliance with state and federal standards at the edge of the waste management area.

# 12.3.4 Alternative 4 - Cap

The existing open area of source areas LF03/FT09 would be capped or covered as part of this alternative. The cap alternative includes the following components:

A soil cover, as described in Alternative 3, would be placed over the LF03/FT09 landfill area, where appropriate.

For areas requiring a composite cover under RCRA Part 264, some areas of LF03 may be capped with a multi-layer geosynthetic cap. Conceptually, the capped area was assumed to be about 8.1 hectares (20 acres) in size. The cap would reduce the release of landfill contaminants to the groundwater through leaching for material existing above the groundwater table. Landfill gas vents to release methane would be installed, as needed, to adequately vent the landfill. A drainage system would be included to direct and manage surface runoff. The capped area would be fenced to minimize traffic over it that could damage the cap components.

The cap alternative would achieve RAOs relating to the elimination of exposure to the waste materials, controlling surface water runoff and erosion, and reducing infiltration to subsequently reduce leaching of contaminants to groundwater.

Currently, no state or federal standards are exceeded at the edge of the waste management area, but institutional controls and groundwater monitoring will be implemented to reduce potential risks from intrusion into the landfill or from use of contaminated groundwater from beneath the landfill. Groundwater monitoring is conducted at the edge of the waste management area to monitor any migration of contaminants from the source areas LF03/FT09 landfill beyond the boundary of the waste management area and Garrison Slough. Monitoring will also be performed to confirm continuing compliance with state and federal standards at the edge of the waste management area.

والمرد

# 13.0 Summary of the Comparative Analysis of Alternatives

In accordance with federal regulations, the cleanup alternatives for each source area were evaluated based on the nine criteria presented in the National Contingency Plan (NCP). The nine criteria are divided into three groups as follows:

Threshold Criteria - Must be met by all alternatives:

- (1) <u>Overall Protection of Human Health and the Environment</u>. How well does the alternative protect human health and the environment, both during and after construction?
- (2) <u>Compliance with Applicable or Relevant and Appropriate Requirements</u>. Does the alternative meet all applicable or relevant and appropriate state and federal laws?

Balancing Criteria - Used to compare alternatives to each other:

- (3) <u>Long-Term Effectiveness and Permanence</u>. How well does the alternative protect human health and the environment after cleanup? What, if any, risks will remain at the area?
- (4) <u>Reduction of Toxicity, Mobility, or Volume Through Treatment</u>. Does the alternative effectively treat the contamination to significantly reduce the toxicity, mobility, and volume of the hazardous substance?
- (5) <u>Short-term Effectiveness</u>. Are there potential adverse effects to either human health or the environment during construction or implementation of the alternative? How fast does the alternative reach the cleanup goals?
- (6) <u>Implementability</u>. Is the alternative both technically and administratively feasible? Has the technology been used successfully at similar areas?
- (7) Cost. What are the relative costs of the alternatives?

Modifying Criteria - Evaluated as a result of public comments:

- (8) <u>State Acceptance</u>. What are the state's comments or concerns about the alternatives considered and about the preferred alternative? Does the state support or oppose the preferred alternative?
- (9) <u>Community Acceptance</u>. What are the community's comments or concerns about the alternatives considered and about the preferred alternative? Does the community generally support or oppose the preferred alternative?

This section contains the results of the comparative analyses for the source areas DP44, WP45/SS57, SS61, DP25, SS35, ST58, and LF03/FT09 where alternatives were evaluated in the Feasibility Study.

# 13.1 DP44 - Battery Leach Field

#### 13.1.1 Threshold Criteria

#### Overall Protection of Human Health and the Environment:

All of the alternatives, except the No Action alternative, would implement institutional controls to prevent exposure to contaminated groundwater until federal and state regulatory levels are achieved. The SVE/extract alternative would be designed to treat contaminated soils and groundwater and could theoretically achieve cleanup objectives more quickly than either Alternative 3, which includes source reduction to treat the soil contamination in the unsaturated zone or Alternatives 1 and 2, which rely on natural processes.

Although the alternatives include varying levels of contaminant removal, all alternatives are expected to require an extended period of time to comply with all state and federal regulatory levels throughout the contaminant plume. For this source area, source reduction using SVE may be warranted, if sufficient contamination remains in the vadose zone.

#### Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The primary ARARs for this source area focus on groundwater protection. Compliance with ARARs include compliance with federal MCLs and with State of Alaska Water Quality Standards (18 AAC 70). All alternatives will eventually comply with groundwater chemical-specific ARARs and, with the exception of the No Action alternative, would prevent exposure to contaminated groundwater through the use of institutional controls. The period for compliance with all state and federal regulatory levels may be decreased through implementation of Alternative 3, SVE, or Alternative 4, SVE/Extract.

It is expected that all action-specific ARARs could be met by all alternatives, including air emission limitations and surface water discharge levels. No action-specific ARARs exist for Alternative 1, No Action, or Alternative 2, Institutional Controls/Groundwater Monitoring.

#### 13.1.2 Balancing Criteria

Table 13.1 includes the comparative analysis among the balancing criteria for source area DP44.

### 4 13.1.3 Modifying Criteria

#### State Acceptance:

The State of Alaska concurs with the selected remedy for this source area.

#### Community Acceptance:

# 13.2 WP45/SS57 - Photo Laboratory/Fire Station Parking Lot

# 13.2.1 Threshold Criteria

### Overall Protection of Human Health and the Environment:

All of the alternatives, except the *No Action* alternative, would implement institutional controls to prevent exposure to contaminated groundwater until federal and state regulatory levels are achieved. Alternative 6 would be designed to treat both contaminated soils and groundwater and could theoretically achieve cleanup objectives more quickly than either Alternatives 3, 4, or 5 which include SVE, bioventing, and/or soil excavation to reduce the source of groundwater contamination in the unsaturated soils. Alternatives 1 and 2 rely on natural processes to achieve groundwater cleanup levels. Recent data indicates that little residual contamination remains in the vadose zone. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination.

Although the alternatives include varying levels of contaminant removal, all alternatives are expected to require an extended period of time to comply with all state and federal regulatory levels throughout the contaminant plume.

Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The primary ARARs for this source area focus on groundwater protection. Compliance with ARARs include compliance with both federal MCLs and with state of Alaska Water Quality Standards (18 AAC 70). All alternatives will eventually comply with groundwater chemical-specific ARARs and, with the exception of the *No Action* alternative, all would prevent exposure to contaminated groundwater through the use of institutional controls. The period for compliance with all state and federal regulatory levels is not expected to be significantly decreased through source removal or groundwater extraction/treatment.

It is expected that all action-specific ARARs could be met by all alternatives, including air emission limitations and surface water discharge levels. There are no action-specific ARARs for Alternative 1, *No Action*, or Alternative 2, *Institutional Controls/Groundwater Monitoring*.

# 13.2.2 Balancing Criteria

Table 13.2 includes the comparative analysis amongst the balancing criteria for source areas WP45/SS57.

# 13.2.3 Modifying Criteria

#### State Acceptance:

The State of Alaska concurs with the selected remedy for these source areas.

#### Community Acceptance:

# 13.3 SS61 - Vehicle Maintenance Building 3213

# 13.3.1 Threshold Criteria

### Overall Protection of Human Health and the Environment:

All of the alternatives, except the No Action alternative, would implement institutional controls to prevent exposure to contaminated groundwater until federal and state regulatory levels are achieved. Alternatives 3 and 4 would be designed to treat either contaminated soils or groundwater and could theoretically achieve cleanup objectives more quickly than either Alternatives 1 or 2 that rely on natural processes to achieve groundwater cleanup levels. Dry wells and surrounding soils were removed during construction of the new building and little residual contamination is expected in the unsaturated soils. Unless additional soil contamination is identified, soil excavation is not expected to decrease the time to reach RAOs. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination within an area below the building.

Although the alternatives include varying levels of contaminant removal, all alternatives are expected to require an extended period of time to comply with all state and federal regulatory levels throughout the contaminant plume.

### Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The primary ARARs for this source area focus on groundwater protection. Compliance with ARARs include compliance with federal MCLs and with State of Alaska Water Quality Standards (18 AAC 70). All alternatives will eventually comply with groundwater chemical-specific ARARs and, with the exception of the No Action alternative, would prevent exposure to contaminated groundwater through the use of institutional controls. The period for compliance with all state and federal regulatory levels is not expected to be significantly decreased through source removal or groundwater extraction/treatment.

It is expected that all action-specific ARARs could be met by all alternatives, including air emission limitations and surface water discharge levels. No action-specific ARARs exist for Alternative 1, No Action, or Alternative 2, Institutional Controls/Groundwater Monitoring.

# 13.3.2 Balancing Criteria

Table 13.3 includes the comparative analysis among the balancing criteria for source area SS61.

# 13.3.3 Modifying Criteria

### State Acceptance:

The State of Alaska concurs with the selected remedy for these source areas.

### Community Acceptance:

# 13.4 DP25 - E-6 Fuel Storage Tank Area

# 13.4.1 Threshold Criteria

### Overall Protection of Human Health and the Environment:

All of the alternatives, except the No Action alternative, would implement institutional controls to prevent exposure to contaminated groundwater until federal and state regulatory levels are achieved. Alternatives 3 and 4 would be designed to treat either contaminated soils or groundwater, if accessible, and could theoretically achieve cleanup objectives more quickly than either Alternatives 1 or 2 that rely on natural processes to achieve groundwater cleanup levels. Groundwater and soil contamination appear to be confined beneath the bermed area containing the tank farm. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination within an area below this tank farm.

The success of the contaminant removal under the various alternatives is dependent upon the accessibility of the contamination and the implementability of a treatment system beneath the tank farm. Given the potential inability to treat a significant amount of contamination in either the subsurface soils or the groundwater, all alternatives are expected to require an extended period of time to comply with all state and federal regulatory levels throughout the contaminant plume.

## Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The primary ARARs for this source area focus on groundwater protection. Compliance with ARARs include compliance with federal MCLs and with State of Alaska Water Quality Standards (18 AAC 70). All alternatives will eventually comply with groundwater chemical-specific ARARs and, with the exception of the No Action alternative, would prevent exposure to contaminated groundwater through the use of institutional controls. Given the likely inability to treat a significant amount of contamination in either the subsurface soils or the groundwater, the period for compliance with all state and federal regulatory levels is not expected to be significantly decreased through source removal or groundwater extraction/treatment.

It is expected that all action-specific ARARs could be met by all alternatives, including air emission limitations and surface water discharge levels. No action-specific ARARs exist for Alternative 1, No Action, or Alternative 2, Institutional Controls/Groundwater Monitoring.

# 13.4.2 Balancing Criteria

Table 13.4 includes the comparative analysis among the balancing criteria for source area DP25.

# 13.4.3 Modifying Criteria

### State Acceptance:

The State of Alaska concurs with the selected remedy for this source area.

#### Community Acceptance:

No public comments were received regarding the alternatives or preferred alternatives included under this Record of Decision

# 13.5 SS35 - Asphalt Mixing and Drum Burial Area

## 13.5.1 Threshold Criteria

#### Overall Protection of Human Health and the Environment:

Alternatives 1 and 2 rely on natural processes to degrade the pesticides and do not prevent surface water runoff into the slough. Alternative 1 also does not prevent direct contact with contaminated soils, because it does not include institutional controls. Alternatives 3 and 4 would effectively prevent direct contact with soils and surface water runoff into the slough. If implementable and cost-effective, Alternative 4 could provide an increased level of permanence by removal of contaminated soils and drums in this area.

## Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The ARARs for this area are the federal and state ambient water quality standards and compliance with these levels for surface waters in Garrison Slough. Alternatives 1 and 2 may result in exceeding these standards. Alternatives 3 and 4 are expected to comply with all ARARs.

It is expected that action-specific ARARs could be met by all alternatives, including land disposal restrictions for Alternative 4, in the event of excavation, and disposal of hazardous waste and surface water discharge levels. No action-specific ARARs exist for Alternative 1, No Action, or Alternative 2, Institutional Controls/Groundwater Monitoring.

#### 13.5.2 Balancing Criteria

Table 13.5 includes the comparative analysis among the balancing criteria for source area SS35.

#### 13.5.3 Modifying Criteria

#### State Acceptance:

The State of Alaska concurs with the selected remedy for this source area.

#### Community Acceptance:

# 13.6 ST58 - Old Quartermaster Service Station

# 13.6.1 Threshold Criteria

### Overall Protection of Human Health and the Environment:

All of the alternatives, except the No Action alternative, would implement institutional controls to prevent exposure to contaminated groundwater until federal and state regulatory levels are achieved. Alternatives 3 and 4 would be designed to treat contaminated soils or groundwater and could achieve cleanup objectives more quickly than either Alternatives 1 or 2, which rely on natural processes to achieve groundwater cleanup levels. Within the groundwater, biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination.

All alternatives are expected to require an extended period of time to comply with all state and federal regulatory levels throughout the contaminant plume.

### Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

The primary ARARs for this source area focus on groundwater protection. Compliance with ARARs include compliance with federal MCLs and with state of Alaska Water Quality Standards (18 AAC 70). All alternatives will eventually comply with groundwater chemical-specific ARARs and, with the exception of the No Action alternative, would prevent exposure to contaminated groundwater through the use of institutional controls. If effective in removing contamination in the smear zone or saturated zone, Alternatives 3 and 4 could achieve ARARs in a shorter period of time than Alternatives 1 or 2 that rely on natural processes to achieve ARARs.

It is expected that all action-specific ARARs could be met by all alternatives, including air emission limitations and surface water discharge levels. No action-specific ARARs exist for Alternative 1, No Action, or Alternative 2, Institutional Controls/Groundwater Monitoring.

# 13.6.2 Balancing Criteria

Table 13.6 includes the comparative analysis among the balancing criteria for source area ST58.

# 13.6.3 Modifying Criteria

#### State Acceptance:

The State of Alaska concurs with the selected remedy for this source area.

### Community Acceptance:

# 13.7 LF03/FT09 - Inactive Base Landfill/Fire Training Area

### 13.7.1 Threshold Criteria

#### Overall Protection of Human Health and the Environment:

Alternatives 1 and 2 do not eliminate any risks associated with exposure to contaminated soils. Alternatives 3 and 4 are effective in preventing exposure to the soil contamination. The contents of the landfill would remain in place; therefore, groundwater monitoring would continue to ensure that levels at the edge of the waste management area do not exceed state or federal standards.

# Compliance with Applicable or Relevant and Appropriate Requirements (ARARs):

Currently, no groundwater at the edge of the waste management area exceeds state or federal standards. Alternatives 3 and 4 would be designed to meet the substantive applicable or relevant and appropriate requirements for RCRA Subtitle C Part 264.

### 13.7.2 Balancing Criteria

Table 13.7 includes the comparative analysis among the balancing criteria for source area LF03/FT09.

#### 13.7.3 Modifying Criteria

#### State Acceptance:

The State of Alaska concurs with the selected remedy for these source areas.

#### Community Acceptance:

~

Table 13.1. Comparison of Cleanup Alternatives for Source Area DP44 Using the Five Balancing Criteria									
DP44 Battery Shop Leach Field D									
Clea	n Up	Alternat	ives		Alternat	ives are ranked by			
1	2	3	4	BALANCING CRITERIA	comparing	g them to each other.			
<ul> <li>Long-term effectiveness and permanence Alternatives 1 and 2 both rely on natural processes, but Alternative 2 prevents us of contaminated groundwater. Alternatives 3 and 4 treat the source in soil to minimize future contaminant migration. In addition, Alternative 4 includes groundwater treatment and therefore addresses more of the contamination. However, its effectiveness in achieving groundwater standards is not well established.</li> </ul>									
0	Reduction of toxicity, mobility, or volume through treatment Neither Alternative 1 or 2 includes treatment. Alternatives 3 and 4 reduce toxicity and volume through treatment, but Alternative 4, which includes groundwater treatment, would treat a larger portion of the contamination.								
0	$\odot$	•	•	Short-term effectiveness Although the effectiveness of groundwate well-established. Alternative 4 might achi Alternative 3.	r extraction ieve cleanup	and treatment is not standards more quickly than			
NA	NA      NA     Implementability Under Alternatives 2-4, institutional controls are readily implementable. Alternatives 3 and 4 include readily available technologies. Alternative 4 is more difficult to implement due to the additional need for installation and construction of a groundwater extraction and treatment system and operational difficulties in an arctic climite								
0	140	'. 600		Cost (\$K) Total cost - capital plus O&M for 30 year	s at 5% inter	r <b>e</b> si.			
Clean	Up Alte	matives	_			KEY:			
<u>1</u> м	o Actio	n 2	] G <sub>7</sub>	oundwater Monitoring/Institutional Cont	alon 🖌	🔵 = best 😁 = good			
3 S	oil Vap	or Extracti	on/G	roundwater Monitoring/Institutional Cont	rols	(● = poor ) = worst			
4 S M	oil Vapo Ionitori	or Extractions ng/Instituti	m/Gro ional	oundwater Extraction and Treatment/Groun Controls	dwater	NA = Not Applicable			

	<b></b>						······································			
L	Tabl	le 1	3.2.	Com	pariso	n of (	Cleanup Alternatives for Source Area WP45/	SS57 Using	g the Five Bal	ancing Criteria
	V	VP4	45/	′SS	57		Photo Lab/Fire Station Parkin	g Lot	WP45	5/SS57
	Cie 1	ean [2]	Up 3	Alte	ernat	ives 6	BALANCING CRITERIA	Alternativ	es are ranked	i by other.
	0	) • • • • • • • • • • • • • • • • • • •								s a continuing y not be s 5 and 6 also ndwater r standards is
	0	0	•	•	•	•	Reduction of toxicity, mobility, o Neither Alternative 1 or 2 includes treatment volume through treatment of soil, but Altern groundwater treatment, would treat a larger	or volum t. Alternat natives 5 an portion of	e through ives 3-6 reduced of 6, which in the contamination	treatment to toxicity and clude ation.
	0	•	•	•	•	•	Short-term effectiveness Although the effectiveness of groundwater e well-established, Alternatives 5 and 6 might quickly than Alternatives 3 and 4. There ma excavation activities.	extraction a chieve cl ay be air en	nd treatment : eanup standar nissions durin	is not ds more g soil
	NA	<ul> <li>Implementability Under Alternatives 2-6, institutional controls are readily implementable. Alternatives 3-6 include readily available technologies. Alternatives 5 and 6 are more difficult to implement due to the additional need for installation and construction of a groundwater extraction and treatment system and the operations difficulties in an arctic climate. For Alternatives 4 and 6, excavation of soils</li> </ul>							able. 5 and 6 are and he operational n of soils	
	0	, <del>8</del> 0	1.200	000',	5.600	°.000	Cost (SK) Total cost - capital plus O&M for 30 years a	u 5% intere	st.	
Ţ.	Clea	n Up	Alte	mati	ves		·······		KEY:	
		No A	Ction	n	2	] Gr	oundwater Monitoring/Institutional Control	5	🔵 = best	🔵 = good
:"	3	Soil Vapor Extraction/Bioventing/Groundwater Monitoring/						🔿 = worst		
	4	Bioventing/Soil Excavation with Offsite Treatment and Disposal/ Groundwater Monitoring/Institutional Controls						Applicable		
	5	Soil Vapor Extraction/Bioventing/Groundwater Extraction and Treatment/ Groundwater Monitoring/Institutional Controls								
	6	Soil Grou Insti	Exca indwa tutio	ivatio ater E nal C	n with Extract Contro	ion s	site Treatment and Disposal/Bioventing/ nd Treatment/Groundwater Monitoring/	ł		

Table 13.3. Comparison of Cleanup Alternatives for Source Area SS61 Using the Five Balancing Criteria							
SS61 Vehicle Maintenance Building 3213 SS61					S61		
Cle:	an Up 2	Aiterna 3	tives 4	BALANCING CRITERIA	Alternati comparing	ves are ranked by them to each oth	ler.
$\overline{\odot}$	•	•	•	Long-term effectiveness and per No continuing source of groundwater con- groundwater contamination is limited to a Institutional controls will reliably preven Alternatives 2-4 provide about the same	rmanence tamination v an area direc t use of the p level of pro-	vas identified in th tly beneath the bui groundwater, then acctiveness.	e soil and lding. efore,
0	0	•	●	Reduction of toxicity, mobility, or volume through treatment Neither Alternative 1 or 2 includes treatment. Alternatives 3 and 4 would reduce toxicity and volume of contamination through treatment if an additional source of soil contamination was identified in the future. Alternative 4, which could include groundwater treatment under the building, could potentially treat a larger portion of the contamination.			
0	0	٢	•	Short-term effectiveness Although the effectiveness of groundwater extraction and treatment is not well-established. Alternative 4 might achieve cleanup standards more quickly than Alternative 3. If a concentrated source of soil contamination was found, there may be air emissions during soil excavation activities.			
NA	•	•	o	Implementability Under Alternatives 2-4, institutional controls are readily implementable. Alternatives 3 and 4 include readily available technologies although a source of soil contamination has not been identified. Alternative 4 is more difficult to implement due to the additional need for installation and construction of a groundwater extraction and treatment system under a building and operational difficulties in an arctic climate. For Alternative 3, if a concentrated source of soil contamination was found, excavation of soils below the water table and near structures would be very difficult.			
0	00	100	3.100	Cost (\$K) Total cost - capital plus O&M for 30 year	n at 5% inter	rest.	
Clean Up Alternatives						KEY;	
1       No Action       2       Groundwater Monitoring/Institutional Controls       Image: Control state of the state of						= good	
3 Soil Excavation with Offsite Treatment and Disposal/Groundwater () = poor () = w Monitoring/Institutional Controls					) = worst		
Groundwater Extraction and Treatment/Groundwater Monitoring/         NA = Not Applicable           Institutional Controls         NA = Not Applicable							

.

2

Table 13.4. Comparison of Cleanup Alternatives for Source Area DP25 Using the Five Balancing Criteria							
DP25				E-6 Fuel Storage Tank Area DI			
Cle	an Up	Altern	atives		Alternatives are ranked by		
1	2	3	4	BALANCING CRITERIA	comparing them to each other.		
0	•	•	•	Long-term effectiveness and permanence Groundwater contamination is limited to to an area directly beneath the bermed area containing the tank farm and the contaminated area does not appear to be expanding. Institutional controls will reliably prevent use of the groundwater; therefore, Alternatives 2-4 provide about the same level of protectiveness.			
0	0	⊙	•	Reduction of toxicity, mobility, or volume through treatment Neither Alternative 1 or 2 includes treatment. Alternatives 3 and 4 would reduce toxicity and volume of contamination through treatment. Alternative 4, which could include groundwater treatment and petroleum product removal under the bermed area, could potentially treat a larger portion of the contamination.			
0	0	٢	•	Short-term effectiveness Although the effectiveness of groundwater extraction and treatment is not well-established, Alternative 4 might achieve cleanup standards more quickly than Alternative 3.			
NA	•	•	•	Implementability Under Alternatives 2-4, institutional controls are readily implementable. Alternatives 3 and 4 include readily available technologies although available soil treatment and disposal facilities are limited. Alternative 4 is more difficult to implement due to the additional need for installation and construction of a groundwater extraction and treatment system under the bermed area and the operational difficulties in an arctic climate. For Alternative 3, excavation of soils below the water table and near structures would be very difficult.			
0	510	*.000	5.500	Cost (\$K) Total cost - capital plus O&M for 30 year	rs at 5% interest.		
Clea	Clean Up Alternatives KEY:						
1 No Action 2 Groundwater Monitoring/Institutional Controls							
3	3 Limited Soil Excavation with Onsite Bioremediation/Groundwater $\bigcirc$ = poor $\bigcirc$ = worst $\bigcirc$ = worst						
4       Groundwater Extraction and Treatment/Groundwater Monitoring/       NA = Not Applicable         1       Institutional Controls							

Table 13.5. Comparison of Cleanup Alternatives for Source Area SS35 Using the Five Balancing Criteria							
SS35 A		A	sphalt Mixing and Drum Burial Area SS35				
Cle	an Up	Alter	natives		Alternatives are ranked by		
1	2	3	4	BALANCING CRITERIA	comparing them to each other.		
0	0	•	•	Long-term effectiveness and permanence Alternatives 1 and 2 rely on natural processes to degrade the pesticides and do not prevent surface water runoff into the slough. Both Alternatives 3 and 4 would prevent direct contact and surface water runoff of the contamination. Alternative 3 would require long-term cover maintenance and monitoring. Alternative 4 would remove the source and would not leave residual contamination.			
0	0	0	٠	Reduction of toxicity, mobility, or volume through treatment Neither Alternative 1, 2, or 3 includes treatment. Only Alternative 4 would reduce toxicity and volume through treatment of contaminated soil.			
0	0	•	•	Short-term effectiveness Both Alternatives 3 and 4 would meet cleanup objectives in about the same amount of time. Under Alternative 4, there may be air emissions during soil and drum excavation activities.			
NA	•	•	•	Implementability Under Alternatives 2-4, institutional controls are readily implementable. Alternatives 3 and 4 include readily available technologies. Under Alternative 4, excavation and handling of a large volume of drums and soil would be very difficult and available soil treatment and disposal facilities are limited.			
0	Cost (SK) 0 0 40 a) 410 Total cost - capital plus O&M for 30 years at 5% interest. b) 490 c) 2,100						
Clean Up Alternatives KEY:							
1 No Action 2 Institutional Controls 3 Soil Cover 🔴 = best 🗢 = good							
4       Soil Excavation/Drum Removal with: a) Onsite Disposal <ul> <li>= poor</li> <li>= worst</li> <li>NA = Not Applicable</li> <li>Offsite Disposal and Treatment</li> </ul>							
.

	Table 13.6. Comparison of Cleanup Alternatives for Source Area ST58 Using the Five Balancing Criteria								
	S	T58	0	ntion Site ST58					
	Clean Up Alternatives				Alternatives are ranked by				
	1	2	3 4	BALANCING CRITERIA	comparing them to each other.				
	. 0	•	rmanence processes to degrade contamination, but d groundwater. Alternatives 3 and 4 treat aminant migration. In addition, sent and therefore addresses more of the as in achieving groundwater standards is						
<ul> <li>Reduction of toxicity, Neither Alternative 1 or 2 incl and volume through treatment treatment, would treat a large</li> <li>Short-term effectiveness Although the effectiveness of well-established, Alternative 4 Alternative 3.</li> <li>Implementability Under Alternatives 2-4, instit Alternatives 3 and 4 include r difficult to implement due to 1 of a groundwater extraction ar arctic climate.</li> </ul>				Reduction of toxicity, mobility, Neither Alternative 1 or 2 includes treatme and volume through treatment, but Alterns treatment, would treat a larger portion of	ction of toxicity, mobility, or volume through treatment r Alternative 1 or 2 includes treatment. Alternatives 3 and 4 reduce toxicity solume through treatment, but Alternative 4, which includes groundwater ent, would treat a larger portion of the contamination. t-term effectiveness igh the effectiveness of groundwater extraction and treatment is not stablished. Alternative 4 might achieve cleanup standards more quickly than ative 3.				
				Short-term effectiveness Although the effectiveness of groundwater well-established, Alternative 4 might achi Alternative 3.					
				Implementability Under Alternatives 2-4, institutional cont Alternatives 3 and 4 include readily availa difficult to implement due to the additional of a groundwater extraction and treatment arctic climate.	trols are readily implementable. able technologies. Alternative 4 is more al need for installation and construction system and operational difficulties in an				
	0	140	350 2,800	Cost (\$K) Total cost - capital plus O&M for 30 years	s at 5% interest.				
ſ	Clea	n Up Alte	ematives		KEY:				
	1	No Actio	n 2G	roundwater Monitoring/Institutional Contr	rols 🕒 = best 😷 = good				
	3	Bioventi	ng/Groundwater	⊙ = poor ) = worst					
Bioventing/Groundwater Extraction and Treatment/Groundwater Monitoring/ Institutional Controls         NA = N					NA = Not Applicable				

.

. . .

- -

~

Tabl	Table 13.7. Comparison of Cleanup Alternatives for Source Area LF03/FT09 Using the Five Balancing Criteria							
LF03/FT09 Inactive Base Landfill/Fire-Training Area LF03/FT09								
Cle	an Up	Altern	atives		Alternatives are ranked by			
1	2	3	4	BALANCING CRITERIA	comparing them to each other.			
0	0	•	٠	Long-term effectiveness and permanence Through institutional controls. Alternative 2 would prevent digging into the waste, but would not completely eliminate direct contact with any surface contam- ination or infiltration to groundwater. Both Alternatives 3 and 4 are effective in isolatiing the landfill contents, but Alternative 4 further minimizes infiltration and contaminant migration to groundwater through use of a composite cap.				
0	0	0	0	Reduction of toxicity, mobility, or volume through treatment None of the alternatives include treatment of the waste. The investigation did not identify "hot spots" of soil or groundwater contamination suitable for treatment within this large landfill.				
0	0	•	⊖	Short-term effectiveness Both Alternatives 3 and 4 would meet cleanup objectives in about the same amount of time.				
NA	Implementability Under Alternatives 2-4, institutional controls are readily implementable. Alternatives 3 and 4 include readily available technologies. Techniques for installation, maintenance, and monitoring of landfill covers are well-established.							
0	230	1,300	,500	Cost (\$K) Total cost - capital plus O&M for 30 years	at.5% interest.			
Clea	Clean Up Alternatives KEY:							
1	1     No Action     2     Groundwater Monitoring/Institutional Controls          = best							
3	3 Soil Cover/Groundwater Monitoring/Institutional Controls							
4	4 Soil Cover/Composite Cover/Groundwater Monitoring/Institutional Controls NA = Not Applicable							

.

·

# 14.0 Selected Remedies

This section presents the source areas that require no further action, followed by a discussion of the selected remedies for the source areas in OUs 3, 4 and 5 that require action. Table 14.1 summarizes the selected remedy for each source area. Table 14.2 provides a summary of the estimated costs of the selected remedies.

The goal of the Superfund approach is to return usable ground waters to their beneficial uses within a timeframe that is reasonable, given the particular circumstances of the site. Reasonable restoration time periods may range from very rapid (1 to 5 years) to relatively extended (several decades). Factors, such as location, proximity to population, anticipated future land use, and mobility of the contaminant plume, are all considered when determining an appropriate restoration timeframe. The use of: (1) natural attenuation with institutional controls, (2) source reduction through treating soil contamination to prevent additional contaminant leaching into the groundwater; and (3) groundwater pumping and treating were all considered viable options for addressing groundwater contamination at Eielson AFB. For source areas within OUs 3, 4, and 5, the following site-specific conditions were considered when determining reasonable restoration timeframes:

- Contaminant plumes in this relatively homogeneous aquifer do not appear to be spreading or are decreasing in size.
- Biodegradation, dispersion, dilution, or adsorption appear to be effectively containing or reducing the size of the contaminant plume.
- Areas impacted by the contamination are relatively small with little likelihood of extended exposure to groundwater anticipated.
- The future land use as a military installation is not anticipated to change in the foreseeable future.

For source areas at Eielson AFB, natural attenuation, in combination with institutional controls or source reduction, was selected in situations where reduction of contamination in the groundwater would be attained in a similar timeframe as active remediation of the groundwater.

## 14.1 No Further Action Sites

As indicated in Section 4.1, the source evaluation screening process found that three areas (LF01, WP32, and DP55) contained contaminants below screening levels or the affected pathway was incomplete; therefore, no further action under CERCLA is required. For the source areas within OUs 3, 4, and 5, the cumulative risk for source areas LF02, LF04, LF06, ST27, WP33, SS36, SS37, SS39/SS63, and SS64 is within acceptable regulatory levels and, therefore, environmental cleanup is not proposed for these 10 source areas under CERCLA. The Air Force will continue to monitor groundwater on the base to ensure compliance with state and federal regulations, and to ensure that groundwater use does not pose an unacceptable risk to human health or the environment.

Under a separate federal program, the Air Force has submitted a closure plan for the ordnance area at LF04 under the Resource Conservation and Recovery Act (RCRA), Section 3008(a). The landfills, LF01, LF02, LF06, and, potentially, LF04 will be addressed in accordance with state of Alaska Solid Waste Regulations under 18 AAC 60.

## 14.2 Recommended Limited Action Sites

Groundwater constituents in five of the source areas (WP45/SS57, ST56, SS61, and DP25) exceed maximum contaminant levels. These source areas are isolated, have no significant contamination or have inaccessible residual contamination in the vadose zone, and are characterized by a stable plume configuration. In the case of DP25, the plume is limited to an active tank farm. Action for these source areas is limited to continued groundwater monitoring and restrictions on the use of the groundwater.

Five of the source areas (WP45/SS57, ST56, SS61, and DP25) will receive limited action including:

- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.
- For groundwater at ST56, wellhead treatment using carbon adsorption or air stripping will be applied, as appropriate, to prevent human exposure to contaminants above regulatory levels.
- If future developments in bioventing technology make implementation practical at DP25, installation of a bioventing system will be re-evaluated at that time.
- Institutional controls to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.

Institutional controls would be used to prevent human exposure to contamination remaining at the source areas at concentrations above state or federal regulatory levels or health-based risk levels. Specific controls would include restrictions limiting access to the source areas, and administrative controls to limit groundwater use and future land use. Access restrictions, including such measures as permanent markers would be used to prevent direct human exposure to contaminants. Groundwater restrictions would be implemented by placing written notification in the base directives prohibiting the use of contaminated groundwater. In addition, all existing and any new wells located within the area of contamination shall be locked to prevent unauthorized use.

The administrative controls for limiting future land use would include placing written notification of these remedial actions in the base land use master plan. The notification shall prohibit any activity that would disrupt aspects of the engineered controls. A copy of the notification would be provided to any prospective transferees of the property and would be included in any transfer documents, including deeds, in the event that the Air Force releases control of the affected property. The Air Force shall provide EPA and the state with written verification that notification(s) have been implemented. ...-

ου	Source Area	Source Area Description	Preferred Clean-up Alternatives	
3	DP44	Battery Shop Leach Field	Soil Vapor Extraction/Groundwater Monitoring/Institutional Controls	
	WP45/SS57	Photo Lab/Fire Station Parking Lot	Groundwater Monitoring/Institutional Controls	
	ST56	Engineer Hill Spill Site	Groundwater Monitoring/Wellhead Protection or Treatment as Appropriate	
	SS61	Vehicle Maintenance Building 3213	Groundwater Monitoring/Institutional Controls	
4	DP25 E-6 Fuel Storage Tank Area		Groundwater Monitoring/Institutional Controls/Bioventing (if feasible)	
1	ST27	E-11 Fuel Storage Tank Area	No Further Clean-up Action	
	WP33	Wastewater Plant Effluent Infiltration Pond	No Further Clean-up Action	
I	SS35	Asphalt Mixing and Drum Burial Area	Soil Cover/Remove Drums in Future (if appropriate)	
	S\$36	Drum Storage Area	No Further Clean-up Action	
	SS37	Drum Storage Area	No Further Clean-up Action	
	SS39/SS63	Asphalt Lake/Asphalt Lake Spill Site	No Further Clean-up Action	
	ST58	Old Quartermaster Service Station Site	Bioventing/Groundwater Monitoring/ Institutional Controls	
	SS64	Transportation Maintenance Drum Storage Site	No Further Clean-up Action	
5	LF02	Old Base Landfill	No Further Clean-up Action	
	LF03/FT09	Inactive Base Landfill/Fire-Training Area	Soil Cover / Composite Cover / Groundwater Monitoring/Institutional Controls	
	LF04	Old Army Landfill and Ordnance Area	No Further Clean-up Action	
	LF06	Old Landfill	No Further Clean-up Action	
SER Sites	LF01	Original Base Landfill and Drum Storage Area	No Further Clean-up Action	
[	WP32	Wastewater Plant Spill Ponds	No Further Clean-up Action	
	DP55	Birch Lake Recreation Area	No Further Clean-up Action	

Table 1	4.1	Summary	of	Selected	Alternatives
---------	-----	---------	----	----------	--------------

#### 14.2.1 Source Area ST56 (OU 3)

Groundwater monitoring/institutional controls constitute the selected remedy for source area ST56. The limited action at ST56 includes supplying drinking water, applying wellhead treatment as applicable, and monitoring groundwater from the water supply well and existing monitoring wells. Subsurface treatment of the groundwater was not selected, because of the complex hydrogeology of the fractured bedrock; the limited extent of the contamination in a remote, restricted area of the base; and the reliability of available institutional controls to restrict the use of the contaminated well.

#### 14.2.2 Source Area WP45/SS57 (OU 3)

Alternative 2, Groundwater Monitoring/Institutional Controls, has been determined to be the most appropriate remedy for source area WP45/SS57. Soil and groundwater investigations previously indicated that groundwater posed a risk to human health or the environment and that residual contamination in the soil could be acting as a source of continuing groundwater contamination. However, data from a report presented by the Utah Water Research Laboratory (UWRL), collected during an independent study of WP45/SS57, indicates that soil contamination is contained around Well 45MW08 and rapid soil contaminant degradation is apparently occurring in the immediate vicinity of that well. In addition, groundwater contamination appears to have reached a steady-state concentration and distribution. The proposed alternative at WP45/SS57 was to implement SVE and bioventing in addition to groundwater monitoring and institutional controls. The selected alternative, groundwater monitoring/institutional controls, is significantly different from the proposed alternative. The reason for changing the remediation alternative for WP45 and SS57 is explained in Section 16.0.

Fire Well C is located in source area WP/57 and is currently connected to the base water supply system. Institutional controls would also apply to this well to prevent use of the contaminated groundwater in a manner that would pose an unacceptable risk to human health or the environment.

#### 14.2.3 Source Area SS61 (OU 3)

Alternative 2, Groundwater Monitoring/Institutional Controls, has been determined to be the most appropriate remedy for source area SS61.

The source of solvent contamination at SS61 was a dry well on the south side of the vehicle maintenance shop. The dry well and surrounding soils were excavated and backfilled. Soil and groundwater investigations indicate that soil above the water table does not pose a significant risk to human health or the environment. Groundwater beneath the vehicle maintenance shop is contaminated with solvent. This plume is limited to the area beneath the building and slightly north beneath the asphalt driveway in front of the building. The majority of the remaining contaminated soil is located below the water table. Therefore, removing this soil would be difficult and of limited usefulness in decreasing timeframes to meet RAOs.

Because of the limited access to the groundwater beneath the building, the action proposed is limited to preventing the use of groundwater and to monitoring the plume to ensure that it does not move from beneath the building. If the groundwater contamination is shown to be moving from beneath the building at concentrations that would pose an unacceptable risk to human health or the environment, the need for cleanup action will be reevaluated.

#### 14.2.4 Source Area DP25 (OU 4)

Alternative 2, Groundwater Monitoring/Institutional Controls, has been determined to be the most appropriate remedy for source area DP25. Currently, all contamination at DP25 is contained within the tank complex berm. The proposed alternative at DP25 is to monitor the groundwater and implement institutional controls. To comply with other state and federal programs, the secondary containment requirement of 18 Alaska Administrative Code 75, the bulk storage fuel tanks will be upgraded with impervious liners in the diked areas. The upgrade project is scheduled to be completed by the state deadline of January 1997. During the installation of the liners, approximately 0.3 m (12 in.) of fuel-contaminated soil from within the berms (21,400 m<sup>3</sup> or approximately 28,000 cu yd) will be excavated. Although the liner is being emplaced to contain possible future fuel spills, it will also serve to prevent infiltration of fuel into the soil, which would otherwise contribute to the spread of contamination at this site.

If monitoring indicates any migration of contaminants outside the tank complex berm in the future, trenches will be emplaced outside the berm to capture any migrating fuel. In addition, if future developments in bioventing technology make implementation practical at DP25, installation of a bioventing system will be re-evaluated at that time.

## 14.3 Recommended Treatment Action Sites

Five source areas in OUs 3, 4, and 5 will require active remediation. Based upon CERCLA requirements, the detailed analysis of the alternatives using the nine EPA criteria, and public comments, the U.S. Air Force, ADEC, and EPA have determined the alternatives that are the most appropriate remedies for each source area. Table 14.1 summarizes the selected remedy for each source area. These remedies were selected as a result of the comparative analysis of the alternatives against the nine EPA criteria. Additional discussion about the alternatives selected for each source area is included in the following subsections.

Cleanup alternatives will be implemented using a phased approach, where design data gathering and ongoing monitoring will continue to be evaluated to confirm the appropriateness of the selected remedy or, once a remedy is implemented, to determine the effectiveness of the technology. This phased approach will accommodate needed selected remedy or system modifications.

#### 14.3.1 Source Area DP44 (OU 3)

The selected remedy for DP44 is soil vapor extraction/groundwater monitoring/institutional controls. This alternative was chosen because of its effectiveness for treating chlorinated solvents that are found at this source area and because it is believed that SVE will reduce risk to human health and the environment sooner than monitoring and institutional controls alone. Groundwater extraction and treatment/SVE is not the preferred alternative because of its difficult implementation, and because biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination.

The primary components of the selected remedy are:

• Install a soil vapor extraction system to remove solvent contamination in soil that is posing a threat to groundwater through leaching.

- Implement institutional controls, as described, to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.
- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.

It may become apparent, during the design phase, implementation, or operation of the SVE system that solvent and fuel-related compounds in the soil and groundwater have declined or have fallen below levels that would pose an unacceptable risk. In such cases, the system performance standards or the remedy may be re-evaluated to allow for the contaminants to naturally degrade.

## 14.3.2 Source Area SS35 (OU 4)

A combination of Alternative 3 (Soil Cover) and Alternative 4 (Possible Removal of Drums) has been determined to be the most appropriate remedy for source area SS35. The placement of a clean soil cover is designed to prevent contact with pesticide-contaminated soil and to prevent runoff of contaminated soil into Garrison Slough. The buried drums will be left in place and the groundwater, surface water, sediments, and aquatic organisms monitored, as appropriate. At this time, excavation of the contaminated soil and drums is not considered cost-effective.

The cover alternative focuses on minimizing DDT migration into Garrison Slough and eliminating the surface soil exposure pathway. The cover is proposed for those areas where DDT has been detected above the risk-based levels in the surface soil. The soil cover is primarily for the purpose of limiting the migration of contaminants into Garrison Slough and to prevent direct soil contact and ingestion by the base personnel and ecological receptors.

The primary components of the selected remedy are:

- Installation of a soil cover over the surface soil contamination to prevent direct contact by humans, animals, and surface water runoff into Garrison Slough.
- Removal of drums in the future, if it is determined that they are a continuing source of .contamination.
- Monitoring of surface water, sediments, and aquatic organisms in this area, as required to verify
  effectiveness of the cover and monitoring of the groundwater to verify that levels remain below
  acceptable screening levels.

## 14.3.3 Source Area ST58 (OU 4)

Alternative 3, an in situ alternative consisting of bioventing/groundwater monitoring/institutional controls, has been determined to be the most appropriate remedy for source area ST58. This alternative will reduce the fuel source in the soils through degradation of fuel hydrocarbons, and will thus reduce the risk to human health and the environment sooner than with monitoring and institutional controls alone. At present, no proven method is known for removing lead from groundwater at a reasonable cost in a reasonable amount of time. However, a treatability test is being performed at another site at Eielson AFB to determine the fate and transport of lead and the most viable option for extraction and treatment, if warranted. Results from this test will be used to further evaluate lead remediation at ST58. Groundwater extraction/bioventing (Alternative 4) is not the preferred alternative

because of its difficult implementation, and because biodegradation, dispersion, dilution, and adsorption appear to be effectively containing and degrading the contamination.

This alternative will reduce the long-term source of contamination by preferentially encouraging the removal of contaminants from the soil through bioventing. Groundwater action will consist of natural attenuation, institutional controls, and monitoring.

The primary components of the selected remedy are:

- Installation of a bioventing system to remove fuels contamination in the soil that poses a threat to
  groundwater through leaching. This system may include air injection within the upper part of the
  groundwater table and smear zone to volatilize and promote bioremediation of the contaminants.
  The system may also include air extraction if deemed appropriate.
- Institutional controls to prevent exposure to contaminated groundwater. In the event of base closure, any remaining contaminated sites will be addressed in accordance with CERCLA Section 120.
- Monitor the groundwater to evaluate contaminant levels and identify changes to contaminant plume configuration until remediation levels are achieved.

## 14.3.4 Source Area LF03/FT09 (OU 5)

Alternative 4, Soil Cover/Composite Cap/Groundwater Monitoring/Institutional Controls, has been determined to be the most appropriate remedy for source area LF03/FT09. FT09 is considered together with LF03, because FT09 is completely contained within LF03.

This alternative was chosen because it is believed that a soil cover/composite cap will be more protective of human health and the environment than will monitoring and institutional controls alone. Groundwater monitoring will be performed at the edge of the waste management area to detect any movement of contaminants.

The cap alternative focuses on eliminating the threat of direct contact with buried landfill debris, and on soil contamination and monitoring of groundwater at the edge of the waste management area to ensure that federal and state standards are met.

The primary components of the selected remedy are:

- For the portion of the landfill where disposal occurred before 1980, RCRA Part 264 is relevant and appropriate. Currently, no groundwater at the edge of the waste management area exceeds regulatory levels; the residual contamination poses a direct contact threat. A cover to address the direct contact threat will be installed and maintained in accordance with relevant and appropriate requirements of Part 264. Groundwater at the landfill will continue to be monitored, as appropriate, to verify that contaminant concentrations, if any, remain within acceptable screening levels.
- For the portion of the landfill where disposal occurred after 1980, RCRA Part 264 is applicable. The final cover will be constructed to: (1) provide long-term minimization of migration of liquids, (2) function with minimum maintenance, (3) promote drainage and minimize erosion, (4) accommodate settling and subsidence, and (5) have a permeability less than or equal to the natural

subsoils present. Post-closure care, including maintenance and monitoring, will be conducted in accordance with 40 CFR 264.117 and 264.228(b).

• Institutional controls will be implemented to restrict land use. In the event of base closure, any remaining contamination will be addressed in accordance with CERCLA Section 120.

## 14.4 Costs of the Selected Remedies

The estimated costs of the selected remedies are presented in Table 14.2

Alternatives by Source Area	Capital Cost	Annual O&M	Present Worth"
Source Area DP44 (Battery Shop Leach Field)			
SVE Alternative SVE components Groundwater monitoring components Additional site investigation TOTAL <sup>(b)</sup>	\$1.280,000	\$65,000 (yrs 1-3) \$8,600 (yrs 1-30) 0	\$1,600,000
Source Area WP45 (Photo Laboratory Building 1	183) and SS57 (Fire Star	tion Parking Lot Spill)	-
Monitor Alternative TOTAL <sup>(*)</sup>	\$5,300	\$11,600 (yrs 1-30)	\$180,000
Source Area SS61 (Vehicle Maintenance Building	3213)		
Monitor Alternative TOTAL <sup>(*)</sup>	\$5,300	\$10,100	\$160,000
Source Area DP25 (E-6 Fuel Tank Studge Burial	Site)		
Monitor Alternative TOTAL <sup>(9)</sup>	\$5,300	\$13,100	\$210,000
Source Area SS35 (Asphait Mixing Area)		· · · · · · · · · · · · · · · · · · ·	
Cover Aliemative TOTAL <sup>(b)</sup>	\$40,000	\$0	\$40,000
Source Area ST58 (Old Quartermaster Service Sta	ition)	· · · · · · · · · · · · · · · · · · ·	
In Situ Alternative Bioventing components Groundwater monitoring components TOTAL <sup>(h)</sup>	\$170,000	\$51,000 (yr 1) \$8.600 (yrs 1-30)	\$350,000
Source Area LF03 (Current Base Landfill) and FT	109 (Fire Training Area)	·	
Cap Alternative Cover (soil) components e Cover (geosynthetic) components Drainage components Groundwater monitoring components TOTAL <sup>th</sup>	\$7,100,000	\$5,000 (yrs 1-30) \$2,500 (yrs 1-30) \$3,750 (yrs 1-30) \$14,600 (yrs 1-30)	\$7,500,000
<ul> <li>(a) The present worth cost is based on a 5 perce</li> <li>(b) TOTAL cost includes mobilization, continge</li> </ul>	int interest rate over 30 y noies, and other costs.	/etrs.	

Table 14.2. Summary of the Estimated Costs for Selected Remedies

# 15.0 Statutory Determinations

The selected remedies meet the statutory requirements of Section 121 of CERCLA, as amended by SARA, and to the extent practicable, the National Contingency Plan. The evaluation criteria for compliance with these statutory requirements are discussed in this section.

## 15.1 Protection of Human Health and the Environment

The selected remedies protect human health and the environment through removal of the sources of groundwater contamination. Institutional controls will eliminate exposure to contaminated groundwater until state and federal standards are achieved. Groundwater monitoring will track the extent of contamination above the MCL. The selected remedy will reduce risks to within the 10<sup>-4</sup> to 10<sup>-6</sup> range for carcinogens and hazard indexes will be less than one. No unacceptable short-term risks or cross-media impacts, resulting from implementation of the remedy, are present that cannot be readily controlled.

# 15.2 Attainment of Applicable or Relevant and Appropriate Requirements of Environmental Laws

CERCLA specifies that remedial actions must attain standards that are defined by EPA and ADEC as applicable or relevant and appropriate requirements (ARARs) for Eielson AFB, unless a waiver is obtained. The selection process for remedial actions may also take into account the *to be considered* (TBC) criteria, if ARARs do not address a particular situation. These criteria may include nonenforce-able criteria, advisories, or guidance issued by federal or state agencies that are not legally binding but are considered, if appropriate, in developing remedial action objectives and PRGs.

The selected remedies will comply with all substantive requirements for ARARs of federal and State of Alaska environmental and public health laws.

## 15.2.1 Applicable or Relevant and Appropriate Requirements

The remedy chosen for each set of source areas will comply with all action-, chemical-, and location-specific ARARs.

## 15.2.2 Chemical-Specific ARARs

The chemical-specific ARARs for remedial actions to be conducted at source areas in OUs 3, 4, and 5 are:

- Maximum contaminant levels (MCLs) and maximum contaminant level goals (MCLGs) established under the Safe Drinking Water Act for area groundwater that may be used as a drinking water supply (40 CFR 141 and 18 AAC 80) (see Table 15.1).
- Federal ambient water quality criteria (AWQC) established under the Clean Water Act for the protection of aquatic life in Garrison Slough and French Creek (see Table 15.1).

- Federal AWOC for the protection of human health from the consumption of fish from Garrison Slough and French Creek (see Table 15.1).
- Alaska water quality standards under 18 AAC 70 for groundwater, designated beneficial use Class (1)(A) for freshwater water supply, including the water quality standards for (1)(A)(i) drinking, culinary, and food processing; (1)(A)(ii) agriculture, including irrigation and stock watering; (1)(A)(iii) aquaculture; and (1)(A)(iv) industrial. At areas where the selected remedy is active remediation, treatment will continue until MCLs are consistantly attained. Natural attenuation is expected to meet the petroleum requirements of 18 AAC 70.
- Alaska water quality standards for fresh waters (i.e., fresh surface water), designated beneficial use Class (1)(A) for fresh water supply, Class (1)(B) for fresh water recreation, and Class (1)(C) for growth and propagation of aquatic life and wildlife. The surface water quality standards include those for (1)(A)(i) drinking, culinary, and food processing; (1)(A)(ii) agriculture, including irrigation and stock watering; (1)(A)(iii) aquaculture; and (1)(A)(iv) industrial; (1)(B)(i) contact recreation; (1)(B)(ii) secondary recreation; (1)(C) growth and propagation of fish, shellfish, other aquatic life, and wildlife (18 AAC 70.020).
- Alaska Oil Pollution Regulations (18 AAC 75)

Under the Alaska Oil Pollution Regulations, responsible parties are required to clean up oil or hazardous releases. The U.S. Air Force has proposed a calculation of soil cleanup levels, based on the findings in the baseline risk assessment (U.S. Air Force 1995b) and a methodology using the EPA SESOIL and AT123D models (Anderson 1992). The proposed soil cleanup levels are based on protecting groundwater in accordance with drinking water standards and are specified in Table 15.2.

Alaska Regulations for Leaking Underground Storage Tanks (18 AAC 78)

Under this regulation, the regional supervisor may identify alternative cleanup standards based on the potential for leaching to groundwater. In accordance with this requirement, alternative soil cleanup standards have been calculated (Table 15.2) based on the findings in the baseline risk assessment (U.S Air Force 1995b) and a methodology using the EPA SESOIL and

- --- AT123D models (Anderson 1992). The soil cleanup levels are based on protecting
- groundwater in accordance with drinking water standards.

### 15.2.3 Location-Specific ARARs

The location-specific ARARs identified in the Draft RI for OUs 3, 4, and 5 include:

- · designation of the underlying aquifer as a sole source aquifer
- flood plain restrictions
- · wetland protection under the Clean Water Act.

Remedial action at source areas within OUs 3, 4, and 5 must consider remediation of contaminated groundwater. Because the aquifer has been identified, but not designated, by the state as the sole source of drinking water supply in the Eielson AFB area, prevention of further water quality deterioration and restoration of water quality to achieve state and federal water quality and drinking water standards are primary objectives of remediation.

Remedial action at source areas within a designated flood plain must be designed to avoid adverse effects, minimize potential harm, and restore and preserve natural and beneficial values of the flood plain (40 CFR 6).

None of the source areas within OUs 3, 4, and 5 contain designated wetlands protected under the Clean Water Act (USAF 1995a).

## 15.2.4 Action-Specific ARARs

Action-specific ARARs are technology- or activity-based requirements or limitations that relate to specific remedial actions. Potential action-specific requirements are identified in Table 15.2. Compliance with action-specific ARARs is evaluated as part of the detailed evaluation of alternatives conducted in the FS process.

	Groundwater	Surface Water		
		AWQC Aquatic Life	AWQC Human Health	
	Drinking Water MCL	Freshwater Chronic	Fish Consumption	
Chemical Compound	(μg/L)	(µg/L)	(μg/L)	
Volatile Organic Compounds		<u></u>	<u></u>	
Benzene	5	5,300	40	
Toluene	000,1	17,500	424,000	
Ethylbenzene	700	32,000	3,280	
Xylenes	10,000	······		
1,4-Dichlorobenzene	75	763	2,600	
1,2-Dichloroethane	5			
cis-1,2-Dichloroethene	70	11,6004	1.85	
trans-1,2-Dichloroethene	100	11,600 <sup>d</sup>	3.2	
Trichloroethene	5	21,900 <sup>b</sup>	81 <sup>c</sup>	
Tetrachioroethylene	5	840	8.85	
Vinyl Chloride	2	······································	525	
Semivolatile Organic Compour	ıds	· · · · · · · · · · · · · · · · · · ·		
DDT		1.14	0.000024	
Chlordane	2	0.0043	0.00048	
Inorganic Compounds		······································		
Lead	154	3.2	T	
Silver	100 <sup>e</sup>	0.12		
(a) EPA action level; MCL er	cpired 12/07/92.			

Table 15.1. Chemical-Specific ARARs for Contaminants of Potential Concern

(b) Insufficient data to develop criterion. Value presented is lowest observed effect level.

(c) Criteria based on carcinogenicity (10<sup>-6</sup> risk).

(d) Freshwater acute criterion; no freshwater chronic criterion exists for this compound.

(e) Secondary MCL.

Because some of the contaminants identified in soils and groundwater within OUs 3, 4, and 5 can be classified as RCRA hazardous wastes. RCRA requirements may be considered action-specific ARARs for those source areas where placement or disposal occurs.

## 15.2.5 Criteria To Be Considered for Protectiveness

In addition to ARARs, federal and state criteria, policy, and guidance have been considered in defining the appropriate level of protectiveness. The TBC criteria identified for source areas within OUs 3, 4, and 5 are discussed in the following subsections.

Several methods, including the MEPAS model, were evaluated for determining soil cleanup levels. Because of the limitations of using the MEPAS model for source areas with floating product or soil contamination that occurs primarily in the smear zone, it was decided that a simpler approach would probably provide more reliable results.

Action	Requirement	Applicability	Reference
Capping Closure with waste in place	Placement of a cap over waste requires a cover designed and constructed to: (1) provide long- term minimization of liquid migration through the cap; (2) function with minimum maintenance; (3) promote drainage and minimize erosion of the cover; (4) accommodate settling so that the integrity of the cover is maintained; (5) have a permeability less than or equal to the	RCRA hazardous waste placed at the site after November 1980 or placement of RCRA waste into another unit. For wastes placed before 1980; RCRA Part 264 is relevant and appropriate.	40CFR264.258(b) for waste piles 40CFR264.310(a) for landfills
- - -	permeability of the bottom liner. Eliminate free liquids, stabilize remaining waste 30-yr post-closure care and monitoring	Applicable to land disposal of RCRA hazartious waste Applicable to land disposal of RCRA hazardous waste	40CFR264.228(a)(2) 40CFR264.310
	Restrict post-closure use of property to prevent damage to the cover	RCRA hazardous waste	40CFR264.117(c)
	cover Protect and maintain surveyed benchmarks used to locate waste cells	RCRA hazardous waste	40CFR264.310(b) 40CFR264.310(b)
Discharge of treatment system effluent	Must comply with ambient water quality criteria and federally approved State water quality standards as appropriate	Point source discharge	40CFR122.44 18 AAC 70 18 AAC 80 18 AAC 72
Land disposal	Attain land disposal treatment standards before placing waste in a land disposal unit to comply with land ban restrictions	RCRA hazardous waste	40CFR268(Subpart D)
Treatment	Must comply with design and operating standards for treatment unit, that is, waste piles, land treatment, water treatment	RCRA hazardous waste	40CFR264.251 40CFR264.273 40CFR264.601
Vapor extraction	Total emissions are restricted under the base permit and the National Emission Standards for Hazardous Air Pollutants.	Air emission control equipment may be required because of the vapors extracted from the fuel layer.	18 AAC 50 40CFR61.93
Solid Waste	Includes requirements for disposal of treated soils that are solid waste.	Solid Waste Disposal	18 AAC 60
Source: CERCLA Complian	nce with Other Laws Manual, OSWER Directive	9234.1-01	

#### Table 15.2. Potential Action-Specific ARARs

The generic leachate pathway soil cleanup levels were developed by the Oregon Department of Environmental Quality using the EPA SESOIL and AT123D models. The cleanup levels are designed to prevent contaminant levels in groundwater from exceeding a health-based safe drinking water level through the leachate pathway. It was decided that although these numbers are based on higher precipitation values than occur in the Fairbanks region, they would provide an adequate margin of safety for cleanup of soils at Eielson. The methodology is described in Appendix A of the FS. The calculated soil concentrations have been included in Table 15.3. However, it is expected that these levels will be refined, if other levels are found to be protective of groundwater and as more site-specific and better fate and transport data become available.

"RCRA ARARs: Focus On Closure Requirements." OSWER Directive 9234.2-04FS, October 1989, is a TBC when RCRA Subtitle C Part 264 is and ARAR.

Chemical Compound	Alternative Soil Cleanup Level (mg/kg)
Trichloroethylene	0.4
Benzene	0.1
Toluene	79
Ethylbenzene	140
Xylenes	760
Sources: Anderson (1992), EPA (1992)	

Table 15.3. Al	Iternative Soil	Cleanup Levels	Based on	Leaching
----------------	-----------------	----------------	----------	----------

## 15.3 Cost-Effectiveness

The selected remedies are considered cost-effective for remediation of the contaminated soils and groundwater, because they have been determined to provide overall effectiveness proportionate to their costs and duration.

## **15.4 Use of Permanent Solutions and Alternative Treatment** Technologies to the Maximum Extent Practicable

The selected remedies protect human health and the environment, comply with federal and state requirements that are legally applicable or relevant and appropriate to the remedial actions, and are cost effective. The remedies use permanent solutions and alternative treatment (or resource recovery) technologies to the maximum extent practicable and satisfy the statutory preference for remedies that employ treatment that reduces toxicity, mobility, or volume as a principal element.

## 15.5 Preference for Treatment as a Principal Element

For source areas DP44 and ST58, the selected remedy includes treatment to address the soil contamination which is the principal threat posed by conditions at the site.

For source areas WP45/SS57, ST56, SS61, DP25, SS35, and LF03/FT09, the selected remedy does not include treatment because the contamination does not meet the definition of a principal threat, as defined in the preamble to the NCP and the contamination can be reliably controlled in place.

## 16.0 Explanation of Significant Differences

CERCLA Section 117(b) requires an explanation of any significant changes from the preferred alternatives originally presented in the proposed plan. Based on recent sampling results, the selected remedy for source areas WP45/SS57 is different than that presented in the proposed plan. The new information indicates that another alternative from the proposed plan provides the best balance of tradeoffs among the alternatives with respect to the nine evaluation criteria.

## 16.1 Proposed Alternative

The proposed alternative for WP45/SS57 presented in the Proposed Plan for Operable Units 3, 4, 5, and Other Areas (May 1995) was Alternative #3: soil vapor extraction (SVE)/groundwater monitoring and institutional controls for WP45, and bioventing/groundwater monitoring and institutional controls for SS57. SVE was chosen for WP45 because it is an effective method of remediation for solvents in unsaturated soils. It was believed that SVE would reduce the risk to human health and the environment sooner than with monitoring and institutional controls alone. Bioventing was the preferred alternative for SS57, because it may be an effective method for treating the fuel-related contaminants in the smear zone, where most of the contamination had been found. These proposed alternatives were based primarily on information presented in the OU 3, 4, and 5 Remedial Investigation, the Baseline Risk Assessment, and the Feasibility Study (U.S. Air Force 1995a, 1995b, and 1995c).

## 16.2 Significant Changes

The selected remedy for both sites WP45 and SS57 has been changed to Alternative #2: groundwater monitoring and institutional controls.

### 16.3 Reason for Change

An independent study of natural attenuation by Utah Water Research Laboratory (UWRL), Utah State University, has been conducted concurrently with the remedial investigation at WP45/SS57. A meeting was held 6 July 1995, during which UWRL presented their findings and modeling of site data collected at WP45/SS57. The soil and groundwater contamination exists at this site in the form of lowlevel sorbed species and dissolved contaminant mass. Currently, no evidence of residual dense nonaqueous phase liquid (DNAPL) is present within the source area at the site; it also does not appear that any residual fuel material exists in the form of light nonaqueous phase liquid (LNAPL). The contamination is adsorbed and contained or in a dissolved phase and not accessible for source removal or treatment. UWRL focused its study on evaluating the current extent of the dissolved TCE plume, investigating evidence of TCE degradation existing throughout the site in the form of anaerobic dechlorination intermediate products, and evaluating the likelibood of biological mediated reactions based on mass balance estimates and known stoichiometric relationships for these anaerobic transformation processes. UWRL field data collection confirmed earlier findings reported by PNL. Low levels of soil contamination (<1 ppm TCE in all samples), an apparently contained groundwater plume (particularly benzene), no free product, and no vinyl chloride detected through DCE was found to be present. New findings include further evidence of TCE anaerobic dechlorination with ethylene and large distribution of DCE product, significantly lower BTEX than previously reported, and rapid transportation of contaminants in the immediate vicinity of monitoring well 45MW08 (suspected source area).

The relative rate and extent of contaminant migration was evaluated through the use of a conventional 3-dimensional advective/dispersive groundwater model that incorporates groundwater flow, contaminant sorption, and contaminant degradation to describe the downgradient movement within the shallow aquifer over time. Model parameters that were not available or measured at the site were estimated using representative literature values.

The results of the UWRL study suggest that groundwater movement from this site is relatively slow (approximately 18 m/yr pore water velocity, with approximately 6 m/yr retarded TCE groundwater velocity based on measured field data). Additionally, with approximately 9 kg of TCE mass apparently lost in the aquifer over a 2-year monitoring period, it appears that TCE degradation is occurring at a first order degradation rate of approximately 0.00027/d (0.027%/d), yielding a TCE half life of approximately 7 years. With these values of contaminant velocity, apparent degradation rate, and an estimated source configuration based on model calibration, the remaining source of TCE contamination is predicted to be exhausted in another 7 years, with the subsequent groundwater plume generated from this source being attenuated within the aquifer to below regulatory limits of 5  $\mu g/L$  within 70 years, and approximately 500 m of the source (Figures 16.1 and 16.2). Comparatively, it was estimated that by implementing SVE, coupled with bioventing, groundwater monitoring, and institutional controls, soils at WP45/SS57 would attain ARARs within 1 to 4 years, and groundwater would take more than 100 years to attain MCLs (U.S. Air Force 1995c). The data presented by UWRL indicates that active remediation of source areas WP45/SS57 will not significantly increase the rate of contaminant degradation from that now occurring naturally.

UWRL also evaluated a source removal scenario. Figures 16.3 and 16.4 graphically show the response of the dissolved TCE groundwater plume over time to complete source removal at source area WP45/SS57. At  $\Delta t$  - 60 years (that is, 60 years after source removal) the TCE plume centerline concentration is still predicted to be above its current MCL 5  $\mu$ g/L. TCE centerline concentrations would drop below the MCL for  $\Delta t$  - 70 years. At  $\Delta t$  - 70 years, the maximum centerline concentration was predicted to be 3  $\mu$ g/L at a distance of 430 to 450 m downgradient from the source. These results, along with the estimate of remaining source lifetime, suggest that source removal alone would be expected to reduce the lifetime of the TCE plume by only approximately 7 years, or 10%, indicating that source removal would not be an effective approach for expediting remediation at source areas WP45/SS57.

In summary, the findings presented by UWRL support several conclusions at WP45/SS57. It is apparent the BTEX plume is contained (see Figures 16.5 and 16.6). The TCE groundwater plume has reached steady-state conditions, and plume containment and TCE degradation are observed (see Figures 16.7 through 16.10). A limited extent of containment distribution was found near the source areas. SVE and bioventing would have limited effectiveness in comparison with the rate of degradation achieved by natural processes.

FINAL

16.3



Figure 16.1. Simulated TCE plume centerline concentrations using input parameter values minimizing the mean square error (MSE).

Source area Y dimension is 70 m and total simulation time is 40 years. Observed TCE centerline concentration data collected from source areas WP45/SS57 by PNL in August to September 1992 and UWRL on May to September 1994 are also included for comparison purposes.



**Figure 16.2.** Simulated TCE plume centerline concentrations using input parameter values minimizing the mean square error (MSE), but with no degradation ( $\lambda = 0/day$ ).

Source area Y dimension is 70 m and total simulation time is 40 years. Observed TCE centerline concentration data collected from source areas WP45/SS57 by PNL in August to September 1992 and UWRL on May to September 1994 are also included for comparison purposes.

T

FINAL



Figure 16.3. Simulated TCE plume centerline concentrations using input parameter values utilizing the calibrated model for source areas WP45/SS57, with the source removal scenario at T = 0 years.

Simulation times from T = 1 to 70 years. Source area Y dimension is 70 m by 40 m.



Figure 16.4. Simulated TCE plume centerline concentrations using input parameter values utilizing the calibrated model for source areas WP45/SS57, with the source removal scenario at T = 0 years.

Simulation times from T = 40 to 70 years. Source area Y dimension is 70 m by 40 m.

FINAL



ì.

16.7







EIELSON AFB SITE 45/57

**BTEX Contours - Sept. 1994** 

Max=393 ppb, Int.=50 ppb

Page 2 of 2

SP 23

SP 24 SP 26

SP 25

1209

324

SP 20

m

T/W NO.

SP 19

16

SP 18

1206







Eielson AFB



Fielson AFB

OUs 3, 4, and 5 Record of Decision



September 1995

16.10

FINAL





ž

Division

SP 7

SP6

Figure 16.9. cDCE Contours, Eielson AFB Source Areas WP45/SS57, September 1994.

Broadway

Wabash

SP27

TP 3

SP 28

100 meters

SP 2

3130

45MW06

TP 22 .

0Us 3,

4

and 5 Record of Decision

.

12:09



FINAL

## 17.0 References

18 AAC 70. 1995. "Alaska Water Quality Standards." Alaska Administrative Code.

18 AAC 75. "Alaska Oil Pollution Regulations." Alaska Administrative Code.

18 AAC 78. "Underground Storage Tank Regulations." Alaska Administrative Code.

40 CFR 6. "Procedures for Implementing the Requirements of the Council on Environmental Quality on the National Environmental Policy Act." U.S. Code of Federal Regulations.

Anderson, M.R. 1992. "Development of Generic Soil Cleanup Levels Based on Analysis of the Leachate Pathway," Report to the Oregon Commission of Environmental Quality, May, 1992.

Bennett, D.A. 1990. "Status of the Uptake Integrated Biokinetic (UBK) Model for Lead." Memorandum. Office of Solid Waste and Emergency Response, U.S. Environmental Protection Agency, Washington, D.C. (August)

CH2M Hill. 1982. Installation Restoration Program Records Search for Elelson Air Force Base, Alaska, Prepared for Air Force Engineering and Services Center, Directorate of Environmental Planning, Tyndall Air Force Base, Florida, and Alaskan Air Command, Elmendorf Air Force Base, Alaska.

Clay, D.R. 1991. "Update on OWSER Soil Lead Cleanup Guidance." Memorandum to Regional Offices. Office of Solid Waste and Emergency Response. U.S. Environmental Protection Agency, Washington, D.C. (August 29)

Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) of 1980, 42 USC 9601 et seq., as amended.

Dames & Moore. 1985. Installation Restoration Program Phase II - Confirmation/Quantification. Stage 1. First Draft Report for Eielson Air Force Base, Alaska, Alaskan Air Command. Prepared by Dames & Moore for the U.S. Air Force Occupational and Environmental Health Laboratory, Brooks Air Force Base, Texas.

Dames & Moore. 1986. Installation Restoration Program Phase II - Confirmation/Quantification. Stage 2, Technical Operations Plan. Administrative Record Doc. 1362, Eielson Air Force Base, Fairbanks, Alaska.

Gilbert, R.O. 1987. Statistical Methods for Environmental Pollution Monitoring. Van Norstrand Reinhold, New York.

Harding Lawson Associates (HLA). 1989. Installation Restoration Program Remedial Investigation/ Feasibility Study, Stage 3, #2, Eielson Air Force Base, Fairbanks, Alaska. Volume III, Draft Remedial Investigation/Feasibility Study (July 1988 - April 1989), Chapter IV. Ad. Record 1390, Prepared by Harding Lawson Associates for the Alaskan Air Command, Elmendorf Air Force Base, Alaska. Harding Lawson Associates (HLA). 1990. Installation Restoration Program Remedial Investigation/ Feasibility Study, Stage 4. Draft Report for Eielson AFB, Alaska. Volume I through V. Prepared by Harding Lawson Associates for the Alaska Air Command, Elmendorf Air Force Base, Alaska.

Harding Lawson Associates (HLA). 1991. Installation Restoration Program Remedial Investigation/ Feasibility Study, Stage 4. Draft Report for Eielson AFB, Alaska. Volume VII to XVII. Prepared by Harding Lawson Associates for the Alaska Air Command, Elmendorf Air Force Base, Alaska.

Hazardous Materials Technical Center (HMTC). 1986. Statement of Work for Phase IV-A Remedial Action Plan, Installation Restoration Program, Eielson Air Force Base, Alaska. Prepared by Hazardous Materials Technical Center for the Air Force Engineering Service Center, Tyndall Air Force Base, Florida.

Liikala, T. L., and J. C. Evans. 1995. Field Investigation, Source Area ST58, Old Quartermaster Service Station, Eielson Air Force Base, Alaska. PNL-10358, Pacific Northwest Laboratory, Richland, Washington.

Nerney, S. R., T. P. Ballestero, L. K. Brannaka, B. Koenen, and I. K. Iskandar. 1994. Draft Final Report on Microwell Investigations of Operable Units 3, 4, & 5 at Eielson Air Force Base Alaska. Prepared by the University of New Hampshire and the U.S. Army Corps of Engineers, Cold Regions Research and Engineering Laboratory. Submitted to Eielson Air Force Base, Fairbanks, Alaska.

Science Applications International Corporation (SAIC). 1988. U.S. Air Force Installation Restoration Program Remedial Investigation of Site 25, The Weathered Sludge Tank Burial Site, at Eielson Air Force Base, Alaska. Science Applications International Corporation, San Diego, California.

Science Applications International Corporation (SAIC). 1989a. U.S. Air Force Installation Restoration Program Remedial Investigation/Feasibility Study of the Fuel Saturated Area at Eielson Air Force Base, Alaska. Decision Document for Selected Sites. Prepared by Science Applications International Corporation for the Alaskan Air Command, Elmendorf Air Force Base, Alaska.

Science Applications International Corporation (SAIC). 1989b. U.S. Air Force Installation Restoration Program Remedial Investigation/Feasibility Study of the Fuel Saturated Area at Eielson Air Force-Base, Alaska. Draft Remedial Investigation Report. Prepared by Science Applications International Corporation for the Alaskan Air Command, Elmendorf Air Force Base, Alaska.

Shannon & Wilson. 1991. Geotechnical Investigation and Foundation Study Vehicle Maintenance Facility Air National Guard Eielson Air Force Base, Alaska. Shannon & Wilson, Inc., Fairbanks, Alaska.

Shannon & Wilson. 1992. Final Report Alaska National Guard UST Removal Building 6128, Eielson Air Force Base, Alaska. Shannon & Wilson, Inc., Fairbanks, Alaska.

U.S. Air Force. 1992. Remedial Investigation/Feasibility Study. Operable Units 3, 4, and 5 Management Plan. Eielson Air Force Base, Alaska. Final Report. Prepared by Battelle Environmental Management Operations for U.S. Air Force Environmental Restoration Program. U.S. Air Force. 1994a. Water Table Elevations at Eielson Air Force Base From September 1991 to September 1993. Eielson Air Force Base, Alaska.

U.S. Air Force. 1994b. Source Evaluation Report Phase 2 Investigation Limited Field Investigation Final Report. Eielson Air Force Base, Alaska.

U.S. Air Force. 1994c. Work Plan 1994 SER Sites and Operable Units 3, 4, and 5. Draft Final, Eielson Air Force Base, Alaska.

U.S. Air Force. 1994d. Sitewide Groundwater Monitoring Program 1993 Report (Draft), Eielson Air Force Base, Alaska. Eielson Air Force Base, Alaska.

U.S. Air Force. 1994e. Environmental Restoration Program, Eielson Air Force Base, Alaska. Volume. 2: Operable Unit I Baseline Risk Assessment Report. (Final).

U.S. Air Force. 1995a. Environmental Restoration Program, Operable Units 3, 4, and 5 Remedial Investigation Report (Final), Eielson Air Force Base, Alaska.

U.S. Air Force. 1995b. Environmental Restoration Program, Operable Units 3, 4, and 5 Baseline Risk Assessment (Final), Eielson Air Force Base, Alaska.

U.S. Air Force. 1995c. Environmental Restoration Program, Operable Units 3, 4, and 5 Feasibility Study (Final), Eielson Air Force Base, Alaska.

U.S. Air Force 1995d. Proposed Plan for Operable Units 3, 4, 5 and Other Areas of Eielson AFB, Eielson Air Force Base, Alaska.

U.S. Army Corps of Engineers (COE). 1991. Trip Transport and Chemical Data Report, Upgrade of the Sewage Treatment Plant. Eielson Air Force Base, Alaska.

U.S. Environmental Protection Agency (EPA). 1988. Guidance Manual for the Integrated Exposure Uptake Biokinetic Model for Lead in Children. EPA/540/R-93/081, PB93-963510, Office of Solid Waste and Emergency Response, Environmental Protection Agency, Washington. D.C.

U.S. Environmental Protection Agency (EPA). 1989a. Risk Assessment Guidance for Superfund (RAGS), Volume 1: Human Health Evaluation Manual. Office of Solid Waste and Emergency Response, U.S. Environmental Protection Agency, Washington, D.C.

U.S. Environmental Protection Agency (EPA). 1989b. Exposure Factors Handbook. EPA/600/8-89/043, Exposure Assessment Group, Office of Health and Environmental Assessment, U.S. Environmental Protection Agency, Washington, D.C.

U.S. Environmental Protection Agency (EPA), Alaska Department of Environmental Conservation (ADEC), and U.S. Air Force. 1990. Federal Facility Agreement Under CERCLA Section 120.

U.S. Environmental Protection Agency (EPA). 1991a. Human Health Evaluation Manual, Supplemental Guidance: "Standard Default Exposure Factors." OSWER Directive 9285.6-03, Office of Solid Waste and Emergency Response, U.S. Environmental Protection Agency, Washington, D.C. U.S. Environmental Protection Agency (EPA). 1991b. Supplemental Guidance for Superfund Risk Assessments in Region 10. U.S. Environmental Protection Agency Region X. Seattle, Washington.

U.S. Environmental Protection Agency (EPA). 1991c. "Update on OSWER Soil Lead Cleanup Guidance." Memorandum from D. R. Clay, Assistant Administrator, Office of Solid Waste and Emergency Response to Regions, U.S. Environmental Protection Agency, Washington. D.C., August 29, 1991.

U.S. Environmental Protection Agency (EPA). 1992. New Interim Region IV Guidance. Communication with Carol Sweeney (Region X), U.S. Environmental Protection Agency, Seattle, Washington, D.C.

U.S. Environmental Protection Agency (EPA). 1994. Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA, Interim Final, EPA/540/G-89/004, Office of Solid Waste and Emergency Response, Environmental Protection Agency, Washington. D.C.

U.S. Environmental Protection Agency (EPA). 1994a (March 8). Uptake/Biokinetic Model for Lead, [CD-ROM], Available from Micromedex, Inc., Denver, Colorado.

# Eielson Air Force Base Operable Units 3, 4, 5, and Other Areas Record of Decision

## **Responsiveness Summary**

## A. OVERVIEW

The proposed cleanup alternatives considered by the U.S. Air Force, Alaska Department of Environmental Conservation (ADEC), and U.S. Environmental Protection Agency (EPA) were presented to the public in a proposed plan (U.S. Air Force 1995d) and discussed in a public meeting on 31 May 1995. This plan proposed the preferred alternative to address contamination in the soil and groundwater at Operable Units 3, 4, 5, and other areas. The preferred alternative restricts groundwater use in the contaminated areas through institutional controls. The controls will remain in effect as long as the contamination persists.

Generally, public comments supported the plan as the best compromise among the clean up options.

These sections follow:

- Background on Community Development
- Summary of the Comments Received During the Public Comment Period and USAF Responses
  - Part I: Summary and Response to Local Community Concerns
  - Part II: Response to Specific Technical and Legal Questions
- Remaining Concerns

## **B. BACKGROUND ON COMMUNITY INVOLVEMENT**

Prior to the addition of Eielson AFB to the EPA National Priority List in 1989, the community was offered little opportunity for involvement in environmental activity. From 1982 until 1989, the USAF used the Installation Restoration Program (IRP) to identify potential contaminated areas and investigate what remedial actions might be required. This process was purely technical and did not evaluate community concerns in the decision-making process. However, after signing a Federal Facility Agreement with the State of Alaska and the EPA in 1991, the Air Force began its Superfund clean up program, which does include extensive community involvement.

A technical Review Committee (TRC), established in 1992, included three representatives from the community (selected by local officials and the University of Alaska Chancellor), industry representatives, and environmental representatives. Many of the TRC participants are members of the professional public. The TRC was converted to a Restoration Advisory Board (RAB) in the Spring of 1995. Three community representatives were selected as RAB co-chairpersons, one each from the communities of Salcha, Moose Creek, and North Pole, Alaska. The RAB met on April 27, 1995 to review OU 3, 4, and 5 information and again on 8 June 1995 during the comment period.

The proposed plan and the public meeting for OU 3. 4. and 5 were advertised twice in each of two local newspapers. In addition, more than 3500 copies of this notice were added as an insert in the base newspaper and delivered to every home in the Eielson AFB housing area on May 19, 1995. Proposed plans were mailed to more than 150 people on the clean up mailing list on May 16, 1995. In addition, copies of the plan were delivered to various information repositories, the Alaska Department of Fish and Game, North Pole City Hall, Moose Creek Fire Department, and several local stores and businesses. Flyers were placed on store bulletin boards in the Moose Creek and North Pole communities.

The Administrative Record is available for public review at:

• Elmer E. Rasmuson Library (Archives Section) Alaska and Polar Regions Department University of Alaska Fairbanks Fairbanks, Alaska 99775 (907)-474-6594

The Index of Administrative Record Documents only, is available at:

- Eielson Air Force Base Library
   3340 Central Ave., Suite 1 Eielson Air Force Base, Alaska 99720-2150 (907)-377-3174
- North Pole Library 601 Snowman Lane North Pole, Alaska 99705 (907)-488-6101

Information is also available at the Information Repositories at:

- Environmental Management Office 354 CES/CEVR
   2258 Central Ave., Suite I
   Eielson Air Force Base, Alaska 99702-2225
   (907)<sub>2</sub>377-1164, Mike Raabe
- Noel Wien Library 1215 Cowles Street Fairbanks, Alaska 99701 (907)-459-1020

# C. SUMMARY OF COMMENTS RECEIVED DURING THE PUBLIC COMMENT PERIOD AND USAF RESPONSES

The public comment period on the Operable Unit 3, 4, and 5 Proposed Plan extended from May 18 through June 17, 1995. Comments received during that period are summarized in Parts I and II. Part I addresses nontechnical concerns; Part II responds to technical and legal questions. Each part is grouped by similar topics.

### PART I - Summary and Response to Local Community Concerns

#### - Topic: Environmental Questions

- Public Comment: One person wanted to know how to get information on clean up work at Eielson, and another asked to be added to the mailing list. A third caller asked for information about source area LF04.

- USAF Response: The Air Force provided fact sheets on work opportunities and how to be considered for this clean up work at Eielson. The mailing list was updated to include the names of those people interested in the environmental clean up work at Eielson Air Force Base. A package of information from the Administrative Record on LF04 was mailed to the interested person.

### PART II - Response to Specific Technical and Legal Questions

#### - Topic: Alternatives Selection

- Public Comment: One person supported the cold mix asphalt process that was used to resurface roads on base with materials recovered from source area SS39 in OU 4. The person commented this procedure should be considered for any diesel-contaminated soil, or any other application where a waste material can be treated and recycled into a useful product, instead of being thrown away.

- USAF Response: The Eielson clean up team appreciates this technology, as demonstrated by having - already used cold mix asphalt paving in local projects. The team will continue to consider this method for future situations where this technology could feasibly be applied. The selected remedies for source areas in OUs 3, 4, and 5 do not include excavation and disposal for any petroleum contaminated soil; therefore, this technology to recycle contaminated soil is not applicable for these areas.

## D. REMAINING CONCERNS

- Topic: Transporting Contaminated Soils Through Moose Creek

- Public Comment: One person was concerned that dust from contaminated soil and heavy traffic to transport the soil could create a risk for residents in Moose Creek. The soil is being transported to an incinerator in Moose Creek. The caller said residents are worried about the potential for an accident due to the poor condition of the narrow roadway between the base and the treatment facility. The caller requested the trucks be diverted to a back road that leads directly from the base to the incinerator and that avoids populated areas.

- USAF Response: The comment was routed through the RAB military co-chairman to the Support Group commander. The contractor was advised of the safety concerns expressed by the area residents. The trucks were directed to drive on the new Richardson Highway, this eliminating excessive traffic in the Moose Creek community. The issue was also addressed at the RAB meeting in North Pole on 8 June 1995.

Attachment A: Community Relations Activities at Eielson Air Force Base, Alaska

## COMMUNITY RELATIONS ACTIVITIES at Eielson Air Force Base, Alaska

- 1982 Eielson conducts records search and interviews to identify environmental problem areas under the Air Force Installation Restoration Program.
  1983-1989 Eielson AFB investigations identify contamination.
  Nov. 1989 Eielson AFB listed on EPA National Priority List for priority cleanup.
  May 1991 Eielson AFB signs Federal Facility Agreement with EPA and ADEC.
  Oct. 1991 Eielson AFB holds first public meeting to announce Superfund cleanup.
- Oct. 1991 Public Relations Plan released.
- Jan. 1992 Administrative Record established at University of Alaska Fairbanks library.
- May 1992 Technical Review Committee established, including three community representatives from North Pole, Fairbanks, and University of Alaska Fairbanks.
- Jun. 1992 Public meeting on Operable Unit 1B proposed plan.
- Dec. 1992 Public meeting on Record of Decision for OU-1B (signed in Sep. 1992).
- 1992-1993 Interviews with 40 community members to update Community Relations Plan.
- Jan. 1993 International Bioventing Symposium held at Eielson AFB to assess innovative technology.
- Sep. 1993 Video documentary on base environmental program released; aired on base TV.
- Nov. 1993 Public meeting on OU-2 Proposed Plan and SER Phase 1 recommendations.
- Apr. 1994 Public meeting on OU-6 Proposed Plan and Removal Actions for three sources.
- Jun. 1994 Public meeting on OU-1 Proposed Plan and Removal Actions for three sources.
- Apr. 1995 Restoration Advisory Board established to include community co-chairs.
- May 1995 Public meeting on OU-3,4,5 Proposed Plan and other areas.

In October 1991 at the first environmental clean up public meeting, Eielson AFB released its Community Relations Plan. In subsequent public meetings from 1992 to 1995, Eielson AFB presented the Proposed Plans for Operable Unit 1 and 1B; Operable Unit 2; Operable Units 3, 4, and 5; and Operable Unit 6, and discussed the Source Evaluation Report areas.

From 1992 through 1993, surveys and interviews of more than 100 community residents were used to update the Community Relations Plan. Eielson AFB prepared fact sheets on such topics as water quality, Technical Assistance Grants, Information Repositories, clean up technologies, and work opportunities to keep the public advised on clean up activity. These publications are available at the information repositories, or from the Eielson AFB community relations point of contact.
Appendix A

	A	8	C	0	[ε_	F	G	н	1	J	K
								:			· · · · · · · · · · · · · · · · · · ·
11	Source Area	Media	Date	Analyte	D٤	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
12	DP25	Water	1992	0.0 0-Triethyl phosphorothioate	10	hð\r	1	1-			
13	DP25	Water	1992	0.0-Dielhyl 0-2-pyrazinyl phosphorothica	10	µg/L	·	-	i <b></b>	-	! 
<u></u>	UP25	Water	1992	1 1 1 Tuchlorgethage	<u>vs</u>	HOLL	13	<b>i</b> − 1	••••••••••••••••••••••••••••••••••••••		
<u> </u>	0925	VValst Mater	1993	1.1.2.Tuenioroethane	05	un/	<u>در</u>	-		! <del>-</del>	r
+	0025	Water	1993	1 1 2-Trichloroeithane	5	- uo/l	13				
1 to 1	DP25	Water	1992	1.1-Dichloroethane		uo/L	9	-  -	E	-	
<u> </u>	0P25	Water	1993	1.1-Dichloroethane	5	ua/L	13	i-	<u> </u>	·····	
10	DP25	Water	1992	1.2.4-Trichlorobenzene	10	ug/L	1		, 		
11	DP25	Waler	1992	1,2-Dichloropenzene	10	µg/L	1			]_	
12	DP25	Water	1992	1.2-Dichloroethane	0.5	µg/L	9	-	-	[_	
13	DP25	.Water	1993	1.2-Dichloroethane	5	µg/L	13	1-	_	-	
14	DP25	Water	1993	1.2-Dichloroethene (total)	1.2	μ <b>g</b> /L	13	-	<del>.</del>	I_	
15	DP25	Water	1992	1.3-Dichlorobenzene	10	µg/L	1	-	-	-	
16	OP25	Water	1992	1.4-Dichlorobenzene	2	µg/L	10		-	-	
17	DP25	Water	1993	1.4-Dichlorobenzene		hð\r	13	-	-	-	
16	DP25	Water	1992	1 4-Naphthoquinone	10	µg/L	1	-	-	-	
19	DP25	Water	1993	1-Butanoi	N/A	hđyr	13	-		-	
20	OP25	Water	1992	1-Naphihylamine	10	ugn_	<u>1</u>	-	<u> </u>	-	
21	DP25	Water	1992	Z.3.4.6-1 etrachtorophenol	10	ug/L	1	-	·		
42	0015	vyater ·	1992		10	P9/L	1				
24	0024	Výdisi	1000.	2.4.0rthlomobeooi		ug/L	1		-		
1	DP25	Water	1992	2 4-Demethylisbertol	10	un/i	1		-	2	
74	OP25	Water	1992	2.4-Dindroohenol	50	Lan	1	_			
27	DP25	Water	1992	2 4-Dinitrotoluane	10		1	-	-		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
28	DP25	Water	1992	2.6-Dichlorophenoi	10	JUGAL )	1	-			
29	DP25	Water	1992	2.6-Dinkrotokuene	10	µg/L	1	-	<b>_</b>	-	
30	OP25	Water	1992	2-Acetylaminofluorene	10	HQ/L	1	-	-	-	
_31	DP25	Water	1993	2-Butanone	100	µg/L	13	-	-	-	
32	DP25	Water	1992	2-Chloronaphthalene	10	µ9/L	1		-	-	
33	DP25	Water	1992	2-Chiorophenol	10	µg/L	1	-	-	-	
34	OP25	Water	1992	2-Methylnaphthalene	10	ugi.	1	-	-	-	
35	DP25	Water	1992	2-Methylphenoi	10	μα/L	1	-	-	-	
36	DP25	Water	1992	2-Naphthylamine	10	PØL.	1	-	-		
37	DP25	vvater	1992	2-Naroaninoe	50	MAL 1					
38	DP25 1	vyater	1992	2-Nill Ophenol	20	HOLE I					
	DP25	Water	1992	3.3-Dimelhyber2title	10				_		
	DP25	Water	1997	3-Methylcholaothcene	10	ud/L	1				
42	DP25	Water	1992	3-Nitroanitine	50	uq/L i	1	_			
43	DP25	Water	1992	4.6-Dinitro-2-methylphenol	50	µ0/L	1	-	-	-	······
44	DP25	Water	1992	4-Aminobiphenyl	10	µ9/L	1	-		-	
45	DP25	Water	1992	4-Bromophenyiphenyi ather	10	µg/L	1	-	÷	-	
46	DP25	Water	1992	4-Chtoro-3-methylphenol	20	µg/L	1	-	-	-	
47	OP25	Water	1992	4-Chloroaniine	20	HQ/L	1	-	-	-	
48	DP25	Water	19921	4-Chlorophenylphenyl ether	, 10	HOL !	1	-	-	-	
49	DP25	Water	1993	4-Methyl-2-pentanone	50	HQ/L	13	-	-	-	
<b>1</b>	UP25	Water	1992		TQ	HOL	1	-	<b>_</b>	-	
51	0825	vvaler :	1992		- 30	1491	1	<u> </u>			
	0F23 NØ25	Water	1992	- Nitrou stating 1-avide			1	<u> </u>	-	-	
	DP25	Water	1992	S-Nitro-o-tokuidine	10	uan (	1		-		
- 55	DP25	Water	1992	7.12-Dimethybenz(alanthracene	10	uo/L	1	_	-	_	
56	DP25	Water	1992	Acenaphthene	10	µg/L	1	-	-	_	
57	DP25	Water	1992	Acenaphthylene	10	μg/L	1	-		- 1	
- 68	DP25	Water	1993	Acetone	10	hô/L	13	3	5.1	15	25402
59	DP25	Water	1992	Acetophenone	10	µ <b>q/L</b>	1	-	-	-	
60	DP25	Water	1992	Alpha,alpha-dimethylphenethylamine	10	ug/L	1	_	-	-	
61	OP25	Water	1992	Aniline	10	hð\r	1	+	-	-	
62	DP25	Water	1992	Anthracene	10	Nov	1			-	
63	DP25	Water	1992	Antimony	200	HON	5	-	_	-	
64	UP25	water .	1992	Aramine	10	HO/L		-		-	260 4
65	UP23 DD26	VVBIET :	1992		1.547.6		2	1		31	428-1
	0025	Weier	1000	анын Велгеле	- WA					210	250-1
<b></b>	DP25	Water	1002	Benzane	2 2	1401	17	ل م		1 705-03	258-18
1.5	DP25	Water	1992	Benzo(a)antivacene	10	ug/L	1		-	_	

	Ä	9	¢	0	Ε	F	G	Н	1	[ ]	K
									******		·
1	Source Area	Media	Date	Analyte	DL	Unit	s #Samples	i #Detect	Min Conc	: Max Conc	Max Location
70	DP25	Wale:	1992	Benzo(a)pyréné		10 µg/L		; <b>-</b>		i <del>.</del>	
71	DP25	Water	1992	Benzo(bifluoranthene		10 µg/L	1	-	-	i-	1
72	DP25	Water	1992	Benzo(ghi)perylene		10 µg/L		!!	·	<u>.                                    </u>	1
73	DP25	Wate:	1992	Benzo(k)fluoranthene		10 µg/L		<b> </b>	· -	<u> </u>	
74	DP25	Water	1992	Benzyt alcohol		20 µg/L		<b>_</b>		. <u> </u>	!
75	DP25	Water	1992	Beryllum		3 µg/L		ii-	;-	<u>!-</u>	
76	DP25	Water	1992	Bis(2-chloroethoxy)methane		10 µg/L	·		-	1	
17	UP25	Water	1992	Bis(2-chloroethyl) ether	····	10 µg/L	·····	-	<b>-</b>	<b></b> •	
78	DP25	water	1995	Bis(2-chioroisopropyi) einer				<b>!</b>	! <b></b>	¦ <b></b> ;	
-79	0P25	vvaler	1992	Bis(2-ethymexyl) phthalate			· · · · · · · · · · · · · · · · · · ·		<del></del>		
	0P25	vyater	1335	Bromkle		10 µg/L			! <b>-</b>	j <b>-</b>	
	DODE	vvater .	1992	Butyibenzyipritralate		10 µg/L		]=			
	0025	VValer	1995		NI/A	io:µg/L		-	4 20E + 04	7 705+04	1
	0025	And Ci	1003			6	12		4.206 104	7.70ET04	ZOMUZ
	0025	VERT	1993		N/A	Jug/L				- 705+02	260 4
	0025	VVACCT	1992		1100	10 upit	4		1.405703	1.70ET03	230+1
66	0625	VValor Molater	1003			10:00/				<b> -</b>	
<b> </b>	D025	Mater	1001	Chloraform	····	5 Uga	1	<u></u>	<u> </u>	<u> </u>	
00	0025	Water	1007	Chleenform	····-	5 00/		1	<u> </u>	<u></u>	·····
	DP25	Mater	1993	Chomern		20.000	. 13				
1-8-	DD36	Nigter	1002	Chovena		10-004					······
	0025	Widter	1000	Cis.1.2.dehloroethylene		1.00/			4 4		25P.16
⊢.÷	DP26	V/ster	1992	Cobalt		20 Jug/					230-13
	DP25	Water	1997	Caoper		20:000	A	1			25M02
	0025	1A/giar	1002	Di. c. butwindth gigta				<b>-</b>		40	EJINUZ
00	0025	Mater	1992	Diaactylophinalate		10					
67	DP25	Water	1992	Diallate		10 00/		t			······
	0925	Water	1992	Dibenzia bianthracene		10:00/	1				
	DP25	Water	1992	Dibertzofuran	<u> </u>	10100/1	1				
100	0975	Water	1992	Diethvi opthalate		10.00/	1 1	1			
101	DP25	Water	1992	Oimethoata		101.00/	1				· · · · · · · · · · · · · · · · · · ·
102	DP25	Water	1992	Dynethyl onthalate		10.00/	1	-	-	-	f
103	DP25	Water	1992	Diphenviamité		10.uu/L	1	-	-		
104	0P25	Water	1992	Ethvi methanesulfonate	·	10:ua/L	1	Ì-	-		
105	DP25	Water	1992	Ethylbenzene	•~~	21007	9	3	15	150	25B-15
106	DP25	Water	1992	Famohur		10 µa/L	1	-		_	
107	DP25	Water	1992	Fluoranthene		10 µg/L	1	1_	-	_	
108	DP25	Water	1992	Fluorene		10 µg/L	1	-	-	-	
109	OP25	Water	1992	Fluonde	N/A	:µg/L	2	2	200	400	25-2
110	DP25	Water	1992	h-Chlorofiuorobenzene	N/A	iµg/L	1	1	11	11	25B-1
111	0925	Water	1992	Hexachiorobenzene		10/µg/L	1	-	-	-	
112	DP25	Water	1992	Hexachloroputadiene		10 JUG/L	1	-	-	-	
113	DP25	Water !	1992	Hexachlorocyclopentadiene		10 µg/L	1	-	-	-	
114	OP25	Water	1992	Hexachtoroethane		10 µg/L	1	-	-	-	
115	DP25	Water	1992	Hexachlorophene	:	101µg/L	1	-		-	
116	DP25	Water	1992	Hexachioropropene	·	101µg/L	1	<u> -</u>	-		
117	DP25	Water i	1992	indeno(1.2.3-cd)pyrene	·	10 µg/L	1	-	-	-	
118	DP25 I	Water	1992	(ron	N/A	ug/L	5	5	710	1.60E+04	258-1
119	DP25 i	Water	1992	Isodnin		10 µg/L	<u> </u>	I=	-	-	
120	DP25	Water	1992	Isophorone	·	10)po/L			-		ļ
121	DP25	Water	1992	1805817Cie	·······	10 µg/L	<u> </u> 1	<u> -</u>		-	
122	DP25	Water	1992	Kepone		10;µg/L		<u> -</u>	-	-	
123	DP25	Water	1992	Kerosene	1.00E+	04 µg/L	<u>1</u>	<b> -</b>		-	050
124	DP25	Water	1992	Lead		5 µg/L		1	11	11	258-1
125	UP25	vvaler	1993.	LE80		2/hð/r	·[······	1	5.7	60	238-17
126	DP25	water	1992	m-Cresol	<u>.</u>	10 HOL	!	<b>-</b>	-		
127	DP25	vvater	1992			IU HO/L	<u> </u>		-	-	251.000
120	UP25	water	1992		,N/A	ING/L		5	9.90E+03	1.70E+04	25802
129	DP25	vvater !	1982		.N/A		5		1.402+03	6.00E+03	238-1
130	DP25	vvater	1992			IN HOLL	- <u> </u>	F		<u> </u>	
131	0025	water	1992		·	SUCT	+	<u> </u>			
132	DP25	water	1992				13	<u> </u>		-	359.4
133	0025	vvale:	1993				13	<u> </u> '	2.3		4-06-4
134	0025	vvaler Mater	1000	n-mudso-or-n-opropylamine		IO UPPL			-		
나꽖	0025	YVALC:	1000	n-mic usop-n-butylanine	· · · · · ·	101pg/L	1	<u></u>	-	<b>~</b>	
110	0925	Alatar	1992	n-Nim Solati yana -		10 POL		<u> </u>			

	A	8	C	0	E	] F	G	H H	1		ĸ
		L		4 <u>.</u>		4		·······	••••••••••••••••••••••••••••••••••••••		*
11	Source Area	Media	Date	Analyte	DL	Units	#Samples	i #Detect	Min Conc	Max Conc	Max Location
138	DP25	Water	1992	n-Nitrosodiphenylamine	10	Jug/L	1	1_	-	1-	1
139	OP25	Water	1992	n-Nitrosomethylethylamine	10	ug/L	1	-	; <b></b>	*	
140	DP25	Water	1992	n-Nitrosomorpholine	10	µg/L	1	· —	_  _	t	
141	0P25	Water	1992	n-Nifrosopiperidine	10	µg/L			j		1
142	DP25	Water	1992	Naphthalene	10	µ0/L	1	! <b></b>	i		
143	DP25	Water	1992	Nickel	30	ua/L	š	ii <b>-</b>	·	-	
144	DP25	Water	1992	Nitrate	N/A	_==_= µa/L	7	2	200	400	25B-1
145	DP25	Water	1992	Nrinte	200	UQ/L		21 <del>-</del>		1	
146	0825	Water	1992	Narobenzene	10			:: 	••••••••••••••••••••••••••••••••••••••	1	
147	DP25	Water	1992	Nitroscovralidioe	10			1	I	<u></u>	
	DB26	Water	1007	o Tolugioa	10	HALL		<u> </u>	<u> </u>	f	÷
140	0723	VVGITI	1992	o Chlorofiveraber 2000	NIZA	ug/l	· · · · · · · · ·				260 4
143	DP 23	Valer	1992	p-Chlorollociobenzene		99/L	<u>.                                    </u>		<u>_</u>		200-1
150	0025	vvaler	1337	p-Dimemylaminoazobenzene	10	- HALL					
191	025	vvater	1335	p-Phenyleneolamine	10	- pgrc			<b>-</b>	ļ <del>.</del>	
152	0P25	Water	1992	Parathion	10	HQ/L	<u>_</u>	[			ļ
153	DP25	Vvaler	1992	Pentachioronitrobenzene (pcnb)	10	hðv		-			
154	DP25	Water	1992	Pentachiorophenoi	50	ug/L	1	-		j=	ļ
155	DP25	Water	1992	Phenacetin	10	HQ/L	1	-	-	<u> </u>	
156	OP25	Water .	1992	Phenanthrene	10	:µg/L	1	<b> -</b>		-	
157	DP25	Water	1992	Phenol	10	µg/L	<u>!</u>	<u> -</u>	-		ļ
158	DP25	Water	1992	Phosphate	400	µg/L	2	-	-	I-	
159	DP25	Water	1992	Potassium	N/A	µg/L	5	<u> </u> 5	3.40E+03	4.50E+03	25M02
160	DP25	Water	1992	Pronamide	10	µg/L	1	-	-	-	
161	DP25	Water	1992	Pyrene	10	µg/L	1	-	-	-	
162	DP25	Water	1992	Safroi	10	µ <b>g</b> /L	1	-	-	-	
163	DP25	Water	1992	Silver	20	ind\r	5	-	-	-	
164	DP25	Water	1992	Sodaum	N/A	µg/L	5	5	4.10E+03	7.00E+03	258-1
165	DP25	Water	1992	Sulfate	N/A	HOL	2	2	1.10E+03	2.20E+04	25-2
166	DP25	Water	1992	Sym-trinitrobenzene	10	HQ/L	1	-	=	-	
167	DP25	Water	1992	Tetrachioroethene	0.5	Hg/L	9	-	-	-	
168	DP25	Water	1993	Tetrachloroethene	5	Pg/L	13	-		-	
169	DP25	Water	1992	Tetraethyl dithiopyrophosphate	10	HQ/L I	1	-	-	-	
170	DP25	Water	1993	Tetrahydrofuran	10	Jug/L	13	-		-	
171	DP25	Water	1992	Tin	100	JUG/L	5	1_		-	
172	0025	Water	1992	Tolvene	2	HO/L i	9	1	210	210	258-15
173	DP25	Water	1993	Toluene	. 5	JUQ/L	13	3	36	8.90E+03	258-18
174	0P25	Water	1992	Total organic carb	N/A	IND/L I	1	Ī	1.00E+03	1.00E+03	25-2
175	DP25	Water	1993	TPH-dieset	100	uu/L	14	6	100	1.100+04	258-4
174	0925	Water	1993	TPH-gasoline	2 00 =+03	LINGA	14	2	9 50E+03	1 905+04	25R_4
177	D825	Weter	1992	Trans-DCF	1	uo/i			-		
	0025	Alatar	1002	Tab da obserbare	10	Lun/L	1			E	
110	0025	1A/eter	1002	Trichtmethere	······································			<u> </u>		<u> </u>	
	0025	Vielen	1002		·			<u> </u>			······
100	DP23	Water	1002		30	100					
101	0025	VVelet	1002		<u> </u>		3			<b>—</b> ———————————————————————————————————	
162	0P25	vvater	1992		······					l	·····
183	DP25	vvater	1993		10		13		-		000 40
1.84	0025	VVAICT	1335	Ayenes (KRI)				<u> </u>	067	400	200-13
185	0225	vvaler	1993	Ayenes (ICCH)			13	ļ	230	3.402+03	230-10
186	0423	vvater	1992		10	<b>HHH</b>	5	ļ	18	66	238-1
	0925	SUTISCE SOIL	1768		1	HANKS	3				2540 -
	UP25	SUITECE SOIL	1988	Benzens	NVA	Induka	1	<b>1</b>	410	410	2040-0
169	DP25	Surface Soil	1988	BHC. OCL		Hours			-	-	
190	DP25 -	Surface Soil	1988	BHC, gamma (Indane)	1	ihðveð	6	-			
191	DP25	Surface Soil	1968	Chlordane	1	Induct	8	-		-	
192	DP25	Surface Soil	1968	DDD, pp'	1	ihð\kð	5	[			
193	DP25	Surface Soil I	1988	DDE. PP	<u>t</u>	have	6	<b> -</b>	-	<u> </u>	
194	OP25	Surface Soil	1988	DOT. 95	<u> </u>	hayro	5	2	1.5	5.7	25M02
195	DP25	Surface Soil	1968	Diekinn	1	iug/kg	8	1	8	8	25TP05
196	DP25	Surface Soil	1988	Diethyl ether	200	hð/kg	5	-	-	-	
197	DP25	Surface Soil	1988	Endosulfan sulfate	1	Pa/kg	6	-	_	-	
198	OP25	Surface Soil	1988	Endosulfan, a	1	ug/kg	6	2	6	9	25TP01
199	DP25	Surface Soil	1968	Endosulfan, b	1	HQ/kg	8	-		-	
200	DP25	Surface Soil	1988	Endnin aldehyde	1	pg/kg !	6	-	-		
201	DP25	Surface Soil :	1988	Ethylbenzene	80	have	9	4	13	3.00E+03	25AB-5
202	DP25	Surface Soil !	1988	Heptachlor	1	ug/kg	5	<u> </u>	-		1
203	OP25	Surface Sol	1968	Heptachior epoxide	1	ipg/kg	6	1	1.4	1.4	25TP01
204	DP25	Surface Soil	1986	Lead	'N/A	µg/ka l	20	20	4.00E+03	1.00E+04	25-2
205	DP25	Surface Soll	1044	Lest	·N/A	ucito	34	34	1.605+01	2 535+04	258-15

.- --

	A	6	<u> </u>	j 0	<u> </u>	ί F	G	н		7	ĸ
1										i	!
	Source Area	Media	Date	Analyte	2L	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
206	DP25	Surface Soil	1988	PCB-1254 (aroclor)	54	hð\kð		1	172	172	25TP05
207	DP25	Surface Soil	1988	Toluene	N/A	µg/kg	3	3	0.022	3.10E+03	25AB-5
208	DP25	Surface Soil	1985	ТРН	1 C2E+04	ug/kg		4	1.16E+04	5.89E+05	25TP01
209	DP25	Surface Sort	1988	Xylenes (total)	N/A	µg/kg		4	i <b>48</b>	1.50E+04	125A8-5
210	DP25	Surface Soil	:988	Xylenes (lotal)	80	µg/kg		51-	-	<u>;                                    </u>	
211	DP25	Soil	1988	Aidrin	1	µg/kg		)i		<u>-</u>	
212	DP25	Sou	1988	BHC, beta	1	µg/kg	<u> </u>	)	i-	-	
213	DP25	Sort	1986	BHC, gamma (lindane)	1	µg/kg	<u>9</u>	H=	i	1-	1
214	DP25	Soil	1988	Chiorgane	1	µg/kg	<u> </u>	)i=	-	i-	
215	DP25	Soil	1988	DDD. pp	1	µg/kg	5	)	-	-	
216	DP25	Soil	1986	ODE pp'	1	µg/kg	. S	li i	2	2	25TP02
217	OP25	Soil	1988	DDT pp	1	ug/kg	9	3	2	1 17	25TP02
218	DP25	Sol	1988	Dieidrin	1	µg/kg	r <b>9</b>	<b> -</b>	-	-	
219	DP25	Soil	1988	Diethyl ether	200	µg/kg	9	1	500	500	25M01
220	DP25	Soil	1988	Endosullan sulfate	1	uo/ko	9	<b>i</b> _	-		
221	DP25	Sol	1988		1	ua/ka	Ś			-	h
222	DP25	Soil	1968	Endosulian b		uo/ko	9			-	
371	0025	Soil	1088	Endrig aldehyde	1	un/kn		······································		<b></b>	
224	0925	Cal	1088	Ethybertene	80	uo/ko		1 1	100	190	251401
1 228	0016	Sal	1900	Langinghior	1	ug/kg		<u></u>	- 130		
443	0025	Sed	1000		·····	Py'ny	÷ 0	1	750		25TDAt
240	0723	500	1900		BU(A	yyr y			9 705 -05	8 705 406	2500 4
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0825	300	1900		50A	HALKS	·	<u></u>	5 205-00	0.100405	2330-7
420	0005	- 30H	1908		1WA	HAVE		D	J.202+U3	3.00E+U3	257004
229	0026	2011	1900		24	aug/Kg		<u> </u>	013	013	231PU1
230	0025	3011 Sed	1986		1.022704	HQ/KQ	9	4	1.025+04	2.09E+02	25MU3
231		501	1968	Xylenes (1018)	00	HQ/KÇ	8	1	110	110	25401
232	DP44	Water	1992	T.1,1-Trichloroetharie	0.5	µg/L	15		-	-	
Z33	DP44	VVater	199Z	1,1,2-Trichloroethane	0.5	ug/L	15		-	-	
234	DP44 .	Water	1992	1.1-Dichloroethane	1	µg/L	15	-	- 	-	
235	DP44	Water	1992	1.2-Dichloroethane	0.5	hð/r	15	) <b>-</b>	-	<u> </u>	
236	DP44	Water	1992	1.4-Dichlorobenzene	2	.µg/L	15	-	-	-	
237	DP44	Water	1992	Antimony	200	Mg/L	4	Į <u> </u>	-	-	
238	DP44	Water	1992	Barium	20	HQ/L	4	3	190	200	44M04
239	OP44	Water	1992	Benzene	2	hðvr	15	2	3.7	5.3	44M05
240	OP44	Water	1992	Beryttium	3	µg/L	4	-	-	-	
241	DP44	Water	1992	Bromde	500	pg/L	4	-		-	
242	DP44	.Water	1992	Cadmum	10	µ9/1	4	-	<b>.</b>	-	
243	DP44	iWater	1992	Calcium	100	ug/L	4	3	5.60E+04	6.00E+04	44M060
244	DP44	Water	1992	Carbon tetrachloride	1	₽g/L	15	-	-	-	
245	OP44	Water	1992	Chioride	200	µg/L	4	3	2.30E+03	2.30E+03	441404
246	DP44	Water	1992	Chloroform	0.5	149/L	15	-	-	-	
247	DP44	Water	1992	Chromium	20	µg/L	4	-	-	-	
248	DP44	Water	1992	Cis-1,2-dichloroethylene	1	HQ/L	15	12	1.1	260	44M04
249	DP44	Water	1992	Cobalt	20	µg/L	4	-		-	
250	DP44	Water	1992	Copper	20	µg/L	4	-	-	-	
251	DP44	Water	1992	Ethylbanzane	2	μ <u>φ/</u> L	15	-	-	-	
252	OP44	Water	1992	Fluoride	100	µg/L	4	3	100	t00	44M04
253	DP44	Water :	1992	h-Chiorofluoroberizene	N/A	µg/L	9	9	9	12	44M04
254	DP44	Water	1992	Iron	20	µg/L	4	3	6.00E+03	6.40E+03	44MQ4
255	OP44	Water	1992	Magnesium	100	µg/L	4	3	1.20E+04	1.20E+04	44M04
256	DP44	Water '	1992	Manganese	10	HQ/L	4	3	3.30E+03	4.60E+03	441406
257	DP44	Water	1992	Methylanechloride	5	uo/t	15	1-	-	-	
258	DP44	Water	1992	Nickel	30	µq/L	4	<b> -</b>	-	-	
259	DP44	Water	1992	Nirate	200	µg/L	4	1	700	700	4414068
280	DP44	Water	1992	Ninte	200	UQ/L	4	t			
261	DP44	Water	1992	p-Chlorofuorobenzene	N/A	UQ/L	9	9	8.5	10	44602
267	DP44	Water	1992	Phasohile	400	Lun/	4	-	-	-	
201	DP44	Water	1997	Polassium	300	100/			3.305+01	3.505-03	4414060
244	0244	Water	1997	Silver		1007			-		
		Water	1002	Sodium	300			-	A ONELON	5.800-100	441404
100	DRAA	Alatar	1007	Suffete	500	147/			1 205-04	1 605-0-0	444408
200	DB44	Water	1007	Terrechlomethene	0.4	PY'L	12		A 7	1.VVE 744	444400
<del>ا</del> ید ا	0044		1992		100	1994 1996	13	1	<u>v.7</u>	<u></u>	
1 400	0044	Alater	1000	Taliasa	100			E			
209	OP44	VIII I	1792	Trana DCE	<u>4</u>		19	Į <b>-</b>			444404
2/0	0744	YVALOT	1992	Trans-UCC	1	HWL	15	<u> </u>	1.3	3.4	
1271	0044	A A A A A A A A A A A A A A A A A A A	1992		1	100	15	3	1.2	2.302+03	
1272	0144	VVOIE!	1992			PO/L	4		-	-	(I
1 273	UP44	vvater	1992		2	DO/L	15	1=	-		( 7

	A	в	C C	D	<u>Ε</u>	F	G	Н		Ţ Ţ	ĸ
<b></b>	<b></b>	<b></b>		<u> </u>			· · · · · ·		••••••••••••••••••••••••••••••••••••••	i	••••••••••••••••••••••••••••••••••••••
1 1	Source Area	Media	Date	Analyte	ÐL	Units	#Samples	: #Detect	Min Conc	Max Conc	Max Location
274	DP44	Waler	1992	Xvienes (total)	5	ud/L	15	. <u> </u>	-		
275	DP44	Adater	1907	7.60	10				·		
276		Sudace Sol	1992	1.1.1.Trichloroethage	0.5	Un/ka					<u>;</u>
2/0	0044	Surface Soll	1002			pyrky		·		÷	<u> </u>
411	UP44	Surface Soll	1992			Para			Į=	<u></u>	
2/8	DP44	Sunace Soil	1995	r 1-Dicrigitettiane		novka					<u> .</u>
279	DP44	Surface Soil	1992	1 2-Dichloroethane	0.5	ug/kg		'i <b>−</b>	!-		
280	DP44	Surface Soil	1992	1.4-Dichlorobenzene	2	µg/kg		!-		<u>i –                                    </u>	1
201	DP44	Surface Soil	1988	2-Methylnaphthalenee	30	µg/kg	3	1	550	550	44M03
282	DP44	Surface Soil	1988	Acenaphthene	10	µg/kg	3	1	2.90E+03	2.90E+03	44M03
283	OP44	Surface Soil	1986	Acenaphthylene	10	µa/kg	3	f	480	480	441403
284	DP44	Surface Soil	1988	Alominum	N/A	uc/Ka	3	i 3	5 21E+06	1 396+07	44M01
285	DD44	Surface Soil	1089	Appress	20	10/20		1 1	5 50E+03	5 505+03	4414073
203		Curface Soli	1049		4 505+03	- www.s		†	8.002+03	8.000-000	44404
200	UP44	Shuace 201	1900		4.306403	hourd	·		3.002703	1 0.002 -03	44NU 1
26/	UP44	Surface Soil	1988	Banum	N/A	hđ\kd			7.90E+04	1.03E+05	44MC1
288	DP44	Surface Soil	1992	Benzene	Z	µg/kg	9	;	-	-	
289	DP44	Surface Soil	1988	Benzo(a)anthracene	10	ug/kg	3	2	i 240	4.80E+04	44M03
290	DP44	Surface Soil	1988	Benzo(a)pyrene	9	µg/kg	. 3	2	470	1.80E+04	44M03
291	DP44	Surface Soil	1968	Senzo(b)/kuoranihenene	30	µg/kg	. 3	2	690	2.10E+05	44M03
292	DP44	Surface Soil	1988	Benzo(g,n,i)parylene	40	ug/kg	3	1 2	630	1.40E+04	44M03
291	DP44	Sufface Soul	1988	Cadmum	N/A	ua/ka	3	3	726	906	44M02
794	DP44	Surface Soil	1988	Саюнт	N/A	. UC/YO	1	1	2.705+04	1 105+07	441401
100	0944	Surface Coll	1001	Carbon tetrachiorida	+				4.792,00		
290	0044	Gunace Soll	1000			PU/KU					444404
296	0144	Surrace Soil	1988		20	hävkä	. 3		1/0	011	44MU)
297	DP44	Surface Soil	1992	Chioroform	0.5	ihð\kð		-	-		
298	OP44	Surface Soil	1988	Спотил	N/A	.hð\xa	3	3	1.53E+04	2.74E+04	44M01
299	DP44	Surface Soil	1988	Chrysene	70	iµg/kg	3	1	2.106+04	2.10E+04	44M03
300	DP44	Surface Soil	1992	Cis-1.2-dichloroethylene	1	µo/kg	9	-	-	[-	
301	DP44	Surface Soit	1986	Cobelt	N/A	ug/kg	, 3	3	4.72E+03	1.37E+04	445401
302	DP44	Sufface Soil	1988	Copper	N/A	:µa/kg	3	3	1.22E+04	3.65E+04	44M01
303	0244	Surface Soil	1988	Dibeazo(a b)anthracene	50	uaña	3	1	6 50E+03	6 50E+03	441401
1 304	DB44	Surface Cod	1000		10	LIONA	;		2 305+03	2 205-03	441407
	DO44	Surface Sol	1000			- pyrag			2.002.00	2.000	
300	UP44	Surrace Soll	1332	Envioenzane		- Marka			4 005 4 0 4	-	
306	OP44	Surrace Sol	1988	Fluoranmene		рака	3	1	4.90E+04	4.90E+04	44.403
307	DP44	Surface Soil	1986	Fluorene	30	HO/KO	3	1	3.80E+03	3.60E+03	44M03
308	DP44	Surface Soil	1968	Indeno(1.2.3-cd)pyrene	50	ihāvkā	3	2	660	1.50E+04	44M03
309	DP44	Surface Soil	1968	Iron	N/A	µg/kg	3	3	9.552+06	2.58E+07	44M01
310	DP44	Surface Soil :	1988	Lead	N/A	µg/kg	6	6	1.17E+04	4.79E+04	44M02
311	DP44	Surface Sod	1988	Magnesum	N/A	:ug/kg	3	3	2.65E+06	7.12E+08	44M01
312	OP44	Surface Soil	1988	Manager	N/A	100/k0	1 3	3	1.635+05	4 455+05	441401
111	DP44	Sudace Soil	1988	Methylene chingle	N/A	un/ka	3	3	1 806+03	1 105+04	444401
	0044	Surface Soil	1000		5	PP MP			1.002.00		
	0044	Sumace Soll	1992			IPP			-		449.802
1313	0244	SUMACE SON	1966		- 6.3/A	INTER	3		000	000	
316	DP44	Surface Soil	1988	NICKEI	INVA	haved	3	3	1.10E+04	3.40E+04	44MU1
317	DP44	Surface Soil	1988	Phenanihrene	20	µg/kg	3	I	3.70E+04	3.70E+04	443403
318	OP44	Surface Soil	1988	Polassium	N/A	i µg/kg	3	3	5.95E+05	9.40E+05	44M01
319	OP44	Surface Sod	1988	Pyrane	60	i µg/kg	3	1	3.20E+04	3.20E+04	44M03
320	DP44	Surface Soil	1966	Sodkum	N/A	,µg/kg	3	3	2.048+05	4.10E+05	44M01
321	DP44	Surface Soil	1992	Tetrachloroethene	0.5	ua/ka	9	1	- 1.8	1.8	445Y504
1 322	DP44	Surface Soil	1997	Toluene	2	uake	9	5	- 21	34	445YS06
L.	DRAA	Curface Cell	1004	······	1 076404	LUDA'S			3.000-00	A 445.04	44402
	0044		1000		1.005+04	Ling and			2 405-04	7.005.00	440010
124	L/P'44	Sumace Soft	1000		2.002704	(PARK)	4	<u></u>	4.400704	1.000 700	
325	LXP44	SUITACE SOIL	1992	Irans-DUE	1	have	ļ		-	-	
326	DP44	Surface Soil	1992	Trichloroethene	1	Induction	9		-	-	
327	OP44	Surface Soil	1988	Vanadium	N/A	hð kð	3	3.	2.03E+04	5.41E+04	44M01
328	OP44	Surface Soil	1992	Vinyl chlonde	2	ug/kg	9	<b> -</b>	-	+	
329	DP44	Surface Soft	1992	Xylenes (totel)	5	iug/kg	9	-	-	-	
330	DP44	Surface Soil	1988	Zinc	N/A	ING/Kg	3	3	3.18E+04	1.46E+05	441402
331	DP44	Soil	1984	2-Methylnaphthalense	N/A	U0/10	1	1	30	30	44M01
1777	DP44	Soil	1944	Acadaphibeos	10	10000	;à				
1000	0044	Sod .	1004	4		LINE PR					
1333	0144	304	1966		10	IN WES			-	-	
1334	UP44	501	1990	Acerophenone	30	have	22	ļ	100	100	44.004
335	DP44	501	1988	Aluminum	N/A	µ9/kg	<u> </u>	1	6.65E+06	6.65E+06	441401
336	DP44	Sol	1988	Anihracene	20	iµg/kg	1	-	-	-	
337	DP44	Sol	1988	Antimony	4.50E+03	µg/kg	1	-	-	-	
338	DP44	Soil	1968	Banum	N/A	:µg/kg	1	1	1.13E+05	1.13E+05	44M01
339	DP44	Soil	1988	Benzo(a)anthracene	N/A	"Jo/kū	1	1	200	200	44M01
340	DP44	Sod	1988	Herizo(a)wrene	N/A	10/10	1	1	390	390	44M01
1341	DP44	Soil	194.8	Benzo(b)fuscanthenece	N/A		<u> </u>		440	440	44M01

*~* .

	A	8	С	0	Τ	E	F	G	н	1		ĸ
					-							
1	Source Area	Megia	Date	Analyte	0	DL	Units	#Samples	#Detect	Min Conc.	Max Conc	Max Location
342	DP44	Soil	1948	Benzora h uperviens	N'A		ua/ko		1	280	280	44101
141	DB44	Sail	1000	Berganbellervil Obhalate		50	ualka					441407
144	0044	- 3011 	1000	Dist 2-eutrinexy i problete	••••		ug/kg		·····	80	60	446006
1	0944	501	1000	Buly benzy promite		334	HY NY				<b></b>	***3809
343	DP44	301	1968		hure.	435	purky	·			-	
346	UP44	501	1988		NIA		µgvkg			4.485+08	4.48E+06	44MU1
347	UP44		:988			20	hð/kð	······		-		 
348	DP44	501	1988	Chromium	N/A		hà/kö		1	1.12E+04	1.12E+04	44M01
349	DP44	SOI	1988	Chrysene	N/A		hð\kð	<u>1</u>	1	280	260	44M01
350	DP44	Soil	1986	Cobalt	N/A		hð/kð		<u> </u>	6.49E+03	6.49E+03	44M01
351	DP44	Soil	1988	Copper	N/A		ug/kg	. 1	1	1.39E+04	1.39E+04	44M01
352	DP44	Soil	1988	Dibenzora.h)anthracene		50	µg/kg	. 1	! <u> </u>	-	-	
353	DP44	Sou	1988	Dibenzoluran		10	µg/kg	1	-	-	-	[
354	DP44	Soil	1988	Fluoranthene	N/A		µg/kg	1	1	770	770	44M01
355	DP44	Soli	1988	Fkiorene		30	ua/ka	1	i-	-	_	
144	0944	Sail	1988	Indeport 2 3-cd/ovrene	N/A		ua/ko	1	1	270	270	441401
167		Soul.	1984	Ino	N/A		UD/20	1	••••••••••••••••••••••••••••••••••••••	1 15E+07	1 15E+07	444401
144		1501	1007	Várocene		033	Ug/kg	10		50	510	AASSAE
330	0044	- 304 Seit	1000		61/A		HALLA I	;		8 5 4 E + 62	1 245 404	44000
133	0044	201	1908		N/A	~~~~	HYRY .			0.042700	1.245404	446000
300	UP44	201	1990		N/A		haved	4		2.0000003	4.07E+04	443808
361	UP44	50#	1498	wagnes(om	NVA		hð Kð		<u> </u>	3.305+06	3.30E+U6	44MUI
J62	UP44	501	1988	manganese	NVA		hā\kā	1		2.156+05	2.15E+05	44M01
363	DP44	Sol	1988	Methylene chloride	N/A		hð, kð	<u> </u>	ļ	1.70E+03	1.70E+03	44M01
364	DP44	Sol	1988	Naphingiene		9	µg/kg	1	i <b>-</b>	-	-	
365	DP44	Soil	1990	Naphthalene		30	µq/kq	22	i 1	80	80	44M04
366	0P44	Soil	1986	Nickel	N/A		µg/kg	. 1	1	1.43E+04	1.43E+04	44M01
367	DP44	Sou	1988	Phenaninrene	N/A		ug/kg	1	1	530	530	441401
368	0P44	Soil	1988	Potassium	N/A		ug/kg i	1	! 1	6.05E+05	6.05E+05	44M01
369	DP44	Sol	1988	Pyrene	N/A		hð/kg	1	1	400	400	44M01
370	DP44	Soil	1988	Sodium	N/A		µa/ka	1	1	2.55E+05	2.55E+05	44M01
371	DP44	Soil	1992	Total dissolved solids	N/A	لم :	%	10	10	83.9	97.5	445507
372	DP44	Soil	1992	Total petroleum hydrocachons	N/A		uc/ko	10	10	0.063	31	445504
171	OP44	Sol	1989	тры	N/A		un/iro	1	1	4 76E+04	A 785+04	44401
		Soil	1000		1 40	1	uniko i	55	30	1 105-04	7 206408	445908
176		<u>QOII</u> Cail	1000		NI/A		HORE			7.445-04	2445-04	444404
	DP44	501	1000		AVA	······	uwny .			2.415-04	2.446 104	
1			1906	<u>Zac</u>	. I'WA		Pyring		<u>├</u>	2.015104	4.0/6704	44401
377	LF02	vvater	1994	( echnical) chiorgane		·····	μφιι	L	-	-		
378	LFOZ	vvater	1994	1,1,1-Trichlorceinane			hðvr		-	<b>-</b>	-	
379	LFOZ	Water	1994	1.1.2.2-Tetrachloroesnane	~	· · · · · · · · · · · · · · · · · · ·	HAL	•	-		-	
380	LF02	Water	1994	1.1,2-Trichloroethane	~~~ <u>~</u>	0.5	POL	6	-	-	-	
381	LF02	Water	1994	1.1-Dichloroethene		<u>1</u>	hđv:	6	-	-	-	
382	LF02	Water	1994	1.1-Dichloroethene		0.5,	µg/L	6	<u> -</u>	-	-	
383	LF02	Water	1994	1.2,4-Trichlorobenzene	<u></u>	10	nðr	4	-	-	-	
384	LFO2	Water	1994	1.2-Dichlorobenzene		101	µg/L	16	-		-	
365	LF02	Water	1994	1,2-Dichloroethane		0.5	µg/L	6		-	-	
386	LF02	Water	1994	1.2-Dichloropropane		0.5	µg/L	6	-	-	-	
387	LF02	Water	1994	1.3-Dichlorobenzene		10	μ <b>9/</b> L	16	-	-	-	
388	LF02	Water	1994	1.4-Dichlorobenzene		10	µg/L	16	-	-	-	
389	LFQ2	Water	1994	2,4,5-Trichlorophenol		10	HO/L	4	-			
390	LFD2	Water	1994.	2.4.6-Trichlorophenol	1	10	µg/L	4	-	-	-	
391	LF02	Water	1994	2.4-Dichlorophenol		10	HQ1	4			-	
392	LF02	Water	1994	2.4-Dimethylphenol	÷	10	ug/L	4	-	-	_	
393	LF02	Water	1994	2.4-Dintrophenol	<u>.                                    </u>	50	JUG/I	4	-			
394	LF02	Waler	1994	2.4-Dinantokuene		50	ua/	i i	<u> _</u>	-		
195	En2	Weter	1994	2.8. Designation serve		10	107/		·····			······
1	1.502	Vister	100/	2. Chiemathuhand athas			105		<u> </u>		-	
Hind Hand	U V2		100/			40	1997 G		<u> </u>			
<b>1</b>	1.500	vvaler	1894		<u> </u>			4				·····
138	LFUZ	vvaler	1994			10	HANT		<b></b>			
283	LFOZ	AAS(et.	1994	z-meutyinaprinaiene		101	hàvr	4	<b>-</b>	-		
400	LF02	Water	1994	2-Methylphenol		10	HQ1	4		-	-	
401	LF02	Water	1994	2-Niroaniline		50	hår	4	-	-	<b>.</b>	
402	LF02	Water	1994	2-Nitrophenol		10	µg/L	4	اا			
403	LFO2	Water	1994	3.3'-Dichiorobenzidine		20	µg/L	. 4	-	-	-	
404	LFO2	Water	1994	3-Nitroaniline		50	µg∕L	4			-	
405	LF02	Water	1994	4.4-000		0.1	µg/L	4	-	-	-	
406	LF02	Water	1994	4.4-DDE		0.1	Pg/L	4	-	-	-	
407	LF02	Water	1994	4.4'-DDT		0.1	µg/L	4				
408	LF02	Water	1994	4-Bromophenyl-ohenviether		10	JUD/L	4	-		=	
409	LFO2	Water	1994	4-Chloro-3-methylphenol		20	ua/L	4	-	-	-	

~

.

· · · ·	· · · · · · · · · · · · · · · · · · ·	1	~	D	T	1 E	7 2	<u>т п</u>	·····	T	r
ļ	<u> </u>		L		<u> </u>	L	1	L <u>. "</u>	<u></u>	L	<u> </u>
										i	
1	Source Area	Media	Date	Analyte	DĻ	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
410	LF02	Water	1994	4-Chloroaniline	20	) µg/L	4	i <del>-</del>	-	i-	!
411	LF02	Water	1994	4-Chlorophenyl-phenylether	1(	Jun/L	4	<b>_</b>	-	+	
412	1 602	Mater	1004	4 Melbulopecol	10	) und	4		!	<u>.</u>	
	1.502	******	1994							·	
413	LFU2	water	1994			hđir		·	-	·	: 
414	LF02	water	1994	4-Narophenol	50	) hdvr		· <u> </u>		j <del>~</del>	••••••
415	LF02	Water	1994	a-8HC	0.05	i µg/L	4	i		: <b>-</b>	
416	LF02	Water	1994	a-Chlordane	0.05	ug/L	4		-	1	·
417	LF02	Water	1994	Acenaphthene	10	) ua/L	4	· _	r <b>–</b>	i	
444	L 602	Water	1004	Acanophylage	16	1000			·····	1	Ļ
140			1004	Alexa Alexa						. <u>.                                   </u>	· · · · · · · · · · · · · · · · · · ·
418	LFUZ	water	1994		0.05	, hðir	4		; == (		·
420	LF02	Water	1994	Aluminum	N/A	µg/L_	. 4	4	2.53E+03	8.57E+03	02M0B
421	LF02	.Water	1994	Anthracene	TÇ	) µg/L	4	-	-	i-	
422	L F02	Water	1994	Ασιτηρόν		UD/L		1	19.8	19.8	021109
422	1.500	Minter	1004	Accento	ALLA		*		21.4	705	024400
463	LFUZ	VVALC:	1974	Arsenic		HAL			21.9	<u> </u>	
424	LF02	water	1994	D-8HC	0.05	. hður	•		-		
425	LF02	Water	1994	Barium	N/A	_µg/L	8	i 8	37.7	392	02M01
426	LF02	Water	1994	Benzene	1	µg/L	. 6	-	-	-	
427	L F02	Water	1994	Renzo(a)anthracene	10	Ua/L	4	-		-	
420		Water	100/	Benzo(a)ovrene	16				······	h	
1400	1.500		1002			HW -	··			t	
429	LFOZ	water	1994	Seuzo(D)IIUoraninené	10	HQVL				/ <del>-</del>	
430	LF02	Water	1994	Benzorg.h.uperylene	10	µg/L	. 4	-	**	-	
431	LF02	Water	1994	Benzo(k)fluoranthene	10	ug/L	4.	-	-	[++	
432	LF02	Water	1994	Senzoic acid	SC	ug/L	4	-	-	-	·······
411	E07	i)A/ster	1904	Renzyl akohol	20					t	
			1334			Par	· · · · · · · · · · · · · · · · · · ·			h	· · · · · · · · · · · · · · · · · · ·
434	LFOZ	(VVater	1994	Beryllium		HOL		-		[ <b>-</b>	·
435	LF02	Water	1994	Bis(2-chloroethoxy)methane	10	ug/L	1 4	-	-	-	
436	LF02	Water	1994	Bis(2-chloroethyl)ether	10	ug/L	4	-	-	I-	
437	LF02	Water	1994	Bis(2-chloroisopropyl)ether	10		4	-	-	-	
418	1 602	Mater	1994	Ris (2-athulhayy) of thaiala	10	-uo/l	4			l	
	- F02		1354	Dist 2-en ymex y yp milaitaic	100			······································		ļ	
439	L+02	vyater	1994	Riouzde	100	-inder	·			ļ <b>-</b>	
440	LF02	Water	1994	Bromodichioromethane	1	∙µg/L	6	-	-	-	
441	LF02	Water	1994	Bromoform	1	·μ <b>g/L</b>	- 6	-	-	-	
442	LF02	Water	1994	Bromomelhane	2	Hg/L	. 6	_	-	[_	
441	1 602	Water	1994	Budylbenzylonthatate	10	un/	4	_		1	·····
	1 500	TALIAS	1004	Cadman		-					····
		VVELOT	1334	Calimum		ibor.			-	-	
445	LF02	Water	1994	Cekcum	N/A	HOL	8	8	2.97E+04	6.17E+04	02M02F
446	LF02	Water	1994	Carbazole	10	) (JUD) (	4		i	L-	
447	LFOZ	Water	1994	Carbon tetrachlonde	0.5	μα/L	6	-	-		_
448	LF02	Water	1994	Chloride	N/A	uo/L	4	4	960	1.60E+04	02M02
440	1 502	Alater	1004	Chiomheatene	1	- and	12				
	1500	Alata.	1004	Chloridenzen						······	
450	LFGZ	vvater	1994	Chloroemane	l	HOL			-		
451	LF02	Water	1994	Chloroform	T	HQ1	6	-	-		
452	LF02	Water	1994	Chloromethane	1	μ <b>α/L</b>	6	-	-	<b> -</b>	
453	LF02	Water	1994	Chromium	N/A	uge.	3	3	11	19.9	02M08
454	F02	Water	1994	Слузеле	10			I_		I	
100	L 602	1 A later	100/	Cir. 1 1 dichlomorane -	0.5		†~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~				· ·
	LTV2	YYALCI	1334		·····	The state				<b>H</b>	
456	LFOZ	vvaler	1994		1	hair .	į <u> </u>		<b>-</b>	[ <b></b> ]	
457	LF02	Water	1994	Copper	1	, uga	7	4	26.2	55,9	021401
458	LF02	Water	1994	d-BHC	0.05	Jug/L	4	-	-	-	
459	LF02	Water	1994	Di-n-butyiphthalate	N/A	JUG/L	4	4	• 1	2	02M01
460	1 502	laister	1994	Dunoctylabilatio	10	unal	4				
	LF V2		1004								
401	LFUZ	1 VVerter	1334	Uperiz(a.n)animi avene		HUYL	·	-			-
462	LF02	Water	1994	Dibenzoturan	10	PO/L	4		-	-	
463	LF02	Water	1994	Dibromochloromethane	1	I pg/L	6	-	-	-	
444	LF02	Weter	1994	Dichlorodifuoromethane	1	Jugh	6		I <b>-</b>	-	
485	1 602	Water	1994	Diekim	0 1	.uni	4		_	_	
	1503	Liferator	1004	Ciedle debth alate							
400	LFUZ	TYANA	1934	CleanAdhungiliere		INDER			-	ļ <b>-</b>	·····
467	1.502	INALEL	1994		10	LING L	4	-	-	<b>[</b> ]	
468	LF02	Water	1994	Endozuifan i	0.05	LING/L	4	-	-	-	
469	LF02	Water	1994	Endosulfan li	0.1	Hg/L	4	-	-	I-	
470	LF02	Water	1994	Endosulian sulfate	D 1	ug/L	4	-	-	[	
171	1 E02	Water	100/	Entrin	0.1	100				<u></u>	
+	LF 02	I FYGIOT	1004			inder -	+				
472	LF0Z	water	1994	EINITI BIGENYOS	U.1	har	4		-	<b> </b> '	
473	LF02	Water	1994	Ethylbenzene	1	1001	6	-	<u>-</u>	[ <b>-</b> ]	
474	LF02	Water	1994	Fluoranihene	10	Jug/L	4	-	-	-	
475	LF02	Water	1994	Fluorene	10	Iua/L	4	-	-	-	
474	1 E02	Water	1904	A-BHC			1				·
F#	LTU2		1334	9-DIC	0.03					f=	·····
[ 477 ]	ILF02	(VValer	1994	d-Chiordane	0.05	JUDIL	1 44			í <del>–</del>	

•

<b></b>	A			B	F	Т	F			H	<del></del>	<b></b>	
		L	استسا		I	_	<u> </u>	L			l'	ļ	<u> </u>
	Source Area	Madin	Dole	An slute			inde			-	i Mie Cono	Mary Case	Man (
478	LEAT		1004	Haplachlor		5	0/11/1	+oampie	4	-		Max Conc	Max Cocation
479		14/ster	1004		0.0	č.	3				· _	1	<u>.</u>
480	1 602	Mator	1054				19/1- 10/1						
481	LEN2	Water	1004	Hey Schlorobulsdieße		<u> </u>	····-		A				
482	L F02	Water	1064	Hexamiorocyclopentagiene					A	······································	· · · · · · · · · · · · · · · · · · ·	+ <u>=</u>	
481	1 602	Water	1004	Heyschloroothane		<u> </u>	9.5	•			· · · · · · · · · · · · · · · · · · ·	+	j
144	1 502	Malac	1004	baedoul 2 3.oduovrege		<u> </u>	φ. 			••••••••	ļ		
485	1602	Mater	1004	Indenial 2.3-00/Direne	NUL	<u> </u>	1941 L.		0	0	4 855 -01	A DEFLO	:021420
400	1602	March	1004			గో			<u>0</u>	•	4.03E703	0.002704	IVENUO
400		water	1394	Isoproidite		<u>, 1</u>	g/L				-	-	
	LFU2	vvater	1394	Leau		<u>. H</u>		······	D		0.1	18.3	02M01
405	LFOZ	vvater	1994	M p-sylene		1.4	ig/L	·······	<u>D!</u>		-	-	
489	LFOZ	water	1994	Magnesium	N/A		<u>9/L</u>				7.15E+03	2.32E+04	021401
490	LFO2	Water	1994	Manganese	N/A	<u></u>	ig/L		8	8	1.02E+03	1.67E+03	02M02F
491	LF02	Water	1994	Methoxychior	0.5	5 <u>µ</u>	<u>9/L</u>		4!-			<u> -</u>	
492	LF02	Water	1994	Methylene chlonde		<u>1 µ</u>	<u>g/L</u>		61	1	1.7	1.7	02M01
493	LF02	Water	1994	n-Nilroso-di-n-propylamine	10	οų	g/L		4 –		-	<u> </u>	
494	LFOZ	Water	1994	n-Ndrosodiphenylamine	1(	<u>9 µ</u>	g/L		41-			i	
495	LF02	Water	1994	Naphthalene	10	μĊ	Q/L		4 -		-	-	
496	LFO2	Water	1994	Nicke!	N/A	ų	g/L		1	1	22.9	22.9	021402
497	LF02	Water	1994	Nitrobenzene	10	) µ	g/L		4 -		-	-	
498	LF02	Water	1994	o-Xylene	1	1 µ	g/L	(	6		-	-	
499	LF02	Water	1994	PC8-1016	1	1.μ	g/L		4 -		-	-	
500	LF02	Water	1994	PCB-1221		γ	g/L		41-		-	-	
501	LF02	Water	1994	PCB-1232	1	Ϊþ	g/L		41-		-	-	
502	LF02	Water	1994	PCB-1242	1	ι.μ	ġ/L		41-		-	-	1.4
503	LF02	Water	1994	PCB-1248	1	1.µ	g/L		4  -			I-	
504	LFO2	Water	1994	PCB-1254	1	Ϊįμ	g/L		41-		-	-	
505	LF02	Water	1994	PCB-1260	1	Iμ	9/L :	4	41-		-	I	
506	LF02	Water	1994	Pentachlorophenol	50	) i ju	9/L	4	41-			<b> _</b>	
507	LF02	Water	1994	Phenamihrene	10	Ū·u	ġ٨.		4-		-		
508	LF02	Water	1994	Phenol	10	μÌ	ol.		4 -			-	······································
509	LF02	Water	1994	Potasskum	N/A	μ	o/L		4	4	7.10E+03	1.14E+04	02M01
510	LF02	Water	1994	Pyrene	10	Σú	a/L		41-	·	_		
511	LF02	Water	1994	Silver	1	1:0	0/L	1	8 -				
512	LFD2	Water	1994	Sodwm	N/A	Linni Li	a/L		6	Б	8 39F+03	2 24E+04	02M01
513	LF02	Water	1994	Sulfate	N/A	 	<u>6</u> 1		<u>,</u>		300	3.90E+04	021401
514	1 F02	Water	1994	Tetrachioroethene (PCE)	0 5	5.5	eA !		RI		-	-	
515	L F07	Water	1994	Tin		1.14	<u>е —                                   </u>		8				
516	1 507	Water	1994	Tohiege	·····		<b>₹</b>				2.2		02402
517		Water	1004	Total dissolved rolids	N/A		8		4		1 505-05	3 105-05	021401
1.11	L 02	Moter	1004			5.11	-		1-			0.102.103	V2/HU1
610		Nater	1004	Toron 1.2 dishigrouthers			¥				-		
620	1502	Neller	1974		~~~					'			V2MK/1
320		Water	1004		0.0	2 H			- <u>-</u>				
221		vvaler	1004		U.3		φ.					}	
324		vvaler ·	19994			<u>. P</u>	<u>v</u> . j		:		-	-	
323	UF02	TYBICT	1004	V (P HENDUTI)	H/A	Ľ	<u> </u>			4	11	35.2	V2MU8
		THE C	1334		· U.:	<u>, 1</u>	<u> </u>				-		
	LFU2	vvaler	1994		NIA	<u>ių</u> : Lietos			<u>-</u>	- 4	76	126	020005
		VValler	1994	(I SCHICE) (THURSEN		<u> 11</u>	<u>₩</u> ∟   #		1)=		-	<u> </u>	<b>-</b>
		VVEIQI :	1394		······	<u>. P</u>			<u>}</u>		-		
525	1-03-40	vvaler	1994	1,1,2,2-1 erachiorosmane		Πh	<u>w</u> .		<u> </u>			-	
329	LF03-80	VVFIOT	1994			P		1	!!			<b>-</b>	
930	L-03-50	vvater	1994	1, 1-LICTIOROETDANS	N/A	<u>. (P</u>	<u>9/L</u> ↓	1	<u></u>	<u> </u>	33	33	BUMLE
531	LF03-sp	Water	1994	1,1-Dichloroethene		510	<u>9/L</u> ]		11-		-	<b> </b>	
532	LF03-sp	Water	1994	1,2,4-Trichkorobenzene	500	<u>, h</u>	ovL ;		21-		-	-	
1 233	LF03-sp	vvater	1994	1.Z-Uichiorobenzene	500	i h	<u>el</u> i	4	<u>•</u>	2	24	25	83M0600
534	LF03-60	Water	1994	1.2-Dichloroethane		5 (µ	<u>ا او</u>		!!-				
535	LF03-sp	Water	1994	1.2-Dichloropropane		<u>P</u>	<u>er</u>		11-		-	-	
536	LF03-sp	Water	1994	1.3-Dichlorobenzene	500	<u> ip</u>	<b>%</b>		<b>4</b> -		-	-	
537	LF03-sp	Water	1994	1,4-Dichlorobenzene	500	) ju	<u>91</u>		•	2	64	82	83M08
538	LF03-sp	Water	1994	2.4.5-Trichlorophenol	500	) JU	<b>q/L</b> ]		2 -	]		<u> -</u>	
539	LF03-sp	Water	1994	2.4.5-Trichlorophenol	500	hΒ	g/L ]	-	2		-	-	
540]	LF03-sp	Water	1994	2,4-Dichlorophenol	500	۱ <u>۱</u>	<b>91</b>		2 -		-	-	
541	LF03-sp	Water	1994;	2.4-Dimethylphenol	500	λip	<b>۵</b> ۲ i		2 -		-	-	
542	LF03-sp	Water	1994	2.4-Dinkrophenoi	2.50E+03	Siμ	g/L		2 -	]	_	-	
543	LF03-sp	Water	1994	2.4-Dinitrotoluene	2.50E+03	Iμ	g/L	4	4	1	-	-	
544	LF03-sp	Water	1994	2.6-Dimitrotoluene	500	) (µ	0 <sup>4</sup> .		2 -		-	-	
645	LF03-sp	Water	1994	2-Chloroethylvinyl ether	20	); ui	φ/L	1	1-			-	

		,		<u> </u>	T F	Ē	<u> </u>	<u> </u>		·····	
<b></b>	<u> </u>	<u></u>	<u>v</u>		I	<u> </u>			L	L	<u> </u>
Ι.	l										
	Source Area	Media	Date	Analyte	_ <u></u>	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
546	LF03-sp	Water	1994	2-Chioronaohthalene	500	µg/L		!`	-	-	l
547	LF03-sp	Water	1994	2-Chiorophenol	500	ug/L		2 -	<b></b>	-	_
548	LF03-sp	Water	1994	2-Methylnaphinalene	500	µg/L	2		-	]-	
549	LF03-\$P	Water	1994	2-Methylphenol	500	µg/L	2	! -	-	-	•
550	LF03-sp	Water	1994	2-Natroandine	2 50E-03	vg/L		2 -		I-	•
551	LF03-10	Water	1994	2-Narophenol	500	ua/L	7	2 -		·····	
552	1 F03-50	Water	1994	3 3'-Dichlorobenzidine	1 00E+03	unA		· _ · · ·			······
651	1.603-80	Water	1994	3-Mitmanine	2 505+03	107/		);	1		<u></u>
	L 607 ep	Mater	+004		2.002.00	M44	~~~~	. : — 		ļ <del>.</del>	
334	CFU3-ap	VVAICT	1994	4 4 - 000	<u>.</u>	hdvr			-	1	
222	LF03-SP	vvaler	1994	44-DUE	<u>V1</u>	µg/L	1	· <del>-</del>	-		
556	LF03-sp	Water	1994	4.4'-DDT	0.1	hð\r	1	i	-	<u>(</u>	
\$57	LF03-50	Water	1994	4-Bromophenyl-phenylether	500	µ9/L	. 2	!!-	i <b>-</b>	<b>-</b>	
558	LF03-sp	Water	1994	4-Chioro-3-methylphenol	1.00E+03	µg/L	. 2	-	-	[-	
559	LF03-sp	Water	1994	4-Chloroaniune	1.00E+03	µq/L	2	! _	-	-	
560	1 F03-so	Water	1994	4-Chiorophenvi-phenviciher	500	ud/i	2	1-			
561	1 F03-so	Water	1994	4-Methylohenol	N/A	1400			4 806-03	6 205 +03	836408
1	L 503-40	Water	1004	4 Alamanina	1 005+01	100/	· · · · · · · · · · · · · · · · · · ·	<u></u>		-	
1002	1.502.00	- FELGT	1334		3 500-00	HUYL .		÷		······	
	LFU3-50	vvaler	1994		2.306+03	HG/L		Į			
564	LF03-SP	vvater	1994	a-6HC	0.05.	hövr	<u> </u>	;= ;	-	-	
565	LF03-sp	Water	1994	a-Chlordane	0.05	µg/L	: 1	-	-	I	
566	LF03-sp	Water	1994	Acenaphthene	500	µg/L	2	!	**	-	
567	LF03-\$0	Water	1994	Acenaphthylene	500	µg/L	2	i-			····
568	LF03-sp	Water	1994	Aldrin	0.05	µq/L	1	-			
569	LF03-\$0	Water	1994	Aluminum	N/A	ua/1	2	2	211	333	83M08
570	LE03-40	Water	1004	Anthracene	500	บณ์เ	· · · · · · · · · · · · · · · · · · ·				
874	1.503.40	Mater	1004	Artenic	M/4	10/			-	-	#3140P
	LF03-sp	1Alata-	1004		A 06	<u></u>	·····	ļ	3.9		COMPO
3/2	LF03-sp	vvaler	1994		0.05.	har			-	-	
\$73	LF03-sp	water	1994	Baunu	N/A	HQ/L	2	<b></b>	375	405	83M08F
574	LF03-sp	Water	1994	Benzene	N/A	HQ/L	1	1	20	20	83M06
575	LF03-sp :	Water	1994	Benzo(a)anthracene	500	µg/L	2	<u> </u>	-	-	
576	LF03-sp	Water	1994	Benzo(#)pyrené	500	µg/L	2	-	-	-	
577	LF03-sp	Water	1994	Benzo(b)fluoranthene	500	HQ/L	2	-	_	-	
578	LF03-40	Water	1994	Benzo(g.h.i)perviene	500	Harl.	2	-		-	
579	LF03-sp	Water	1994.	Benzo(k)fluoranthene	500	ua/l	2	1-			
500	1603-40	Water	1994	Renzok acid	2 50E+03	uni		1			
	1 503 40	18/01c-	1004		1 005+03	and a					
- 241	LFUJ-ap 1	VYALCI	1334		1.002 -001	POL	<b>4</b>				
		vvater	1994	Beryihum		hour		-		-	
583	LF03-sp	waler	1994	Bis(2-Chloroethoxy)methane	500	HOL.	2	<b>-</b>	-	-	······································
584	LF03-SP	Water	1994	Bis(2-chloroethyl)ether	500	HQ/L	2	-		-	
585	LF03-sp	Water	1994	Bis(2-chloroisopropyl)ether	500	HOV.	2	i-	-	-	
586	LF03-sp	Water	1994	Bis(2-ethylhexyl)phthalale	500,	HQ/L	2	-	-	-	
587	LF03-sp	Water	1994	Bromide	100	µ <b>g/L</b>	1	-	-	-	
588	LF03-sp	Water	1994	Bromodichloromsthane	10	HQ/L	1	1-	-	_	
589	LF03-sp	Water	1994	Bromoform	10,	HQ/L	1			-	
590	LF03-50	Water	1994	Bromomethane	20	.∞ar.⊒	1	† <u> </u>			
501	I FO3-en	Water	1004	Butvibenzytohthalste	500	10/		+		-	
502	FO2.so	Water	1004	Cadmium	1.	UN/			1 0.	10	#1440#
	(E02	taleter	1004		N/A	1007	<b></b>		5 245-07	4 105-14	\$340AF
Les	LF US-ap I		1334		200			4	7.0-5-700	0.705-00	~~~~~
1984	LFU3-#P	TYRICI	1994		500	MAR		ļ <b>-</b>			······································
595	LP03-60	vvaler	1994		5	HOL .	1	<b>!</b>	_	-	
596	LF03-sp	Water	1994	Chloride	,N/A	Poll	1	ļt	9.90E+04	9.90E+04	53MC8
597	LF03-sp	Water	1994	Chlorobenzene	10	µg/L	2	<u> -</u>	-	-	
598	LF03-sp	Water :	1994	Chioroethane	10	MO/L	1	-		-	
599	LF03-10	Water :	1994	Chloroform	10	HQL.	1	i-	<b>.</b>	-	
600	LF03-#0	Water	1994	Chioromethane	10	µg/L	1	<b> _</b>	-	_	
601	LF03-\$0	Water	1994	Chromium	NA	ua/L	2	2	20.7	26	83MOBF
603	1 F03.40	Weier	1994	Chrysene	500	ມດກໍ		1		-	
1	1 603.00	Water	1004	Cie. 1 3 dichloroprogene	E.	1995 1997		<u> </u>			
	LE02 4-		1001		DI/A			<u> </u>	-		+ 1+404C
604	LPUS-SP	TANT	1554			μονι	2	2	20.2	1/2	4-3MK/61*
605	LF03-40	vvater	1994	а-внс 	0.05	HOL	1			-	
608	LF03-sp	Water	1994	D-n-butyiphthalate	500	µq/L	2	<u> </u>	-	-	
607	LF03-sp	Water	1994	Di-n-octylphthalate	500;	µg/L	2		-	-	
608	LF03-\$0	Water	1994	Dibenz(a,h)anthracene	500	µo/L	2	-	-	<b></b>	
809	LF03-sp	Water	1994	Dibenzofuran	500	HO/L	2	-	-	-	
610	LF03-SD	Water	1994	Dibromochloromethane	10:	Joyl	1	[]	-	-	
1111	LF03-20	Water	1994	Dichlorodifluoramethane	N/A	ua/i	1	1	650	650	834405
1 1 1 1	E FOSten	Water	1994	Diakton	01	ual		<u></u>			
1	L 603	Water	1004	Netto dable	5.1 ·	104 C					
1414	U-V-SP i	*****	1204		3003	and the second	- 4	ו – ו	-	-	

-

FINAL

-

	<u> </u>	6	<u> </u>	D	Ē	<u></u> F	<u> </u>	<u>н</u>	1	J	×
									:		ì
1	Source Area	Media	Date	Analyte	DL	Units	#Samples	*Detect	Min Conc	Max Conc	Max Location
614	LF03-sp	Water	1994	Ormethylphthalate	500	µg/L	Z		i		I
615	LF03-Sp	VVater	1994		0.05	ug/L	1		-	. <u>-</u>	······
417	LF03 co	Water	1004		0.1		1		[	<u> -</u>	
618	LF03-sp	Water	+994	Endosaler suigite	0 1	ug/L	<u> </u>		· _		
619	LF03-s0	Water	1994	Endric aldehvde	0 1	100/L					+
620	LF03-sp	Water	1994	Ethylpenzene	N/A	10/1	·	1	34	38	83MO8
621	LF03-so	Water	1994	Fluoranthene	500	uq/L		· · · · · · · · · · · · · · · · · · ·	-	t	
622	LF03-sp	Water	1994	Fluorene	500	ug/L	2	-		/  -	l
623	LF03-sp	Water	1994	g-BHC	0.05	µg/L	1		<b>_</b>	†	+ <u>-</u>
624	LF03-sp	Water	1994	g-Chiordane	0.05	µg/L	1	1		-	
825	LF03-sp	Water	1994	Heptachlor	0.05	ug/L	1	i <b>—</b>	-	ļ <b>.</b>	
626	LF03-sp	Water	1994	Heptachlor epoxide	0.05	µg/L	t	:_	-	-	
627	LF03-sp	Water	1994	Hexachlorobenzene	500	µg/L	2	-	**	[-	
528	LF03-sp	Water	1994	Hexachlorobulagiene	500	µg∕L	2	-	-	-	
629	£F03-\$p	Water	1994	Hexachlorocyclopentadiene	500	µg/L	2	I	-	<u> -</u>	
630	LF03-sp	Water	1994	Hexachioroethane	500	µq∕L	2	i <b></b>		-	
631	LF03-sp	Water	1994	Indeno(1,2,3-cd)pyrene	500	hðv	2	-	-	<u> </u>	
632	LF03-sp	Water	1994	tron	N/A	µg/L	2	2	5.45E+05	5.68E+05	83M08F
633	LF03-sp	Water	1994	Isophorone	500	hð/r	2		-		
634	LF03-50	VValer	1994	Lead	N/A	µg/L	2	2	11	18.6	83MOBF
635	LF03-50	water	1994	M,p-xylene	N/A	<u>µg/L</u>	1	<u> </u> ]	44	44	83M08
636	LF03-5P	vvater	1994	Magnesium		µg/L	2	2	4.99E+04	5.27E+04	S3MOEF
	LFU3-50	VV2LCT	1994	Mangañese	N/A A #	HQ/L	Z	2	7.428+03	6.29E+03	GJMU8F
810	LFUS-sp	Water	1004		U 3	ug/L				-	#24.4/10
640	( 503-sp	Water	1004		500				230	230	034608
841	LF03-sp	Water	1004	n-Nikosoticheowiamine (1)	500	- HAL	2				
642	(FOLAD	Water	1994	Nachthalene	500	un/L					
643	LF03-so	Water	1994	Nickel	N/A	ud/L	1	1	27.2	27.2	834404
644	LF03-50	Water	1994	Nitrobenzene	500	:00/1	2	-		-	
645	LF03- so	Water	1994	0-Xvient	N/A	HO/L	1	1	47	47	834408
646	LF03-sp	Water	1994	PC8-1016	1	ING/L	. 1	I-	-	I_	
647	LF03-10	Water	1994	PCB-1221	2	ug/L	; 1	-	-	-	
648	LF03-sp	Water	1994	PC8-1232	1	:µg/L	1	I-	-	j <u> </u>	
649	LF03-sp	Water	1994	PC8-1242	1	;µg/L	1	<b> _</b>	-	-	
650	LF03-sp	Water	1994	PCB-1248	1	,µg/t_	1		-	-	
651	LF03-sp	Water	1994	PCB-1254	1	JUSIL	1	!-	-	-	
652	LF03-sp	Water	1994	PCB-1260	1	jug/L	1	-	-	-	
653	LF03-sp	Water	1994	Pentachlorophenol	2.50E+03	ind\r	2		=	-	·
654	LF03-sp	Water	1994	Phenanthrene	500	ug/L	2	-	-	-	
655	LF03-sp	Water	1994	Phenol	500	10.	2	1	250	250	831408
656	LF03-sp	Water	1994	Potasskum	N/A	INDIC	2	2	4.66E+04	4.85E+04	83MO6F
93/	LF03-\$P	water	1994	Pyrene Situa	500		2				
030	LFU3-sp	Vyaler	1994	Skyer	1 N//A	HONT.	<u> </u>	-	-	-	031404E
1450	LFW-SP	rvaler Woler	1994	Suidata	100	: HAL		<u> </u>	9.130404	7.00E+04	ojmuor
641	E Clen	Water	1004	Terrachimethene (PCS)	N/A	1995	······			- 63	
662	LF03-m	Water	1994	Toluene	N/A	ud/L			460	480	83408
663	LF03-so	Water	1994	Total dissolved solids	N/A	100/L			4.60E+0F	4.50E+04	63404
564	LF03-so	Water	1994	Toxaghene	5		1	_ · · · ·	-	-	
665	LF03-sp	Water	1994	Trans-1.2-dichloroethene	10	µg1	1		-	-	·i
666	LF03-sp	Water	1994	Trans-1.3-dichloropropene	5	;µg/L	1		-		
667	LF03-sp	Water	1994	Trichloroethene (TCE)	N/A	μς/L	1	1	150	150	BOMCB
668	LF03-sp	Water	1994	Trichlorofluoromethane	N/A	UQ/L	1	1	13	13	83M08
669	LF03-sp	Water	1994	Vinyi chionde	N/A	µg/L	1	1	17	17	63M06
670	LF03-sp	Water	1994	Zinc	N/A	HO/L	2	2	172	249	83M06
671	LF03/FT09	Water	1994	(Technical) chiordane	<u> </u>	HQ1	22	-	-	-	
672	LF03/FT09	Water	1994	1,1,1-Trichloroethane	1	ug/L	22	1	1.5	1.5	03M10
673	LF03/FT09	Water	1994	1,1,2,2-Tetrachloroethane	!	HQ1	22		-	-	
674	LF03/FT09	Water	1994	1,1,2-Trichloroethane	0.5	ind.	22		-	-	
675	LF03/FT09	Water	1994	1.1-Dichloroethane		HQ.	22	1	33	33	USMOB
6/6	LF03/F109	vvater	1994		0.5		ZZ		-		
874	LEGARTON	VY SIGT	1994		10		35	<b></b>	-	-	
878	ENS/FTOD	Water	1004			IN C	79	2	24	<u>2</u> 2	USMUBUU
100	LEAVETON	Water	1004	1.2-Gradien oou en ee		i Piret		<u> </u>			
1001	1 F03/FT09	Water	1904	1.3-Dichlombenzene	10	International In	70	<b>_</b>	-	_	
						a property for		,	_	_	

	A	Б	L C	D	E	F	G	н	1	1 7	ĸ
			<b></b> ,	······································	4,					h	·····
1 1	Source Area	Media	مردر	Anavie	51	Líoils	#Samples	#Detect	Min Cone	May Conc	Max Location
682	L COTVETOO	Matatar	1004	1.4. Displorabe07808		ue/l	70	2	64		ASHOP
802		VVd(S:	1004		10	ugre mgre	10				VJINUO
003	LF03/FT09	YVALC:	1994			<u></u>			· · ·		
004	LF03FT09	2V815:	1004		10	POP 5					
600	LFUSIFIUS	vvaler	1994			μον.	33		! <b>-</b>		
000	LF03/F109	vvater	:994		10	μανι	Jo		· —		
687	LF03/F109	VValer	1994	2.4-Dinstophenoi	50	HOL	35	-		· <b>-</b>	
688	LF03/F109	Water	1994		50	hdir			1	i —	
689	LF03/FT09	Water	1994	2,6-Dinifrotoluene	10	<u>µq/L</u>	35	<b>-</b>	i <b>-</b>	i	
690	LF03/FT09	Water	1994	2-Chloroethylvinyl ether	Z	<u>µ</u> g∕L	22	_	-	1-	1
691	LF03/FT09	Water	1994	2-Chioronaphthalene	10	hâvr	35	-	<u> </u>	1	/
692	LF03/FT09	Water	1994	2-Chlorophenol	10	_µg/L	35	-	i –	i	
693	LF03/FT09	Water	1994	2-Methylnaphthalene	10	µg/L	35		-	-	
694	LF03/FT09	Water	1994	2-Methylphenol	10	µ <b>g</b> /L	35	_	-	[	
695	LF03/FT09	Water	1994	2-Nitroaniline	50	µg/L	35	-	-	-	f
696	LF03/FT09	Water	1994	2-Nilrophenol	10	µg/L	35		-	-	
697	LF03/FT09	Water	1994	3 3-Dichloropenzidine	20	µg/L	35	-	-	-	
698	LF03/FT09	Water	1994	3-Naroanime	50	Hg/L	35	-	-	-	
699	LF03/FT09	Water	1994	4 4'-000	0.1	ug/L	22	f	0.21	0.21	03M01
700	LF03/FT09	Water	1994	4.4'-DDE	0.1	ug/L .	22	_	-		
701	LEO3/ET09	Water	1994	4.4-DDT	01	ud/L	22		_		
707	LF03/FT09	Water	1994	4-Bromophenyl-onenviether	10	u <b>o</b> /	35		<sup></sup>	i	
703	LE03/ET09	Water	1994	4-Chioro-3-methylphenol	20	uu/L	351	-			
704	1 F03/FT09	Water	1994	4-Chlorozouine	20	uali	34	_	_	-	
705	L FO3/FTOS	Water	1904	4-Chlomoheovi-oheoviether		<u>r₹</u> ; ua/i	361	_			······
706	LE03/6709	Water	1004	4.Methylonenol	10.	<u></u>	35	2	4 80E+03	6 200+03	03408
707	LF03/FT09	'IA/ater	1004		20	000	74	-	-	-	
708	LEON/ETOS	Mater	1004	4-Neroobenoi	50	HALL	36	_			
700	( E01/ET09	Avaler	1994	a.BHC	0.05	1000	22				
710	E03/ET09	Water	1004	a Chiordana	0.05		221			<u> </u>	
711	LEDT/ETD9	Matar	1904		10	und '	35)			<u> </u>	
747	LE03/6T09	Mater	1904		10		35	_		<u> </u>	
112	LF03/F109	Minter	1004	Aldren	0.05	HAPL .	301				·····
744	LF03/F103	VYaler	1004	Aluque	11	ug/L	22	~ ~		2 705-04	021410
7.6		VValei	1004	Actherage	10	<u>μφις :</u>	201	- 27	211	3.785	
712	LF03/F1V9	Water	1994	Antennav		<u></u>	79		-		
100	LEONETOS	vvater	1994	Amonio		HANT -		-		-	-
717		water	1994		ېد د مد د	HQ/L	37	30	4.3	302	USM12F
718	LFU3/F109	vvater	1994		0.05	Havr (	22			-	A
119	LF03/F109	vvater .	1994			HOVE	34		120	1.07E+03	
1 120	LEONETOO	vvaler	1994		10	HOL I	24	•	1.7	<u> </u>	USMUB
141	LFUJ/FTU9	I YVBIET	1994:		10	HOL I	30				
722	LFUSFTUS	vvaler	1334	Denzo(a)pyrane		POL .	30				
12	LFUSPETUS	water	1994		10	HON I	351	-			
1/24	LFU3/FT09	vvater	1994:	oenzo(g.n.)perviene	101	HANT :	35	-	-		
125		vvaler	1994		101		35	-		<b>-</b>	
726	LF03/F109	vvaler	1994	Senzoic acid	50	HQVI.	35	-			·····
127	LF03/FT09	vvater	1994		20	HQ/L	35	<b>-</b>	-	<u> </u>	
/ 28	LF03/FT09	vvater	1994		<u>l</u> i	HOVL I	43	1	1.5	1.6	USIMITZE
729	LF03/FT09	vvater	1994	Dis(2-chloroethoxy)methane	10	ug/L	35	-			
730	LF03/FT09	water	1994		10	HQVL.	35	-	<b>-</b>	[ <b>-</b> ]	
731	LFOSFT09	vvaler	1994	SIN Z-Chlorotsopropyleifver	10	HOVL	35	-			
/32	LF03/F109	water	1994		10		35	2	3.5	61	USMUS
123	LPUSE 109	VVIII .	1994		1001	HOVL	22	4	120	460	WW01
134	LF0.VF109	weter	1994	Stoniodichiorometrians		MUL	22			<b></b>	
132	LF03#109	VVIII (M	1994			HUL ]	Z2	-			
/36	LF03/F109	AASIAL.	1994		2	HON !	22	-	-	-	
137	LFUSPIUS	YVELC	1994		10	ugvi i	35	-		-	
738	LFOWFTD9	VVMC	1994		1!	HUL I	40	4	1.1	11.7	USHOZ
739	LF03/FT09	water	1994		IN/A	hđu	44		3.32E+04	6.36E+05	USMOSF
740	LF03/FT09	water	1994	Carbazola	10	1101	35	-	-		
741	LF03/F709	vvater	1994	Carpon teirachionde	0.51	hður i	22	-	-	-	
742	LF03/FT09	Water	1994	Chionde	N/A	NO/L	22	22	1,00E+03	\$.90E+04	USIM06
743	LF03/FT09	Water	1994	Chlorobenzene	tj	HQ/L	44	-	-	÷	
1744	LF03/FT09	Water	1994	Chloroethane	1	Mar !	22	-	-	-	
745	LF03/FT09	Water	1994	Chioroform	1	197. j	22(	-	-	-	
746	LF03/FT09	Water	1994	Chloromethane	1	191	22	-	-	-	
747	LF03/FT09	Water	1994	Chromum	٩ ا	har	29	13	10.1	92	03M12F
748	LF03/FT09	Water	1994	Chrysene	10	HQ/L	35	-	-		
749	LF03/FT09	Water	1994	Cis-1.3-dichloropropene	0.5	uon/Li	22	- 1	- i		

A.11

	A	В	[c	D	<u>E</u>	F	G	н	1	[	ĸ
1.				Analida	-	Index	#C				
	Source Area	Media	Date	Analyte	<u>UL</u>	Unas	#Samples	SUBLECT	Min Conc	Max Conc	Max Location
750	LF03/F109	Vyater	1994	Cobar	ا <del>ب</del>	UQ/L			110	267	031410
757	LEONETOS	14/ster	1004		0.15	09/0	22	<u> </u>			
753	LEO3/ETO9	Water	1994	Di-n-butvionihalate	10	ug/L	35	5	2.8	5.2	03Mm02
754	LF03/FT09	Water	1994	Di-n-octylphthalate	10	ug/L	35	-	-	1	
755	LF03/FT09 .	Water	1994	Dibenzia.hianthracene	10	µ9/L	35	· · · · · · · · · · · · · · · · · · ·	_	i-	· · · · · · · · · · · · · · · · · · ·
756	LF03/FT09	Water	:994	Dibenzofuran	10	µg/L	35	-	-		I
757	LF03/FT09	Water	1994	Dibromochloromethane	1	µq/L	22	!-			
758	LF03/FT09	Water	1994	Dichlorodifluorometnane	1	µg/L	22	5	1.2	650	03M08
759	LF03/FT09	Water	1994	Dieldrin	01	µg/L	22	;		-	
760	LF03/FT09	Water	1994	Diethylphthalaite	10	µg/L		1	1.1	1.1	03Mm02
761	LF03/FT09	Water	1994	Dimethylphthalate	10	µg/L	35	<u> </u>	- 	i <b>-</b>	
782	LF03/FT09	Water	1994		0.05	<u>µq/L</u>		ļ=	• <b></b>	<b>_</b>	·
763	LF03/F T09	Water	1994			µg/L		·		·	
764	LF03/FT09	vyaler	1994			ug/L	27			-	
745	LEONETOS	Water	1994		01	100/J	22				
767	LE03/ET09	Water	1994	Fibvibenzene	1	uo/L	22	3	1.5	38	03M08
768	LF03/FT09	Water	1994	Fluorantiene	10	ug/L	35	-		-	
769	LF03/FT09	Water	1994	Fluorene	10	μ <b>g/L</b>	35	i	-	-	
770	LF03/FT09	Water	1994	g-BHC	0.05	µg/L	22	-	-	-	
771	LF03/FT09	Water	1994	g-Chlordane	0 05	µg/L	22	-	-	-	
772	LF03/FT09	Water	1994	Heptachlor	0 05	µg/L	22	-	-	-	
773	LF03/FT09	Water	1994	Heptachior epoxide	0.05	µ <b>g/</b> L	22	<b></b>	-	-	
774	LF03/FT09	Water	1994	Hexachiorobenzene	10	ug/L	35	-	-	-	
775	LF03/FT09	Water	1994	Hexachlorobutadiene	10	ug/L	35			<del>ات</del>	·
776	LF03/FT09	Water	1994	Hexachiorocyclopentadiene	19	µg/L	35	-	-		
777	LF03/F109	vvaler	1994	Hexachioroeinane	10	HO/L	30			·	
778	LF03/FT09	VValer	1994	Ingeno(1.2.3-co)pyrene	Al/A	1µ9/1.	30		1.826403	5 885405	MALACAE
780	1 E03/ET09	Water	1994	(sophorphe	110		35	47	1.046 103	J. 60E 703	
781	LE03/ETD9	Water	1994	Lead	1	La/L	34	20	3.4	61.9	03402
782	LF03/FT09	Water	1994	M p-xviene	1	uq/L	22	2	3.1	44	03MO8
783	LF03/FT09	Water	1994	Magnesium	N/A	µg/L	44	44	7.82E+03	5.37E+04	03401
784	LF03/FT09	Water	1994	Мапдалезе	N/A	μ <b>g/L</b>	44	44	295	8.29E+03	03MOBF
785	LF03/FT09	Water	1994	Methoxychior	0.5	ιμg/L	22	-	-	-	
786	LF03/FT09	Water	1994	Methylene chloride	1	µg/L	22	1	250	250	03M08
787	LF03/FT09	Water	1994	n-Nitroso-di-n-propylemine	10	109/2	35	-	-	-	
768	LF03/FT09	Water	1994	n-Nitrosodiphenylamine	10	IN/L	35	-		-	
789	LF03/FT09	Water	1994	Naphinatene	10	indur.	35	1	1.3	1.3	
790	LEUSVETOP	vvater	1994	Nickel	10	HOL.	36	9	23.5	<b>20.3</b>	0.514118
707	LF03/F109	Water	1994	A Yriege	····	i port.	20		1.8	47	011409
791	LEO3/ETO9	Water	1994	PC8-1016	······	udi	22		-	-	0,000
794	LF03/FT09	Water	1994	PCB-1221		100/L	22		_		
795	LF03/FT09	Water	1994	PCB-1232	1	iµg/L	22		_		
796	LF03/FT09	Waler	1994	PC8-1242	1	HQ/L	22	-	_	-	·····
797	LF03/FT09	Water	1994	PCB-1248	. 1	iug/L	22		-	-	
798	LF03/FT09	Water	1994	PCB-1254	1	Ing/L	22		-	-	
719	LF03/FT09	Water	1994	PC8-1260	1	ug/L	22	-	-	-	
800	LF03/FT09	Water	1994	Pentactviorophenol	50	ING/L	35	-	_		
801	LF03/FT09	Water	1994	Phenanthrene	10	491	35	<b>_</b>		-	
802	LF03/FT09	Water	1994	Phenol	10	HQ/L	35	1	250	250	03M08
603	LEUXETOS	VValer 16/stac	1994	Poressium		HONE -	23		5.042+03	4.836+04	UJWUGP
804	LEWARTING	I VVALUT	1994	· Clause	<u></u>		<u>دد</u>	<u> </u>	-		
805	EFUSIFIUS	Water	1004	Sodium	N/A	uo/i			5 335+02	9 885+04	TOMORE
107	1.503/5100	Water	1994	Sulfate	100	ud/L	22	21	3.80F+01	2.305+04	03Mm020
100	LF03/FT09	Water	1994	Tetrachioroethene (PCE)	0.5	ug/L	22		53	53	03MC8
809	LF03/FT09	Water	1994	Ţ'n		μο1.	22		-		
810	LF03/FT09	Water	1994	Toluene	1	μα/L	22	2	2.5	460	03M08
811	LF03/FT09	Water	1994	Total dissolved solida	N/A	HQ/L	22	22	1.80E+05	4.60E+06	031408
812	LF03/FT09	Water	1994	Toxaphene	5	indvr	22	-	-	-	
813	LF03/FT09	Water	1994	Trans-1,2-dichioroethene	1	µg/L	22	-		-	
814	LF03/FT09	Water	1994	Trans-1,3-dichloropropene	0.5	HQVE.	22	-	-	-	
115	LF03/FT09	Water	1994	Trichlergethene (TCE)	0.5	pg/L	22	6	0.64	150	O3MO8
816	LF03/FT09	Water	1994	Trichlorofluoromethans		µg/L	22	1	13	t3	03M08
017	LF03/FT09	vvater	1994	Venadium	1	Jug/L	25		10.0	43	U3M12

<b></b>	A	<u>} B</u>	C	0	E	F.	G	н	<u> </u>		ĸ
1					-		_	_			1
1	Source Area	Media	Date	Analyte		Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
618	LF03/FT09	'Avater	994	Vinyl chioride	0.5	µg/L		5	0.54	17	03M08
819	LF03/FT09	water	1994	Zes	2	µg/L	25	21	20.4	1.03E+03	03M02
820	LF03/FT09	Surface Soil	1992	0.0.0-Triethyl phosphorothioate	660	µg/kg	9	) - <u></u>	<u></u>	1-	
821	LF03/FT09	Surface Soil	1992	0 0-Diethyl 0-2-pyrazinyl phosphorolfiloa	660	µg/kg	9		<u>!                                    </u>	-	
822	LF03/FT09	Surface Soli	1985	* 1 1-Trichloroethane	50	µg/\g	10	1	240	240	103M02
823	LF03/FT09	Surface Soil	1988	1 1-Dichloroethylene	100	µg/kg	10	1	300	300	103M02
824	LF03/FT09	Surface Soil	1992	1.2.4-Trichlorobenzene	560	_µ9/kg	9	-	-	· <b>-</b>	
825	LF03/FT09	Surface Soil	1992	1 2-Dichloropenzene	660	µg/kg	9	-		:-	1
626	LF03/FT09	Surface Soil	1992	1,3-Dichlorobenzene	660	µg/kg	9	li—	-	-	1
827	LF03/FT09	Surface Soil	1992	1 4-Dichlorobenzene	660	µg/kg	9	). <u></u>	-	!-	1
828	LF03/FT09	Surface Soil	1992	1,4-Naphthoguinone	650	µg/kg	9		i—	I_	1
829	LF03/FT09	Surface Soil	1992	1-Naphthylamine	660	µg/kg	9		1-	i <b>_</b>	1
630	LF03/FT09	Surface Soil	1992	2.3.4.6-Tetrachloropheno!	660	µa/kg	: 9	+i <b></b>	-	-	1
831	LF03/FT09	Surface Soil	1992	2.4.5-Trichlorophenol	660	ug/kg	9		; <b>—</b>	·	
832	1F03/FT09	Surface Sort	1992	2.4.6-Trichlorophenol	660	µq/kg	9	i		I—	1
833	LF03/FT09	Surface Soil	1992	2.4-Dichiorophenol	660	ua/ka	9			I_	
634	LF03/FT09	Surface Soil	1992	2 4-Dimethylohenol	560	ua/ka	9	1-	-	_	<u>.</u>
835	LF03/FT09	Surface Sod	1992	2 4-Dingrophenol	3 30E+03	uo/ka	9	=	-		<u>,</u>
836	E03/ET09	Surface Soil	1992	7 4-Dimitrotoluene	660	uo/ka	9	÷			
837	LEONETON	Surface Sort	1992	2.6-Dichlorooftenal	660	uo/ko					·····
838	LEOVETOS	Surface Soil	1992	2.6-Districtionene	660	un/kn					
A10	LE03/ET09	Surface Soil	1907	7-Acetylaminofluorene	550	Ua/ka	·				
440	Ent/ETOS	Surface Sol	1002	2-Active antihalene	660	up/kg			E		
1 144	LEONETOO	Curface Coll	1002	2-Chicroshandi	600	HANNA Nalpa			[	E	<u> </u>
	LF03/FT09	Surface Soil	1992		660	uoleg.			4 105-03	4 105+02	0364500
	1603/5103	Surface Soll	1001			MANA			4.106700	4.10ETU3	0331303
	LF03/FT09	Surface Son	1002	2-Mennyiphenoi	000	ug/kg			j=}		
	LEONETOS	Surface Soul	1002		2 205+02	HYTNY				······································	
1	LF03/F109	Surface Soll	1000		3.302-03	UWAY				,	
	LFUSFIUS	SURACE SON	1335		4 305 + 03	have		·		<b>_</b>	
	LF03/F109	Surface Soll	1992	3.3-Dichorobenzioline	1.305403	DONG	· · ·			F	
546	LF03/F109	Surface Soil	1992	3.3-Dimenyoenzidine	000	NUNKO		·		ļ	
649	LF03/FT09	Surface Sol	1992	3-Meinyicholantinene	000	HQ/KQ	8			·	
850	LF03/F109	Surrace Soil	1992	3-Nitroanilline	3.30E+03	HO/Kg	i 9			, <b>**</b>	ł •
351	LF03/F109	Surface Soil	1992	4.6-Dinitro-2-methylphenol	3.30E+03	hðvæð	9	-	=	-	
852	LF03/FT09	Surface Soil	1992	4-Aminobiphenyi	560	hā/kā		[-		<b></b>	
853	LF03/FT09	Surface Soil	1992	4-Bromophenylohenyl ether	660	hð\kð			-	-	
854	LF03/FT09	Surface Soil	1992	4-Chloro-3-methylphenol	1.30E+03	HOURS .	9	i <del></del>	( <b>-</b> )	l <b>-</b> !	
855	LF03/FT09	Surface Soil	1992	4-Chloroaniline	1 30€+03	indyred	9	<u>i –                                    </u>		-	
856	LF03/FT09	Surface Soil	1992	4-Chiorophenylphenyl ather	660	µg/kg	9	!-	-		
857	LF03/FT09	Surface Soil	1989	4-Methylphanol	30	uarka	9	2	30	600	03M08
858	LF03/FT09	Surface Sod	1992	4-Methylphenol	660	ug/kg	9	-	-	-	
859	LF03/FT09	Surface Soil	1992	4-Nitroaniline	3 30E+03	µg/kg	9	-	-	-	
660	LF03/FT09	Surface Soil	1992	4-Nitrophenal	3.30E+03	µQ/kg	9	-	-	-	
861	LF03/FT09	Surface Soil	1992	4-Nilroquincline-1-oxide	660	ug/kg	9	(I	- 1	-	
862	LF03/FT09	Surface Sol	1992	5-Nitro-o-toluidine	660	µg/kg	9	-	-	-	
863	LF03/FT09	Surface Soil	1992	7,12-Dimethylbenz(a)anthracene	660	µg/kg	9	-		i	
864	LF03/FT09	Surface Soil	1988	Acenaphthene	10	µg/kg	10	1	110	110	03M01
865	LF03/FT09	Surface Soit	1992	Acenaphthene	660	ug/kg	; 9	}		-	
866	LF03/FT09	Surface Soil	1958	Acenaphthylene	10	µg/kg	10	-	-	-	
867	LF03/F709	Surface Soil	1992	Acenaphthylene	660	µg/kg	9	<b>I</b> _	- 1	-	
868	LF03/FT09	Surface Soil	1992	Acetophenone	660	ugika	9	-	-	-	
469	LF03/FT09	Surface Soil	1992	Alphe,alpha-dimethylphenethylamine	660	'µg/kg	9	1-		-	
870	LF03/FT09	Surface Soil	1988	:Aພານທານທາ	5.07E+06	µg/kg	11	2	5.16E+06	8.27E+06	03M03
871	LF03/FT09	Surface Sol	1992	Anline	660	ua/ka	9	-	-	-	
872	LF03/FT09	Surface Soll	1992	Antivacene	660	µo/ko	9	(		<b>-</b>	
873	L F03/FT09	Surface Soil	1992	Aramite	660	uc/kg	9	-			
874	LF03/FT09	Surface Soll	1968	Arsenic	6.39E+03	uana	9	1	1.23E+04	1.23E+04	03405
175	LF03/FT09	Surface Soll	1988	Barlum	5.33E+G4	uaña	11	2	8.76F+04	9.07F+04	03M01
876	L F03/FT09	Surface Sol	1984	Benzene	44	Luo/ko	10		65	65	03TP01
177	LE03/ET09	Surface Sold	1988	Benzolalanthracane	10	unito	10		1.505+02	1.505+01	031401
878	LF03/FT09	Surface Soil	1992	Benzo(a)anthracene	660	uo/u	9			181	0157505
170	LE03/ET09	Surface Sol	1988	Benzo(a)ovrece	9	uo/ka	10		1 705+03	1 705+01	03401
880	LEO3/ETO9	Surface Seit	1997	Henzo(a)ovrece	860	uofen	0		70 1	200	0357505
1 201	LENVETOD	Surface Coll	1000	Bearo(bifu)/20/bane	30	HAR DE			2 105403	2 105403	munt
1	EN3/ETOP	Surface Coll	1007	Banzor bill was thene		- Pigraw		;	6. IVE TUJ	2.105703	M3SVEAF
	LEGA/ETOO	Surface Cold	1000	Benzola h ()Gendere	400	Marken .		<u>├</u>	1 400-00	1 405-00	0001303
	L-03/FT09	Curface Cold	1001	Benev(grijporyene		PWTW .	······································	·		1.702703	MACYCOL
	LFU3F 199	Sunace Sol	1992		000	1 Ward	¥		200	200	0331303
1.005	LF03/F109	SULACE SOIL	1992	Benzo(K)Nugraninené	660	parta	9	. 3	59.6	190	0051305

L	Α	8	C	0	Ε	<u>i f</u>	G	H	Î	<b></b>	ĸ
<u>۱</u>											
3	Source Area	Media	Date	Analyte		Units	#Samples	*Delect	Min Cone	Max Conc	Max Location
686	LF03/FT09	Surface Soil	1992	Benzyi alcohol	: 30E+03	µg/kg		' <b>-</b>	-	-	
867	LF03/FT09	Surface Soil	1968	Servilium	2.44E+03	µg/kg	11	5	320	1.26E+03	03M01
888	LF03/FT09	Surface Soli	1992	Bisi2-chioroethoxyiMethane	660	µg/kg	9		_		1
889	LF03/FT09	Surface Soil	1992	Bis(2-chloroethyl) ether	560	µg/kg	9	· <b></b>		['	
890	LF03/FT09	Surface Soil	1392	Bisi 2-chloroisoprobyl) ether	660	⊨µg/kg		• •••	<u>-</u>	-	
891	LF03/FT09	Surface Soil	1992	Bis(2-e(nylhexyl) phthalate	660	µg/kg		<u> </u>	142	180	035Y\$09
892	LF03/FT09	Surface Soil	1988	Bis(2-ethylhexyliphthalate	100	µg/kg	10	2	340	1.70E+03	103M01
893	LF03/FT09	Surface Soil	1989	Bis(2-ethylhexyl)phthalate	50	⊨µ⊊/kg	9	· 1	590	590	03M08
894	LF03/FT09	Surface Soil	1989	Butylbenzylohthalate	30	µq/kg	9	2	40	1 150	03M08
895	LF03/FT09	Surface Soil	1992	Butylbenzylphthalate	660	hð/kð	. 9		-	<b>[-</b> .	
896	LF03/FT09	Surface Soil	1988	Cadmium	455	µg/kg	9	1	1.28E+03	1.286+03	03M01
897	LF03/FT09	Surface Soil	1988	Сакит	N/A	µg/kg	11	11	1.78E+06	5.65E+07	03M02
898	LF03/FT09	Surface Soil	1992	Chlorinated benzenes	660	µg/kg	18	i <b>—</b>	_	[_	
899	LF03/FT09	Surface Soil	1992	Chlorobenzilate	660	µg/kg	9.	I —	-	-	i
900	LF03/FT09	Surface Soil	1988	Chromum	N/A	µg/kg	. 11	11	6.53E+03	3.57E+04	03M02
901	LF03/FT09	Surface Soil	1988	Chrysene	70	µa/ka	10	1	2.30E+03	2.30E+03	03M01
902	LF03/FT09	Surface Soil	1992	Chrysene	660	ug/kg	9	4	75	250	035YS05
903	LF03/FT09	Surface Soil	1988	Copart	N/A	UQ/kg	11/	11	4.36E+03	3.07E+04	03402
904	LF03/FT09	Surface Soil	1988	Copper	N/A	uo/ko	11	11	1.01E+04	8.51E+04	03M02
905	LEO3/ET09	Surface Sou	1988	DDD pp	1	uo/ko	3		-	-	
906	1 F03/FT09	Surface Soil	1988	ODE op	1	ua/ka	3	-	-		
907	LE03/ET09	Surface Soil	1988	DOT op		<u>ng/ko</u>	3		-		
908	LF03/FT09	Surface Sol	1989	Den-bulyiphthalate	30		9	1	 60	60	03M08
909	1 F03/FT09	Sudace Soil	1997	Orn-butylopthalare	00 1038		a	_			
910	1 F03/FT09	Sudace Soil	1988	Di-n-octvinhihalate	150	Ho/ka	10	1	270	270	03401
811	: F03/FT09	Sudace Sol	1997		660	vo/kg			4/¥	210	0.000
1	LE03/ET09	Surface Sol	1007	Definite	860	PANA .				Ē	
414	LF03/F103	Surface Soil	1007		000	ug (ka					
81.3	LEO3/FTUS	Surface Soil	1088	Dibenzia histitizzane		ug/kg	10		400		034404
946	E03/F109	Sudace Soil	1000	Dibenzolutto		hàng l			+		
313	LEONETOO	Surface Sol	1000		000	- Ong		-		·	
916		Surface Soil	1992	Cretary photoste	200	pgrxg					
917	2F03/F109	Surface Soa	1969		<u> </u>	hðurð	¥		180	1001	USMOS
918	LF03/F109	Sumace Soil	1995		000	UOXO	3		-	j <b>=</b>	····
919	LF03/F109	Surface Soil	1992		660	.hð\kg	3	-		<b>,</b>	······································
920	LF03/F109	Surrace Sou	1992	Dipnenylamine		hđ, kđ	3	-	<b>-</b>		····
921	LF03/FT09	Surface Soil	1988	Endosulian, a		HOW	3	-		<b></b>	
92Z		Surface Soil	1985	Engan aldenyde	1	hð kð	3				·
923	LF03/FT09	Surface Soil	1992	Ethytmethenesuronale	000	hāvikā	9		-		
924	LF03/F109	Surface Soil	1988	Enybenzene	08	пралка	TU .	1	100	100	03462
925	LEO3/E109	Surface Soil	1992		660	hđ/kg	8		-	-	
926	LEUSVETOS	Sunace Sou	1988	Filloranthene	50	howd	10	1	4.502+03	4.50E+03	USM01
927	CF03/FT09	Surface Soil	1992	Fluoranhene	660	ihð\kð	9	4	140	410	0357505
928	LF03/FT09	Surface Soil :	1992	Fluorene	660	havka	¥	-	-		
929	LF03/FT09	Surface Soil	1992	Hexachlorobenzene	660	hðvið	8	-		F	
930	LF03/FT09	Surface Soil .	1992	hexachioroputadiane	560	ind\ka		-	-		·····
331	LP03/FT09	Surface Soil	1992	Hexachiorocyclopentadiene	660	hð kö		-			
932	L-03/FT09 '	Surface Soil	1992	rtexachioroethana	660	1 <b>10/kg</b>	9	-		f	
823	LF03/FT09	Surface Soil	1992	HEXECIODDRene	660	indured.	9		-		
934	LF03/FT09	Surface Soit	1992	Hexachioroproperte	660	hđyka	9		-	-	
935	LF03/FT09	Surface Soil	1988	Indeno(1,2,3-cd)pyrene	50	ugring .	10	1	1,50E+03	1.50E+03	03M01
936	LF03/FT09	Surface Soil	1992	indeno(1,2,3-cd)pyrene	660	-Fa/kg	9	1	200	200	03SY\$05
937	LF03/FT09	Surface Sol	1988		N/A	µg/kg	11	11	9.05E+06	4.20E+07	U3M01
938	LF03/FT09	Sumace Soil	1992	Isodan	660	HORKS	9	-			
839	LF03/FT09	Surface Soil	1992	anonde	660	µg/kg	9	-	-	-	
940	LF03/FT09	Surface Soil	1992	(SOSAfroie	660	140/kg	9	-		<b></b> _	
941	LF03/FT09	Surface Soil !	1992	Kepone	660	ind, rd	9	<b>-</b> l	-	-	
842	LF03/FT09	Surface Soil	1992	Kerosene	660	hð ka	9	6	137	1.10E+05	03SYS09
943	LF03/FT09	Surface Soil I	1968	Lead	4.56E+03	µq/kg	11	2	1.07E+04	1.32E+05	03M01
944	LF03/FT09	Surface Soil	1989	Leed	N/A	No/kg	14	14	2.94E+03	4.13E+04	03M06
945	LF03/FT09	Surface Soil	1992	m-Cresol	660	µg/kg	9	-		-	
946	LF03/FT09	Surface Soil	1992	m-Dinirobenzerie	660	HQ/kg	9	-	-	-	
947	LF03/FT09	Surface Soil	1988	Magnesium	N/A	Parkg	11	11	2.12E+06	1.48E+07	03M02
948	LF03/FT09	Surface Soil	1985	Manganese	N/A	µg/kg	11	11	1.40E+05	6.25E+05	03M01
949	LF03/FT09	Surface Soil	1988	Mercury	19.6	ug/kg	9	3	70.8	161	034403
950	LF03/FT09	Surface Soil	1992	Methapyrilene	660	-µg/kg	9	-	-		
951	LF03/FT09	Surface Soil	1992	Methyl methanesulfonate	660	µa/ka	9	-	-	-	
952	LF03/FT09	Surface Soil	1988	Methylane chloride	150	indvid	10	2	2.60E+03	4.40E+03	03M02
953	LF03/FT09	Surface Soil	1988	Molybdenum	2.40E+03	µg/kg	9	2	3.58E+03	7.25E+03	03M02

A.14

FINAL

	A	e	¢	0	5	F	G	H	1	J	ĸ
									······	**************************************	
L	Source Area	Vezia	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	I Max Conc	Max Location
954	LF03/F709	Surface Soil	1992	n-Nitroso-di-n-dipropylamine	660	) µg/kg	9		-	-	
955	LF03/FT09	Surface Soil	1992	n-Nitrosodi-n-butylamine	660	) µg/kg	9			-	
956	LF03/FT09	Surface Soil	1992	n-Nilrosodiethylamine	66	) hð\kð	9		ļ <b>-</b>	-	
85/	LF03/FT09	Surface Sol	1992		500 66/	) ug/kg			1 <del></del>	· ••	
940	1 603/6709	Surface Soil	1992	n-Nitrosometrytethylamine	66/	) un/ko		,		·	
960	LEO3/ETO9	Surface Soil	1992	n-Nitrasomoroholme	660	:e.x.::x ) ua/ka			/		••••• <b>-</b>
961	LF03/FT09	Surface Soil	1992	n-Nitrosoppendine	660	) ua/ka	9		-	-	i
962	LF03/FT09	Surface Soil	1992	Naphthalene	550	) µg/kg	9	: 2	74.5	1.50E+03	035Y509
963	LF03/FT09	Surface Sol	1988	Nickel	N/A	µg/kg	. 1\$	11	9.37E+03	6.43E+04	03M02
964	LF03/FT09	Surface Soil	1992	Nitrobenzene	660	) µg/kg	9	i <del></del>	1_	-	
965	LF03/FT09	Surface Soil	1992	Nitrosopyrrolidine	660	) µg/kg	9	-	-	-	
966	LF03/FT09	Surface Soil	1992	o-Toluidine	560	) µg/kg	. 9	-	-	<b></b>	
967	LF03/FT09	Surface Soil	1992	o-Dimethylaminoazobenzene	660	) µg/kg	9	i <b>-</b>	<u></u>	-	
968	LF03/FT09	Surface Soil	1992	p-Phenylenediamins		ug/kg		· —		<b></b>	
870	LE03/F 109	Surface Soll	1992	Parankin Bagiachlosonikobenzene (ocób)	66/			<u></u>	-		
871	LF03/FT09	Surface Soil	1992	Pentachioronnenol	3 30F+0	ua/ko		!	_		
972	LF03/FT09	Surface Soil	1992	Phenacetin	660	) ua/ka	. 9	<del>-</del>	_		· · · · · · · · · · · ·
973	LF03/FT09	Surface Soil	1988	Phenanthrene	20	) µg/kg	10	1	2.20E+03	2.20E+03	03M01
974	LF03/FT09	Surface Soil	1992	Phenanihrene	660	µg/kg	9	5	81	130	03SYS05
\$75	LF03/FT09	Surface Soil	1992	Phenol	660	µg/kg	9	<u> </u>	-	-	
976	LF03/FT09	Surface Soil	1988	Potassium	N/A	:µq/kg	11	11	4.28E+05	3.45E+06	03M02
977	LF03/FT09	Surface Soil	1992	Pronamide	660	i:µg/kg	9	<b> -</b>	-	-	
978	LF03/FT09	Surface Soil	1988	Pyrene	60	ug/kg	10	1	2.40E+03	2.40E+03	03M01
2/3	LF03/F109	Surface Sol	1992	Safol	000	ineyed	, <u> </u>	<b>1</b>	<b>V</b> II	330	0331505
100	LFU3/FTU9	Surface Soil	1992	Sodiam	N/A	un/ko	11	- 11	1395+05	- 771E+06	03402
382	LF03/FT09	Surface Soil	1992	Sym-trinërobenzene	660	lua/ka	9	_	-	_	
963	LF03/FT09	Surface Soil	1992	Tetraethyl dithiogyrcohosphate	660	lua/ka	9		-	-	
984	LF03/FT09	Surface Soil	1988	Thallium	1.798+04	µg/kg	9	2	2.51E+04	2.94E+04	03M01
985	LF03/FT09	Surface Sori	1988	Toluene	160	i µg/kg	10	1	470	470	034402
986	LF03/FT09	Surface Soil	1988	ТРН	1.025+04	lµĝ∕kg	i 9	4	2.81E+04	1.30E+06	034401
\$87	LF03/FT09	Surface Soil	1989	ТРН	1.00E+04	µq/kg	39	19	1.00E+04	4.90E+06	035812
888	LF03/FT09	Surface Soil	1992	Tributyl phosphate	560	µg/kg	9	<u> -</u>	-	-	
889	LF03/FT09	Surface Soil	1988		30	HQ/KG	10	2	60	160	03M02
001	1603/6109	Surface Soil i	1048	Vanade m	N/4	Juglika	11	11	1 215+04	1 555405	03402
992	LF03/FT09	Surface Soil	1988.	Zinc	4.63E+04	iuo/ka	11	10.	1.01E+04	1.68E+05	03M01
993	LF03/FT09	Soil	1988	1,1,1-Trichlorosthane	60	μ <b>g/kg</b>	6	-	-	-	
994	LF03/FT09	Soil	1988	1,1-Dichloroethylen#	100	lµg/kg	6	-	-		
995	LF03/FT09	Soil	1988	Acenaphinene	10	i ugʻkg	6	1	60	60	03M01
996	LF03/FT09	Soil	1988	Acenaphthylene	10	,µg/kg	5	-	-	-	
997	LF03/FT09	Soil	1988	Aluminum	1.50E+07	iug/kg	8	2	8.99E+08	1.02E+07	03M03
898	LF03/FT09	. 50# Sod	1988		0.39E+03	HQ/Kg	6	1	8.55E+03	6.55E+03	03M03
1000	LFUSETAS	Soil	1988	Benzens	+.91E+U3	LUCKS	<u>e</u>		1.450703	1.002405	
1001	LF03/FT09	Soil	1988	Benzo(a)anthracene	. 10	UDO/KO	6		-		· · · · · · · · · · · · · · · · · · ·
1002	LF03/FT09	Sort	1988	Benzo(a)pyrene	· · · · · ·	μα/κα	6	-		-	
1003	LFOXFT09	Soil	1988	Benzo(b)fluoranthene	30	lug/kg	6	-		-	
1004	LF03/FT09	Sod	1988	Benzo(g.h.i)perylene	40	µg/kg	6	-	-	-	
1005	LF03/FT09	Soil	1988	Beryllium	1.53E+03	µq/kg	B	2	320	1.22E+03	03M01
1006	LF03/FT09	Soil	1988	Bis(2-ethylhexyl)phthalate	100	HQ/KQ	ļ	-	-	-	AN1 64-
1007	LF03/FT09	Sol	1968		455	Havka	8	3	537	768	031401
1000	EDVETO9	504 I	1082	Choose	N/A	porto .	e e		1 375-14	4.78E+04	03401
1010	LF03/FT09	Soil	1988	Chrysene	70	iuo/ko	6		-	-	
1011	LF03/FT09	Soil	1988	Cobalt	N/A	's g/ka	8	6	7.20E+03	1.73E+04	03M01
1012	LF03/FT09	Soil	1968	Соррег	N/A	i) jugi ka	8	6	1.91E+04	6.14E+04	03M01
1013	LF03/FT09	Soil	1968	000, pp'	1	iµg/kg	3	-		-	
1014	LF03/FT09	Soil	1988	ODE. pp'	1	Hg/kg	3	1	2	2	03M02
1015	LF03/FT09	Soil	1988	ODT. pp'	1	Ling Kg	3	1	5	5	03M03
1016	LF03/FT09	Soil	1988	Di-n-octylphthalate	150	hð kð	6	<b> -</b>	-	-	
1017	LF03/FT09	Sol	1965	Dibenzo(a.h)anthracene		parka	6	<del>-</del>		-	
1018	LEONETOD	Soil	1965	Enguadran, a		ug/kg		<u> </u>	-	-	
1020	LF03/FT04	Sol	1988	Eihvibenzene	RC	UPWING .	i R	_	_		
1021	LF03/FT09	Soil	1988	Fluoranthene	50	ug/ko	. 6	-		-	·····
				······································						L	

	A	8	C	D	ξ		F	G	н	1	Ļ	ĸ
1	_				_			· .			1.	
1	Source Area	Media	Date	Analyle	DL		Unds	#Samples	#Detect	Min Cone	Max Conc	Max Location
1022	LF03/FT09	Soil	1988	Indeno(1,2,3-cd)pyrene		50	ug/kg	. 6		-		
1023	LF03/F109	501	1988		N/A	-	ug/kg	10	10	1.376+07	2.98E+07	03MU2
1025	LEONETOS	Sol	1984	Lead	4 56E+	63	ua/ko		5	6 90 5+03	1 185+04	033309
1026	LE03/ET09	Soll	1989	Lead	N/A		uā/ka	6	6	7.06E+03	1.12E+04	035809
1027	LF03/FT09	Sail	1988	Magnesium	N/A		µa/ka	8	ß	4.11E+06	8.02E+06	103M01
1028	LF03/FT09	Sail	1988	Manganese	N/A		µg/kg	8	. 8	8.15E+04	4.25E+05	03M04
1029	LF03/FT09	Sort	1988	Mercury	19	36	µg/kg	6	<b></b>	-	-	
1030	LF03/FT09	Sol	1988	Methylene chloride	1	50	µg/kg	6	3	340	2.20E+03	03M01
1031	LF03/FT09	Soil	1988	Molypdenum	2.40E+	03	µg/kg	5	1	3.99E+03	3.99E+03	03M01
1032	LF03/FT09	Soil	1986	Nickel	N/A		µg/kg	8	8	1.59E+04	4.41E+04	03M01
1033	LF03/FT09	Soil	1988	Phenanthrene		20	µą/kg	6		-	<b> _</b>	
1034	LF03/FT09	Soil	1968	Potassium	N/A		µg/kg			6.56E+05	3.74E+06	03M01
1035	LF03/FT09	Sod	1982	Pyren¢		БO:	µg/xg	5	-	-	-	034105
1036	LF03/F709	501	1988	Sodium	NVA		ug/kg		5	3.37E+05	0.30E+03	UJMU2
103/	LF03/F109	- 500 Fool	1988		1/90-	60 -	ug/kg	;		-	<b></b>	
1030	LE03/ET09	- <u>50</u> %	1997		N/A		Harva K	10	- 10	85.2	97.4	035502
1040	LEOVETOS	Sol	1992	Total permisum hydrocarboos	01	02	ua/ka	10	6	0.4	B 7	035506
1041	LF03/FT09	Soil	1988	ТРН	1.02E+0	04	<u></u>	6	2	1.57E+04	7.82E+04	034402
1042	LF03/FT09	Sori	1989	ТРН	1.00E+	04	ug/ka	28	13	1.40E+04	1.50E+07	035809
1043	LF03/FT09	Soil	1988	Trichloroethene		30	µg/kg	6	-	-	-	
1044	LF03/FT09	Soil	1988	Trichlorofluoromethane	_1:	50.	µg/kg	6	-	-	-	
1045	LF03/FT09	Sou	1988	Vanadivm	N/A	ند.	µ <b>q/kg</b>	8	8	2.87E+04	1.42E+05	03M01
1046	LF03/FT09	Soil	1988	Zinc	N/A		ug/xg	8	8	8.53E+03	6.82E+04	03M02
1047	LF04	Water	1992	1,1,1-Trichloroethane	<u>C</u>	).S	ua/L	2		-	-	
1048	LF04	Water	1992	1,1,2-Trichloroethane	0	).5.	µg/L	2		<del>.</del>	-	
1049	LF04	Water	1992	1,1-Dichloroethane		1	μ <b>ο/L</b>	2		=		
1050	LF04	Water	1992	1.2-Dichloroethane		1.5	µ <b>9/</b> L	2	<u>-</u>		-	
1051		vvater	1992	1.4-DKDKDBBEDZERE		<u>~</u>	ugyr.	4			-	
1052		Water	1007	Anumeny			pgrt.		- 1	- 120	- 120	041402
1053		Water	1992	Baoum	N/A		HYPE.			120	1905+01	044402
1055	LF04	Water	1992	Banzane		2:	ug/1	2	-	-	-	
1056	LF04	Water	1992	Berylium	·	3,	ua/L	4	1	4.8	4.8	04M02
1057	LF04	'Water	1992	(Bromide	50	00	µg/L	1	_	-	-	
1058	LF04 .	Water	1992	Cadmium	·······	10	μ <b>α</b> /L	4	<u> </u>	-	-	
1059	LF04	Water	1992	Calcium	N/A		µg/L	4	4	8.90E+04	2.40E+05	04M02
1060	LF04	Water	1992	Carbon tetrachloride		1	µ <b>g/</b> 1_	2	-	-	-	
1061	LF04	Water	1992	Chlonde	N/A	1	<u>, 10/L</u>	1	1	1.60E+04	1.60E+04	04M02
1062	LF04	Water	1992	Chloroform	0	0.5	<b>µ\$/L</b>	2	-		-	
1063	LF04	Water	1992	Chromum		20	hð/r	4	3	30	200	04MC2
1064	LF 04	vvaler	1992	Caball		20	HQAT.	2	-			041402
1063	LP04	Water	1992	Concer	·····-	201	uovi			/0  40	120	041402
1047	I FOA	Water	1992	Chylenzene		21	<u>rar -</u>			- 2	-	
1068	LF04	Water	1992	Fluoride	N/A	ļ	uu/L	1	1	1.70E+03	1,70E+03	04M02
1069	LF04	Water	1992	h-Chlorofuorobenzene	N/A		µg/L	1	1	9	9	04M02
1070	LF04	Water	1992	Iron	N/A	<del>i</del>	WG/L	4	4	8.50E+04	2.70E+05	04402
1071	LF04	Water	1992	Lead	N/A		µg/L	1	1	25	25	04M02
1072	LF04	Water	1992	Magnesem	N/A	1	ug/L	4	4	4.00E+04	1.406+05	04M02
1073	LF04	Water	1992	Mangariese	N/A		µg/L	4	4	1.50E+03	5.50E+03	04M02
1074	LF04	Water	1992	Methylenechloride		5	μ <b>η/</b> L	2	-		-	
1075	LF04	Water	1992		N/A	<u>, i</u>	POIL .	4	4	40	330	04M02
1076	LF04	VValer	1992		Z		HOAL	1			-	
1077	1 604	Water	1002			<u>u</u>			-	-	-	044402
1078	UT 04	Water	1001	:pros not thinke (Geniger) e	-19874			1		0.8	5.8	1474 C
1040	L FOA	Water	1992	Pote saum	N/A		HOUT			5 105+03	1 605-04	044407
10.11	LF04	Water	1992	Silver		201	ug/L	4		-	-	
1082	LF04	Water	1992	Sodrum	'N/A			4	4	1.40E+04	6.70E+04	04M02
1083	LF04	Water	1992	Sulfate	NA		μο/L	1	1	2.70E+03	2.70E+03	04M02
1084	LF04	Water	1992	Sullides	1.00E+	04	Hart.	2	-	-	-	
1065	LF04	Water	1992	Tetrachioroethene	Ç	).5	µg/L	2	-	-	-	
1086	LF04	Water	1992	Tin	1	00	har	4	-	-		
1087	LFOM	Water	1992	Touene		21	µg/L	2	-	-	-	
1086	LF04	Water	1992	Total dissolved solids	N/A	, ,	mG/L	2	2	640	1.00E+03	04M02
1089	LF04	Water	1992	Trans-DCE		<u>t i</u>	µg/L	2		- 1	-	

Ľ,

ĸ

	A	8	l c	0
1	Source Area	Media	Date	Analyle
1090	LF04	Water	1992	Trichloroethene
1091	LF04	Wate'	1992	Vanadium
1092	ĻF04	Water	1992	Viñyi chiande
1093	LF04	Water	1992	Xylenes (total)
1094	LF04	Water	1992	Zinc
1095	LF04	Surface Soil	1989	2,4-Dinitrotaluene
1096	LF04	Surface Soil	1989	2-Methyinaphthalene
1097	LF04	Surface Soil	1992	Antimony
1098	LF04	Surface Soil	1992	Barium
1095	LF04	Surface Soil	1992	Beryllium
1100	LF04	Surface Sod	1989	Bis(2-ethylnexyl)phthalat
1101	LF04	Surface Soil	1992	Cadmium
1102	LF04	Surface Soil	1992	Calcium
1103	LF04	Surface Soil	1992	Chromium
1104	LF04	Surface Soil	1992	Cobalt ·
1105	LF04	Surface Soil	1992	Copper
1106	LF04	Surface Soil	1988	Di-n-butylonthalate
1107	LF04	Surface Soil	1989	Dec-butyiphihalate
1108	LF04	Surface Soil	1992	Irón
1109	LF04	Surface Soil	1992	Magnesium
1110	LF04	Surface Soil	1992	Manganese
1144	LE04	Surface Soil	1098	n-Nitrosodioheoviamine

1	Source Area	Media	Date	Analyle	ÐC	_	Units	#Samples : #Oetect	Min Conc	Max Conc	Max Location
1090	LF04	Water	1992	Trichloroethene		1	µg/L	2 –	-	-	
1091	LF04	Water	1992	Vanadium		30	ug/L	4 2	60	350	04M02
1092	1 504	Water	1992	Vinyi chionde		4	ug/L	<u> </u>	-	-	<del>~</del>
1094	LF04	Water	1992	Zinc		10	ua/L	4: 3	80	- 510	04M02
1095	LF04	Surface Soil	1989	2,4-Dinitrotaluene		70	µg/kg	4	;	-	
1096	LF04	Surface Soil	1989	2-Methyinaphthaiene		30	µg/kg	4. 1	210	210	045803
1097	LF04	Surface Soil	1992	Animony	2.00E+	04	hð, kð	8:-	-	-	
1098	LF04	Surface Soil	1992	Barium	N/A		hð\ků	8 8	9.20E+04	1.60E+06	04SYS02
1099	LF04	Surface Sol	1992		<b>د</b>	00	ug/kg	8 1	1.40E+03	1.406+03	0457508
1100		Surface Soil	1969	Cadmum	1.005+	30 27	un/ko		10	/0	045808
1102	1 F04	Surface Soil	1992	Calcium	N/A		ua/ka	<u><u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	4.60E+06	1.10E+07	04SYS08
1103	LF04	Surface Soil	1992	Chromium	N/A		µg/kg	8 8	1.40E+04	2.30E+04	04SYS06
1104	LF04	Surface Soil	1992	Cobait ·	N/A		µg/kg	6: 8	6.00E+03	9.00E+03	04SYS04
1105	LF04	Surface Soil	1992	Copper	N/A	··	µg/kg	8 8	2.10E+04	4.40E+04	04SYS04
1108	LF04	Surface Soil	1988	Di-n-butylphthalate	· · · · · · · · · · · · · · · · · · ·	60	µg/kg :	3 -		-	
1107	LF04	Surface Soil	1989	Den-butyiphihalate	AUZA .	30	hāvkā .	4!	-	-	ALCYCOD
1106	1 604	Surface Soil	1992	Maanesuum	N/A		ug/kg		1.405+06	1.00E+07	0451502
1110	1 F04	Surface Sol	1992	Manganese	N/A		uu/ka	8 8	2 50E+05	3.70E+05	045YS02
1111	LF04	Surface Soil	1986	n-Nitrosodiphenylamine	~~~	20	ug/kg	3j-	-	-	
1112	LF04	Surface Soli	1992	Nickel	N/A		µg/kg	8 8	1.30E+04	2.60E+04	04SYS08
1113	LF04	Surface Soil	1992	Potassium	N/A		µg/kg	8 8	1.40E+06	2.10E+06	04SYS08
1114	LF04	Surface Sol	1992	Silver	2.00E+0	03	µq/xg	8 -	-	-	
1115		Surface Soil	1992			~	HANKO	<u> </u>	3.50E+05	5.80E+05	04SYS08
1111		Surface Sol	1992		1.0024	<u>.</u>	ug/kg	3 7		2405405	041403
1118	LF04	Surface Soll	1989	TPH	1 37E+	04 i	uo/ke	22 17	1.10E+04	3.70E+05	04\$806
1119	LF04	Surface Soil	1992	Vanadium	N/A		µo/kg	8 8	2.70E+04	4.60E+04	045Y508
1120	LF04	Surface Soil	1992	Zinc	N/A		µg/kg	8 8	4.40E+04	1.00E+05	04SYS04
1121	LFO4 .	Soil	1989	2,4-Dinitrololuene		70,	µg/kg i	2 1	120	120	04SB02
1122	LF04	Şoil	1989	2-Methyinaphthalene	<u>.</u>	<b>30</b> i	hðyrd	2 -	-	-	
1123	LF04	Son	1989	Bis(2-ethymexyl)phthalate		50	µg/kg	14 2	160	190	04508
1124		Sol	1968	Di-n-outypontnatate	······	10.	ug/xg ug/ko	14 1	2.002+03	2.00E+03	
1125	L FD4	Soil	1988	n-Nitrosodiobenviamine		20.	ua/ka j	4 1	530	530	04402
1127	LF04	Soil	1988	ТРК	N/A		µa/ka	4 4	4.98E+04	1.88E+05	041403
1126	LF04	Soil	1989	трн	1.37E+	04	µq/xq	20 11	1.80E+04	5.14E+05	04507
1129	LF06	Water	1994	(Technical) chlordane		1	μανί	4 –	-	-	
1130	LF06	Water	1994	1,1,1-Trichloroethane		1	ug/L	4-	-	-	
1131	LF06	Water	1994	1,1.2,2-Tetrachioroethane		1		4 -	-		
1132		VVater .	1994	1.1.2-Inchloroestane	<u>v</u>	1.31				<u> </u>	
1134	1 F06	Water	1994	1 1-Dichloroelhene	0	៍ទាំ	ua/L	4-	-		
1135	LF06	Water	1994	1,2,4-Trichlorobenzene		10	ug/L	4	-		
1136	LF06	Water	1994	1,2-Oichlorobenzene		10	191	12 -	-		
1137	LF06 I	Water	1994	1,2-Dichloroethane	0	.5	HQ1_	4	-	-	
1138	LF06	Water '	1994	1,2-Dichloropropane		51	μg/L	4 -	=	-	
1139	LF06	vvat er	1994	T.3-Dichlorobenzene		10		12 -		<u>-</u>	
1140	LFV0	Water	1994	1,		101	uari	4	-		
1142	LF06	Water	1994	2,4,6-Trichlorophenol	·····	10	ug/L	4	_		
1143	LF06	Water	1994	2,4-Dichlorophenol		10	μα/L	4		-	
1144	LF06	Water	1994	2,4-Dimethylphonol		10	µg/L	4 -	-		
1145	LF06	Water	1994	2,4-Dinitroptienol		50	<b>µq/L</b>	4 -	-	-	
1146	LF06	Water	1994	2.4-Dinitrotoluene		50	101	8 -	-	-	
1147		water	1994	2.0-LMTRICIOUETE							
1144		vvaler Water	1004			10			<u> </u>	<b>-</b> ł	
1150	t F06	valer i	1994	2-Chiorochenol		10/					
1151	LF06	Water	1994	2-Methyinaphthalene		101	uo L	4-	<b>_</b> }	<u></u> †	
1152	LF06	Water	1994	2-Methylphenol		10	µq/L	4 -	-	-	
1153	LF06 • ·	Water	1994	2-Nitroaniline		50	10/L	4 -	-	-	
1154	LF06	Water	1994	2-Nirophenol		10	ug/L	4 -	-	-	
1155	LF06	Water	1994	3,3'-Dichlorobenzidine		20 !	ug/L	4 -	-	-	
1156	1506	VV216r	1994	j-NRCSPILLO		0		4-	<b>-</b>	<b>-</b> }	
112/	L	AASICL	1334	Tex-metrikihinguna (intel)		(Q)	PAR -	≠ <u> </u> −	-	- 1	

Ę

F

G

н{

J

1 1158 1159	Source Area	Media	Date	Anaivte	21		#Samoles				1
1 1158 1159	Source Area	Media	Date	Anaivte	D1	1.1.0	#Samoies		• • - · _		1
1158 1159	II ENC					Unas	Houmpies	PUBIECI	Min Conc	Max Conc	Max Location
1159	LFUG	Water	1994	4 4'-DDD		hð\r	4	. <del></del>		-	<u> </u>
14480	LF06	Water	1994	4 4-00E		µg/L			. <b>-</b>	ļ <del>.</del>	
1160	100	Vvale*	1004	4 4-001	10 10	Ug/L		-			
1162	LFV6	Water	1994	A-Chinro-3-methylohenol		HY'L		: <u>-</u>	. <u> </u>	+	1
1163	1 606	Water	1991	4-Chlorpanäne	20	u0/L					+
1164	LF06	Water	1994	4-Chlorophenyl-phenylether	10	uo/L	4	-		I_	<u>.</u>
1165	LF06	Water	1994	4-Nitroanime	20	ua/L	4	_		-	
1166	LF06	Water	1994	4-Nnropheno!	50	ug/L	4	_	······································	<b>_</b>	
1167	LF06	Water	1994	₽-8HC	0 05	μg/L	4	. =	+	; <b>-</b>	
1166	LF06	Water	1994	a-Chiordane	0 05	µg/L	4	· —	-	-	••••••••••••••••••••••••••••••••••••••
1169	LF06	Water	1994	Acenaphthene	10	µg/L	4	!-	-	-	
1170	LF06	Water	1994	Acenaphthylene	10	ug/L	4	-	-	-	
1171	LF06	Water	1994	Aidon	0 0\$	µg/L	4	-	-	-	
1172	LF06	Water	1994	Aluminum	N/A	µg/L	4	4	2.05E+03	9.36E+03	064406
1173	LF05	Water	1994	Anthracene	10	µg/L	4	-	-	-	
1174	LF06	Water	1994	Antimony	1	µg/L		- i		-	
1175	LF06	Water	1994	Arsenic	3	µg/L	8:	7	12.6	38.3	06M04
1176	LF06	Water	1994	b-BHC	0.05	µg/L	4!	-	-	-	
1177	LF06	Water	1994	Banum	N/A	µg/L	8	8	112	384	06M04
1178	LF06	Water	1994	Benzené	1	µ9/L	4	-	-		1
1179	LF06	vvater	1994	Benzolajanihracene	10	µg/L	4,	-		-	
1180	LEVO	vvaler	1994	Benzokajpyrene	01	port.	4			·	
1181	LFUG	vvater Moter	1994	Senzoi Difuoramiene	10	han.		-	<del>.</del>		
1182	LFU6	Water	1994	Benzo(g.n.i)perviene	10	Hdvr.		-	-	-	
1103		vvaler	1004	Benzok and	50	pg/L		-	_		
4496	LEVO	Water	1004	Benzyl skobol	<u></u> n¢	ио/і					····
1186	1 606	Water	1994	Bewlinm		uo/	A		-		·····
1187		Water	1994	Bis/2-chiomethory/methane	10	un/L	·	_	_	-	·····
1188	EF06	Water	1994.	Bis(2-chloroethvi)ether	10	UO/L	4				
1189	LF06	Water	1994	Bis/2-chloroisopropyl)ether	10	ua/L	4	-	-		
1190	LF06	Water	1994	Bis(2-ethvinexyl)phthalate	10	ud/L	4	1	1	1	064405
1191	LF06	Water	1994	Bromide	100	µg'L	4	-	-	-	
1192	LF06	Water	1994	Bromodichioromethane	1	uu/L	4	-	_	=	
1193	LF06	Water	1994	Bromoform	1	ugl	4	-	-	-	
1194	LF06	Water	1994	Bromomethane	2	µq/L	4	- ]	-	-	
1195	LF06 -	Water	1994.	Butylbenzylphthalate	10	µ9/L	4	-	-	-	
1196	LF06	Water	1994	Cadmium	1	µg/L	8	1	1	1	061406
1197	LF06	Water	1994	Calcium	N/A	hð\r	8	8	4.58E+04	7.36E+04	061406
1198	LF06	Water	1994	Carbezole	10	ug/L		-		-	
1199	LF06	Water	1994	Carbon tetrachloride	0.5	HOVL	4	-	-	-	
1200	LF06	Water	1994	Chloride	N/A	horr .	<u>4</u>	4	1.50E+03	4.70E+03	05M02
1201	LF06	vvater	1994	Chlorocenzene	<u></u>	ug/L	8	<b>=</b>	-		}
1202	LF05	vvaler	1994	Chloredame	1	HQ/1	4		-	-	
1203		VVECOF Mater	1994	Chippristhese	1		4		-		
1204	1 FA6	Water	1004	Chromium	N/A	US/L			- 14	- 20.2	051404
1200	1 506	Water	1994	Chrysene	10	ual	A			4V.2	
1207	1 F06	Water	1994	Cis-1 3-dichlorg progene	0.5	<u></u>					
1208	LF06	Water	1994	Cobat	1	JUNI	3	-		-	
1209	LF06	Water	1994	Copper	1	JO/L	6	4	25.3	82.1	OSMO6
1210	LF06	Water	1994	4-BHC	0.05	LO/L	4				
1211	LF06	Water	1994	Di-n-butylphthalate	10	HO/L	4	3	1	2	OGM02
1212	LF06	Water	1994	Di-n-octylphthatate	10	HOAL	4		-	-	
1213	LF06	Water	1994	Dibenz(a,h)anthracene	10	µ <b>g</b> ⁄i,	4	-	-	–	
1214	LF06	Water	1994	Dibenzofuran	10	ug/L	4	-	-	-	
1215	LF06	Water	1994	Dibromochloromethane	1	µg/L	4	-	-	-	
1216	LF06	Water	1994	Dichlorodifiuoromethane	1	µg∕L.	4	-	-		
1217	LF06	Water	1994	Dieklrin	0.1	µq/L	4		-	-	
1218	LF06	Water	1994	Diethylphihalate	10	μ <b>ο/</b> L	4	1	4	4	06M02
1219	LF06	Water	1994.	Dimethylphthalate	10	ug/L	4	-	-	-	
1220	LF06	Water	1994	Endosulian 1	0.05	Hg/L	4	_	-		
1221	LF06 1	Water	1994		0.1	HO1	4	-	-	-	
1222	1506	vvater	1994	Eurosukan sunale	0.1	HOW	4	-	-	-	
	U-00	**Bief	1354	Engral Redain stdebude	0.1	HOVE	4	=	-		
499	1600	i fi (mi ma							- '	-	

,

FINAL

---

	A	В	C C	0	E	F	G	н		J	K
	1								:	i	
1	Source Area	Media	Date	Analyte	ΟL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1226	LF06	Water	1994	Filloranthene	10	) hð\r	4	• ••	i	-	
1227	LF06	Water	1994	Fluorene	10	0 μg/L		. <b>_</b>		-	·
1228	LF06	Water	-99-	g-970	0.0	sµg/L			; <b>-</b>		
1229		Water	1994	g-Chordane	0.0	sug/L	4		· · · · · · · · · · · · · · · · · · ·	: <b>-</b> T	
1234	LFUG	Vedte:	1004			5 µg/L	4			·•••	~~~~~
11232		Water	1994	Hexachinonenzêne		o ugal	·	~ <b>_</b>	<u>.                                    </u>	<u></u>	T
1232	LF 00	Waler	1994	Hexachioroputadiene		) ug/l					
1734	EF06	Water	1994	Hexachlorocyclopentachene		0 µ0/l	7				
1235	1.F06	Water	1994	Hexachloroethane	1(	) µ0/L	4		∲	+	
1236	LF06	Water	1994	Indeno(1,2,3-cd)pyrene	10	) ua/L	4		·	∮····	
1237	LF06	Water	1994	iroñ	N/A	uo/L	7	7	1.30E+03	2.15E+04	06404
1238	LF06	Water	1994	Isophorone	10	) µg/L	4	· <b>_</b>	-	-	
1239	LF06	Waler	1994	Lead	1	µg/L	7	5	3.7	23.7	064404
1240	LF06	Water	1994	M p-xylene	1	µg/L	4	i <b>—</b>			
1241	LF06	Water	1994	Magnesium	N/A	µg/L	8	8	8.04E+03	2.12E+04	06M04
1242	LF06	Water	1994	Manganese	N/A	µg/L	8	8	475	2.79E+03	064406
1243	LF06	Water	1994	Methoxychlor	0.9	μ <b>g/L</b>	4	-		-	
1244	LF06	Water	1994	Methylene chloride		ug/L .	4	-		-	!
1245	LF06	Water	1994	n-Naroso-di-n-oropylamine	10	) µg/L	4	-	-	ļ <u> </u>	
1245	LF06	Water	1994	n-Nilrosodiphenylamine	1(	).µ <b>0/</b> L	4	I <b>—</b>	-	<b> -</b>	
1247	LF06	Waler	1994	Naphihalene	10	) h <b>ů</b> vr	4	.—		-	
1248		Water	1994	NICKO	N/A	hð\r		2	30	31.87	051404
1249		Vvater	1994	Nitrobenzene	· ···	HQ/L	4				
1490	LFUG	vvale:	1994	O-Aylene		UQ/L	4	-	<b>-</b>	-	·····
1221	LFUG	VValer	1004	PC8-1016				-			
1251		Water	1004	PCB-1221		ug/L					
1254	1 606	Water	1004	PCB-1242		1995 100/	4		-		
1255	LF06	Water	1994	PCB-1248		UCUT.	4				
1256	LF06	Water	1994	PCB-1254		1 JUC/L	4		······································		
1257	LF06	Water	1994	PCB-1260		µg/L	4	_		_	
1258	LF06	Water	1994	Penlachlorophenol	50	) µg/L	4	-		-	······
1259	LF06	Water	1994	Phenanihrene	10	) µg/L	4	**	-	-	
1260	LF06	Water	1994	Phenol	10	hug/L	4	-	-	-	
1261	LF06	Water	1994	Potassium	N/A	µg/L	1	1	5.44E+03	5.44E+03	06M04
1262	LF06	Water	1994	Pyrene	10	) µg/L	4	-	-	-	{
1263	LF06	Water	1994	Silver	1	1 <b>µ9</b> ⁄	8		-	-	
1264	LF06	Water	1994	Sodium	N/A	iug/L	5	5	5.44E+03	9.52E+03	06M04
1265	LF06	Water	1994	Sulfate	N/A	ug/L	4	4	5.50E+03	1.80E+04	06MC4
1266	LF06	Water	1994	Tetrachloroethene (PCE)	0.5	Pg/L	4	-	-	-	
1267	LF06	Water	1994		·	HOVL	5		<del>_</del>	-	
1200		vvater Minter	1994	Touriers	-61/A	INGN I	4		-	-	061402
1209		vvater Minter	1994	Toxes dissored solids		hove :	•	•	2.100700	2.705703	
1271		Weter	1994	Trans-1 2-dichlomethene							
1777	LFD6	Weler	1994	Trans-1.3-dichloroprocene	0.9	i ua/l	4				
1273	LFOS	Water	1994	Trichlorgethene (TCE)	0.5	jug/L	4		_		
1274	LF06	Water	1994	Trichtorofluoromethane	1	IUg/L	4	-	-	[	
1275	LF06	Water	1994	Vanadium	N/A	:µq/L	4	4	11.8	30.1	OGM04
1276	LF06 i	Water	1994	Vinyt chloride	0.9	ijug/L	4	-	-	-	
1277	LF06	Water	1994	Zinc	2	Jug/L	\$	4	43.1	64.6	06M06
1278	5535	Water	1992	1,1,1-Trichloroethare	0.5	iug/L	12	-	-	-	
1279	SS35	Water	1992	1,1.2-Trichloroethane	0.5	JUG/L	12	-	-	1-	
1280	SS35	Water	1992	1,1-Dichloroethane	······	I UQ/L	12	-	-	-	
1281	5\$35	Water	1992	1.2-Dichloroethane	0.5	ING/L	12	1	3.2	3.2	35GP03
1282	SS35	Water	1992	1,4-Dichlorobenzene		jugit	12	-	· <b>—</b>	-	
1253	5535	VVater	1892		Q.1	HOVL	13		-	-	350000
1244	0000	THERE	1992		0.0		13	1	0.12	0.12	350902
1200	3333	TVELE:	1000	4.4 *00 l	. U.1	PO/L	13	¥	<u>v.14</u>	0.16	3339902
1287	5535	Water	1907	Anna-RHC	0.00		13	-	-		
1210	SS35	Water	1992	Aracior-1016	1		13	-	_		····
1280	SS35	Water	1992	Arocior-1221			13			<u></u>	
1290	SS35	Water	1997	Arocior-1232		ug/L	13				
1291	SS35	Water	1992	Arocior-1242	1	UQ/L	13	-	-	-	
1292	5535	Water	1992	Arockir-1248		Jug/L	13		_		
1293	SS35	Water	1992	Arockor-1254		ING/L	13		-	-	

	A	6	C	D	E	F	G	H	1	1	К
	1		··		****	· · · · · ·		+			!
1 1	Source Area	Media	Oate	Analyte	OL.	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1294	SS35	Water	1992	Arocior-1260	1	µg/I_	13		· _	•	
1295	ISS35	Water	1992	Benzene	2	µg/L	12	1	3.5	3.5	35GP03
1296	SS35	'Water	1992	Beta-BHC	0.05	ug/L	13	1	0.05	0.05	35GP01
1297	SS35	Water	1992	Carbon tetrachloride	1	µg/L	12		-	!_	1
1298	\$\$35	Water	1992	Chlordane	01	µg/L	13		i <b>_</b>	·	;
1299	SS35	Water	1992	Chloroform	0.5	ug/L	12	_		·_	
1300	SS35	Wale:	1992	Cis-1 2-dichloroethylene	1	un/L	12	1	: 11	11	35GP01
1301	SS35	Water	1992	Deita-BHC	01	Jo/L	13	**	-	i ##	
1302	5535	Water	1992	Dieldrin	0.05	uc/L	13	· •••	· ····	+	**************************************
1303	\$\$35	Water	1992	Endosuitan	01	ug/1	13			· <del></del>	· ······
1304	\$\$35	Water	1992	Endosultan lu	0.5	Lon/l	13				
1105	15535	1A/ater	1992	Endocultan sultaie	0.5	LIO(	13				
\$ 206	9936	Moter	1002	Fodoo		00/	13	·			·
1107	5635	Minter	1002			Pyrc Nati	13				
1307	5035 6036	VYdle:	1002			PY'L		·	<b>_</b>		
1300	CC 75	Water	1000	Commo PHC (Codena)		por-	12		-	-	250004
1308	3333	vvaler	1992		0.05	P9/-		······	0.07	0.07	336P04
1310	3335	vvater	1992	Meptachior		μ <del>α</del> ντ	13	·	-	-	········
1311	3335	vvater	1992	Heptachlor epoxide		:havr	13			-	
1312	5535	vvater	1992			HQ/L	12	7	5.7	68	35GP01
1313	5535	water	1992	Methoxychior	2	μ <b>α</b> /L	13		-	-	
1314	5535	vvater	1992	Methylenechloride	5	hð\r	12				
1315	SS35	Water	1992	Tetrachloroethene	0.5	µg/L	12	<b>_</b>	-	<b>_</b>	
1316	5535	Water	1992	Toiuene	2	µg/L	12	1	3.3	3.3	35GP01
1317	5\$35	Water	1992	Toxaphene	2	µgA.	13	-	-		
1318	\$\$35	Water	1992	Trans-DCE	1	hðv	12:	-	-	-	
1319	SS35	Water	1992	Trichioroethene	1	Harl	12	-	-	-	
1320	\$\$35	Water	1992	Vinyi chloride	2	.μ <b>g/</b> L	12	-	-	-	
1321	\$\$35	Water	1992	Xylenes (total)	5	HG/L	12	1	9.4	9,4	35GP02
1322	S\$35	Surface Soil	1990	2-Methylnaphthalene	30	uq/kg	4	2	50	2.10E+04	355804
1323	SS35	Surface Soil	1992	4 4'-000	20	' µg/kg	8	6	29	5.10E+03	35DIR05
1324	SS35	Surface Soil	1992	4.4'-DDE	10	µg∕kg :	8	7	11	1.90E+04	35DIR05
1325	SS35	Surface Soil	1992	4.4'-DDT	20	hð kö	8	7	69	4.90E+04	35DIR05
1326	SS35	Surface Soil	1986	Aldrin	1	ipg/kg	3	-	-	-	
1327	SS35	Surface Soil	1992	Aidan	10	µg/kg	81			-	
1328	SS35	Surface Soil	1992	Alpha-BHC	10	µg/kg	8	1	17	17	35DIR05
1329	SS35	Surface Soil	1988	Alumnum (sed)	N/A	µg/kg	3	3	4.59E+06	9.25E+06	35M02
1330	\$\$35	Surface Soil	1992	Arocior-1016	100	JOA/KO	8		-	-	
1331	SS35	Surface Soit	1992	Aroctor-1221	200	ua/ka	8	-	-	-	····
1332	SS35	Surface Soil	1992	Arocior-1232	200	jua/ka j	Bi	_		••••••	
1333	5535	Surface Soil	1992	Arocior-1242	100	ua/ka	8			-	······································
1334	\$\$35	Surface Soil	1992	Arociar-1248	100	uarko	8			-	
1315	\$\$35	Surface Soil	1992	Araciar-1254	100	ua/ka i	8	_		_	
1336	5535	Surface Soil	1992	Ameior-1260	100	un/ka	A			_	·····
13.7	SS35	Surface Soil	1988	Barium (sed)	N/A	ua/ka	3	3	5 17E+04	1 105-05	35M02
1310	\$\$35	Surface Soil	1988	Received (sed)	320	uc <u>erce</u> (	3		461	1015-01	35402
1110	\$\$35	Surface Sol	1902	Reta-RHC	10	⊥CX^V.i	\$	4	100	1012703	3501004
1170	5535	Surface Coll	1988	AHC beta	1	10000	1			-	
1.17	\$\$35	Surface Coul	1988	Cadmum (sed)	254	DOVA			847	2045-02	354402
1142	SS35	Surface Soil	1988	Calcum (sed)	 N/A	Londen -	3		2416-04	5 BREAM	35402
1111	5535	Surface Col	1984	Chiomane						-	
1	5535	Surface Coll	1900	Chivitan		Lug/ing	3		1	3	355804
	CC15	Surface Call	1002						5 790	3	350000
		Surface Soll	1040			HIG/KG		٤	102	410	3501700
1.340	5535	Surface Set	1055	Cabali (200)	N/A	- Paper Lig	3	3	CUT2CEUS	LONGTUR	351102
1.34/	3333 <u>-</u>	Surface Sol	1000	Connec (Red)	N/A		3		3.205403	0.112403	301402
1.140	3333	Surrace Soll	1900	Copper (sec)		howd			1.136+04	2.302+04	35002
1343	3333	Surface Soll	1900	000, #P		Paved	3	2	4	16	JONNUT
1350	3335	SUITACE SOI	1880		U.4	PB/Kg	3	2	Q.4	390	335804
1351	5535	Surface Soil	1988	DDE. PP	N/A	havka	3	3	5	39	35M01
1352	5535	Surface Solt	1990	UUE PP	N/A	How a	3	3	0.09	306	355804
1353	5535	Surface Soil	1988	OOT, pp'	N/A	ihðika	3	3	16	573	35M01
1354	SS35	Surface Soil	1990	ODT, pp	N/A	hðvrð	3	3	4	356	355804
1355	SS35	Surface Soil	1992	Deta-BHC	20	pg/kg	8	-	-	-	
1356	SS35	Surface Soil	1988	Dieldnin	1	portg	3		_	-	
1357	\$\$35	Surface Soil	1992	Dieldrin	10	ug/kg	8	-	-		
1358	SS35	Surface Soil	1992	Endosulfan I	10	iµg/kg	8	ÿ	31	31	35DIR05
1359	SS35	Surface Soil	1992	Endosulfan fi	10	pg/kg	8			-	
1360	\$\$35	Surface Soil	1988	Endosu#an suifate	1	µg/kg	3	-		-	
1361	\$\$35	Surface Sod	1992	Endosulfan sulfate	20	ua/ka	8	-	-	-	

4	-		

. - .

j	A	9	<u> </u>	D	Ē		[ _ F	<u> </u>	н	<u> </u>	l	ĸ
	i i				_			_				
	Source Area	Media	Date	Analyte	OL		Units	#Samples	#Delect	Min Conc	Max Conc	Max Location
1362	SS35	Surface Soil	1992	Endna		10	hð\kð		<u> </u>	20	20	135DIR05
1363	SS35	Surface Soil	1988	Endnn aldehyde		1	hð\kð	3		<b>.</b>	i —	<u> </u>
1364	\$\$35	Surface Soit	1992	Endrin aldehyde		0.2	µg/kg	8	-	-	-	<u> </u>
1365	5535	Surface Sorl	1990	Fluoranthene		30	µg/kg		**	-	_	
1366	SS35	Surface Soil	1992	Gamma-BHC (Indane)		10	µg/kg	8	1	1.10E+03	1.10E+03	350(R05
1367	\$\$35	Surface Soil	1990	Heptachlor	N/A		µg/kg	3	<u> </u>	0.5	0.6	355804
1368	SS35	Surface Soil	1992	Heptachlor		10	µg/kg	. 8		-	-	•
1369	SS35	Surface Soil	1992	Heptachior epoxide		60	hð\kð	. 8	. 1	13	13	35DIR05
1370	\$\$35	Surface Soil	1988	Iron (sed)	N/A		µg/kg	3	3	9.00E+06	1.82E+07	35M02
1371	\$\$35	Surface Soil	1992	Lead	N/A		µg/kg	6	8	6.00E+03	1.20E+05	35DIR07
1372	\$\$35	Surface Soil	1988	Lead (sed)	4 S6E	+03	ug/kg	3		-	-	
1373	\$\$35	Surface Soil	1990	Lead (sediment)	N/A		µg/kg	. 4	4	2.64E+03	6.16E+03	355807
1374	SS35	Surface Soil	1988	Magnesium (sed)	N/A		µg/kg	3	3	2.84E+06	5.38E+06	35M02
1375	SS35	Surface Soil	1968	Manganese (sed)	N/A		ug/kg	3	3	1.90E+05	3.38E+05	35M02
1376	SS35	Surface Soil	1992	Methoxychlor		100	µ¢/kg	8	-	_	-	
1377	\$\$35	Surface Soil	1990	Naphthalene		30	ua/ka	4	- 1	3.60E+03	3.60E+03	35SB04
1378	\$\$35	Surface Soil	1988	Nickel (sed)	N/A		ua/ka	3	3	1.04E+04	1.99E+04	35M02
1379	\$\$35	Surface Soil	1990	Phenanthrene		30	ua/ka	4	-	-	_	
1380	\$\$35	Surface Soil	1988	Po(assam) (sed)	N/A		ua/ka	3	3	4 72E+05	7 68E+05	35402
1384	6616	Sudace Sol	1990	Purene		30	ualka			-	-	
1387	5535	Surface Soll	1989	Sodum (sed)	N/A		uo/ko	1	9	1985+04	3 685-05	35402
1383	5535	Surface Coll	1003	Torachène		400	ug/kg		_	-	4.046703	
1384	5515	Surface Sol	1989	ТРи	1 02E	+04	uo/ko		4	2 75E-04	2 755-04	351402
1	6635	Surface Coll	1000	ТРы		20	uo/ke	ر الار	1	40	3.805-04	365804
1300	2032	Surface Sof	1000	Vagadium (sed)	N/A	20	walke		23	1 675-04	3.002703	33384
1.00	6635	GuildGe 308	1040	Zine (4ed)	N/A		LIG(PA	3		2 77E+04	5.000404	351102
130/	3333	Sed	1000	2 Mathuineanthalana	110	20	ug/kg			40	3.415704	33802
1305	2032	Soll	1000	2-methykiaphthalens					4	40	10	355803
1389	2222	SOR	1300		NIZA	-	- ugrig	•	1	0.4	1.405.07	35001
1390	3333	201	1960	Alumnum (sed)	N/A	·	howe	•		0.43E+00	1.196+07	33802
1381	2222	204	1989:	Banum (200)	IN/A		hànd			7.4/6404	3.882+05	35002
1392	5535	201	1966			320	i hônkô		3	(92	1.03E+03	35501
1393	5535	SOI	1968:	BHC, Deta		1	-hðræð		1	1.8	1.8	35M01
1394	5535	Sol	1988	Cadmum (sed)		433	hävid	4	-	-	-	
1395	\$\$35	Sol	1988	Calcum (3ed)	N/A		µg/kg		4	3.23E+06	1.14E+07	35M02
1396	\$\$35	Soil	1988 '	Chlordane		_1	hðvkð	4	1	12	12	35\$01
1397	\$S35	Soil	1990	Chlordane		Z	hð\kð	5	4		269	35\$\$03
1398	SS35	Sail	1988	Chromium (sed)	N/A		hð kð	4	4	1.19E+04	2.13E+04	35501
1399	SS35	Soil	1988	Cobalt (sed)	N/A	İ	i pavka	4	4	6.97E+03	1.12E+04	35501
1400	SS35	Soil :	1988	Copper (sed)	N/A	i	µg/kg	4	4	1.57E+04	3.30E+04	35801
1401	\$\$35	Sol	1988	DDD, pp'		1	µg/kg	4	3	43	1.12E+03	35M01
1402	\$\$35	Sol	1990	DDD, pp	N/A		µg/kg	5	5	39	5.85E+04	355503
1403	S\$35	Soil	1988	ODE, pp			hð kö	4	3	107	1.35E+03	35M01
1404	SS35	Soil	1990	DDE. pp'	N/A		µg/kg	5	5	25	9.71E+03	355503
1405	SS35	Soil ·	1988	DDT, pp	N/A		µq/kg	4	4	5.6	3.21E+04	35M01
1406	SS35	Soil	1990	DDT, 99'	N/A		pg/kg	\$	5	111	3.96E+05	35\$\$03
1407	SS35	Soil	1988	Diekinis		1	iµg/kg	4	-	-		
1408	\$\$ <b>3</b> 5	Soil	1988	Endosulfan sulfate		1	µg/kg	4	-	-		1
1409	S\$35	Soil :	1968	Endrin aldenyde		1	µg/kg	4	-	-	=	[
1410	SS35	Soil :	1990	Fluorenthene		30	µg/kg	5	2	50	70	355503
1411	SS35	Soil i	1990	Heptachior	N/A		µg/kg	5	5	0.5	0.6	355808
1412	\$\$35	Soil	1968	fron (sed)	N/A		pyg/kg	4	4	1.20E+07	2.40E+07	35501
1413	5535	Soil	1968	Lead (sed)	4.56E	+03	iµg/ka	- 4	1	4.51E+04	4.51E+04	35M02
1414	SS35	Soil	1990	Lead (sediment)	N/A		µg/ka	5	5	6.34E+03	1.61E+04	355503
1415	SS35	Soil	1988	Magnesium (sed)	N/A		ug/ka	4	4	3.73E+06	6.42E+04	35501
1416	SS35	Soil	1988	Manganese (sed)	N/A		µg/ka	4	4	2.00E+05	9.00E+05	35501
1417	SS35	Soil	1990	Naphthalene		30	µa/ka	5	-	_		[
1418	SS35	Soll	1968	Nickel (sed)	N/A		La/ka	4	4	1.405+04	2495+04	35501
1410	SS35	Soil	1990	Phenanthrane		30	UO/ro	5		30	30	355803
1420	\$\$35	Soil	1988	Potessium (sed)	NA		uo/to			6 215-05	9 345-04	35501
1427	5535	Soil	1990	Pyrana		30	un/en	5		70	70	355503
1477	\$\$15	Soil	194.6	Sodium (sed)	N/A	~~	LANKO			2 625-05	ARSELOR	35501
1755	\$\$35	Soil	1000	ТРы	1 075	م	LING AND			1 395-04	7 975-05	15501
172	5535	Soil	1000	ТРМ	1.946	10	100/4-0	14	اد.	1,000	2 685-05	34580
1111	\$\$35	Soil	1988	Vanadium (sed)	N/A		INCOM	5- 1-	1 <b>4</b> 1	2425-04		15402
140	9916	Sol	1069	Time (and)	N/A		HALL REAL			1.74CTV4	7 945-04	36201
	0000	Woter 1	1007	1 1 1 Trichlonathane		0.41					1.07574	
1.7	9930 6616	Sinter	1000			0.3		2		-		
120	3339	I A COLOR	1892			<u>U.Ə</u>	HUL .	2	-	-	-	
1429	3330	AA <b>X16</b> 1	1992			11	HOVE !	2			-	l

	A	8	¢,	0	E	F	G	н		L 1	ĸ
1					_		<b>.</b> .				
1	Source Area	Media	Date	Analyte	L	Units	#Samples	#Delect	Min Conc	Max Conc	Max Location
1430	5536	Water	1992	1 2-Dichloroelhane	0.5	µg/L	2		i <b></b>	<u>j-</u>	
1431	5536	Waler	1992	1 4-Dichiorobenzene		: µg/L	Z		<u> </u>	<b> -</b>	
1432	5536	Water	1992	Barum	200 M14			· · · · ·			26.2
1434	S\$36	Water	1992	Ben zene	<u> </u>	Linu/I				1 <u>330</u>	
1435	5536	Water	1992	Bervilum		ual	3	. <u> </u>		[	
1436	SS36	Water	1992	Cadmum	10	ud/L	3				
1437	\$536	Water	1992	Çalcıum	N/A	µg/L	3	3	6.50E+04	1 1.60E+05	36-2
1438	SS36	Water	1992	Carbon tetrachioride	1	µg/L	2	. —	-	-	
1439	SS36	Water	1992	Chloroform	0.5	µg/L	2	i <b>-</b>	·	1-	
1440	SS36	Water	1992	Chromum	20	ug/L	. 3	i <b>-</b>	-	–	
1441	\$\$36	Water	1992	Cis-1,2-dichloroethylene	1	µg/L	2	<u>]</u> _	-	-	
1442	SS36	Water	1992	Coban	20	ŀµg/L	3	[-	-	-	]
1443	SS36	Water	1992	Copper	20	ι,μ <b>ο/L</b>	3	!—	-	-	
1444	SS36	Water	1992	Ethylbenzene	2	:µg/L	2	i-	-	-	
1445	SS36	Water	1992	lron	N/A	µg/L	3	3	2.90E+03	1.90E+04	36-2
1446	SS36	Water	1992	Magnesium	N/A	µ9/L	3	3	1.50E+04	2.50E+04	36-2
1447	SS36	Water	1992	Manganese	N/A	µg/t	3	3	1.70E+03	4.40E+03	36-2
1448	\$\$36	Water	1992	Methylenechloride	¢	µg/L	Z	]	<u> -</u>		
1449	8836	Water	1992	Nickel	30	HQ/L	3	-	-	-	
1450	5536	vvater	1992	Polassium	N/A	ug/L	3	3	3.70E+03	4.90E+03	36-2
1451	2220	Water	1992	Codum	20	ug/L	3		 E 005-02	- 8 305 403	76.7
1452	5550	. VVdlef	1001	Tetraphorosthese	0.5	ug/		; J	3.302703	0.30C+U3	30-2
1454	3330	Water	1002	Tie	100			<u> </u>			······
1455	5536	Water	1992	Tokiege	2	uo/i i	2	-	<u> </u>		
1456	SS36	Water	1992	Trans-DCE	 1	uo/L	2				······
1457	SS36	Water	1992	Trichloroethene	ī	ua/L	2			-	
1458	SS36	Water	1992	Vanadium	30	µg/L	3	i_	<u> </u>	-	······································
1459	SS36	Water	1992	Vinyl chloride	2	ue/L	2		-	-	
1460	SS36	Water	1992	Xylenes (lotal)	5	ug/L	2	-	-	-	
1461	SS36	Water	1992	Zinc	10	HQ/L	3	1	12	12	36-2
1462	SS36	Surface Soil	1986	Lead	N/A	hove	11	11	4.00E+03	7.00E+03	36-1
1463	SS36	Soil	1986	Antimony	5.00E+03	.hovka	1	-	-	-	
1464	SS36	Soil	1992	Antimony	2.00E+04	ind ka	2	-	-	-	
1465	SS36	Soil	1986	Arsenic	:N/A	ihð ká	1	1	2.40E+04	2.40E+04	36SD-2
1466	\$\$36	Soil	1992	Barum	N/A	navið	2	2	3.40E+04	5.20E+04	36MW03-2
1467	5536	501	1986	Servilium	NVA 1000	INDARD	1	<u>                                     </u>	1.402+03	1.40E+03	36SD-2
1468	5536	Sol	1992		500	Lugare I	4	<b>I</b> -			
1403	0030	Soil	1001	Cadimy um	1.005+03	inclea		-			
1479	5536	Sol	1992	Caicing	N/A	un/ka			2 305+06	2 405+08	3640403-2
1472	SS36	Sol	1946	Chromeen	N/A	uo/ka			1.70E+06	1.70E+06	365D-2
1473	5536	Sal	1992	Chromium	N/A	ug/ka	2	2	5.00E+03	7.00E+03	36411/03-2
1474	SS36	Soil	1992	Cobat	N/A	luo/ka	2	2	4.00E+03	4.00E+03	36MM03-1
1475	SS36	Soil	1986	Copper	N/A	IHG/kg	1	1	8.30E+04	8.30E+04	36SD-2
1476	SS36	Soil	1992	Copper	N/A	µg/kg	2	2	4.00E+03	1.40E+04	36MW03-2
1477	S\$36	Soil	1986	DOD, pp	N/A	µg/kg	1	1	15	15	36SD-1
1478	SS36	Soil	1986	DDE pp	:N/A	iµg/kg	1	1	80	80	3650-1
1479	SS36	Soil	1966	DDT, pp'	, N/A	µg/kg	1	1	240	240	36SD-1
1480	SS36	Soil I	1992	iron	N/A	HO/KQ	2	2	5.10E+06	7.60E+06	36MW03-2
1481	SS36	Soil	1986	Lead	<u>N/A</u>	µg/kg	3	3	1.00E+04	7.80E+06	36SD-2
1482	5536	501	1992	Magnesum		hðvka	2	2	1.80E+06	2.30E+06	3000003-2
1483	5536	Sod	1992	Manganese	N/A	haved	2	2	7.80E+04	1.20E+05	36MW03-2
1484	3330	Sol	1905	Net Cury	. 100 N/A	199.49	1	ļ <del>.</del>	-		7460.0
1/405	0000	- 30ii - 1	1001	Nietal	.M/A	WWW			+.00C+U4	9.005-04	3030-2
1/17	5536	Soil	1002	Pristan	N/A	100/60			3 205-04	4.00CTU3	3610403-1
1400	SS36	Sol	1944	Selensum	-N/A	iuo/ko			600	AUD 800	3650-2
1489	SS36	Soil	1986	Silver	500	jug/ka		<u> </u>	-		
1490	SS36	Soil	1992	1 Silver	2.000+01	Jug/ka	2			_	·
1491	SS36	Sor	1992	Sodium	.N/A	jug/ka l	2	2	2.50E+05	2.90E+05	3640403-1
1492	SS36	Soil	1986	Thallium	5.00E+03	N9/Kg	1		-	-	
1493	SS36	ISoil ;	1992	Tin	1.00E+04	HQ/kg	2	-		-	
1494	SS36	Soil	1992	Vanadium	N/A	149/kg	2	2	9.00E+03	1.30E+04	36MW03-2
1495	\$\$3 <del>6</del>	Soil	1986	Zinc	N/A	ING/kg	1	1	9.60E+04	9.60E+04	36SD-2
1496	SS36	Soil	1992	Zinc	N/A	Hg/kg	- 2	2	1.00E+04	1.60E+04	36MW03-2
1497	SS37	Water	1994	(Technical) chlordane	1	HIGAL I	5	-	-	-	

FINAL

. .

	A	9	С	Ū.	Ē	F	G	Ĥ	1	<u> </u>	ĸ
		•									
1	Source Area	Media	Date	Analyte	ΟL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1498	SS37	Water	1994	1,1 1-Trichtoroethane	1	µg/L	5	. –	-	-	
1499	SS37	Water	1994	1 1.2.2-Tetrachloroethane		µg/L	5		·	-	-
1500	SS37	Water	1994	1 1 2-Trichloroethane	05	µg/L		<u>.                                    </u>	! <u>-</u>	-	
1501	S\$37	Water	1994			µg/L	5	· —		•••••••••••••	· · · · · · · · · · · · · · · · · · ·
1502	SS37	vvater	1994			HQ/L	5	-			·
1503	3331	Warer	1994		10	<u>µg/L</u>	0		ļ <del>.</del>	!**	
1604	5331	Water	1004	1.2-Dichloroethage	0.5	ugit.	<u>،</u>			·	·
1506	8837	Water	1994		0.5	ua/L	ś	· · · · · · · · · · · · · · · · · · ·			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
1507	5537	Water	1994	1 3-Dichlorabenzene	10	ud/L		1_		+	
1508	SS37	Water	1994	1.4-Dichlorobenzene	10	ua/L	16	-	·	•	
1509	5537	Waler	1994	2.4.5-Trichlorophenol	10	uc/L	6	  -	÷	-	
1510	SS37	Water	1994	2.4.6-Trichlorophenol	10	µg/L	6	<u> </u>	I —	-	
1511	SS37	Water	1994	2 4-Dichlorophenol	10	µg/L	6	-	-	_	
1512	SS37	Water	1994	2.4-Dimethylphenol	• 10	µg/L	6	-	-	-	
1513	SS37	Water	1994	2.4-Dinitrophenol	50	µg/L	6	-	-	-	
1514	SS37	Water	1994	2.4-Dinarotoluene	50.	µg/L	12	-	-		
1515	SS37	Water	1994	2.6-Dinitrotoluene	10	µg/L	6	-	-	-	
1516	S\$37	Water	1994	2-Chloroethylvinyt ether	2	ug/L	5	-		-	
1517	SS37	Water	1994	2-Chioronaphthaiene	10	ug/L	6	-	-	-	
1518	5\$37	Water	1994	2-Cnicrophenol	10.	μgν	6	=	••		······
1519	SS37	Water	1994	2-Methylobasol	10	ug/L	6	-		-	
1020	333/	Water	1994	2-Mitmaniline	10:	PPL 1	b	_			······
1827	2337	Water	1994		30* 10:	ug/L			-		
1571	9937	Water	1904	3 3'-DichlorobenZirline	20		Â				
1524	SS37	Water	1994	3-Nitroantine	50	ua/L	6	-	-		
1525	SS37	Water	1994	4.4'-DDD	0.1	µ0/L	5	-	-	-	
1526	SS37	Water	1994	4.4'-ODE	0.11	HO/1	5	-		-	
1527	SS37	Water	1994	4.4-ODT	0.1	JQ/L	5	_	-	-	
1528	SS37	Water	1994	4-Bromophenyl-phenylether	10,	µg/L	6	i—	-	-	
1529	SS37	Water	1994	4-Chioro-3-methylphenol	20	µg/L _	6	-	-	-	
1530	SS37	Water	1994	4-Chloroaniline	20	hðyr :	6	-	-	-	
1531	\$537	Water	1994	4-Chlorophenyl-phenylether	10	ug/L	6	-	-	-	<b>-</b>
1532	\$\$37	Water	1994	4-Methylphenol	10j	navr				-	·····
1533	SS37	Water	1994	4-Nitroanime	20	HOL		-		-	· · · · · · · · · · · · · · · · · · ·
1534	5537	Water	1994	4-Nitrophenol		Hou :	5:	-	<b>-</b>		
1333	383/	VY2R6F	1994		0.05	port.		-			
1330	5537	Water	1004	Arenanbinene	101						
1538	SS37	Water	1994	Acenaphiliviene	10		6	-			
1539	SS37	Water	1994	Aldrin	0.05	uan	5	-		_	· · · · · · · · · · · · · · · · · · ·
1540	SS37	Water	1994	Alumnum	N/A	µg/L	10	10	46.8	387	37-1
1541	S\$37	Water :	1994	Anthracene	10	ug/L	6	-		-	
1542	S\$37	Water	1994	Antimony	1	,Jg/L	10	-	-	-	
1543	SS37	Water	1994	Arsenic	N/A	HQ/L	10	10	6.7	56.7	37-1
1544	S\$37	Water	1994	b-BHC	0.05	HQ/L	5	-	-	-	
1545	S\$37	Water	1994	Barium	N/A	µg/L	10	10	97.7	207	37-1
1546	5537	Water	1994		<b>1</b> 1	uovL (	5	-		<b>-</b>	
1547	553/	water	1994	Sector(s)anglessie	10		5	-	-		
1948	353/	vvater Mater	1004	Benzy/bit warthers	10						·····
1660	5537	Water	1004	Renznia h unerdene	101	100	C 4		-		
1551	\$\$37	Water	1004	Benzo(k Muorzothene	10)	Lunvit					·····
1552	5537	Water	1994	Benzoic acid	501	ua/L	6	-			
1553	SS37	Water	1994	Benzyi skohol	20	POL I	6	-	-	-	
1554	SS37	Water	1994	Beryllium	1	ug/L	10	-		-	
1555	SS37	Water	1994	Bis(2-chloroethoxy)methane	10	µg/L	8	-	-	-	
1556	\$\$37	Water	1994	Bis(2-chloroethyl)ether	10	MQ/L	6	-	-		
1557	SS37	Water	1994	Bis(2-chloroisopropyl)ether	10	HQ/L	6	-	-		
1558	S\$37	Water	1994	Bis(2-ethythexyl)philialate	10	µg4_	5	2	2	2.3	37-1
1559	SS37	Water	1994	Bromodichloromethane	1	ugit	5	-	-	-	
1560	\$537	Water	1994	Bromoform	<u>1</u>	µg/L	5	-	<b>_</b>	-	
1561	5537	Water	1994	Bromomethane	2	HOL	5	-	-	-	
1552	5537	vvaler .	1994		10	HOL	5	-		-	
1003	1003/	vvaler Water	1004		1) N/A	100	101	-	-		37-26
1646	5537	Water	1004	Camazole	10.		۱۷ غ			3.13ETUR	31.42
							Q1				

.

	A	8	С	D	E	F	G	] н	1	[	к
											1
1	Source Area	Media	Date	Analyte	01	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1566	SS37	Water	1994	Carbon tetrachioride	0.5	iµq/L	5	-	' <b></b>	-	
1567	SS37	Water	1994	Chlorobenzene	1	µ9/L	10	· ••	_	-	
1568	<b>S\$</b> 37	Water	1994	Chioroethane		µg/L_	5	-	-	-	
1569	\$\$37	Water	1994	Chloraform		µg/L	5		. <u> </u>	-	
1570	SS37	Water	1994	Chloromethane	1	µ\$/L	5		-		
1571	\$\$37	Water	1994	Chromium	1	µg/L	10	5	1.7	4.2	37-4D
1572	\$\$37	Water	1994	Chrysene	10	μ <b>g/L</b>	6				i
1573	\$\$37	Water	1994	Cis-1.3-dichloropropene	0.5	µg/L	5		, <b>-</b>	<u> </u>	
1574	S\$37	Water	1994	Cobail	1	pg/L	10	2	1.1	1.1	37-1
1575	SS37	Water	1994	Copper		µg/L	10	5	1.1	9	37-1
1576	\$\$37	Water	1994	d-BHC	0.05	µg/L	5		-	L	
1577	\$\$37	Water	1994	Di-n-butyiphthalate	10	ug/L	6	i <b>-</b>	-	-	
1578	SS37	Water	1994	D-n-octylphihalate	10	µ9/L	6			; <del></del>	
1579	SS37	Water	1994	Dibenz(a,h)anthracene	10	µg/L	6	-	-	-	1
1580	\$537	Water	1994	Dibenzofuran	tú	µg/L	6	-	-	-	
1581	\$\$37	Water	1994	Dibromochloromethane	1	ug/L	: 5		-	-	
1582	SS37	Water	1994	Dichlorodifluoromethane	1	hđ/r	5	-	-		
1583	S\$37	Water	1994	Dieldrin	0 1	µg/L	\$	i-	-	-	
1584	\$\$37	Water	1994	Diethylphthalate	10	µg/L	6	i-	-	-	· · · · · · · · · · · · · · · · · · ·
1585	\$\$37	Water	1994	Dimethylphthalate	10	µg/L	6	-		-	
1586	SS37	Water	1994	Endosulfan !	0 05	μ <b>g/</b> L	5	-	-	-	
1587	ISS37	Water	1994	Endosulfan I:	0.1	ug/L	5	r —	-	-	
1588	S\$37	Water	1994	Endosuifan sulfale	01	µg/L	5	, <b></b>	_	-	
1589	S\$37	Water	1994	Endrin	0.1	µĝ/L	5	-	-	i <b>—</b>	
1590	SS37	Water	1994	Endrin aldehyde	0.1	Ha/L	5		-	-	
1591	\$\$37	Water	1994	Ethylbenzene	1	ug/L	5	-	-		
1592	SS37	Water	1994	Fluoranthene	10	JQ/L	6		=		
1593	SS37	Water	1994	Fluorene	10	µg/L	6	-	-		
1594	SS37	Water	1994	a-8HC	0.05	ua/L	5		-		<b></b> -
1595	\$\$37	Water	1994	o-Chlordane	0 05	ua/L	5	r	-	•	
1596	SS37	Water	1994	Heplachior	0.05		5	-		-	
1597	SS37	Water	1994	Heptachlor spoxide	0.05	uo/L	5	i—		_	
1598	SS37	Water	1994	Hexachlorobenzede	10	uo/L	6	-		_	······································
1599	5537	Water	1994	Hexachloroputadiene	01	JUJL	B	-	-		
1600	\$\$37	Water	1994	Hexachiorocyclopentadiene	10		6	)	<b>_</b>		·······
1601	\$\$37	Water	1994	Hexachloroethans	10	uc/L	6	-	-	-	
1602	\$\$37	Water	1994	indeno(1 2.3-cd)pyrene	10		6	i	••		·
1603	\$\$37	Water	1994	Iron	N/A	UD/L	10	10	1.04E+03	1.70E+04	37.1
1604	\$\$37	Water	1994	ISODIOCOTE	tÖ	iud/L	6	-	_	-	
1605	SS37	Water	1994	Lead	1	ua/L	10	1	3.4	3.4	37-1
1606	SS37	Water	1994	M.D-xviene	1	UQ/L	5	-	-	-	
1607	SS37	Water	1994	Magnesium	N/A	ud/L	10	10	1.04E+04	1.16E+04	37-2F
1608	ISS37	Water	1994	Manganete	N/A	ua/L	10	10	886	2.59E+03	37-2
1609	SS37	Water	1994	Methoxychior	0.5	Jug/L	5	-	-	-	
1610	S\$37	Water	1994	Mathviene chloride	1	100/L	5	-	-		······································
1611	SS37	Water	1994	n-Nitroso-di-n-propylamine	10	UNL.	6				
1612	SS37	Water	1994	n-Närosodiphenylämine	10		6	_	-		
1613	SS37	Water	1994	Naphthalene	10	;ud/L	6	-	-	-	
1614	SS37	Water	1994	Nickel	1	ud/L	10	5	1.3	3.3	37-1
1615	SS37 *	Water	1994	Narobenzene	10	UGA.	6	-			· · · · · · · · · · · · · · · · · · ·
1616	SS37	Water	1994	o-Xviene	1	IUO/L	5	<b>.</b>	-	-	·
1617	SS37	Water	1994	PCB-1015	1	ug/l	5	_		-	
1618	SS37	Water	1994	PC8-1221		Jud/L	5		_	_	
1819	5517	Water	1994	PCB-1232	1	und.	5	_			·
1620	\$\$37	Water	1994	PCB-1242	1	:00/	5				
1621	\$\$17	Water	1994	PCB-1248	•		5		••••••••••••••••••••••••••••••••••••••	_	· · · · · · · · · · · · · · · · · · ·
1822	SS37	Water	1994	PCB-125		ugA.	4		-		······
1671	5537	Waler	1994	PCB-1260	1						
1624	5537	Water	1994	Pentachiotophenoi	50	iua/L	L				·······
1674	5537	Water	1994	Phenanitrine	10	inalia Juni					
1676	5537	Water	1004	Dhend		1007		-			
14.77	597	Water	1904	Dotatta m	N/A	inon i	10		1 305-02	3 625-02	37.7
1827	6637	Water	100/	Purana		I MARKE	V.			U.VEETUJ	····
1020	555	Water	1004	Silver	······	1049	0				
1029	5331	Traller	1004	Soview	1/A	- <u>port</u>	10		4715-00		17.15
1030	3331	v valiçî Moter	1004		1¥A 0 5	- MQ/	10		/IC+03	0.04E+U3	31-1P
1931	3337 Cea7	10/200	1004		U.3	-pg/L	3			-	
14.14	5337	Vine-	1004			1001		<u> </u>	1.4	2.2	
1933		*****	1334		1		. 31	, I	_ '	- 1	

ų

 $\sim$ 

•

	A	<u> </u>	Ċ	0	Ε	F	G	н	1	J	ĸ
									1	,	
1	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1634	<u>SS37</u>	Waler	1994	Totai dissolved solids	100	µg/L	20	15	2.10E+03	2.40E+05	37-3
1635	ISS37	Water	1994	Toxaphene		<u>µg/L</u>	5			<b>_</b>	
1636	5537	Water	1994	Trans-1.2-dichioroethene		ug/L			·		
1637	5031	Water	1094		C V 2 A	UQ/L				·	·····
1630	6917	Water	1094	Techorofuoromethane	1	ucA					
1640	5537	Water	1994	Vanadium		ug/L	10	7	11	4.5	37-1
1641	SS37	Water	1994	Vinyi chloride	0.5	µg/L	5	· _		-	
1642	S\$37	Water	1994	Zinc	2	µg/L	tO	7	2.3	15.3	37-1
1643	SS37	Surface Soil	1986	2-Butanone	NiA	µ@/kg	1	1	75	75	37-2
1644	\$\$37	Surface Soil	1988	2-Butanone (mek)	N/A	µg/kg	6	6	19	1.30E+03	3748-6
1645	S\$37	Surface Soil	1986	2-Methylnaphthalene	N/A	µg/kg	1	1	1.20E+03	1.20E+03	37-1
1646	\$\$37	Surface Sorl	1987	2-Methylnaphthalene	N/A	µg/kg	2	2	26	1.10E+04	37AB-1
1647	\$\$37	Surface Soil	1988	2-Methylnaphthalene	N/A	Pg/kg	10	10	46	3.00E+04	37AB-9
1648	5\$37	Surface Soil	1988	Anthracene	N/A	hð\kð	1	1	240	240	37AB-11
1649	5537	Surface Soil	1987	Benzoic acid	N/A	hðukð			65	66	37AB-4
1650	5337	Surface Sol	1990	Dibeo 70 trac		ug/kg	1	1	430	430	3748-11
1631	5537	Surface Sol	1907	Dibenzolucin	N/A	Haliko -	2	4	140	390	3740-0
1653	ISS37	Surface Sol	1986	Ethylbenzene	N/A	ho/ko	1	1	110	110	37-1
1654	SS37	Surface Soil	1987	Ethylbenzerie	N/A	µg/kg		1	4.20E+03	4.20E+031	37AB-1
1655	SS37	Surface Soil	1988	Elhyibenzene	N/A	µQ/kg	. 4	4	3	570	3748-9
1656	SS37	Surface Soil	1988	Fluoranthene	N/A	µg/kg	1	1	570	570	37AB-11
1657	5537	Surface Soil	1968	Fluorene	N/A	µg/kg	4	4	70	3.10E+03	37A8-9
1658	SS37	Surface Soil	1986	Lead	N/A	µg/kg	t1	11	4.00E+03	9.00E+03	37-1
1659	S\$37	Surface Soil	1987	Lead	N/A	µg/kg	35	35	3.00E+03	2.00E+04	37-3
1660	\$\$37	Surface Soil	1986	Lead	N/A	µg/kg	17	17	4.10E+03	2.50E+05	3748-10
1661	5537	Surface Soil	1980		N/A	how Kg	1	1	670	570	37-1
1002	2531	Surface Sol	1086		N/A	Long to	3	9	37	1.202-04	37.1
1664	5537	Surface Soil	1987	Phenallihrene	N/A	Judiko	1	1	1.50E+03	1 505+03	37AR.1
1665	SS37	Surface Soil	1988	Phenanthrene	N/A	ua/ka	3	3	52	2.30E+03	37AB-9
1666	SS37	Surface Soil	1987	Phenol	N/A	µg/kg	2	2	26	35	37AB-3
1667	SS37	Surface Soil	1988	Pyrene	N/A	howo	2	2	200	640	37AB-11
1668	SS37	Surface Sol	1986	Toluene	N/A	ug/kg	4	4	28	115	37-2
1669	\$\$37	Surface Sol	1987	Tolvene	N/A	ug/kg j	4	4	2	3.50E+03	37AB-1
1670	\$\$37	Surface Soil	1988	Tokiene	N/A	µg/kg j	6	6	2	1.40E+03	37AB-6
1671	SS37	Surface Soil	1987	Trichloroethene	<u>N/A</u>	,µg/kg	51	5	1	3	3748-4
1672	5537	Surface Soil	1986	Xylenes (lotal)	NVA NVA	PG/Kg	1	1	32	32	37-2
1673	5537	Surface Sol	1987	Aylenes (lotal)	N/A	ugrkg :	21	2	3	4 205+04	37AB-1
1675	6637	Sollace Soll	1988	1 1 1-Tochomethane	N/A	waka i	1	1	73	4.205,100	37_50_1
1676	5537	Soil	1988	2-Butenone (mek)	N/A	uarka	1		17	17	37AB-8
1677	S\$37	Soil	1987	2-Methyinaphthalene	N/A	uarka	2	2	3.60E+03	2.50E+04	37AB-1
1678	SS37	Soil	1988	2-Meth yinaphthalene	N/A	µg/kg	2	2	480	4.30E+03	37AB-0
1679	SS37	Soil ,	1986	4.4'-DOT	N/A	µg/kg	1	1	70	70	37SD-1
1680	SS37	Soil	1988	Benzo(a)anthracene	N/A	µg/kg	1	1	210	210	37AB-8
1681	\$\$37	Sor	1985	Beta-BHC	N/A	how	<b>†</b>	1	5	5	37SD-1
1682	<u>SS37</u>	504	1988	Chrysone	N/A	ihovka	1	1	-290	290	3/AB-8
1683	553/	Sol	1097		IWA MA	LING A	2	2	1 505-04	1.00±+03	3740.2
1624	15537	Soil	1988	Finymestere	.N/A	100/20	<u>، (</u>		2	190	3748.6
1644	SS37	Sod	1987	Fluoranthrene	N/A	ug/ka	3	3	16	66	37-3
1687	SS37	Sot	1988	Fluoranthrene	N/A	10%g	1	1	290	290	37AB-8
1688	\$\$37	Soil	1987	Fluorene	N/A	µg/kg	2	2	67	1.20E+03	37AB-1
1689	\$\$37	Soil	1986	Leed	N/A	µg/kg	2	2	1.00E+04	1.90E+04	37-1
1690	S\$37	Soil	1987	Lead	-N/A	µg/kg	7	7	6.00E+03	3.80E+04	37AB-4
1691	SS37	Sof	1965	Lead	N/A	ihðyka	9	9	5.10E+03	1.80E+05	3748-10
1692	SS37	Soil	1967	Naphrillione	IN/A	µg/kg	2	2	1.90E+03	1.20E+04	37AB-1
1693	5537	501	1958	Naphthalana		(Harka	1	1	3.00E+03	3.00E+03	3/AB-6
1054	333/ 6637	Soul	1097	Chan to https	N/A	UNITED IN	1	1	<b>30</b>	50 410	3748 1
1695	5537	Sort	1988	Phenetitrene	N/A	UQ/XO			200	200	3748.4
1697	SS37	Seil	1987	Pyrene	NA	UGAL	3	3	200	61	37-3
1694	5537	Soil	1988	Pyrene	N/A	µg/ka	1	1	360	360	3748-8
1699	\$\$37	Şoil	1987	Toluene	N/A	ugha	2	2	4.30E+03	7.90E+04	37AB-2
1700	SS37	Sou	1988	Toluene	N/A	191kg	3	3	4	120	3748-8
1701	SS37	Soil	1987	Trichloroethene	N/A	pg/kg	2	2	2	3	37AB-3

	_

	A	B	] ¢ ]	D	٤	F	G	Н	1	3	ĸ
	_				_						
	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1702	\$\$37	Sod	1988	Trichloroelhene	N/A	µg/kg	1	<u>.                                    </u>	3	3	37-SD-2
1703	SS37	Son	1987	Xylenes (lotai)	NA	µg/kĝ	2	~2	1.00E+05	2.10E+05	3748-2
1704	353/	530	1983	A yienes (total)	10	ug/kg			<del>ر</del>	/00	3/40-6
1705	2239/03	Water	1992	0.0.0. methyl phosphorothicate	10	HW-	0 2		-	······································	·····
4702	5339/63	Alater	1992	1 1 Tricploroethane	0.5	ugit			· · · · · · · · · · · · · · · · · · ·	;	
1708	SS39/63	Water	1992	1 1 2-Trichloroethane	0.5	ua/L			••		······································
1709	SS39/63	Water	1992	1 1-Dichloroethane	·	µa/L	S		·	4	·
1710	SS39/63	Water	1992	1.2.4-Trichloropenzene	10	µg/L	6		<b></b>	•	
1711	SS39/63	Water	1992	1.2-Dichlorobenzene	10	µg/L	6	-	-	ţ	· · · · · · · · · · · · · · · · · · ·
1712	\$\$39/63	Water	1992	1.2-Dichloroethane	05	µg/L	5		-	-	· · · · · · · · · · · · · · · · · · ·
1713	SS39/63	Water	1992	1,3-Dichlorobenzene	10	µg/L	6	-	-	-	
1714	SS39/63	Water	1992	1,4-Dichlorobenzene	2	µg/L	11	·	-	-	
1715	SS39/63	Water	1992	1.4-Naphinogumone	10;	µg/L	6	I <b>-</b>	-	-	
1716	SS39/63	Water	1992	1-Naphthylamine	10	ug/L			-	-	
1717	SS39/63	Water	1992	2.3.4.6-Tetrachlorophenoi	10;	µg/L	6	<u>-</u>			
1718	SS39/63	Water	1992	2.4 5-Trichlorophenol	10.	hdvr	6	<u></u>		-	
1719	SS39/63	Water	1992	2.4.6-Trichlorophenoi	10	µg/L	6	<b>-</b>			
1720	SS39/63	Water	1992	2.4-Dichlorophenol	10	ug/L	6	ļ		-	
1721	5539/63	Water	199Z	2,4-Dimethylphenol	10.	μg/L		!-	<b>-</b>		
1722	5539/63	water	1992			HG/L	B		<b>-</b>	L	·
1723	2232/03	VValer	1992		10,						
1724	5630/63	Water	1992	2 5-Dichioropheno/		<u>1975</u>					····
1725	5539/63	Water	1992		10	<u>#915</u>		-	-		
1727	5539/63	Water	1992	2-Chiorona00thale08	10				<b>–</b>		
1728	SS39/63	Water	1992	2-Chlorophenol	10	ua/L	<u> </u>	i=	-	-	
1729	SS39/63	Water	1992	2-Methylnaphthalene	10	µg/L	6	•		-	
1730	\$\$39/63	Water	1992	2-Methylphenol	10,	UG/L	6	-		-	
1731	\$\$39/63	Water	1992	2-Naphthylamine	10,	µg/L	6	<b>I</b> -	-	-	
1732	SS39/63	Water	1992	2-Niroanime	50.	µg/L i	6	-	-	-	
1733	SS39/63	Water	1992	2-Nitrophenol	10	µg/L	6	-		-	
1734	SS39/63	Water	1992	3.3'-Dichlorobenzidine	20	µg/L	6	<b> -</b>	-	-	
1735	SS39/63	Water	1992	3.3'-Dimethylbenzidine	101	ug/L '	6	-			······
1736	SS39/63	Water	1992	3-Methylcholanthrene	10:	µg/L	6	-	-	-	
1737	SS39/63	Water	1992	3-Niroaniline	50-	µg/L		ļ <b>-</b>			
1736	5539/63	water	1992	4.6-Denitro-2-methylphenol	301	HAL		<u> -</u>			
1/39	5539/03	VV2ler	1992	4-Ammoophenyi	10		0 6				· · · · · · · · · · · · · · · · · · ·
1741	530/61	Water	1992	4-Chlom-3-methylphenol	20		<u>.</u>				·····
1742	\$\$39/63	Water	1992	4-Chloroaniine	20	ua/L	6	-			
1743	SS39/63	Water	1992	4-Chiorophenylphanyl ether	1D:	HO/L !	6				
1744	SS39/63	Water	1992	4-Methylphenol	101	HO/L	6	-	_		
1745	SS39/63	Water	1992	4-Nitroanilme	50	µg/1	6	-			
1746	SS39/63	Water	1992	4-Nitrophenol	50	ug/L	6	-	-	-	
1747	SS39/63	Water	1992	4-Nitroquinoline-1-oxide	10	µg/L	8	-	-	-	
1748	SS39/63	Water	1992!	5-Nitro-o-lokuidine	10	HO/L I	6	<u> -</u>		-	
1749	SS39/63	Water	1992	7.12-Dimethylbenz[a]anthracene	10	191		ļ <b>-</b>	-	-	
1750	SS39/63	Water	1992	Accaphiliene	10	hðvr †		ļ <del>.</del>	-	-	
1751	5539/63	water	1992.		101	HOVL		<u> </u>			
1752	0039/63 6610/61	vvater	1992		101		5		<b>-</b>	-	
1.33	003000	Water	1007	Andige	101	HALL !				<u> </u>	
1744	5539/63	Water	1992	Anthracane	10	ua/L	0 R	-	-		
1756	SS39/63	Water	1992	Antimony	2001	µg/L	7	i			
1757	SS39/63	Water	1992	Arame	10:	uo/Li	ß		-		
1758	SS39/63	Water	1992	Arsenic	5	µg/L		1-	-	-	
1759	SS39/63	Water	1992	Baram	201	ug/L	7	5	120	210	39M04
1760	SS39/63	Water	1992	Benzene	2:	hâvr	5	I-	-	-	
1761	SS39/63	Water	1992	Benzo(a)unthracene	10	Mar 1	6	<u>-</u>	-	-	
1782	SS39/63	Water	1992	Benzo(a)pyrane	10	HQL !	6	-	-	-	
1763	SS39/63	Water	1992	Benzo(b)fluoranthene	10	håvr	6	-	-	-	
1764	SS39/63	Water	1992	Benzo(ghi)perviene	101	ug/L İ	6	-	-	-	
1765	SS39/63	Water	: 1992:	Benzo(k)/Junranthene	10;	HO/L	8		-	-	
1766	SS39/63	Water	1992	Benzyl alcohol	201	WWL 1	6	j-	-		
1767	5538/63	- Water	1992	Beryikum	<u>.                                    </u>	HOL		-		-	
1768	12239/63	vvater	1992		10,	HAL		Į <b>-</b>	-	-	
11769	3338/03	: ¥¥æl@ľ	1992	CH5(4+CTIIC) CREATIVE) #CTI#F	10,	ucuri.	6	1-	-	-	

. .

	( A	8	Ċ	D	E	۶	G	н	<u> </u>	J	к
		·····								· · · · ·	
L 1	Source Area	Media	Date	Analyte	<u> </u>	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1770	SS39/63	Water	1992	Bist2-chloroisoBropyl) ether	10	µg/L	6	<u> </u>	i <del>-</del>	-	_
1771	SS39/63	Water	1992	Bis(2-ethylhexyl) phthalate	10	HQ/L	5	<u>.                                    </u>		<u> -</u>	
1772	SS39/63	Water	1992	Bromide	500	havr	1	-		· •	·····
1773	5539/63	Water	1992	Butylbenzylphinalale	10	hđ/r		·	· —	· · · · · · · · · · · · · · · · · · ·	! 
1774	5539/63	vvater	1992	Calence	100		<u>{</u>	. +** 	-	-	-
1775	2272/27	vvater	1992	Carbon tetrachiotide	100.	HQ/L	<u>_</u>		4.702+04	5.105+04	135MQ1
1779	2272/07	Water	1992		200	<u>µg/</u>		<u> </u>	-	·	
1770	2239/03	VVdlet	1002	Chloringted hanzedas	200	LON	17	· <b>_</b>	<b></b>		
1770	6630/63	Wrater	1002	Chlorobenziane	10	P9'5		<u> </u>		·	
1780	553565	Water	1007	Chioroform	 	va/l	×				
1781	5510/61	Water	1997	Chrometra	20	ua/i	7	-			
1787	5539/63	Water	1992	Chorsene	10	J0/1	6				
1781	5539/63	Water	1992	Cis-1 2-dichlorOethviene	1	ua/L	5	:   ==			
1784	5539/63	Water	1992	Coban	20	ug/L	7				
1785	SS39/63	Water	1992	Copper	20.	μαλ	7	1	36	36	39MO4
1786	SS39/63	Water	1992	Di-n-bulyiphinalate	10	µg/L	6		-	-	
1787	SS39/63	Water	1992	Di-n-octylphthalate	10:	ug/L	5	-			
1788	\$\$39/63	Water	1992	Dialiate	10	µg/L	6	-		-	
1789	SS39/63	Water	1992	Dibenz(a.h)anthracene	10	µç/L	6	-	-	—	
1790	SS39/63	Water	1992	Dibenzofuran	10	µg/1_ :	6	-	-	-	
1791	SS39/63	Water	1992	Diethyl phthalate	10	µy/L	6	-	-	-	
1792	SS39/63	Water	1992	Dimethoate	10	µg/L	6	-	-	-	
1793	SS39/63	Water	1992	Osmethyl phthalate	10	µg/L	6		-	-	
1794	SS39/63	Water	1992	Diphenylamine	10:	nd/r	8	-	-	-	
1795	\$\$39/63	Water	1992	Ethyl methanesulfonate	10;	ug/L		-	-	-	······
1796	SS39/63	Water	1992	Ethylbenzene	Z:	µg/L	5	<b>-</b>			
1797	\$\$39/63	Water	1992	Famphur	10:	hovr.				-	
1798	5539/63	Water	1992	Fluoraninene	10	ug/L		<u> -</u>	-	-	· · · · · · · · · · · · · · · · · · ·
1799	5539/63	vvater i	1992		101			-			
1800	5538763	Valer	1992		1001				-		
1001	5535/63	Autor	1002		10-		A				
1802	SS1061	Water	1002	Herachiorocyclopen(adjene	10;	und i	×	L			
1804	5539/63	Water	1992	Hazachiomethane	10:	uo/L	6	-			·····
1805	SS39/63	Water	1992	Herachorochene	10	ua/L	6				
1806	\$539/63	Water	1992	Hexachioronconelle	10.		6	-			
1807	SS39/63	Water	1992	Indeno(1,2,3-cd)ovrene	10	ua/L	6				
1808	SS39/63	Water	1992	Iron	20	µq/L ;	7	5	1.00E+03	9.10E+03	39M04
1809	SS39/63	Water	1992	Isodrim	10	µg/t (	6		-	-	
1810	SS39/63	Water	1992	Isophorone	10)	ug/L	6	-	-	-	
1811	SS39/63	Water	1992	Isosafroie	10	µg/L	6	-	-	-	
1812	SS39/63	Water	1992	Kepone	10	ha/L	6	-		-	
1813	\$\$39/63	Water	1992	Kerosene	1.00E+04	HQ/L	6	-	-	-	
1814	SS39/63	Water .	1992	Lead	5	µg/L	1	-	-	-	
1815	5539/63	Water	1992	m-Cresol .	10	µq/L	6	-	-	-	
1816	SS39/63	Water	1992	m-Dinkrobenzene	10	µg/L	-8	I-		-	
1817	5539/63	Water	1992	Magnesium	100	havr	7	5	9.80E+03	1.10E+04	39MO4
1111	5539/63	Water -	1992	Mang <b>anese</b>	10	149AL		5	T.40E+03	3.30E+03	397403
1815	3339/63	vvaler	1892		10		5	<del>~</del>	-	-	
1020	233063	Vialas	1007	Man IV: (Ingular)esuaultale	101		0 2			<u> </u>	
	5539/63	Water	1992	Manual property of the second se		unni i					
1823	5530/63	Visio	1992	- Altronotica traviania	10	100			-	-	
1874	SS39/63	Water	1992	n-Nitrosodisthylamne	10	uc/L		-			
1825	SS39/63	Water	1992	n-Nitrosodimethylamine	10		6	-	-	-	······
1826	SS39/63	Water	1992	n-Nitrosodiphenviamme	. 10	บอโ	6	) —	-		
1827	SS39/63	Water	1992	n-Nitrosomethylethylemine	10	ug/L		-	-	_	
1828	SS39/63	Weter	1992	n-Nitrosomorpholine	10	UC/L	8	-	-	-	·····
1829	SS39/63	Water	1992	n-Nirosopiperidine	10	Hg/L	8	-	-	-	
1830	SS39/63	Water	1992	Nephihaiane	10	Mart I	6	-	-	-	
1831	SS39/53	Water	1992	Nickel	30	µQ/L	7	-	-	-	
1832	SS39/63	Water	1992	Nitrate	200	HQ/L	1	-	-	_	
1833	SS39/83	Water	1992	Nitrile	200	ug/L	1	-		-	
1834	SS39/63	Water	1992	Nitrobenzene	10	ug/L	6	-	-	-	
1835	SS39/63	Water	1992	Nilrosopyrrolidine	10	µç/L ∐	6	-	-	-	
1836	5539/63	Water	1992	o-Toluidine	10	ug/L	6	-	-	-	
1837	\$\$39/63	Water	1992	p-Dimethylaminoazobenzene	10	µ.y/L	8	-	-	-	

	A	в	c	D	E	F	G	н	1	1 3	K
_											
1_1_	Source Area	Media	Date	Analyte	, DL	Unit	s #Samples	#Oetect	Min Conc	Max Conc	Max Location
1838	SS39/63	Water	1992	p-Phenylenediamine		10 µg/L		i _	· —	-	1
1839	SS39/63	Water	1992	Parathion	· · · · · · · · · · · · · · · · · · ·	10 µg/L	(	<u>i -                                    </u>			
1840	SS39/63	Wate:	1992	Pentachioranitropenzene (pCRD)	··	10 µg/L		<u>} -</u>	<u> </u>	!+- 	
1841	SS39/63	Water	1992	Pentachiorophenol	···	50 µg/L		<u>i                                    </u>	.=	· <b>-</b>	<u>!</u>
1842	SS39/63	Water	1992	Phenacetin		10 µg/L		<u>.</u>	<u> </u>	·	·
1843	SS39/63	Water	1992	Phenanthrene		10 µq/L		<u>-</u>			i
1844	SS39/63	Water	1992	Phenol		10 µg/L		). —	~		
1842	2272/02	vvare:	1992	Prosphate					-	- 2 805+02	1201104
1840	5539/63	water	1335	Porassium		10 µg/L			3.20E+03	3.80E+03	23M04
1041	2030403	vvate:	1954	Pronaimoe	· · · · · · ·				· ~	·	
1040	5039/03	YVdiCi	1992	ryiene Sofrai	•••••••				— 		<u> </u>
1840	5539/63	Avalet	1007	Ciluar		20 00/				[,	[
1841	5539/65	Water	1092	Sodarm	34	10 un/l		- -	300	4 605+03	139404
4867	5510/63	Water	1992	Sulfate	Ść	10 unit				4.000000	
1251	5519/63	Wals	1992	Sym-tonizobeozene		10 uo/l	 6				
1854	5519/61	Water	1992	Tetrachlomethene	õ	5 µg/L			-		
1855	5539/67	Water	1992			0.00/2			_	_	
1856	SS39/63	Water	1992	Tin	10	X0 µa/L		_	······································		
1857	SS39/63	Water	1992	Toluene		2.µ0/L	5			,	·······
1858	SS39/63	Water	1992	Total organic carb	N/A	µg/L	: 1	! 1	5.00E+03	5.00E+03	39-5
1859	SS39/63	Water	1992	Trans-DCE	,,`	1 µg/L	5	j_	-	-	
1860	SS39/63	Water	1992	Tributyl phosphate	1	ιο μογί	6	<u>i</u> –	_	_	
1861	SS39/63	Water	1992	Trichloroethene		1 µg/L	5	1 <b>-</b>	-	-	
1862	5539/63	Water	1992	Vanadium		30 µg/L	7	-		-	
1863	SS39/63	Water	1992	Vinyi chloride		2 µg/L	5	i —	-	-	
1864	SS39/63	Water	1992	Xylenes (total)	· ··· · ··· · ·	5 µg/L	<u>; 5</u>	, i <b></b>	-	-	
1865	5539/63	Water	1992	Zinc	1	l0 µg/L	7	3	13	32	39M04
1866	SS39/63	Surface Soil	1989	2-Meinyinaphinalene	3	10 µg/kg	17	2	30	1.20E+03	39SB15
1867	\$\$39/63	Surface Soil	1968	Akuminum	N/A	µg/kg		4	2.85E+06	7.26E+06	395601
1868	SS39/63	Surface Soil	1988	Arsenic	5.39E+0	3.µg/kg			-	-	
1869	5539/63	Surface Soil	1985	Sanum	N/A	havka		4	4.43E+04	9.20E+04	39M03
1870	\$\$39/63	Surface Soil	1988	Beryilum	32		4	2	565	750	39M01
1871	5539/63	Surface Soil	1968	BHC, Deta		1 µgvkg	·	1	2		39MU1
18/2	5539/63	Surface Sol	1906	BHU, gamma (Indane)	4 6				-	-	201404
10/3	5610/63	Surface Sol	1900	Calorum		U. PORC		<u>, 3</u>	1 205-06	1.000403	39401
1074	SC30/63	Surface Soil	1900	Calculate	<u></u>	1.00%		•	1.200 100	3.125700	30403
1876	5535/63	Surface Soil	1988	Chromium	N/A	Jugarag		ų	5 45E+03	3 235+04	39401
1877	5339/63	Surface Sol	1988	Cobalt	N/A	-ua/ka			3.50E+03	8 25E+03	394401
1878	SS39/63	Surface Soil	1988	Copper	N/A			3	9 30E+03	2 21E+04	39401
1879	\$\$39/63	Surface Sol	1988	000 op		1 u0/ko		1	8	8	39M03
1880	SS39/63	Surface Soil	1989	DDD po		1:µa/ka	17	9	1	328	39SB14
1881	SS39/63	Surface Soil	1988	DDE. pp	~·······	1:µg/kg	1 4	1	2	2	39M03
1882	SS39/63	Surface Soil	1989	DDE, pp'		1 µg/kg	17	8	1	127	395802
1883	SS39/63	Surface Soil	1988	DDT, pp		1.µg/kg	4	3	4	13	39M03
1884	SS39/63	Surface Soil :	1989	DDT. pp		1.µg/kg	17	9	1	437	39SB02
1885	SS39/63	Surface Soil	1988	Diekinn		1 pang	1	11-	-	-	
1886	SS39/63	Surface Soil	1988	Endosulfan sulfate		1 µg/kg	j 4	-	-	-	
1887	SS39/63	Surface Soil	1988	Endosulfan, a		1;µg/kg	4	1	2	2	394/03
1888	SS39/63	Surface Soil	1988	Endosultan_b		1 jug/kg	4	-	-	-	
1889	SS39/63	Surface Soil	1986	Endrin aldehyde		1 µg/kg	4	1	2	2	39M03
1890	5539/63	Surface Solt	1989	Fluorene		KO I HO/Kg	17	3	30	680	39SB15
1891	5539/63	Surface Soil	1988	Heptachlor epoxide		T HO/kg	4	-		-	
1692	5539/63	Surface Soil	1988	Iron	N/A	HQ/Kg	4	4	1.50E+06	1.53E+07	30000
1893	5539/63	Surface Soil	1909	Sophorone		17. µg/kg	17	2	30	1.30E+04	395801
1494	0039/03	Surface Soil	1992		13	N INDAKO	13	3	930	J.30E+03	3435URA
1895	2219/03	Surface Soil	1986			how		4	4./96+03	1.15E+04	399007
1896	3339/63	Surface Sol	1908	Magnessum	N/A	have		4	1.096+06	4.346+06	304404
1091	223303	Surface Cold	1000	manyef 1920 Nanhih siens		U U U U U U U U U U U U U U U U U U U			1.410400	6.000405	1008204
1090	5539/63	Surface Coll	1092	Nickal	N/A		<u></u>	4	3U	330	30401
1900	SS39463	Surface Soil	1988	Polassium	N/A	10/200		4	2.636404	5.445-04	39401
1001	SS3963	Surface Sol	1989	Sodium	N/A	LUD/In A		· · · · · ·	1.3RE-OF	2.855-04	39401
1907	SS39/63	Surface Soil	1992	Total petroleum hydrocarbons	2.00F+0	H µa/ka	7	13	2.20F+04	6.90F+05	39SS07A
1903	SS39/63	Surface Soil	1988	TPH	N/A	UQ/ka		1	2.05E+04	2.15E+04	39403
1904	SS39/63	Surface Soil	1989	ТРН	1	0:00/10	56	32	20	1.78E+07	395903
1905	\$539/63	Surface Soil	1988	Vanadium	N/A	µa/ka	4	4	1.18E+04	2.69E+04	39M01

 $\sim$ 

<b></b>	A	8	<u> </u>	D	E	F	<u> </u>	<u> </u>	1	<u> </u>	ĸ
Ι.	l		_		-						
	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
1906	\$\$39/63	Surface Soil	1988		N/A	hđyka		4	2.21E+04	4.52E+04	39M01
1907	SS39/63	504	1989	2-Methyinaphthalene		ia häved	Z		/	-	l
1908	SS39/63	Sof	1988		N/A	µg/kg	3	3	6.47E+06	1.04E+07	391403
1909	SS39/63	Sal	1988	Arsenic	6 39E+0	13 µ <b>g/kg</b>		1	8.89E+03	0.696+03	139M03
1910	\$\$39/63	So:	1988	Banum	N/A	hð\kð		3	7.522+04	1 9.92E+04	139M03
1911	SS39/63	Soil	1988	Beryllum	N/A	ug/kg	3	3	462	1.45E+03	39M03
1912	SS39/63	Soil	1988	BHC. beta		1 µg/kg	<u>:</u> 3	·		<b> -</b>	ļ.
1913	SS39/63	Şoil	1988	BHC. gamma (lindane)		1 μ <b>g/kg</b>	1 3	i-	-		
1914	SS39/63	501	1988	Cadmium	49	is µg/kg	3	2	743	1.29E+03	39M03
1915	5539/63	Sod	1988	Calcum	N/A	µg/kg	3	3	2.70E+06	4.28E+06	391403
1916	\$\$39/63	: Soil	1988	Chiordane		1 µg/kg	3	1	11	11	39M03
1917	SS39/63	Sol	1968	Chromium	N/A	ug/kg	: 3	. 3	1.17E+04	1.87E+04	39403
1918	SS39/63	Soil	1988	Cobalt	N/A	µg/kg	3	3	6.228+03	9.39E+03	39M03
1919	SS39/63	Sod	1988	Copper	N/A	ug/kg	4	1 4	1.32E+04	2.19E+04	39M03
1920	5539/63	Soil	1988	DDD. pp	N/A	µg/kg	3	3	4	262	39M03
1921	SS39/63	Soil	1989	DDO, pp		1 ug/kg	2	I_	-	I_	1
1922	SS39/63	Soil	1988	DDE. DE'		1 ua/ka	3	2	7	56	39M03
1923	5539/63	Sol	1989	ODE. 00'	····	1 µg/ko	2		<b>—</b> .		
1924	\$\$39/63	Soil	1988	DDT. po	N/A	uo/ka	3	1 3	6	356	39403
1925	5539/63	Soil	1989			tuo/ka					1
1926	SS39/63	Spil	1988	Diektrin		1 µ0/ko	1	l <b></b>			
1977	5539/67	Sol	1988	Endosulfan sulfate		1 uo/xe					
1078	CC30/63	Sal	1089	Endosultan	·	1 unika					204402
1020	0030/03	Soul	1098	Sndotoffan b	···-	1 uadka	1 3	<b>_</b>	····		3376433
1023	2230403	Sol	1089			1 worke		-	- 50		201402
1034	0010461	Sal	1080	Elionate		D LIGHTO			×	<b></b>	338405
1931	5533703	Sou	1000	Fillerene		1 usika					
1332	222802	Sol	1000		AU/A	- Party			-	-	201400
1333	223302	- 301 Call	1000			JUP NG	3		1.200 +0/	2.00E+0/	JSMNUJ
1834	5539/03	. 301	1303	Isophorone	J	O HONG	4	-	-		
1935	2239/03	301	1992	Kerosena	13	/:µg/kg	21		2.70E+03	8.20E+04	395502C
1936	5539/63	50#	1968	Lead	N/A	ihavka	3	3	9.92E+03	1.40E+04	39M03
1937	5539/63	Sol	1988		- N/A	hours	3	3	3.74E+06	5.45E+06	39M03
1938	\$\$39/63	Soil	1980	Manganese	N/A	Though	3	3	2.03E+05	2.84E+05	39M03
1839	\$\$39/63	Soil	1989	Naphthalene	3	0.µg/kg	2	-	-		
1940	SS39/63	Sol	1988	Nickel	N/A	hð, kð	3	3	1.55E+04	2.01E+04	39M03
1941	SS39/63	Sol	1988	Potassium	N/A	hð ver	3	3	5.80E+05	7.05E+05	39M02
1942	SS39/63	Soil .	1968.	Sodium	<u>N/A</u>	µg/kg	3	3	2.49E+05	3.89E+05	39M03
1943	SS39/63	Sol	1992	Total petroleum hydrocarbons	2.00E+0	4 iµg/kg	47	14	8.90E+04	3.20E+07	39SS06C
1944	SS39/63	Soil	1988	ТРН	N/A	pg/kg	3	3	6.21E+05	4.10E+08	391403
1945	\$\$39/63	Son	1989	ТРН	1	0 µg/kg	33	27	42	5.82E+04	39501
1946	SS39/63	Soil	1968	Vanadium	'N/A	µg/kg	3	3	2.45E+04	3.78E+04	391403
1947	SS39/63	Soil	1968.	Zinc	-N/A	pg/kg	3	3	4.18E+04	5.18E+04	39M03
1948	\$\$57	Water	1992	1,1,1-Trichloroethane	0.	5 pg/L	3	-	-	—	
1949	SS57	Water	1992	1,1.2-Trichloroethane	0.	5 µg/L	3	-	- 1	-	(
1950	SS57	Water	1992	1,1-Dichloroethane		1109/1	3	I–		-	
1951	SS57	Water	1992	1,2-Dichloroethane	C.	SiverL	3	1	5.3	5.3	575802
1952	SS57 i	Water	1992	1.4-Dichlorobenzene		2'µc/L	3	-	-	-	
1953	SS57	Water	1992	Benzene		2 401	3	2	5	530	575802
1954	SS57	Water	1992	Carbon letrachloride	1	1:001	3	-	_		
1955	SS57	Water	1992	Chloroform	. 0.	SING/L	3	_	-	-	
1956	\$\$57	Water	1992	Cis-1.2-dichloroethylene	INVA	Jug/L	3	3	1.1	73	57SB02
1957	\$\$57	Water	1992	Ethylberizene		2 401	3	1	180	180	575802
1958	5557	Water	1997	1 eet	· · · · · · · · · · · · · · · · · · ·	51001	3	2	5.8	93	575802
1959	5557	Water	1992	ListhyleGechiowie		Siud/	3				
1960	\$\$57	Water	1907	Tetrachiomethene	<u> </u>	51.000		1		1	575902
1641	6057	Water	1002	Tohuse		2	3	·	1 005-02	1 005-02	675802
1045	0057	Water	1007	Trent DCE		1			1.00	1.872103	513002
4041	0057	Water	1001	Techimethese	·						
1001	000/	Water	1002	Transford General	, m		3			<u> </u>	
	0001	TOUS	1000	Villeast (Minis	<u> </u>	A POL			1000.00	-	670000
1205	3331	TVALUES :	1992	Ayenes (ICAR)	•	SINGL	3	└ <u></u>	1.002403	1.00E+03	375802
1366	\$\$57	SUITACE SOI	1992	1,1,1-Irichioroethane	0.	SINGING	3	<u> </u>	<b>-</b>	-	ļ
1967	355/	SUITROE SOIL	1992	T, T, Z- I IICHIOTOETHERE	<u> </u>	o have	3		<u> </u>	÷	
1968	5\$57	Surface Soil	1992	T.1-Dichloroethane			3	-	-		
1969	SS57	Surface Soil	1992	1.2-Uichlorosthane	0.	5 µg/kg	3		-		[
1970	SS57	Surface Soil	1992	1.4-Dichloroberizene		2 pono	3	<u> l</u>	<u> </u>		
1971	S\$57	Surface Soil	1992	Benzene		2 µ <b>9/k</b> 9	3	1	2.2	2.2	57SB03-A
1972	SS57	Surface Soil	1992	Carbon tetrachloride	_ 100 000000000000000000000000000000000	1 ug/kg	3		-	-	
1973	SS57	Surface Soil	1992	Chloroform	<b>O</b> .	Siµg/kg	3		-	-	í

. — —

	A	В	C	0	Ε	F	G	н	l i	<u> </u>	K
· · · ·			<b></b>								!
1	Source Area	Media	Date	Anaryte	DL	Ųnits	#Samples	#Detect	Min Conc	Max Conc	Max Location
1974	SS57	Surface Soil	1992	Cis-1.2-dicnloroethylene	1	µq/kg	3	<b></b>	-	i—	
1975	SS57	Surface Soil	1992	Ethylbenzene	2	µg/kg		2	6.7	2.60E+04	575802-A
1976	5557	Surface Soil	1992	Lead	N/A	µg/kg		1	4.20E+03	4.20E+03	575801-A
1070	0007	Surface Soil	1992	Teirachioroetheae	ت ۸ ۵	µg/kg			· ······························	• •	
1070	3337	Surface Sol	1002		2	notice			19	3 805+04	675903 A
1980	SS57	Surface Soil	1992	Trans-OCE		uo/ko		<u> </u>	-	-	310002-1
1961	SS57	Surface Soil	1992	Trichloroethene	1	ud/xa					
1982	SS57	Surface Soil	1992	Vinyi chloride	2	µg/kg	3		i <del></del>	_	
1983	SS57	Surface Soil	1992	Xylenes (total)	5	µg/kg	3	2	11	1.50E+05	57\$802-A
1984	SS57	Soil	1992	1 1 1-Trichloroethane	0.5	µg/kg	3	i	-	-	
1985	S\$57	Soil	1992	1 1 2 Trichloroethane	0.5	µg/kg	3	<u></u>	-	-	
1986	SS57	Sou	1992	1.1-Dichloroethane	1	ug/kg	3		-	-	
1987	\$\$57	Soil	1992	1.2-Dichloroethane	0.5	µg/kg	3	·		-	
1988	\$\$57	Sort	1992	1.4-Dicfilorobenzene		µg/kg		1 <b></b>	<b>_</b>	-	
1989	\$\$57	500	1992	Benzene	2	hð\kd			-	-	
1990	333/	500 Sed	1992			ug/kg	3	! <u>-</u>		-	
1007	0007	Sol	1992	Cis.1 2-deblocosthylene	<u>v.</u> 1	uo/ko	3		· · · · · · · · · · · · · · · · · · ·	-	
1993	SS57	Sol	1997	Ethylpenzane	2	ua/ko	3	1	2 40E+03	2 406+03	57SB02-B
1994	SS57	Soil	1992	tead	N/A	ug/ka	1	1 1	3.90E+03	3.90E+03	57\$801-8
1995	\$\$57	Sol	1992	Methylenechlonde	5	µ@/kg	3	i=		-	
1996	SS57	Soil	1992	Tetrachloroethene	0.5	µg/kg	3	i <b>-</b>	-	[ <del>_</del>	
1997	SS57	Şoil	1992	Toluene	2	µg/kg	. 3	2	16	1.608+03	57S802-8
1998	SS57	Son	1992	Trans-DCE	1	μογκο	. 3		-		
1999	\$ <b>\$</b> 57	Soil	1992	Trichloroethene	<u>1</u>	hð ykð	<u> </u>		-	-	
2000	SS57	Soil	1992	Vinyi chionde	2	µg/kg	3		-	-	
2001	<u>SS57</u>	Sol	1992	Xylenes (total)		pa/kg	3	1	8.30E+03	8.30E+03	575802-8
2002	5561 5561	water	1994	1.1.2.2.Tetrachienethane	·····	HOUL	3	, <u> </u>		-	
2003	2261	Water	1994	1 1 2-Trobleroethane	- <u></u>		3				
2005	SS61	Water	1994	1 1-Dichloroethane	1	·uu/L	3			_	
2006	SS61	Water	1994	1.1-Dichtoroethene	0.5	ua/L	3	i–	-	-	
2007	SS61	Water	1994	1,2.4-Trichlorobenzene	10	ipg/L	5	I–	-	-	·····
2008	SS61	Water	1994	1.2-Dichlorobenzene	10	µg/L	11	5	1.9	19	61MW01DD
2009	SS61	Water	1994	1,2-Dichloroethane	0.5	hð.	3	-	-	-	
2010	SS61	Water	1994	1.2-Dichloropropane	0.5	ug/L	3	-	-	-	
2011	SS61	Water	1994	1,3-Dichlorobenzene	. 10	ug/L	11	-	-		
2012	SS61	Water	1994.	1.4-Dichicrobenzene	10	ug/L	11	<u>i –</u>	<u></u>	-	·
2013	5561	VValer	1994	2.4.5-1 richiorophenol	10	IPO/L	2	<u> -</u>	-		
2014	3301 CCC1	VValer Minter	1994	2,4.0-17kChiorophenor	10	HOVL	5	-		-	
2015	2001	Water	1004	2.4-Dimethutobergi	10	LING A	5	-		<u> </u>	
2017	SS61	Water	1994	2.4-Dinitrophenol	50	UQA	5	-		_	
2018	SS61	Water	1994	2.4-Dinitrololuene	. 50	ug/L	10			-	
2019	SS61	Water	1994.	2.6-Oinstrotoluene	10	µg/L	; 5	1-	-	-	
2020	SS61	Water	1994	2-Chloroethylvmyl einer	2	µg/L	3	1-	-	-	
2021	SS61	Water	1994	2-Chioronaphthalene	10	iµg/L	5	i-	-	-	
2022	SS61	Water	1994	2-Chlorophenol	10	µg/L	5	-	<b>→</b>	-	
2023	SS61	Water	1994	2-Meinyihaphthalene	10	HQ/L	5	4	6	16	61MW02
2024	3361	VVater	1984	2-Memphine	10			1-	-	-	
2023	3301	Water	100.4	2-Nitrophenol	10	100		1	-		
2027	SS61	Water	1994	3.3'-Dichlorobenzidine	20		5			-	
2028	SS61	Water	1994	3-Nitroaniline	50	μο/L	5	1-	-	-	
2029	SS61	Water	1994	4-Bromophenyl-phenylether	10	µg/L	5	1_	-	-	
2030	SS61	Water	1994	4-Chloro-3-methylpheno!	20	µg/L	5	-	-	-	
2031	SS61	Water	1994	4-Chloroaniline	20	HQ/L	5	-	-	-	
2032	S\$61	Water	1994	4-Chlorophenyi-phenyiether		µg/L	5	-	-	-	
2033	SS61	Water	1994	4-Methylphenol	10	ug/L	5	2	18	16	61MW01DD
2034	SS61	VValer	1994	4-NRTGENHING	20	1 <b>40/L</b>	5	<b> </b>	-	-	
2035	0001	vvale!	1994				3	<u> </u>	-	<u> </u>	
2012	5561	Water	1994	Acensonthviene	10	ug/l		-		<u> </u>	
2018	SS61	Water	1994	Akiminum	N/A	UC/L	4	4	661	2.42E+04	61MW01
2039	SS61	Water	1994	Anthracene	10		5	i-		-	
2040	\$\$61	Water	1994	Antemony	1	µg/L	4	]	—	-	
2041	SS61	Water	1994	Arsenic	N/A	μc/L	6	6	12.7	81.2	61MW02

· ·

	1			n	Т Б		r		1	<b></b>	1
	<u>+^</u>	<b></b>	<u> </u>	l	J	<u> </u>	<u> </u>	<u>.                                    </u>	<u>ا ا</u>	<u></u>	<u>K</u>
1				• • •	-						
	Source Area	Media	Date	Analyte	DL	Units	#Samples	i #Detect	Min Conc	Max Conc	Max Location
2042	SS61	Waler	1994	Banum	N/A	µg/L	S	. 5	228	1.34E+03	61MW01
2043	SS61	Water	1994	Benzene	2	µ9/L	35	. 2	2.1	2.8	61MW01
2044	SS61	Water	1994	Benzo(alanthracene	10	ug/L	5	-	·	1-	1
2045	SS61	Water	1994	Benzoralovrene	10	un/l		-	<u>+</u>	h	<u> </u>
2045	ICC61	Minter	1904	Dentoi bili joranihene	10					1	+····
2040	0.001	4444C	1004			1975	··			ļ <del>.</del>	+
2047	3301	vvate:	:994	Benzo(g.n.i)perviene		µg/L		. <b></b>	<u> -</u>	<u> -</u>	i
2048	ISS61	Water	1994	Benzo(Kitivoranthene	10	µg/L	5		!	<b></b>	
2049	SS61	Water	1994	Benzoic acid	50	ug/L	5	-	<u> -</u>	-	
2050	SS61	Water	1994	Benzyl alcohol	20	µg/L	5	:		-	
2051	SS61	Water	1994	Beryllum	1	µg/L	6	. 1	1.8	1.8	61MW01
2052	SS61	Water	1994	Bis(2-chloroethoxymethane	10	ua/L	Ś	·	i -	1 <del></del>	
2053	ISSAI	Water	1994	Bis/2-chiproethyliether	10	un/l			t	1	†
2054	CC41	Water	1004	P.c(2.chiwo:corpoylieiber	10	<u></u>		·		1	
20.54	0001	4441G1	1001			PAL.	~~~	<u> </u>		-	
2055	3301	vvater	1994	Bistz-einymexyliphinalate		μçγι	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	<u> </u>	5./	10.1	161 MWO2
2056	5561	Water	1994	Bromodichloromethane	1	µg/L	3	i —	İ	<b></b>	
2057	SS61	Water	1994	Bromotorm	1,	µ9/L	3	:	(—!	<u> </u>	<u> </u>
2058	SS61	Water	1994	Bromomethane	2	μÇ/L	. 3	·	-	-	
2059	SS61	Water	1994	Butylbenzylphthalate	10	µg/L	5	1	2.1	2.1	61MW02
2060	SS61	Water	1994	Cadmium	1	µg/L .	6	1 1	. 1.1	1.1	61MW03
2081	ISS61	Waler	1994	Calcium	N/A	ud/L	6	6	5.57E+04	9.68F+04	61MW03
2042	SS51	Water	1004	Carbazole	10	00/	<u>د</u>	i			
2044	19941	Mater	1004	Carbon tetrachianda	0 E	1000				<u> </u>	<u> </u>
2003	10001	v Validi	1934				3		/ <b></b>	ļ <b>-</b>	L
4004	13361	water	1994		1	HAL		<u> -</u>		E	
2065	5561	Water	1994	Chloroethane	1	HQ/L	3	i—		-	
2066	\$\$61	Water	1994	Chloroform	1:	HQ/L	3	j <b></b>		-	
2067	SS61	Water	1994	Chloromethane	1	µg/L	3	i-	-	]-	
2068	SS61	Water	1994	Chromium	1	μgΛ	4	3	11.8	56.1	61MW01
2069	SS61	Water	1994	Chrysene	10.	µg/L	5	-	_	-	
2070	ISS61	Water	1994	Cis-1,2-dichloroethylene	5	ua/L	32	23	9.9	3.20E+03	61-PS-3A
2071	SS61	Water	1994	Cis-1 3-dichlorontopene	0.5	<u></u>	3	ا میں میں میں ا ا میں	-	_	
2072	SSE1	Mater	1004	Cobat	N/A	100 C	6	A	54	-	6110401
10070	0001	Vidici	1004	Coost					50.0		C10000
2013	3361	vyater	1224		10	<b>yu</b> r			30.6	09.0	C INVVUS
2074	5561	vvaler	1994	Di-n-bunyiphthalate	10	HOL 1	2	2	2.5	3.7	61MW02DD
2075	SS61	Water	1994	Di-n-octylphthalate	10,	µg/L	<u>5</u>	3	1.1	1.5	61MW02
2076	SS61	Water	1994.	Olbenz(a.h)anihracene	10.	µg/L	5	-	-		
2077	SS61	Water	1994	Oibenzofuran	10.	µg∕L ∖	5	-	-	-	
2078	SS61	Water	1994	Dibromochloromethane	1:	µg/L	3	-	-	-	
2079	SS61	Water	1994	Dichlorodifluoromethane	1	uo/L	3	-	-	-	
2080	\$561	Water	1994	Diesel	700	LIGAL	32	_	-	-	
2081	5561	Water	1994	Diethylohthalate	10	LINI I	5				
2082	CCC1	Afater -	1004	Cutorflyinhib slote			5		[]	[	
2002	3301	Vedici	1334					-			AL 00.00
2083	3301	vvaler	1334	Eurydenzene	3	HOL			3.3	<b>5.</b> 0	01-7-3-38
2084	3561	vvaler	1994		10	µ91	5	-	-		
2085	5561	Water	1994	Fluorene	101	µg/L	5	-	-	-	
2086	SS61	Water	1994	Gasoline	120)	havr	32	4	400	2.00E+03	61-PS-3A
2087	SS61	Water	1994	Hexachiorobenzene	10;	HQ/L	5	-	-	-	
2088	SS61	Water	1994	Hexachlorobutadiene	10	µq/L	5	-	-	-	
2089	SS61	Water	1994	Hexachlorocyclopentadiene	10	ua/L	5	-		-	
2090	SS61	Water	1994	Hexachloroethane	10	u <b>a/</b> L	5		······	_	
2004	SS61 1	Water	1004	Indepo(1.2.3_cd)outsta	101	und .	2				
	0001		1004	······································	- MIA	ener.			7 885	1.245	A118104
2092	1000	vvaler	1834		10//		6		1.002403	1.245700	
2093	5561	vvater	1994	ISOPHOLOAC	10	hđvr (	5	<u> </u>	5.7	6.7	01MVV02
2094	5561	Water	1994	Feag	<u> </u>	µg/L	5	4	15.2	40.4	51 <b>MW03</b>
2095	SS61	Water I	1994	M.p-xylene	1;	µg/L j		2	17	290	61MW02
2096	SS61	Water	19941	Magnesium	N/A	µg/L	6	6	1.27E+04	2.64E+04	61MW01
2097	SS61 1	Water	1994	Manganese	N/A	µg/L :	6	6	2.91E+03	8.82E+03	61MM01
2098	SS61 (	Water	1994	Methylene chionde	1	UCIL	3	i_	-	-	
2099	SS61 '	Water	1994	n-Nitroso-di-n-oropylamine	10;	uori	5	_		_	
2100	SS61	Water	1994	o-Nitrosodiobenviamine	1/1	un/			27	3 8	61M640200
17101	SSE1	Water	1004	Nanhthalana	1/1	1997 E.	5		4.7	4.0	61LAACTOON
1100	5501	Water	1004	n reșt le ligigi tă	AUA	1101	2		14 2		
2102	0001	vvaler	1994		19/A	HOLL	6	6	(1.2	153	TUVYMIO
2103	5561	vvaler	1994		101	hður	5			<b></b>	
2104	5561	Water	1994	0-Xyiene	<u>1</u>	µg/L	3	1	6,8	8.8	51MW01
2105	SS61	Water	1994:	Pentachiorophenol	50	HO/L	5	1	2.3	2.3	61MW02
2106	SS61 1	Water	1994	Phenanthrene	10	µg/L	5	1	1.7	1.7	61MW02
2107	S\$61	Water	1994	Phenol	10	1'g/L	5	-		-	
2108	SS61	Water	1994	Potasaium	N/A	HO/L	6	6	6.49E+03	9.746+03	61MW01
2109	ISS61	Water	1994	Pyione	10	ud/L	5	_	-		

FINAL

A.31

.

September 1995

	A	a	C C	0	E	F	Ğ	H	t	1	K J
				······································						·	
1 1	Source Area	Media	Date	Analyte		Units	#Samples	#Oelect	Min Cone	Max Conc.	May I acation
2110	SSR1	14/cura	1004	Silver		U0/1	5	_	-	-	The COLLOCATION
1 111	1996	Walt:	1004	Sadara	NI/6	Pyre .	ž		0 075+01	7 765-01	e 18454000
244.2	0001	YVait:	1334			Pare			9.925403	2.305704	O INIYUZ
2112	2301	VValet	1004			- PG/L		**		· · · · · · · · · · · · · · · · · · ·	
2113	2301	AASIG.	1004	Tation -		<u> </u>				-	
2116	5561	vvater	1994	loivene	····· ····	HQ/L	35		6.8	250	61MW02
2115	3561	vvaler	1994	Trans-1.2-dichloroethene		μ <b>α</b> /L	35	20	3.2	140	61-PS-88
2116	SS61	Water	1994	Trans-1.3-dichloropropene	0.5	p µg/L	3		-	<b></b>	
2117	5561	Water	1994	Trichloroethene (TCE:	1	hð\r	35	15	1	1.10E+03	61-PS-3A
2118	S\$61	Water	1994	Trichlorofluoromethane		µg/L	3	-	-	-	
2119	SS61	Water	1994	Vanadium	1	µg/L	5	4	10.7	165	61MW01
2120	SS61	Water	1994	Vinyi chionde	0.5	ug/L	3	-	-	-	
2121	SS61	Water	1994	Xylenes (total)		µg/L	32	9	3.3	35	61-PS-3A
2122	SS61	Water	1994	Zinc	N/A	μ <u>c</u> /L	3	3	99.1	340	61MW01
2123	SS61	Surface Soil	1994	1.1.1-Trichloroethane	1	µg/kg	1	-	**		
2124	SS61	Surface Soil	1994	1.1.2.2-Tetrachioroemane	1	ua/ka	1	-	-		
2125	SS61	Surface Sort	1994	1 1 2-Trichlorgethane	0 52	ua/ka	1				
2126	SS61	Surface Soil	1994	1.1.Dichloroethane		ua/ka	1			-	
2127	SS81	Surface Soil	1994	1 1-Dichiomethene	0.52	ua/ka	1	-			
2498	5561	Sudace Col	1904	1 2 4-Trichtorobenzene	670	UD/YO		<b>_</b>		-	
1110	SSEI	Surface Col	1004	1.2. Detrivohenzane	270	un/ka					
1	19961	Surface Coll	1002		D10	Harris		{			
14130	0001	Surfects 204	1004			Junite -					
2131	3301	Surrace Soil	1994		0.32	µg/kg				-	
Z132	3301	Surrace Soil	1994		6/0	µg/Kg	3,		<del>-</del>		
Z133	0001	SUITACE Soil	1994		6/0	hði kð	3	-	-	-	
2134	5561	Surface Soil	1994	2.4.3-1000000000000	6/0	hđurđ	<b>1</b>	-		-	
2135	SS61	Surface Soil	1994	2.4.6-Trichlorophenol	670	ug/kg			-	-	
2136	SS61	Surface Soil '	1994	2.4-Dichlorophenol	670	ug/kg	1	-	-	-	
2137	SS61	Surface Soil	1994	2,4-Dimethylphenol	670	hävkä	11	-	-	-	
2138	\$\$61	Surface Soil	1994	2.4-Dinitrophenol	3.40E+03	µg/kg	1	-	-	<u> </u>	
2139	\$\$61	Surface Soil	1994	2.4-Dinitrotoluene	3 40E+03	ug/ko	2	- 1	-	-	
2140	SS61	Surface Soil	1994	2.6-DinArololuene	670	.ug/kg	1,	-	-	-	
2141	SS61	Surface Soil	1994	2-Chloroethylvinyl ether	2.1	µg/kg	1	-	-	-	
2142	SS61	Surface Soil	1994	2-Chioronaphthaiene	670	:µg/kg	1	-	-	-	
2143	SS61	Surface Soil	1994	2-Chiorophenol	670	µg/kg ;	1	-	-	-	1
2144	SS61	Surface Sol	1994	2-Memyinaphthalene	670	.ug/kg	1	-	-	-	1
2145	SS61	Surface Soil :	1994.	2-Methylohenol	670	uu/ka	1	_ 1	_	-	
2146	SS61	Surface Soil	1994.	2-Nitroankine	3.40E+03	ua/ka !	1	_	<del>_</del>	-	
2147	SS61	Surface Soil	1994	2-Nitrophenof	670	uo/ko i	1		-		
2140	5561	Surface Soil	1994	3 3'-Dichlorobenzidine	1 30E+03	uona	1		-		
2149	SS61	Surface Soit	1994	3-Nitroanine	3 40E+03	iuo/ka	1	_	-	-	
2150	SS61	Surface Soil	1994	4-Aromoneny-openyisther	670	ua/ka	1		_	_	
2161	5581	Surface Soil -	1994	4-Chion 3-melbylohenol	1 305+03	10000					
2462	9961	Surface Soil	1004		1 30E+03	10000		_	i		
9424	9981	Surface Call	1004	A Chierrohany chan dethar	270	Under 1	······································				
2481	C261	Currace 201	1004		010 270	Langer Page 1					
4124	5001	Curface 7 -	1004		3 405-00	- Province					
4133	5501	Surface Soll	1004		3.405-03	ING NO					
4130	0001	Surface Coll	1004		J.40E+03	PY NU j			-	-	
2187	5361	Surface Soil	1994		6/0	Parka	1				
2156	3361 ***	Surface Soil	19.24		6/U	IN DRG L			-	-	
2155	3301	Surrace Soil	1994		N/A	have	1		2.33E+06	2.93E+06	ALOVVINIO
Z160	3561	Surrace Soil	1994	Annracene	670	HQ/KQ	1	-	-	-	
Z161	5561	SUITECE Soil (	1994	Antimony	1.00E+03	hðukg	1	-	-	-	
2162	5561	Surface Soil ;	1994	Arsenic	N/A	hđi ka i	1	1	2.70E+03	2.70E+03	61MW01A
2163	SS61	Surface Soil	1994	Barum	N/A	ug ko	1	1	3.86E+04	3.66E+04	61MW01A
2164	SS61	Surface Soil	1994	Benzene	1	µg/kg	1			-	
2165	SS61	Surface Soil i	1994	Benzo(a)anthracene	670	pg/kg	1	-		-	T
2166	SS61	Surface Soil	1994	Benzo(a)pyrene	670	pa/ka	1		-	-	
2167	SS61	Surface Soil :	1994	Benzo(b)fluoranthene	670	iug/kg i	1	- 1		-	
2168	S\$61	Surface Soil	1994	Benzo(g.h.i)perylene	670	µg/kg	1	-	-	-	
2169	S\$61	Surface Soil	1994	Benzo(k)/luoranthene	670	iµg/kg i	1	-	-	- 1	
2170	SS61	Surface Soil	1994	Senzoic acid	3.40E+03	µg/kg i	1		-	-	
2171	SS61	Surface Soil	1994	Benzyl elcohol	1.30E+03	ug/kg	1	-	_	- 1	
2172	SS61	Surface Soil '	1994	Beryllum	1.00E+03	ug/kg	1	_	-	-	
2173	SS61	Surface Soil	1994	Bis(2-chloroeinoxy)methane	670	ua/ka	1		-	_	
2174	SS61	Surface Soil	1994	Bis/2-chloroethyliether	670	UG/kg	1		_		
2175	SS61	Surface Soil	1994	Bis(2-chloroisoproevi)ether	670	Jug/ko	1			<b>_</b>	
2170	SS61	Surface Soil	1994	Bis/2-ethylhexyllohthelate	670	uo/ka	1			_	
7477	9541	Curface Court	10041	Bromodichiotomaticae	•••	LUCARA	+				

FINAL

ц.,	$\sim$

	A	B	C	D	E	F	G	H	I I .	J	K
	1								· · · · · · · · · · · · · · · · · · ·	1	
1	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
2178	SS61	Surface Soil	1994	Bromoform	1	µg/kg		-		-	
2179	SS61	Surface Soil	1994	Bromomethane	21	hðyka	1			-	
2180	\$\$6!	Surface Soil	1994	Butyibenzyiphthalate	6/0	hð\xd				i <del>-</del>	
2181	5561	Surface Soll	1994	Cagmium	10000403	have		<u> </u>	-	-	
2182	0001	Surface Sol	1004		670	Darka			1.37 2700	1.9/2100	ALDWAND
2103	0001	Surface Sol	1994	Carbon tetrachiotide	0.52	LO/ko		-	·	••••••••••••••••••••••••	
2185	ISSA1	Surface Soil	1994	Chlorobenzene	1	ua/ka			+		1
2186	SS61	Surface Soil	1994	Chloroethane	1	ua/ka			-		<b>*</b>
2187	SS61	Surface Soil	1994	Chloroform	1	µg/kg	1				
2188	SS61	Surface Soil	1994	Chioromethane	1	ug/kg	1	-		-	
2189	SS61	Surface Soil	1994	Chromium	N/A	ug/kg	1	. 1	8.10E+03	8.10E+03	61MW01A
2190	SS61	Surface Soil	1994	Chrysene	670	µg/kg	1	ı <b>—</b>			**************************************
2191	SS61	Surface Soil	1994	Cis-1,3-dichioropropene	0 52	µg/kg	1	<u> -</u>	-	-	
2192	SS61	Surface Soil	1994	Cobalt	N/A	µg/kg	1	1	4.10E+03	4.10E+03	61MW01A
2193	SS61	Surface Soil	1994	Copper	N/A	µQ/kg	1	1	1.24E+04	1.24E+04	61MW01A
2194	5561	Surface Soil	1994	Di-n-butylphtnalate	670	hð\kð	1	<u> </u>	-	I <u>—</u>	
2195	SS61	Surface Soil	1994	Di-n-octylphthatete	670	hðykð	1	<u> _</u>	-	<u></u>	l
2196	SS61	Surface Soil	1994	Dibenz(a.h)anthracene	670	hðvkð	1	<u>i-</u>	-	-	i 
2197	5561	Surface Sol	1994	Dibenzofuran	670	µg/kg	1			-	
2196	3301	Surface Soll	1004	Displand (Responsible Sec	· 1	µgritg usition	1				· · · · · · · · · · · · · · · · · · ·
2139	2001	Surface Soil	1004	District Doll World The Control Contro	E7A	HUMAN .		·		-	
2200	5561	Surface Sol	1994	Dimethylophhalate	670	ud/ka	1	-	_	-	
2202	SS61	Surface Soll	1994	Elhybenzene	1.	va/ko	· · · · · · · · · · · · · · · · · · ·				
2203	SS61	Surface Sol	1994	Fluoranthene	670	polico	1	<u></u>			<b></b>
2204	SS61	Surface Soil	1994	Fluorene	670	µg/kg	1	_	· —	-	
2205	SS61	Surface Soil	1994	Hexachlorobenzene	670	µg/kg	1	-		-	
2206	SS61	Surface Soil	1994	Hexachlorobutadiene	670.	ug/kg	1	-	=	-	
2207	\$\$61	Surface Soil	1994	Hexachiorocyclopentadiena	670	ug/kg	1		-	-	
2208	S\$61	Surface Soil	1994	Hexachiorosthane	670	µg/kg	1	-		-	
2209	SS61	Surface Soil	1994	Indeno(1.2.3-cd)pyrene	670	µg/kg	1	-	-		
2210	SS61	Surface Soil	1994	Iron	N/A	hðykð	1	1 1	5.20E+06	5.20E+06	61MW01A
2211	SS61	Surface Soil	1994	isophorone	670	µg/kg	1	<u> -</u>	-	-	
2212	SS61	Surface Soil	1994	Lead	N/A	hðykö i	1	1	5.70E+03	5.70E+03	61MWQ1A
2213	SS61	Surface Soil	1994	M.p-xylene	1	hove	1	<u> -</u>	-	-	A
2214	5561	Surface Soil	1994	Magnesium	N/A	howe		1	1.656+06	1.655+06	BIMYVUTA
2210	3301	Surface Soll	1994	Manganese Manganese	N/A 1	t-gykg	I	<b>'</b>	1.032+03	1.055705	ALOWWALD
2210	9301	Surface Soil :	1994	n-Mimeo-di-o-Oronylamine	670	ug/kg		E	-		
2218	SS81	Surface Soil	1994	n-Nitrosodiphenviamine (1)	670	uarka i					
2219	SS61	Surface Soil	1934	Naphinalene	670	ua/ka	1	-		=	
2220	SS61	Surface Soil	1994	Nicket	INVA	ua/ka i	1	t .	1.03E+04	1.03E+04	61MW01A
2221	SS61	Surface Soli	1994	Nitrobenzene	670	µo/kg	1	-	-	-	
2222	SS61	Surface Soil	1994	0-Xylene	1	yorka	1	-		-	
2223	SS61 1	Surface Soil	1994	Pentachiorophenol	3.40E+03	ug/kg	1	-	-	-	
2224	SS61	Surface Soil	1994	Phenanthrene	670	ugric	1	-	-	-	
2225	SS61	Surface Soil	1994	Phenol	670	uo ko	1			-	
2226	5561	Surface Soil	1994	Pyréne	670	hðurð	1	=		-	
2227	5561	Surface Soil	1994	Siver	1 00E+03	µg/kg	1	=		-	
2228	5561	Surface Soil I	1004		0.32	HUNKO	1	5		-	
2220	1320	Surface Soll	1004	Trate 1 2 dichiometrese	<u>-</u> 1		1	5	-	-	
2734	SS81	Surface Soil	1004	Trans. 1 3. dichiorozonawa	0.52	ua/ko		[ <u> </u>	-	-	
2237	SS61	Surface Sol	1984	Trichlorgethene (TCE)	0.52	uq/ka	1	<u> </u>		_	
2233	SS61	Surface Soil	1994	Trichlorofluoromethume	1	µg/ka	1	1			
2234	SS61	Surface Soil	1994	Vanadium	N/A	µg/kg	1	1	1.50E+04	1.50E+04	61MW01A
2235	S\$61	Surface Soil	1994	Vinyl chloride	0.52	µg/kg	1	-	-	-	
2236	SS61	Soil	1994	1,1,1-Trichloroethane	1	µg/kg	13	1	4.3	4.3	61MW02B
2237	SS61	Soil i	1994	1.1,2.2-Tetrachioroethane	1	µg/kg	13	-	-	-	
2238	SS61	Soil	1994	1,1,2-Trichloroethene	0.5	ug/kg	13	-	-	-	
2239	SS61	Soll	1994	1,1-Dichloroethane	1	ug/kg	13	1	2.7	2.7	51MW028
2240	SS61	Soil	1994	1.1-Dichloroethene	0.5	hàyrà	13	1	0.76	0.76	61MW02B
2241	SS61	Soil	1994	1.2.4-Trichlorobenzens	1.32E+04	hðykð	11	<u> </u>	-	-	
2242	SS61	So#	1994	1.2-Dichlorobenzene	1.32E+04	µg/kg	37		2.9	11	61MM028
2243	5561	Sol	1994:	1.2-Dichioroethane	0.5	have	13	=		-	
2244	5561	Soil	1994	1.2-Dichioropropene	0.5	have	13		-	<b>_</b>	
2245	\$561	Soil	1994	1.3-Dichlorobenzene	1.32E+04	pg/kg i	37	2	2.6	26	61MW02CD

	<u> </u>	8	¢	D	E	L E	G	H		J	<u>к</u>
Ι.			_							1	
1	Source Area	Media	Date	Analyte		Units	#Samples	#Delect	Min Conc	Max Conc	Max Location
2246	5561	So'!	1994	1.4-Lichlorobenzent	32E+04	hð kð		1	. 76	7.6	61MW02A
2247	3301	50	1994	2.4 5-Trichlorophenol	1 925-04	ugrkg	11	. <b>-</b>	· ••		······
2240	000	Sal	1004		1 325+04	HO/KO		· ····			<u>.</u>
2250	SS61	Soul	1994	2 4-Dimethylobegol	1 32F+04	uo/ko	11		·		·
7751	SS61	Sol	1994	2 4-DiortrooBenol	6 60F+04	uJ/ko		 	+		
2252	SS61	Sal	1994	2.4-Dinitratoluene	6 60E+04	ua/ko	22	2 -		 i=	
2253	SS61	Sol	1994	2.6-Dinifrotoluene	1 32E+04	ug/kg	11	<b>_</b>	1 <b>-</b>	i	
2254	SS61	Sol	1994	2-Chioroethylvinyl ether	2	µg/kg	13	_	i <b>-</b>	1-	4
2255	SS61	Sol	1994	2-Chloronaphthalene	1 32E+04	µg/kg	. 11	1-		i-	P
2256	SS61	Sot	1994	2-Chlorophenol	1.32E+04	µg/kg	11	1-		1_	1
2257	SS61	Soil	1994	2-Methyinaphihalene	1.32E+04	µg/kg	11	2	13	1.30E+04	61MW028
2258	SS61	Sof	1994	2-Methylphenol	1.32E+04	µg/kg	11		1_	[-	¥
2259	SS61	Soil	1994	2-Niroaniine	6 50E+04	µg/kg	11	!-	-	) <b></b>	
2260	SS61	Soil	1994	2-Nitrophenol	1.32E+04	µg/kg :	11	í <b>-</b>	[-	-	
2261	SS61	Sol	1994	3.3 -Dichlorobenzidine	2.60E+04	hð\kà			-	<u> -</u>	
2262	SS61	Sor	1994	3-Nitroaniline	6.60E+04	µg∕kg	11	1-	-	l=	
2263	SS61	Sol	1994	4-Bromophenyl-phenylether	1.32E+04	hðvyð	11	·	<b>!</b>	<u> -</u> !	
2264	SS61	Sol	1994	4-Chloro-3-methylphenol	2.60E+04	µg/kg		-	[ <del>_</del>	-	·······
2265	15561	Soil	1994	4-Chloroandine	2.60E+04	µQ/kg	11	<b>-</b>	<b> -</b>	<b></b>	
2266	5561	Soll	1994	4-Uniorophenyi-phenyiciher	1 32E+04	hävkä		1-	[ <del>-</del>	í	
2257	5561	Sou	1994		1.32E+04	have	11		•••••••••	<u>↓</u> )	
2260	5361	501 Sed	1994	4-NR/Danjime		UG/KQ	11		-		
2203	2001	Sou	1004		1 326+04	ug/kg	·	: <b>-</b>		<b>↓</b> ¯	······
2271	5561	Sal	1994	Acepaphitylene	1.326+04	LID/KO	11		E	┟═─────┤	
2772	5561	Sol	1994	Akimoum	N/A	ualka	11	11	2 93E+08	8 48F+08	61LEAD2C
2273	SS61	Sort	1994	Anthracene	1 32E+04	uo/ko	11	1-	_	-	C INITIOZO
2274	SS61	Soll	1994	Animony	200	uu/ke !	11	i <b>_</b>		1	······································
2275	SS61	Soll	1994	Arsenic	N/A	ua/ka	11	11	2.40E+03	1 19E+04	61MW02C
2276	SS61	Soll	1994	Banum	N/A	µo/ka	11	1 11	3.22E+04	7.05E+05	61MW02C
2277	SS61	Soil	1994	Benzene	1.6	µo/kg	13	2	1.8	38	61MW02C
2278	S\$61	Soil	1994	Benzo(a)anthracene	1.32E+04	HQ/kg	11	-	-	-	······
2279	SS61	Soll	1994	Benzo(aipyrene	1.32E+04	ug/kg	11	-	-	-	
2260	SS61	Soil	1994	Senzo(b)fluoranthene	1 32E+04	µg/kg	11	i-	-	-	
2281	SS61	Soil	1994	Benzo(g,h,i)perylene	1.32E+04	µg/kg	11	<u> -</u>	-	-	
2282	SS61	Sol :	1994	Benzo(k)Ruoranthene	1.32E+04	have		<b>∤_</b>	-	-	
2283	SS61	Soli	1994:	Benzoic acid	6.60E+04	µg/kg	<u></u> <u>f1</u>	<u></u>	-	-	
2284	SS61	Soil	1994	Benzyi alcohol	2.60E+04	Horkg	11	<b>-</b>	<b> -</b>	-	
2285	5561	Sor	1994	Berylium	400	Horkg		-		I=	
2286	SS61	Sol	1994	Bis(2-chioroethoxy)methane	1.325+04	havkg	11	<b>↓</b>	-	<b> </b>	
2287	5361	50# Call :	1994	Bis(2-crick beingt)ether	1.325+04	Porkg	11	<b>!-</b>	-		
2200	3301	501	1004		1.325+04		+1		-		
2203	5361	Soil	1004	Bis 2-milys exy() a lineato	1.320704	HO/FO	11 F1		-		
2294	SS61 ·	Sal	1994	Bronotoen	······································	undro	13	<u> </u>	Ξ		
2292	SS61	Sol	1994	Bromomethana	2	ua/ka	13		-	-	
2293	SS61	Sad	1994	Butyloenzyiphthalate	1.32E+04	wa/ka	11	-			
2294	SS61	Soil	1994	Cadmium	1.00E+03	ug/kg	11	1-	-	-	
2295	SS61	Soil	1994	Calcium -	N/A	µg/kg	11	11	1.62E+06	5.45E+06	61MW028
2296	SS61	Soil	1994.	Carbazole	1.325+04	µg/kg	11	1-	-		
2297	SS61	Solt	1994	Carbon tetrachloride	0.5	ug/kg (	13	1-	-	-	
2298	SS61	Sori	1994	Chlorobenzene	1	µg/kg	26	1	100	100	61MW02C
2299	SS61 .	Sort	1994	Chloroethane	1	ugikg	13	-	-	-	
2300	SS61	Soil	1994	Chloroform	1	induxă	13	1	2.4	2.4	61MW02B
2301	<u>SS61 j</u>	Sol	1994	Chiorometisane		have	13	-	-	-	
2302	SS61	Sol	1994	Chromium	N/A	Ha/kg		11	7.00E+03	1.44E+04	61MW02C
Z303	5561 (	SOI	1994		1.32E+04	µg/k0	11	<b> -</b>	-	ļ <b>-</b>	
2304	3361	50# Carl	1994	Cis-1,3-Sichloropropene	U.5	MQ/Kg	13	<b>-</b>	-	-	044 A 4
2303	2301	Soul	19941	Canoor	IN/A	have	11	11	2.00E+03	4.606+03	PIMWUIC
2306	5351 6681	501	1994	Copper Di a tetricitta altra	1 20E+04	hävkä	11	<u> </u>	1.1/2404	4.395+04	o IMWUZC
2307	SS61	994 Sail	1984		1 175+04	PUNG		<u>[</u>		[	
2200	SSA1	Soil	1994	Diberzia hisnihescana	1 325+04	Uraliza	44			<u> </u>	
2310	SS61	Sal	1994	Dibenzofuran	1.32E+04	LO(ko	11	<u> </u>	-	[	
2111	SS61	Sol	1994	Dibromochloromethane	1	UD/EO	11		_		
2312	SS61	Sol	1994	Dichlorodifluoromethene	······	UDER	5				
2313	SS61	Sol	1994	Diethylphthalale	1.32E+04	uq/ka	11				

FINAL

•

	A	в	C	D	E	F	Ğ	Η_	<u> </u>	<u> </u>	ĸ
			·····							 	1
1	Source Area	Media	Date	Analyte	OL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
2314	SS61	Soil	1994	Dimelhylphthalate .	1 32E+04	µ9/kg	11	**		-	I
2315	SS61	Sort	1994	Elhylbenzene	57	µg/kg	13	3	6.1	57	61MW02CD
2316	SS61	Son	1994	Fluoranthene	: 32E+04	µg/kg	11		-	-	· ····································
2317	5561	Soil	1994	Fluorene	1.32E+04	µg/kg	11	2	1	1.50E+03	61MW02B
2318	SS61	Sort	:994	Hexachlorobenzene	1.32E+04	µç/kg	11	-		-	
2319	SS61	Soil	1994	Hexachiorobutagiene	1 32E+04	µg/kg	11	-	-	i—	
2320	SS61	Sout	1994	Hexachlorocyclopentadiene	1 32E+04	ug/kg	11	- 1	-	-	·
2321	5561	Sor	1994	Hexachloroetnane	1.32E+04	ug/kg	11	-	_	-	
2322	\$\$61	Soil	1994	Indeno(1,2.3-cd)pyrene	1.32E+04	µg/kg	11		-	-	1
2323	SS61	Soil	1994	Iron"	N/A	µg/kg	5	5	5.20E+08	1.03E+07	61MW01C
2324	5561	Soil	1994	Isophorone	1 32E+04	H2/kg	11	2	10	1.00E+04	61MW028
2325	ISS61	Soil	1994	Lead	N/A	µg/kg	11	11	3.50E+03	1.95E+04	61MW02C
2326	SS61	Soil	1994	M,p-xylene	62	µg/kg	13	2	62	110	61MW02C
2327	SS61	Soil	1994	Magnesium	N/A	µg/kg	11	11	1.65E+06	5.28E+06	61MW02B
2328	SS61	Soil	1994	Manganese	N/A	µg/kg	11	11	7.19E+04	1.96E+05	61MW01C
2329	SS61	Soil	1994	Methylene chloride	1.	µg/kg	13	2	1.7	2.3	61MW029
2330	SS61	Sor	1994	n-Nitroso-di-n-propylamme	1.32E+04-	µç⁄kg	11	-	-	-	·····
2331	SS61	Sol	1994	n-Nitrosodiphenylamine (1)	1.32E+04	µg/kg	11	-	-	-	
2332	SS61	Sol	1994	Naphthalene	1.32€+04	µg/kg	11	21	35	3.50E+04	61MW029
2333	SS61	Sol	1994	Nickel	N/A	ug/kg	11	11	6.00E+03	1.24E+04	61MW02A
2334	SS61	Soil	1994	Nitrobenzene	1.32E+04	µg/kg	11	-	**	-	
2335	5561	Sol	1994	o-Xylene	44	µg/kg	13	3	1.7	44	61MW02CD
2336	5561	501	1994	Pentachiorophenol	5.60E+04	ug/kg	11	- i	-	-	
2337	\$\$61	Soll	1994	Phenanifirene	1.32E+04	µg/kg	11	2	3	3.90E+03	61MW028
2338	SS61	Soil	1994	Phenol	1.32E+04	µg/kg	11		-	i—	
2339	SS61	Soil	1994	Pyrene	1.32E+04	µg/kg	11	- 1			
2340	\$\$61	Soil	1994	Silver	1.00E+03.	ug/kg	11	- 1	-	- 1	
2341	\$\$61	Soil	1994	Tetrachloroethene (PCE)	7.	hàyka ,	13	4	7	140	61MW02B
2342	SS61	Soil	1994	Toluene	44	µg/kg	13	3	1.B	100	61MW02C
2343	SS61	Soil	1994	Trans-1,2-dichloroethene	1,	hðykð	13	1	2.9	2.9	61MW028
2344	SS61	Soil	1994	Trans-1.3-dichloropropene	0.5	µg/kg	13	-	-	-	
2345	\$\$61 ;	Şoil	1994	Trichloroethene (TCE)	12;	µg/kg i	13	4	1.8	250	61MW028
2346	\$\$61	Soil	1994	Trichlorofluoromethane	1:	µg/kg	13	1	2.3	2.3	61MW028
2347	S\$61	Soil	1994	Vanadium	N/A ·	µg/kg	11	11	1.18E+04	2.22E+04	61MW02C
2348	S\$81	Soil	1994	Vinyl chloride	0.5	µg/kg	13	-	- 1	—	
2349	S\$61	Soil	1994:	Zinc	N/A	10110	12	12	1.39E+04	4.59E+04	61MW02C
2350	SS61-sp	Water	1994	Benzene	21	µg/L	9	-	-	-	
2351	SS61-sp	Water	1994	Cis-1.2-dichloroethylene	6;	µg/L	9	7	9.9	3.20E+03	91-PS-3A
2352	SS61-sp	Water	1994	Diese	700	µg/L i	8	-	-	-	
2353	SS61-sp 3	Water	1994	Ethylbenzene	3	µq/L	9	1	5.8	5.8	91-PS-38
2354	SS61-sp	Water	1994	Gasoline	120.	μ <b>9/</b> L	9	3	490	2.00E+03	91-PS-3A
2355	\$\$61-sp	Water	1994	Tetrachloroethene	1.	µg/L	10	-	-	-	
2356	SS61-sp !	Water	1994	Toluene	2.	harr i	9	-	-	-	
2357	SS61-sp	Water	1994	Trans-1.2-dichloroethylene	31	µq∕L	9	5	6.4	56	91-PS-6A
2358	SS61-1p	Water	1994	Trichloroethene	1	MO/L	9	8	3.6	1.10E+03	91-PS-3A
2359	SS61-10	Water	1994	Xylenes (lotal)	. 3(	JQL	9	3	15	35	91-PS-3A
2380	SS64	Water	1994	(Technical) chlordane	1,	µç/L	4	-	-	-	
2361	SS64	Water	1994	1,1,1-Trichloroethane	1;	µg/L	4	-	-	- 1	
2362	SS64	Water	1994	1.1.2.2-Tetrachloroethane	. 1	µ9/L	4	-	-		
2363	SS64	Water	1994	1,1,2-Trichlorpethane	0.5	µgA.	4	-	-	-	]
2364	\$\$64	Water	1994	1,1-Dichloroethane	1	hðr 1	4	-	-	<b></b> _	
2365	SS64 1	Water	1994	1,1-Dichloroethene	0.5	HOL !	4	-		<u> </u>	
2366	SS64	Water	1994	1,2,4-Trichlorobenzene	10	HQ4	5	-	-	-	
2367	SS64	Water	1994	1,2-Dichlorobenzene	10,	hõr	13	-	-		
2368	\$\$64	Water	1994	1.2-Dichloroethane	0.5,	NO/L	4	-	-	l	
2369	SS64	Water	1994	1,2-Dichloropropane	0.5	<u>hor</u> į	4			<u> </u>	
2370	\$564	Water	1994	1,3-Dichlorobenzene	10	HQL )	13	-		-	
12371	5564	Water	1994	1,4-Dichiorobenzene	10	µg/L	13	-	-	-	
2372	SS64	Water	1994	2.4,5-Trichlorophenol	10,	µg/L	5	-	-		
2373	SS64	Water	1994	2.4,6-Trichlorophenol	10	PCL	5		-	-	
2374	SS64 1	Water	1994	2.4-Dichlorophenol	10.	ugn	5	-	-	-	
2375	5564	Water	1994	2,4-Dimethylphenol	10	HWL	5	-		-	
2376	SS64	Water	1994	2,4-Dinitrophenol	50	HAL	5	-	-	-	
2377	SS64	Water	1994	2.4-Dinitrololuene	501	HOL	10	-			
2378	SS64 1	Water	1994	2.5-Dinitrotokuene	10	por l		- 1	÷	-	
2379	SS64	Water	1994	2-Chloroethylvinyt ether	21	µ9/L	4	-	<u>-</u>	-	
2380	5564	Water	1994	2-Chloronaphinaiene	10 <sub>1</sub>	por [	5	-		-	
2381	SS64	Water	1994	2-Chiorophenol	10	ug/L 🗄	5	- 1	- 1	- 7	
	A	в	c	<u> </u>	E	F	G	н	Ϊ.	J	ĸ
------	--------------	-----------------	------	-----------------------------	---------------------------------------	--------------	---------------------------------------	---------------------------------------	----------	-----------	--
	-				_			_	•	1	[
1	Source Area	Media	Cate	Analyte	<u></u> _L	Unds	#Samples	#Detect	Min Çanç	Max Conc	Max Location
2382	\$\$64	Water	1994	2-Methyinaonthaiene	10	µg/L		<u> </u>		! <b></b>	
2383	5564	Water	1994	2-Methylphenol	10	µg/L	······	<u>-</u>	· •	+- 	·····
2364	5564	_vater	1004	2-Nuroaniine		µg/L		: <b>-</b>	<u> </u>	<b>ب_</b>	
2386	5564	Water	1994		20	uo/l		·			······
2387	5564	Aater	1994	3-Nitroandine	50	un/L				<u> </u>	
2388	SS64	Water	1994	4 4'-DDC	01	ug/L	4	<u>-</u>			
2389	SS64	Water	1994	4 4'-DDE	0.1 D 1		4		-	<u> </u>	
2390	SS64	Water	1994	4.4'-DDT	01	Jug/L	4		-	i	
2391	SS64	Water	1994	4-Bromophenyl-phenylether	10	Ug/L	5	· · · · · · · · · · · · · · · · · · ·			····
2392	SS64	Water	1994	4-Chloro-3-methylphenol	20	µg/L	5	 _	<b></b>		
2393	SS64	Water	1994	4-Chloroaniline	20	µg/L	5		-	<u>+</u>	
2394	SS64	Water	1994	4-Chlorophenyi-phenylether	10	µg/L	5	-	-	-	······································
2395	SS64	Water	1994	4-Methylphenol	10	µg/L	5	-	+-	-	······································
2396	SS64	Water	1994	4-Nitroaniline	20	hð/L	5	-	-	I-	
2397	SS64 -	Water	1994	4-Nitrophenol	50	µg/L	5	-	-	-	
2398	ISS64	Water	1994	a-BHC	0.05	-µg/L	4	1-	-	-	
2399	SS64	Waler	1994	a-Chiordane	0.05	µ9/L		-	-		
2400	5564	Water	1994	Acenaphthene	10	µg/L	<u> </u>	1	<u> </u>	-	
2401	SS64	Water	1994	Acenaphihylene	10	hð\r	5	-	**	-	
2402	5564	Water	1994	Akirin	0.05	hđyr	4	<b>!-</b>	-	-	
2403	5554	Water	1994		N/A /-	HQ/L	4	4	1.08E+03	1.46E+03	54MW01
2404	5364	vvaler Water	1994	Antoracene	10	hð\r hð\r	<u>5</u>				
2405	0004	vyaler Moter	1994	Апинопу	1 N/A	197L			-	- 13 4	6416404
2400	2304	Water	1004		0.05	1997		<b>.</b>		13.1	04/11/1
2408	5004	VVdie:	1904	Bathim	N/A	HON	A	e	- 165		RAIMUNI
2409	5564	Water	1994	Renzens	1	uall .	· · · · · · · · · · · · · · · · · · ·	)0)	-		
2410	S\$64	Water	1994	Benzo/augibracette	t0	uo/l	5		-		
2411	SS64	Water	1994	Benzo(a)ovrene	10	JUO/L	5		_		
2412	SS64	Water	1994	Benzo(b)fluoraninene	10	'IQ/L	5	<u> </u>		-	
2413	SS64	Water	1994	Benzo(g.h.i)perviene	10	µg/L	5	-	-	-	
2414	SS64	Water	1994	Senzo(k)fluoranthene	10	µç/L	5	-	-	_	
2415	\$\$64	Water	1994	Benzoic acid	50	HUL	5	-	-	-	
2416	SS64	Water	1994	Senzyl alcohol	20	µg/L	5	-	-	-	
2417	SS64	Water	1994	Beryllium	1	hô/r	8	-	-	-	
2418	S564	Water	1994	Bis(2-chloroethoxy)methane	10	Pg/L	5	-	_	-	
2419	SS64	Water	1994	Bis(2-chioroethył)ether	10	HQ1	<u> </u>	<u>-</u>	<u></u>	L <u></u>	
2420	SS64	Water	1994	Bis(2-chloroisopropyi)ether	10	.µg/L	<u>\$</u>	- 	-	-	
2421	5564	Water	1994	Bis(2-ethylnexyl)phin#late	10	HOL	5	1	1.9	1.9	64MW01
2422	\$564	Water	1994	Bromodichloromeinane		ug/L	4			-	
2423	5304	vvaler Mater	1004	Bromemethage	·				·····		
2424	2304 CCE4	Water	1004	-Di driben viniti siste	10	ugi j	2			-	
2475	SS64	Water	1994	Carbonium	,, t		,				·····
2427	S\$64	Water	1994	Calcium	N/A	ua/L	8	A	5.305+04	5.66E+04	64MW03DF
2426	S\$64	Water	1994	Carbazole	10	ug/L	5	-	-	-	
2429	SS64	Water	1994	Carbon tetrachloride	0.5	Hg/L	4	I-	=	-	
2430	SS64	Water	1994	Chiorobenzene	1	µq/L	8	-			
2431	S\$64 ·	Water	1994	Chioroethane	1	µg/L	4	-	-	-	
2432	SS64	Waler	1994	Chloroform	1	µ <b>g/L</b>	4	-	-	-	
2433	S\$64	Water	1994	Chloromethane	1	µ9/L	4	_	-	-	
2434	S\$64	Water :	1994	Chromum	1	µg/L	4	-	-	-	
2435	SS64	Water	1994	Chrysene	10	.ug/L	5	-			
2436	5564	Water	1994	Crs-1.3-dichloropropene	0.5	101	4	! <b></b>			
2437	5364	vvater	1994		NVA	HOR.	2	2	5	75	OANEWO2
2438	0304	Valer	1994		1	PU/L	8	4	16.1	36.1	SAWW01
4439	0004	VVBIET I	1384		0.05	HQ/L			-	-	64164CA
2440	3394 SSA	TVALE!	1004		-1975		<u> </u>	5	2.7	3.6	ORMAYVU3
222	5564	Water	1004	Nibenzia hisnikratana	· · · · · · · · · · · · · · · · · · ·	. Mayr			-		
244	\$564	Valci Water	1994		10	uori		E	_		····
7444	SS64	Water	1994	Dibromochlommethane		ua/	4				
2445	SS64	Water	1994	Dichlorodifucconthane		ua/i	4	-		_	
2445	SS64	Water	1994	Dieldrin	01	uol		-	·····		
2447	SS64	Water	1994	Disthviphthalate	10	µo/L	5		-		<b>-</b>
2448	SS64	Water	1994	Dimethylphthalate	10	µg/L	5	-	-		······································
2449	S\$64	Water	1994	Endosulfan I	0.05	ua/L	4	-			

.

^	-	

	A	8	C	0	<u> </u>	F	G	н	1	J	ĸ
		·				······	*****			•••••••	·····
1	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
2450	S\$64	Water	1994	Endosulfan k	01	HQ/L	4	ļ <b>—</b>	<b> -</b>	-	!
2451	S\$64	Water	1994	Endosullari sulfate	0.1	hð\r	4		•	<u>i – </u>	
2452	SS64	Water	1994	Engrin	0,1	hð\r	4	1-	<b> -</b>	I	······································
2453	5564	Water	1994	Endrin aldenyde	<u>.</u>	HQ/L	•••••••••••••••••••		1 ma 4	<b>_</b>	L
2454	5564	vvale:	1994	ELNYIDERZERE		HQ/L	4		<b>-</b>	<b></b> +	ļ
2433	3304	YValc:	1994							ļ <del>.</del>	L
2457	CCEA	Water	1004	A-BHC	0.05	un/l		-		-	
2458	SSEA	Water	1994	o-Chloriane	0.05	uo/t	4	i <b>-</b>	<u> </u>		1
2459	5564	Water	1994	Heptachlor	0.05	un/L	4	i_	+		<u>.</u>
2460	SS64	Water	1994	Heptachior epoxide	0.05	va/L	4		-		· · · · · · · · · · · · · · · · · · ·
2461	SS64	Water	1994	Hexachlorobenzona	10	ua/L	5	i	-		
2452	SS64	Water	1994	Hexachlorobutadiene	10	μα/ι	5	-	-	-	i
2463	SS64	Water	1994	Hexachlorocyclopentadiene	10	.µg/L	5	i-	-	-	
2444	SS64	Water	1994	Hexachioroethane	10	μ <b>g/L</b>	5	—	-	-	
2465	SS64	Water	1994	Indeno(1.2.3-cd)pyrene	10	·µg/t	5	-	-	<b> _</b>	
2464	SS64	Water	1994	Iron	N/A	µg/L	8	8.	1.33E+03	7.77E+03	64MW01
2447	SS64	Water	1994	Isophorone	10	.µg/L	5	-	-	-	
2468	SS64	Water	1994	Lead	1	µg/L	7	5	5.7	12.8	64MW01
2469	SS64 .	Water	1994	M.p-xylene	1	μ <b>g/L</b>	4	-	-	-	
2470	SS64	Water	1994	Magnesium	N/A	JQ/L	8	8	1.21E+04	1.33E+04	64MW03
2471	SS64	Water	1994	Manganese	N/A	HO/L	8	8	1.51E+03	5.77E+03	64MW02
2472	SS64 /	Water	1994	Methoxychlor	0.5	HQ/L	4	-	-	-	,
2473	SS64	Water	1994	Methylene chioride		HQ/L	4	-	-	-	
2474	5564	Water	1994	n-Niroso-di-n-propylamme	10	μ <b>μ/</b> ι.	5		-	-	
2475	SS64	Water	1994	n-Nerosodiphenylamine	10	Pg/L	5	-			·····
2475	SS64	Water	1994	Naphinaiene	TU	HQ/L	5		-	-	A
2477	5564	vvater	1994			HAL		•	0.3	11.8	64MIWO2
24/8	3304	Valer	1004			- DGVL		-	-		
2400	3304	VARIO	1004	0-Ayreile		UC A	-				
2400		Waler	1004	PCB-1016	2	i pyrt.					
2407	3304	Woter	1994	PCB-1227	<u>_</u>						······································
2483	SS54	Water	1994	PCB-1242	······	ua/l.	4		_		
7484	SS64	Water	1994	PC8-1248	1	uovi	4		-	=	
2485	SS64	Water	1994	PCB-1254	1	uc/L	4	_		-	······
2486	SS64	Water	1994	PCB-1250	, 1	UQ/L	4				·····
2487	SS64	Water	1994	Pentachiorophenol	50	INON.	5	-	-	-	
2488	SS64	Water !	1994	Phenanthrene	10	lug/L	5	-	-	-	
2489	SS64	Water	1994	Phenol	, 10	JIPG/L	5	-	-	=	
2490	SS64	Water	1994	Рутеле	10	IPOR.	5	-	-	-	
2491	SS64	Water i	1994	Silver	. 1	lugit.	8	-	-	4	
2492	SS64	Water !	1994	Sodium	INA	Hg/L	8	8	5.03E+03	5.78E+03	64MW03
2493	SS64	Water	1994	Tetrachioroethene (PCE)	0.5	JUG/L	4	2	0.82	1.7	64MW01
2494	\$\$64	Water	1994	Tin	1	H91	4	-	-	-	
2495	S564	Water	1994	Toluene	<u> </u>	HOL.	Į	-	<u> </u>	-	
2496	5564	Water	1994	Toxephene	÷5	IND/	4	-	-	-	
2497	5564	vvaler	1994		1	1997L			1.5	1.6	O-AMIVID1
Z498	5564	vvaler (	1994	Irans-1,5-dichicropropene	U.5		<b>4</b>		-		\$ 41.54 P-
2433	0004	VValigi Mater	1334		······································	Line A	<b>_</b> _		<u>v.e</u>	<u></u>	0-100 YUU1
2504		Vision	1004		·····	1100		-	7.4	7 :	641616730
2000	SSA	Water	1004	View chioride		1001	A	-			
2601	SSA	Water	1904	Zioc	N/A	LUC /		1	20.9	214	64MMM
2504	ST77	Water	1992	1.1.1-Trichlomethase	0.5	un/	13	_			
2505	ST27	Water	1997	1.1.2-Trichlorgethane	0.5	UOA.	13			-	
2506	ST27	Water	1992	1.1-Dichloroethane	1	NO/L	13	_			
2507	ST27	Water	1992	1.2-Dichloroethane	. 0.5	HOL	13		-	-	
2505	ST27	Water I	1992	1,4-Dichlorobenzene	2	HOIL	13	-	-	-	
2509	ST27	Water	1992	Antimony	200	ipg/L	13	-	-		
2510	ST27	Water	1992	Araenic	N/A	HO/L	9	9	5.5	30	27B-11
2511	ST27 I	Water	1992	Barium	'N/A	Hgr.	13	13	77	590	278-13
2512	ST27	Water	1992	Benzene	2	HQ/L	13	-	-	<b>-</b>	
2513	ST27	Water ;	1992	Beryllium	3	ING/L	13	-	-	-	
2514	ST27 I	Water	1992	Bromide	500	HOL	4	-	-	-	
2515	ST27 i	Water	1992	Cadmum	10	W91	13	-	-	-	
2516	ST27	Water	1992	Calcium	N/A	ING/L	13	13	3.50E+04	5.60E+04	278-13
2\$17	ST27	Water	t <b>99</b> 2	Carbon letrachloride	1	Jugit	13	-	-	-	

Ń

	A	6	C	D	<u></u>		F	Ç (	<u> </u> H	L	1	ĸ
1	Source Area	Media	Dat€	Analyte	D	L	Units	#Samples	#Oetect	Min Conc	Max Conc	Max Location
2518	ST27	Water	1992	Chionde	<u>_N/A</u>		µg/L	4	4	1.10E+03	1.40E+03	278-8
2519	ST27	Water	1992	Chioroform		05	µg/L	13		! <u></u>	<u></u>	
2520	ST27	Water	1992	Chromium		20	µg/L	13	1	21	21	278-13
2521	ST27	Water	1992	Cis-1,2-dichioroethylene	···	1	µg/L	13	· <del>-</del>			
2522	ST27	Water	:992	Cobail		20	µg/L	13	1	32	32	278-13
2523	ST27	Water	1992	Copper		20	Hð\r	13	6	48	: 430	278-13
2524	ST27	Wate:	1992	Ethylbenzene		2	µg/L	13		. <u> </u>	i <b>-</b>	
2525	ST27	Water	1992	Fluoride	<u>N/A</u>		µ9/L	4	4	100	200	27-1
2526	ST27	Water	1992	h-Chlorofluorobenzene	N/A		µg/L		7	í <b>8.3</b>	11	278-6
2527	ST27	Water	1992	iron	N/A		µg/L	13	13	33	2.30E+04	27B-13
2528	ST27	Water	1992	Lead		5	µg/L	15	9	5.4	120	278-13
2529	ST27	Water	1992	Magnesium	N/A		µg/L	13	. 13	7.70E+03	1.50E+04	27B-13
2530	ST27	Water	1992	Manganese	N/A		µg/L	13	13	940	1.20E+04	278-8
2531	ST27	Water	1992	Methylenechlonde		5	µg/L	13		-	-	
2532	ST27	Water	1992	Nickel		30	µg/L	13	2	32	82	27B-13
2533	ST27	Water	1992	Nitrate	N/A		µg/L	4	4	200	1.10E+03	278-11
2534	ST27	Water	1992	Nilrite		200	ug/L	4	-	-	-	
2535	ST27	Water	1992	p-Chlorofluorobenzene	N/A		µg/L	7	7	9.2	10	278-10
2536	ST27	Water	1992	Phosphate		400	µg/L	4	-	-	-	
2537	ST27	Water	1992	Potassium	N/A		hðyr	13	13	2.40E+03	1 3.90E+03	278-13
2536	ST27	Water	1992	Silver		20	µg/L	13	-	-	-	
2539	ST27	Water	1992	Sodium	N/A		µg/L	13	13	3.20E+03	5.60E+03	278-8
2540	ST27	Water	1992	Sulfate	N/A		µg/L	4	. 4	9.00E+03	1.50E+04	278-8
2541	ST27	Water	1992	Tetrachioroethene		0.5	µg/L	13	-	I_	-	
2542	ST27	Water	1992	Tin		100	µg/L	13	-		_	,
2543	ST27	Water	1992	Toluene		2:	ug/L	13	-	-	-	
2544	ST27	Water	1992	Total organic carb	N/A		µg/L	2	! 2	4.00E+03	5.00E+03	27-1
2545	ST27 -	Water	1992	Trans-DCE		1	µg/L	13	i <b>–</b>	L	-	
2546	ST27	Water	1992	Trichloroethene		1,	µg/L	, 13	-		-	
2547	ST27	Water	1992	Vanadium		30	µg/L	13	1	91	91	278-13
2548	ST27	Water	1992	Vinyl chlonde		2	µg/L	13	i—	-	-	
2549	ST27	Water	1992	Xylenes (total)		5	µg/L	13.	-	_		
2550	ST27	Water	1992	Zinc		10.	µg/L	13	10	12	400	27B-13
2551	ST27	Surface Soil	1988	Benzene	N/A		µg/kg	1	1	6	6	278-18
2552	ST27	Surface Soil	1988	Ethylbenzene		23	ug/kg	1	-	-	-	
2553	ST27	Surface Soil	1986	Lead	N/A		µg/kg	18	18	4.00E+03	7.00E+03	27-2
2554	ST27	Surface Soil	1988	Lead	N/A		µg/kg	3	3	3.80E+03	3.76E+04	27B-14
2555	ST27	Surface Soil	1988	Toluene	N/A	1	µg/kg	, 1	1	17	17	278-18
2556	ST27	Surface Soil	1988	Xylenes (total)		23	µg/kg	. 1	-		-	
2557	ST27	Soil	1986	Lead	N/A		μg/tg	. 4	4	7.00E+03	8.00E+03	27-2
2558	ST56	Water	1994	1.1.1.Tachloroethane		1,	ug/L	3	_	-	1-	
2559	ST56	Water	1994	1.1.2.2-Tetrachiorpethane		1	µg/L	3	L	_	-	
2560	ST56	Water	1994	1,1,2-Trichlorgethane		0.5	µg/L	3	_	<u> </u>	_	
2561	ST56	Water	1994	1,1-Dichloroethane		1	µg/L	3	-	-		
2562	ST56	Water	1994	1.1-Dichloroethene	·^/	0.5	NO/L	3	-	-	-	
2563	ST56	Water	1994	1,2-Dichlorobenzene		1,	WO/L	6	-	-	-	
2564	ST56	Water	1994	1,2-Dichloroethane	-	0.5	ug/L	3	1-	-	-	
2565	ST56	Water	1994	1.2-Dichloropropane		0.5	ug/L	. 3	<u> </u>	<u> </u>		1
2566	ST56	Water	1994	1.3-Dichlorobenzene		i.	µg/L	6	-	-	-	
2567	ST56	Water	1994	1.4-Dichlorobenzene		- ti	µg/L	6	-	-	-	
2568	ST56	Water	1994	2-Chloroethylvinyl ether		2	µg/L	3		_	-	
2569	ST56	Water	1994	Aluminum	N/A		JQ/L	4	4	385	1.47E+03	564/1/05
2570	ST56	Water	1994	Antimony		1	µg/L	4	-	–	-	
2571	ST55	Water	1994	Arsenic	N/A		µg/L	6	. 6	5	14.7	SEMWOS
2572	ST56	Water	1994	Banum	N/A		ug/L	6	6	250	413	56MMO4F
2573	ST56	Water	1994	Benzene		1	µg/L	. 3	i–	-	-	
2574	ST56	Water	1994	8eryllium		1.	Ha/L	6	-	I_	-	
2575	ST56	Water .	1994	Bromodichioromethane		1	µg/L	3	-			
2576	ST55	Water	1994	Bromotorm		1	עסג	3	-	-	I <b>-</b>	]
2577	ST56	Water	1994	Bromomethane		2	µg/L	3	; <b>-</b>	-	-	
2578	ST56	Water	1994	Cadmium		1:	HQ/L	6	_	-	-	
2579	ST56	Water	1994	Calcium	N/A	•	ug/L	6	6	9.70E+04	1,17E+05	SEMINOSO
2580	ST56	Water	1994	Carbon tetrachionde	<u> </u>	0.5	ug/L	3	i		-	
2561	ST56	Water	1994	Chlorobenzene	****	ī	ug/L	6	-	_		······
2587	ST56	Water	1994	Chloroethane			μ <b>ο</b> /L	1	-	-		
2583	ST56	Water	1994	Chioroform	· · · · · ·	1	uu/L	1			_	
2584	ST56	Water	1994	Chloromethane		1	uo/L	1	L	-		
2645	ST56	Water	1904	Cis-1.3-dichiorgemotene	~· <u> </u>	0.5			L			
											( )	

. .

<b></b>	A	E I	C	0	Ξ	F	G	Тн	<u>, , , , , , , , , , , , , , , , , , , </u>	J J	ĸ
h					<b></b>	-			· <b></b>		
11	Source Area	Media	Date	Analyte	DL	Unit	s : #Samples	: #Delect	Min Cone	May Conc	Max Location
2586	ST56	Water	1994	Cobat	N/A	uo/L		5 5	5.3	137	56MW04
2587	ST56	Water	1994	Copper	N/A	ug/L		3 3	13.4	27	56MW05
2588	ST56	Water	1994	Dipromochloromelhane		ua/L			i_		
2589	ST56	Water	1994	Dichlorodifluoromethane		ua/L			h	-	
2590	ST56	Water	1994	Ethylpenzene		ua/t		3:-	i		
2591	ST56	Water	1994	1000	N/A			5' <b>6</b>	709	8 895+03	SEMMOS
2592	5756	Water	1994	Lead		un/L			37	10.5	SALALOS
2591	STSE	Water	1994			uoll		·		<u> </u>	100001103
2504	STSE	Water	1994	Mannesum	N/A				2745+04	3 27EAAA	SCARADED
2846	CTEE	Alare/	1004	1190330494	N/4				700	2 205-03	56104000
2000	3130	VVALOI	1004			- HARL			149	2.296700	DOMINUA
2030	3130	VVale	1004		A1/A	- HUL				-	
2397	3136	vyaler	1994	NICET		Hg/L			0.4	20.4	CUVVIMOC
12348	5156	VVater	1994	0-Aylene		ug/L				-	
2599	ST56	Water	1994	Polassium	N/A	hôn -		0	6.72E+03	1.03E+04	56MVV04
2600	ST56	Water	1994	Silver		hdvr		<u>1</u>	1-		
2601	ST56	Water	1994	Sodium	N/A	HQ/L		6	1,37E+04	2.39E+04	56MW050
2602	ST56	Water	1994	Tetrachloroethene	0.5	HQ/L		2	13.8	25.1	56WHD
2603	ST56	Water	1994	Toluane	1	.µg/L	3	<u>  -</u>		[ <del>-</del>	
2604	ST56	Water	1994	Trans-1.2-dichloroethene	1	µg/L	; 3	-		-	
2605	ST56	Water	1994	Trans-1.3-dichloropropene	0.5	i;µg/L		<u>;-</u>	-	-	
2606	S756	Water	1994	Trichloroethene	0.5	µg/L	5	2	0.4	0.6	56WHD
2607	ST56	Water	1994	Trichlorofluoromethane	1	µg/L	3	-	-	-	
2608	ST56	Water	1994	Vanadium	N/A	Jug/L	5	5	5.6	19	56MW05
2609	ST56	Water	1994	Vinyl chionde	0 5	µg/L	3	i <b> _</b>	-	I-	
2610	ST56	Water	1994	Zinc	N/A	µq/L	1 4	4	24.2	46.2	56MW05
2611	S758	Water	1993	1,1,1-Trichlorgethane	5	ug/L	14	l-	-	_	
2612	ST58	Water	1993	1 1.2-Trichloroethane	5	ua/L	14		-		
2613	ST58	Water	1993	1 1-Dichioroethane		uari	14		-		
2614	STSR	Waler	1993	1 2-Dichloroethane	5	-un/t	14		_		
2415	STER	Water	1991	1.2.Dichlomethene	1 2	und	14	: !			
2414	CT50	Meter	1001	1.4.Dichicheotene		und	1		<u> </u>		
2617	CTC	Mater	1001	1 Budonosa	100	PPP L	1				
	3130 STCA	VVale:	1993		50	1001		ļ	74		6010100
	3136 ·	AABIGL	1393		100		5	<b>4</b>	10		SOWAAD
1 10 10	5136 CT50	Veller	1001			HOL			37	180	E MALERA
2020	0760	VValgi	1990			CHURL .			9.1		
2021	S128 -	VVAler	1993			HOL			-		
4042	5158	vvater	1993			indur-			-	-	
2623	5158	vvater	1993		100	INGVI.		1	100	9.905+04	SOMANOR
2624	ST58	vvater	1993	Gasoline	2.00E+03	har	14	1	2.60E+05	2.60E+05	58MW09
2625	ST58	vvaler	1993	Lead		HOL.	14	13	35	180	50MW12
2626	ST58	Water	1993	Methylens chloride	N/A	Ind.	14	14	1.8	8	SEMWIC
2627	ST58	Water 1	1993	Tetrachioroethene	5	JONT	14	-	-	-	
2628	ST58	Water	1993	Terahydrofuran	10	<b>µg</b> L	14	-	-	-	
2629	ST58	Water	1993	Toluene	5	Lon.	14	2	2.8	140	55MW10
2630	ST58	Water	1993	Trichloroethent	5	por.	14	-	-	-	
2631	ST58 :	Water	1993	Vinyl chloride	10	ijig L	14	-	-	-	
2632	ST58	Water	1993	Xylenes (lotal)	5	ipg/L	14	- 4	28	830	58MW10
2633	WP33 !	Water	1994	(Technical) chlordane	1	INC.	1	-	-	-	
2634	WP33	Walter	1994	1,1,1-Trichloroethene	1	HO/L	4	-	-	-	
2635	WP33	Water	1994	1,1,2,2-Tetrachloroethane	1	INO/L	4	-	_	-	
2636	WP33	Water	1994	1,1,2-Trichlorgethane	0.5	HOL.	4	-	-	-	
2637	WP33	Water	1994	1,1-Dichloroethane	1	INON.		-	L <b>-</b>	-	
2638	WP33	Water	1994	1,1-Dichloroethene	0.5	ING/L	4	-		-	
2639	WP33	Water	1994	1.2-Dichlorobenzene	t	HO1	1 8	-	-	-	
2640	WP33	Water .	1994	1,2-Dichloroetnane	0.5	HUG/L	1 4	-	-	-	
2641	WP33	Water	1994	1,2-Dichloropropane	0.5	-ug/L	4	-	-		
2642	WP33	Water	1994	1,3-Dichlorobenzene	1	HOL	8	-	-	-	
2643	WP33	Water	1994	1.4-Dichlorobenzene	1	HOL	6	1_		-	
2844	WP33	Water	1994	2-Chloroethyhrinyi ether	2	ING/L	1	I-			
244	WP33	Water	1904	4 4-000	01	.uot					
244	14033	Water	1004	4 4-DDE	0 1	LUNA.	7	l	-		
28.47	W913	Water	1904	4 4-DOT	0.7 0.1	LUNA N		1_	_		
1	14033	Water	1004		0.05	LUNCE IN					
		- THUM	1004		0.00	1 Mar P	+	1			
2048	117-33		1004	A Julia	0.05		· []				•••••
14000	VTP 33		1994		0.05	- HOAT		Į	-		
1	vr 33 1	THE CONTRACT	1334			19/			233	808	33860)
2007	WP33	VVIII OF	1994	Antandiy		HON	4	<u> </u>	-		
2653	VVP33	water	1994	Arsenic	N/A	101	<u> </u>		6.5	62.1	33MOD

÷,

	-	

<u> </u>	A 3			<u> </u>	Г <sup></sup>	F	G	<u> </u>	[	<u></u>	<b>R</b> 1
			<u> </u>		L	لسبنسة		<u> </u>	ļ		
	_		_		-					1	
1	Source Area	Media	Date	Analyte	DĻ	Units	#Samples	#Delect	Min Conc	Max Conc	Max Location
2654	WP33	Water	1994	b-BHC	0.05	HQ/L	4	-	) <u> </u>	!-	]
2655	WP33	Water	1994	Banum	N/A	µg/L	8	8	107	784	33M00
2656	WP33	Water	1994	Benzene	1	µc/L	4	1-	-		
2657	WP11	Water	1994	Beryllium	2	uc/1_	8		_	ř-	
74.58	14/022	thister.	1004	Bromodichincomethane	1	 				t	
2030	VVF-33	VVAICI	1994			Page L					
2659	VVP 33	vvaler	1994	Bromaturm		POL.	<b>•</b>	·		L	
2660	WP33	Water	1994	Bromomethane	2	<u>ug/L</u>	4	+ <del>-</del>	-	-	
2661	WP33	Water	1994	Cadmum	1	µg/L	8	-	-	i <b>-</b>	
2662	WP33	Water	1994	Calcium	N/A	µ9/L	8	: 8	3.93E+04	9.44E+04	33MOC
2683	WP33	Water	1994	Carbon tetrachioride	05	un/l	4	l <b>-</b>			
	14/1222	Vilage.	1004	Cherch-pases				· ····		······································	
4904	VVP33	vya(c:	1334			PUL				<b>[</b>	
2655	WP33	Water	1994	Chioroethane		-ug/L	4		-	-	
2666	WP33	Water	1994	Chloroform	1	µq/L	4	i=	-	-	
2667	WP33	Water	1994	Chloromethane	1	µg/L	4	-	_	-	
2665	WP33	Waler	1994	Cis-1 3-dichlorogropene	0.5	Jug/L	4	Í			
2440	14/0111	Mainr	1004	Conner	1	un/	6	3	116	15	174402
2007	14000	TYDIE:	1000		0.05		······································				
20/0	VYP33	VVBIEF	1994			PAL			-		
2671	WP33	Water	1994	Upromochioromethane	1	HQ1	4			L	<b>]</b>
2672	WP33	Water	1994	Dichlorodifluoromethane	1	µg/L	4	1		L	]
2673	WP33	Water	1994	Dieldrin	0.1	µq/L	4	-	-	-	
2674	WP33	Water	1994	Endosullan 1	0.05	Jug/L	4	-	-	1-	
7876	WP13	Water	1004	Endosultan li		uo/L		í	<b>_</b>		
12013	10022	-vere:	1004		A 4	100	7				·····
20/6	**** 33	vvaler	1994		· •	JUNC		Į		بت	<b>_</b>
2677	WP33	Water	1994		0,1	hor	4	<b>-</b>	-		
2678	WP33	Water	1994	Endrin aldehyde	0.1	HQ/L	4	<b>-</b>	-		
2679	WP33	Water	1994	Ethylbenzene	1	.ug/L	4	! <b>-</b>	-	[=	
2680	WP33	Water	1994	o-BHC	0.05	UCL	4	1-		[]	
7681	WP11	Water	1994	g-Chiomiane	0.05	UNI/	4				
200	14034	*****	1004		0.00						·····
2902	WF33	VVERT	1994		0.05	PAR -					
2683	WP33 !	Water	1994	Meptachior epoxide	0.05	HOL	4	-	-	-	
2684	WP33	Water	1994	(ron	N/A	µg/L	8	8	298.	1.51E+04	33M00
2685	WP33	Water	1994	Lead	1	µg/L :	6	4	2	4,1	33MOCF
2686	WP33	Water	1994	M.p-xviene	1	HO/L	4	-		-	
2487	WP33	Water	1994	Mannesium	N/A	uni	6	8	8 31E+03	1.605+04	334400
200		186-tee	1004	Lie const	6//A				3 636-03	9.125.02	22142005
1000	VVF-33	vvator	1334		100	PPL			2.535703	8.12ETU3	JOHNOCH
2689	WP33 (	Water	1994	Methoxychior	<u>Ç.Q</u>	indvr i	4	-			
2690	WP33	Water	1994	Methylene chloride	1	lug/L	4	-	-	-	
2691	WP33	Water	1994	Nickel	N/A	UQ/L	1	1	20.9	20.9	33M0C
2692	WP33	Water	1994	0-Xviene	1	UQ/L	4	-	-	-	
2693	WD31	Water	1994	PC8-1016		un/i	4	-			
20000	45333		1004	000 1010			······				
2034	VVP 33	VValet	1794	FVD-(22)	***** <u> </u>						
2695	WP33	Water	1994	PCB-1232	<u>,                                    </u>	HOL					
2696	WP33	Water	1994	PCB-1242	. 1	ug/L i	4	- 1	-		
2697	WP33	Water	1994	PC8-1248	1	INGA.	4	-	-	]	
2698	WP33	Water	1994	PCB-1254	1	Hgr.	4	-		-	
2640	WP33	Water	1994	PC8-1260	•		Å	-			······
2700	MP11	Wmer	1004	Potpash at	.N/A	1.50			7 345-02	1405404	111400
6190	14019		10041			1.100		······	1.0000700	1	
2/01	VYP33	vvaler	1954		1	I ST			-		
2702	WP33 (	Water	1994	Sodium	IN/A	HQ/L	8	8	2.79E+04	3.69E+04	33M01
2703	WP33	Water	1994	Tetrachioroethenis (PCE)	, 0.5	ug/L	4	-	<b></b>	• <b></b>	
2704	WP33	Water	1994	Tin	1	HOL	4	-	-	-	
2704	WP33	Water	1994	Toluene	1	Jug/L	4	[	-	-	
2700	WP33	Water	1004	Toxaphane	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	and the second s	4				
27.00	14012	1. Alatas	1004	Trans. 1 9 disblocostines.		100	7	h			
2/0/	VYP JJ		13/34				4				
2708	WP33	Water	1994	Trans-1,3-dichloropropene	0.5	INO/L	4	-		-	
2709	WP33	Water	1994	Trichlorgethene (TCE)	0.5	149/	4	-	-	-	]
2710	WP33	Water	1994	Trichlorofluoromethane	1	HO/L	4	-	-	-	
2711	WP33	Water	1994	Vinyl chloride	0.5	ug/L	4	-	-	-	{
2743	WP33	Water	1904	Zinc		uo/	4	A	21 2	48.2	33400
1774	14044	Afeter	1003	1 1 1 Trichlomathane		1000			0.4		451401
1413	CP TH		382						U.3	U.3	
j2714	VVP45	water	1992	1,1,2-171CNG/00/01/14	0,5	HOIL !	12	-	-		
2716	WP45	Water	1992	1,1-Dichloroethane	1	jugA.	12	1-	-		
2716	WP45	Water	19921	1,2-Dichloroethane	0.5	.pg/	12	1	0.6	0.6	451402
2717	WP45	Water	1988	1.4-Dichlorobenzene	N/A	UOL I	1	1	0.0022	0.0022	4SWMC
2718	WP45	Water	1992	1.4-Dichlorobenzene	2		13				
2744	WD45	Minter	1002	Antimony	- 200		1				
<b>11</b>	10046	T Faler	4000	B	200		<u>4</u>				
2720	VVP45	water	1992	Banumi	N/A	uc/L	Z	2		140	454405
2721	WP45	Water	1992	Benzene	2	lµg/L i	12	I <b>-</b>	-	-	

1

ŝ

٠

	A	a	C	<u> </u>	E	F	G	Н	E	J	ĸ
		<u></u>							i		
1	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc	Max Conc	Max Location
2722	WP45	Water	1992	Beryllium	3	3 μ <b>g/L</b>	2		·	-	1
2723	WP45	Water	1992	Bromide	500	) µg/L	. 2	-	<b>⊦_</b>	_	
2724	WP45	Water	1992	Cadmium	10	) µg/L	2	· •••	-		
2725	WP45	Water	1992		_ <u>N/A</u>	P0/L	2	<u> </u>	4.00E+04	5.30E+04	45M06
2726	WP45	Water	1992	Carbon tetrachionde		HQ/L	12		-	i	
2/2/	VVP45	vvater	1992	Chloroform	N/A	HQ/L	2	<u> </u>	2.90E+03	5.20E+03	45M06
2/26	WP45	vvater	1992	Cherenom			12		<b></b>	<b>-</b>	1
2729	WP45	VValer	1335				4.2			47	1
2734	MADIAS	Water	1007	Cohatt		<u></u>	· · · · · · · · · · · · · · · · · · ·	-	1.9 	•	43MU1
2712	WD45	Water	1007	Copper	20					[	
2733	WP45	Water	1992	Ethylbeozene		> ug/L	12			1— 1—	
2734	WP45	Water	1993	Ethylbenzene	N/A	un/L	1	1	0.00385	0.00385	ASMANC
2735	WP45	Water	1992	Fluonde	N/A	ua/L	2	2	100	200	45M02
2736	WP45	Water	1992	h-Chiorafluorobenzene	N/A	uo/L	6	6	8		45M04
2737	WP45	Water	1992	lron	20	) µg/L	2	1	420	420	45M06
2738	WP45	Water	1992	Lead	5	I Ugil	2	-	-	-	
2739	WP45	Water	1992	Magnesium	N/A	µg/L	2	2	8.60E+03	1.10E+04	45M06
2740	WP45	Water	1992	Manganese	10	).µg/L	2	<u>t</u>	2.40E+03	2.40E+03	45M06
2741	WP45	Water	1992	Methylenechloride	5	µg/L	12	-	-	-	
2742	WP45	Water	1992	Nickel	30	) µg/L	2	-	-	-	
2743	WP45	Water	1992	Nitrate	N/A	hð\r	2	2	1.50E+03	6.60E+03	45M02
2744	WP45	Water	1992	Ninte	200	μ <b>α</b> /L	2	-		-	
2745	WP45	Water	1992	p-Chlorofluorobenzene	N/A	µg/L	6	6	8.5	10	45M04
2746	WP45	Weter	1992	Phosphale	400	μ <b>ο/</b> Ι		-	-	-	
2747	WP45	Water	1992	Polassium	-N/A	INGU	Z	2	2.905+03	3.00E+03	451406
2748	WP45	Water	1992	Silver	20	.ug/L	Z	-		-	
2749	WP45	Water	1992	Sodium	N/A	P01-	2	2	4,20E+03	4.80E+03	45002
2750	VVP45	vvater .	1992			hou	<u> </u>	Z	5.JUE+U3	1.402+04	43800
2/31	WP45	vvater	1992		0.0	ugut.	12	-	-	-	
2/34	W745	Water Motore	1992		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		43		·	-	······
2754		Mater	1002	Trana-D/CE	<u>_</u>		12		- 17	- 30	4544711
2756		Water	1002	Techoroethere	!		12	í	1.4	370	451401
2756	W045	Water	1007	Vensdiam		14100	2	1	190	190	451402
2757	MD45	Water	1997	Vavi chioride	2	uovi	12	-	_		
2758	WP45	Water	1997	Xvienes (total)	5	100/	12			_	·····
2759	WP45	Water	1992	Zinc	10	uo/L	2			······	
2760	WP45	Surface Soil	1992	1,1,1-Trichloroethane	0.5	ug/kg	3				
2761	WP45	Surface Soil	1992	1,1,2-Trichloroethane	0.5	ug/kg	3	-	-	-	
2762	WP45	Surface Soil	1992	1,1-Dichloroethane	t	pg/kg	3	-	-	-	
2763	WP45	Surface Soil	1992	1.2-Dichicroethane	0.5	µg/kg	3	-	-		
2764	WP45	Surface Soil	1992	1,4-Dichiorobenzene	2	have	3	-	-	-	
2765	WP45 i	Surface Soil	1989	Acenaphthene, soil	760	ind wa	17	2	69	452	45\$809
2768	WP45	Surface Soil :	1988	Aluminum	IN/A	10/14	4	4	6.48E+06	8.83E+06	45M03
2767	WP45	Surface Soil	1989	Anthracene, soil	1	h0.k0	17	4	86	921	458505
2768	WP45	Surface Soil	1968	Antimony	4.50E+03	ho/kg	4	2	5.90E+03	8.90E+03	454403
2769	WP45	Surface Soil	1968	Banum	N/A	hô kà	4		8-24E+04	1.50E+05	45M03
2770	WP45	Surface Soil	1992	Benzene		ihðykð	3		-	-	
2771	WP45	Surface Soil I	1968		10	HOW	4	1	210	210	431653
2/72	VW-43	Surrace Soil	1869			pg/kg	17	17	470	535	400000
21/3	VVP43	Currece SOI	1990	Benzo(alumne		LINE AND	4	2	4/0	000	440004
21/4	CPTYY	Surface Soll	1203	Benzolh Buorantheae		IN MARCE	1/	19	200	389	453300
7770	WOAS	Surface Col	1080	Renzorh Surranibane enil	JU + 0	LING REG	47	42	47V	484	445504
;;;;;;	VADAS	Surface Soll	1046	Benzo(a h licendece	40	LING/POL	(/ 		510	530	45402
2775	WP45	Surface Soll	1980	Benzola h iloerviene		Lugito	17	13	0.4	400	455903
2779	WP45	Surface Sol	1989	Benzorkhluoranthene	04	LINO/10	17	15	0.2	264	455905
2780	WP45	Surface Soil	1968	Bis/2-ethyliteryliphthalate	100	UDANO		1	650	650	451403
2781	WP45	Surface Soil	1968	Cadmium	455	uaka	4	3	655	2.81E+03	451402
2782	WP45	Surface Soil I	1968	Calcium	N/A	,µo/ka	4	4	3,70E+06	7.16E+08	454403
2783	WP45	Surface Soil :	1992	Carbon tetrachionide	1	lµg/Kg	3	_	-	-	
2784	WP45	Surface Soil	1992	Chioroform	0.5	µg/kg	3	-	-		
2786	WP45	Surface Soil	1968	Chromium	IN/A	ind yo	4	4	1.22E+04	1.495+04	45M03
2784	WP45	Surface Soil	1988	Chrysene	70	pong	4	2	350	380	45M03
2787	WP45	Surface Soll	1989	Chrysene	0.5	·µg/kg	17	10	1	627	455505
2768	WP45	Surface Soil :	1992	Cis-1,2-dichloroethylene	1	iµg/kg	3.		-	-	
2784	WP45	Surface Sol	1968	Copar	N/A	10000	4	4	6 74E+01	8 73E+03	454413

٠.

	A	8	C	D	Ε	E E	G	н	1	J	к
<b></b>	1								!	-	
F 1	Source Area	Media	Date	Analyte	DL	Units	#Samples	#Detect	Min Conc.	Max Conc.	Max Location
2700	LADAS	Sudara Soul	1000	Canner	N/A	unika	4	4	1 636-04	1 825-04	451403
2130	VVP45	Sufface Soll	1990	Copper	·*·	parka.			1.332-04	1.030+04	43403
2/91	WP45	Surface Soil	1989	Digenzo(a,n)aninracene	0.1	ugikg	1/	14	Ų.4	<b>6</b> 0	455505
2792	WP45	Surface Soil	1988	Diethyiphthalate	60	µg/kg	4	1	210	210	45M01
2793	WP45	Surface Soil	1992	Ethylbenzene		hð\kð	3	. –	-	<u>1</u>	
2794	WP45	Surface Soil	1988	Fluoranthene	50	µg/kg	4	2	570	590	45M02
2795	WP45	Surface Soil	1989	Fluoranthene	Dí	uq/kq	17	15	0.07	1.81E+03	45SS05
2798	10/245	Surface Soil	1080	Ekjorene	9	La/ka	17	1	60	Â0	4555508
2747	1400 45	Curtace Cod	1000			ualka			E10	630	46400
2191	VVP45	Sunace Sol	1900		30	PY Y			310	330	40WIU2
2798	WP45	Surface Soil	1888	Indeno(1.2.3-cd)pyrene	0.3	hðurð	17	13	0.3	253	455503
2799	WP45	Surface Soil	1988	Iron	N/A	hð\xð	4	4	1.21E+07	1.45E+07	45M03
2800	WP45	Surface Soil	1988	Lead	N/A	µg/kg	4	- 4	9.39E+03	1.89E+04	45M02
2801	WP45	Surface Soil	1992	Lead	N/A	uo/ko	2	2	4 60F+03	4 70E+03	455807-A
3802	14/045	Sudace Soil	1096	449004FHUD	N/A	uo/ko	4	4	1855-08	5.025+08	451403
2002		Currace Soll	1000		*//*				0.000 .00	4.025.05	451400
2503	WP45	SULLACE 201	1968	Manganese	NIA	haved	4		2.065+05	4.03E+05	45MU2
2804	WP45	Surface Soil	1988	Mercury	19.6	havka	4	1	228	228	45M03
2805	WP45	Surface Soil	1989	Mercury	80.5	hð\kð	17	2	318	3.17E+03	45SS07-
2806	WP45	Surface Soil	1992	Methylenechloride	\$	µg/kg	3	_	' <b>_</b>	-	
2807	WP45	Surface Soil	1988	Nirke	N/A	ua/ka	4	4	1445+04	1855+04	45M03
2808	WP45	Surface Soll	1989	Phenanthrene	20	uci/ko			<b>∡</b> 001	6201	45402
2000	4045	Curdence Cold	1000			10/				1 205 . 25	AFROM
11903	1444-40	SUMACE SOIL	1993		1	- unit	1/1	14	6	1.295+03	00000
2610	WP45	Surface Soil	1988	Potassium	™A	hð\kð	4	4	5.38E+05	8.07E+05	45M02
2811	WP45	Surface Soil	1988	Pyrene	60	µg/kg	4	1	530	530	451402
2812	WP45	Surface Soil	1989	Pyrene	4	µg/kg	17	15	4	3.77E+03	45\$505
2813	WP45	Surface Soil	1988	Sodium ·-	N/A	ug/ka	4	4	2.77E+05	3.35E+051	451402
2814	WP45	Surface Sol	1997	Tetrachioroethene	0.5	ud/ko	1	1	620	A20	45SBOR-A
2842	14/045	Surface Col	1007	Tokiene		ug/ko	2.				
4013	14045	Surface Soll	1000		<u> </u>	Party I	<u> </u>				
2816	VVP45	Surface Soil	1992	Irans-UCE	··	HWKg	3	-	-	-	
2817	WP45	Surface Soil	1992	Trichloroelhene	1	µ0/kg	3	1	3.30E+03	3.30E+03	455808-A
2618	WP45	Surface Soil	1989	Trichiorofluoromethane	300.	µQ/kg	17	1	320	320	455503
2619	WP45	Surface Soil	1988	Vanadium	N/A	uc/kg	4	4	2.54E+04	3.29E+04	45M03
2820	WP45	Surface Soil	1992	Vinyl chloride	2	ua/ka	3	-		-	
28.21	10/045	Surface Soil	1097	Yvienes (total)	<u>-</u>	u0/ko	3	t	3 305+03	3 30E+03	45SBOR.4
2824	14046	Surface Coll	1000	7	.N/A	Joko	4		1 385+04	4 686-04	4514012
2022	VVI- 40	301808 308	1900		190	µwng			3.300 -04	4.00004	
2823	WP45	501	1885	1,1,1-1 ncniorosunane	0.5	haved	4	-	-		
2824	WP45	Soil	1992	1,1,2-Trichloroethane	0.5	µg/kg	4	-	-	-	
2825	WP45	Sort	1992	1,1-Dichloroethane	1.	µg/kg	4	-	-		
2826	WP45	Sail	1992	1,2-Dichloroethane	0.5	µg/kg i	4	-	-	-	
2827	WP45	Sail	1992	1 4-Dichlorobenzene	2	ua/ke	4	-	-		
78.78	14/045	Sol	1088	AL MINIO	'N/A	uning	1	1	5 865+06	5 865-06	451401
2020	140.45	- 30ii	1000		4 605 .02		·····	~~~~~	0.002.00	0,002.100	
1613	VVP 43	501	1900		4.302 -03	Maring		-	-	-	1.81404
2530	WP45	201	1988		-N/A	HQ/Kg	1;	1	6.41E404	0.41E+04	eoMU1
2831	WP45	Sot	1992	Benzene	2,	parkg :	4	<u>_</u>	<u> </u>	<b>—</b>	
2832	WP45	Sod	1988	Benzo(a)anthracene	10	µq/kg i	1		-	-	
2633	WP45	Sol	1988	Benzo(a)pyrene	9	ug/kg :	1	-	-	-	
2814	WP45	Soil	1985	Benzo(b)fuoranthene	30	uo/ka	1	-	-	_	
20.00	14045	Soil	1004	Benzola h inecidente	10-	uoire					····
40.33		Goli	1000		402						
2036	VVP 43	204	1960		100,		1			-	
2837	WP45	204	1958			HANKO .	1		3.28E+U3	3.28E+03	(UNCA
2835	WP45	Sol	1988	Calcium	NVA I	µ¢/kg	1	1	3.95E+06	3.95E+06	45M01
2639	WP45	Soil	1992	Carbon tetrachioride	1	parka	4	-	-	-	
2840	WP45	Soil	1992	Chloroform	0.5	vg/kg	4		-		
2841	WP45	Soil	1944	Chromium	N/A	10/kg	1		1.16E+04	1.16E+04	45M01
2841	W045	Sout	1049	Chrysene	70	up/ka					
44444	100745	Call	1000			<u></u>					
4043	CP4V	300	1992		1:	HALL I		,			
Z844	VVP45	SOI	1968		IWA .	haved	1	1	0.78E+03	5./8E+03	TUNCA
2845	WP45	Soil	1958	Copper	N/A	parka	1	1]	1.68E+04	1.58E+04	45M01
2846	WP45	Sol	1955	Ciethylphthalate	60	ug/kg	1	-	-	-	
2847	WP45	Soil	1992	Enylbenzene	2:	µg/kg	4	1	770	770	455808-8
2848	WP45	Sail	1984	Fluoranthene	50.	ua/ka	1	-		_	
7840	W.P45	Soil	1084	Indep0/123-cd/overse	£0.	un/re					
1043	14045	Call	1000	here	M/A				1 376-07	1 377	
2550	1412-40	301	1860	1011	INA ·	PARKS :	1	<u> </u>	1.2/240/	1.4/E+0/	
2851	WP45	Soit	1968	La20	N/A	Have	1	1	7.51E+03	7.\$1E+03	45M01
2852	WP45	Soil	1992	Lead	N/A	ug/kg	2	2	4.20E+03	4.30E+03	45SB08-8
2863	WP45	Sod	1986	Magnesum	N/A	ug/kg	1	1	3.82E+06	3.82E+06	45M01
2854	WP45	Soil	1968	Manganese	N/A ·	µg/ka i	1	1	2,15E+05	2.15E+05	45M01
2844	WP45	Soil	1968	Mettery	19 6.	ua/ko	1		-		
7944	14045	Soil	1001	Lethionechicowie	10.01	unite :					
1000	11F 70	GUN .	1992		÷C	+04.41	4		-		
2057	147F43	300	· 12404		, NVA	UDAKG	1	1	1.345+04	1.34E+04	420001

A.42

	A	₿	¢	<u> </u>	Ε	F	G	н	<u> </u>	J	ĸ
1	Source Area	Media	Date	Analyte	Dı.	Units	#Samples	#Detect	) Min Conc	Max Conc	Max Location
2858	WP45	Soil	1988	Phenanthrene		20 µg/kg	1	i		-	
2859	WP45	Sou	:988	Potassium	N/A	µg/kg	1	1	6.14E+05	6.14E+05	45M01
2860	WP45	Soil	1988	Pyrene	(	60 µg/kg	1	_	(_	(_	
2861	WP45	Soil	1988	Sodium	N/A	µg/kg		1	2.65E+05	2.65E+05	45M01
2862	WP45	Soil	1992	Tetrachloroethene	٥	).5 µg/kg	4	1	250	250	45SB08-8
2863	WP45	Soil	1992	Toluene		2 µg/kg	4	r <u>1</u>	1.10E+03	1.10E+03	455808-8
2864	WP45	Soil	1992	Trans-DCE		1 µg/kg	. 4	L	-	i <b>-</b>	
2865	WP45	Soil	1992	Trichloroethene		1 µg/kg	4	1	1.20E+04	1.20E+04	455B08-8
2866	WP45	Soil	1968	Vanadium	N/A	µg/kg	1	1	2.94E+04	2.94E+04	45M01
2867	WP45	Soil	1992	Vinyl chlonde		2 µg/kg	4	i –	·	j	
2866	WP45	Soil	1992	Xylenes (total)		5 µg/kg		1	4 60E+03	4 60E+03	455808-8
2869	WP45	Soil	1988	Zinc	N/A	µg/kg	· 1	1	3.92E+04	3.92E+04	45M01

1